VIAD CORP Form 4

February 21, 2007

FORM 4

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

STATEMENT OF CHANGES IN BENEFICIAL OWNERSHIP OF

SECURITIES

OMB Number:

3235-0287

Expires:

January 31, 2005

0.5

Estimated average burden hours per

OMB APPROVAL

response...

subject to Section 16. Form 4 or Form 5 obligations

may continue.

See Instruction

if no longer

Check this box

Filed pursuant to Section 16(a) of the Securities Exchange Act of 1934, Section 17(a) of the Public Utility Holding Company Act of 1935 or Section

30(h) of the Investment Company Act of 1940

1(b).

(Print or Type Responses)

1. Name and Address of Reporting Person * 5. Relationship of Reporting Person(s) to 2. Issuer Name and Ticker or Trading DYKSTRA PAUL B Issuer Symbol VIAD CORP [VVI] (Check all applicable) (Last) (First) (Middle) 3. Date of Earliest Transaction (Month/Day/Year) _X_ Director 10% Owner X_ Officer (give title _ Other (specify VIAD CORP, 1850 NORTH 02/20/2007 below) CENTRAL AVENUE, SUITE 800 President & CEO / Viad Corp (Street) 4. If Amendment, Date Original 6. Individual or Joint/Group Filing(Check Filed(Month/Day/Year) Applicable Line) _X_ Form filed by One Reporting Person Form filed by More than One Reporting PHOENIX, AZ 85004-4545 Person

(City)	(State)	(Zip) Table I - Non-Derivative Securities Acquired, Disposed of, or Beneficially C			y Owned					
1.Title of Security (Instr. 3)	2. Transaction Date (Month/Day/Year)	2A. Deemed Execution Date, if any (Month/Day/Year)	3. Transa Code (Instr.	8)	4. Securitie nor Disposed (Instr. 3, 4 a	d of (Ľ))	5. Amount of Securities Beneficially Owned Following Reported Transaction(s) (Instr. 3 and 4)	6. Ownership Form: Direct (D) or Indirect (I) (Instr. 4)	7. Nature of Indirect Beneficial Ownership (Instr. 4)
Common Stock	02/20/2007		F	•	780 <u>(1)</u>	D	\$ 37.835	55,171	D	
Common Stock	02/20/2007		J	V	81.8102	A	\$ 0 (2)	2,039.3744	I	401k Plan

Reminder: Report on a separate line for each class of securities beneficially owned directly or indirectly.

Persons who respond to the collection of SEC 1474 information contained in this form are not (9-02)required to respond unless the form displays a currently valid OMB control number.

Table II - Derivative Securities Acquired, Disposed of, or Beneficially Owned (e.g., puts, calls, warrants, options, convertible securities)

1. Title of Derivative Security (Instr. 3)	2. Conversion or Exercise Price of Derivative Security	3. Transaction Date (Month/Day/Year)	3A. Deemed Execution Date, if any (Month/Day/Year)	4. Transacti Code (Instr. 8)	5. onNumber of Derivative Securities Acquired (A) or Disposed of (D) (Instr. 3, 4, and 5)	:	ate	7. Titl Amou Under Securi (Instr.	nt of lying	8. Price of Derivative Security (Instr. 5)	9. Nu Deriv Secun Bene Owno Follo Repo Trans (Instr
				Code V	(A) (D)	Date Exercisable	Expiration Date	Title	Amount or Number of Shares		

Reporting Owners

Reporting Owner Name / Address				•	
	Director	10% Owner	Officer		Other

X

DYKSTRA PAUL B **VIAD CORP** 1850 NORTH CENTRAL AVENUE, SUITE 800 PHOENIX, AZ 85004-4545

President & CEO Viad Corp

Relationships

Signatures

Scott E. Sayre, 02/21/2007 Attorney-in-Fact

**Signature of Reporting Person

Explanation of Responses:

- If the form is filed by more than one reporting person, see Instruction 4(b)(v).
- Intentional misstatements or omissions of facts constitute Federal Criminal Violations. See 18 U.S.C. 1001 and 15 U.S.C. 78ff(a).
- (1) Shares were surrendered for payment of taxes in connection with vesting of restricted stock granted February 18, 2004.
- (2) The information is current as of this filing date.

Note: File three copies of this Form, one of which must be manually signed. If space is insufficient, see Instruction 6 for procedure. Potential persons who are to respond to the collection of information contained in this form are not required to respond unless the form displays a currently valid OMB number.

Named Executive Officers and Directors

Vincent J. Milano

2.134.027(4) 1.08%

Sudhir Agrawal, D. Phil.

Reporting Owners 2

Lagar Filling. Vivid Corti 1 Citi 1
3,976,343 ⁽⁵⁾ 1.99%
Louis J. Arcudi, III
1,408,248(6) *
Julian C. Baker
38,956,462 ⁽⁷⁾ 18.01%
Mark J. Casey
548,199(8) *
R. Clayton Fletcher
649,374 ⁽⁹⁾ *
James A. Geraghty
1,056,621(10) *
Mark Goldberg
157,500 ⁽¹¹⁾ *
Maxine Gowen
79,917 ⁽¹²⁾ *
Kelvin M. Neu
231,184 ⁽¹³⁾ *
William S. Reardon
250,473 ⁽¹⁴⁾ *
All current executive officers and directors as a group (11 persons)
45,278,246 ⁽¹⁵⁾ 20.41%
* Denotes less than 1% beneficial owner.
Except as otherwise noted, the address for each person listed above is c/o Idera Pharmaceuticals, Inc., 167 Sidney Street, Cambridge, Massachusetts 02139.
155

Table of Contents

(2)

Consists of (i) 2,090,125 shares of common stock held by Pillar Pharmaceuticals I, L.P., or Pillar I, (ii) 9,959,956 shares of common stock held by Pillar Pharmaceuticals II, L.P., or Pillar II, (iii) 2,871,839 shares of common stock held by Pillar Pharmaceuticals III, L.P., or Pillar III, (iv) 200,000 shares of common stock held by Pillar Pharmaceuticals IV, L.P., or Pillar IV, (v) 875,000 shares of common stock held by Pillar V, (vi) 8,875,973 shares of common stock held by Participations Besancon, or Besancon, and over which Pillar Invest Corporation has investment discretion, pursuant to an advisory agreement between Pillar Invest Corporation and Besancon, or the Advisory Agreement, (vii) 1,200,000 shares of common stock issuable upon exercise of warrants to purchase common stock held by Besancon and over which Pillar Invest Corporation has investment discretion pursuant to the Advisory Agreement, (viii) 540,681 shares of common stock held directly by Mr. El Zein and (ix) 285,000 shares of common stock subject to issued outstanding stock options that are exercisable within 60 days after March 9, 2018 held by Mr. El Zein. Mr. El Zein, a member of the Idera board until October 31, 2017, is a director and controlling stockholder of Pillar Invest Corporation, which is the general partner of Pillar I, Pillar II, Pillar III, Pillar IV and Pillar V and is a limited partner of Pillar I, Pillar III, Pillar III, Pillar IV and Pillar V. Mr. El Zein expressly disclaims beneficial ownership over shares held directly by Pillar I, Pillar II, Pillar III, Pillar IV, Pillar V and indirectly by Pillar Invest Corporation, including the warrants to purchase common stock issued in connection therewith held by Besancon, or the Besancon Warrants. Pillar I, Pillar II, Pillar II, Pillar IV and Pillar V expressly disclaim beneficial ownership of the Besancon Warrants. Besancon is an investment fund having no affiliation with Mr. El Zein, Pillar I, Pillar II, Pillar IV, V or Pillar Invest Corporation. The information in this footnote is based on a Schedule 13D/A filed with the SEC on October 17, 2016; Form 4s filed with the SEC on May 3, 2017, October 17, 2017, December 15, 2017 and January 19, 2018; and on information provided to us by Pillar Invest Corporation and Mr. El Zein. Pursuant to the terms of the warrants to purchase common stock issued to the Pillar Investment Entities, the warrants to purchase common stock issued to the Pillar Investment Entities cannot be exercised by the holders thereof with respect to any portion of the shares, to the extent that such exercise would result in the Pillar Investment Entities beneficially owning in the aggregate more than 19.99% of (x) the number of shares of common stock issued outstanding or (y) the voting power of our securities outstanding immediately after giving effect to the exercise of the warrants to purchase common stock.

(3)

Consists of (i) 1,723,224 shares of our common stock owned by 667, L.P., (ii) 16,438,080 shares of our common stock owned by Baker Brothers Life Sciences, L.P., (iii) 35,105 shares of our common stock owned by 14159, L.P., (iv) (a) 55,042 shares of our common stock held directly by Mr. Baker and (b) 73,684 shares of our common stock held directly by Dr. Neu, and in which each of 667, L.P., Baker Brothers Life Sciences, L.P. and 14159, L.P. (collectively, the "*Baker Brothers Funds*"), has an indirect pecuniary interest and may be deemed to own a portion of these shares, and (v) (a) 2,250,846 shares of common stock issuable upon exercise of warrants to purchase common stock owned by 667, L.P., (b) 17,626,445 shares of common stock issuable upon exercise of warrants to purchase common stock owned by Baker Brothers Life Sciences, L.P., and (c) 439,036 shares of common stock issuable upon exercise of warrants to purchase common stock owned by 14159, L.P. (vi) (a) 157,500 shares of common stock subject to outstanding stock options that are exercisable within 60 days after March 9, 2018 held by Mr. Baker and (b) 157,500 shares of common stock subject to outstanding options that are exercisable within 60 days after March 9, 2018 held by Dr. Neu. As a result of the application of the Beneficial Ownership Cap, as described below in this footnote, the table above does not include the following as being beneficially owned by the Baker Brothers Funds: (a) 2,389,927 shares of common stock issuable upon exercise of warrants to purchase common stock owned by 667, L.P., (b) 19,280,570 shares of common stock issuable upon exercise of warrants to purchase common stock owned by Baker Brothers Life Sciences, L.P. and (c) 480,555 shares of common stock issuable upon exercise of warrants to

Table of Contents

purchase common stock owned by 14159, L.P. The information in this footnote is based on a Schedule 13D/A filed with the SEC on March 6, 2018; a Form 4 filed with the SEC on January 4, 2018; and on information provided to us by the Baker Brothers Funds and Mr. Baker. Mr. Baker, a member of the Idera board, is a managing member of Baker Bros. Advisors LP and is a principal of Baker Bros. Advisors (GP), LLC, the sole general partner of Baker Bros. Advisors LP. Baker Bros. Advisors LP serves as the investment advisor to the Baker Brothers Funds. Accordingly, Mr. Baker may be deemed to have sole power to direct the voting and disposition of the shares of common stock held directly by the Baker Brothers Funds and indirectly by Baker Bros. Advisors LP and Baker Bros. Advisors (GP), LLC. Mr. Baker expressly disclaims beneficial ownership over shares held directly by the Baker Brothers Funds and indirectly by Baker Bros. Advisors LP and Baker Bros. Advisors (GP), LLC, except to the extent of his pecuniary interest therein, if any, by virtue of his pecuniary interest therein. Dr. Neu, a member of the Idera board, is an employee of Baker Bros. Advisors LP. Under the terms of the warrants issued to the Baker Brothers Funds, the Baker Brothers Funds are not permitted to exercise such warrants to purchase common stock to the extent that such exercise would result in the Baker Brothers Funds (and their affiliates) beneficially owning more than 4.99% of the number of shares of our common stock issued and outstanding immediately after giving effect to the issuance of shares of common stock issuable upon exercise of such warrants to purchase common stock. This limitation on exercise of the warrants to purchase common stock issued to the Baker Brothers Funds is referred to in this footnote as the "Beneficial Ownership Cap." The Baker Brothers Funds have the right to increase this beneficial ownership limitation in their discretion on 61 days' prior written notice to us, provided that in no event are the Baker Brothers Funds permitted to exercise such warrants to purchase common stock to the extent that such exercise would result in the Baker Brothers Funds (and their affiliates) beneficially owning in the aggregate more than 19.99% of the number of shares of our common stock issued and outstanding or the combined voting power of our securities outstanding immediately after giving effect to the issuance of shares of common stock issuable upon exercise of such warrants to purchase common stock. On March 5, 2018, the Baker Brothers Funds provided Idera with 61 days' notice in regards to warrants to purchase up to 20,316,327 shares of our common stock at an exercise price of \$0.47 per share, which the Baker Brothers Funds subsequently exercised on March 15, 2018.

- (4) Includes 1,887,500 shares of common stock subject to outstanding stock options that are exercisable within 60 days after March 9, 2018.
- (5) Includes 3,850,066 shares of common stock subject to outstanding stock options that are exercisable within 60 days after March 9, 2018.
- (6) Includes 1,363,706 shares of common stock subject to outstanding stock options that are exercisable within 60 days after March 9, 2018.
- Consists of shares reported under footnote 3 to this table above. Mr. Baker is a managing member of Baker Bros. Advisors LP and is a principal of Baker Bros. Advisors (GP), LLC, the sole general partner of Baker Bros. Advisors LP. Baker Bros. Advisors LP serves as the investment advisor to the Baker Brothers Funds. Accordingly, Mr. Baker may be deemed to have sole power to direct the voting and disposition of the shares of common stock held directly by the Baker Brothers Funds and indirectly by Baker Bros. Advisors LP and Baker Bros. Advisors (GP), LLC. Mr. Baker expressly disclaims beneficial ownership over shares held directly by the Baker Brothers Funds and indirectly by Baker Bros. Advisors LP and Baker Bros. Advisors (GP), LLC, except to the extent of his pecuniary interest therein, if any, by virtue of his pecuniary interest therein.
- (8) Includes 513,749 shares of common stock subject to outstanding stock options that are exercisable within 60 days after March 9, 2018.

Table of Contents

- (9) Consists of 649,374 shares of common stock subject to outstanding stock options that are exercisable within 60 days after March 9, 2018.
- (10) Includes 621,252 shares of common stock subject to outstanding stock options that are exercisable within 60 days after March 9, 2018.
- (11)

 Consists of 157,500 shares of common stock subject to outstanding stock options that are exercisable within 60 days after March 9, 2018.
- Includes 72,917 shares of common stock subject to outstanding stock options that are exercisable within 60 days after March 9, 2018, and 7,000 shares of common stock held in the name Brian Macdonald for Maxine Gowen Trust, for which Dr. Gowen is a beneficiary and trustee.
- (13) Includes 157,500 shares of common stock subject to outstanding stock options that are exercisable within 60 days after March 9, 2018.
- (14) Includes 217,500 shares of common stock subject to outstanding stock options that are exercisable within 60 days after March 9, 2018.
- Includes 5,867,559 shares of common stock subject to outstanding stock options held by the directors and executive officers as a group that are exercisable within 60 days after March 9, 2018 and shares reported in clauses (i) through (iv) of the first sentence of footnote 3 to this table above.

DESCRIPTION OF HOLDCO CAPITAL STOCK

This section of this joint proxy statement/prospectus summarizes the material terms of Holdco's capital stock that will be in effect if the mergers are completed and is not complete. You are encouraged to read (1) the Holdco charter, which is attached as Annex D to this joint proxy statement/prospectus and is incorporated herein by reference, (2) the Holdco bylaws, which will be in effect as of the effective time of the mergers and a form of which is included as Annex E to this joint proxy statement/prospectus and is incorporated herein by reference and (3) the applicable provisions of the DGCL. The following summary should be read in conjunction with the section entitled "Comparison of Rights of Idera Stockholders, BioCryst Stockholders and Holdco Stockholders" beginning on page 162 of this joint proxy statement/prospectus.

After giving effect to the Holdco charter, Holdco's authorized capital stock will consist of shares made up of:

380,000,000 shares of Holdco common stock, par value \$0.01 per share; and

20,000,000 shares of Holdco preferred stock, par value \$0.01 per share, the rights and preferences of which may be established from time to time by the Holdco board.

At the effective time of the mergers, each issued and outstanding share of Idera common stock (other than shares held by Idera, BioCryst, Holdco, Merger Sub A or Merger Sub B, or any subsidiary of each of the aforementioned entities, or by Idera as treasury shares, which will be canceled and retired and cease to exist) will be converted into the right to receive 0.20 fully paid and nonassessable shares of Holdco common stock (plus cash in lieu of fractional shares).

At the effective time of the mergers, each issued and outstanding share of Idera preferred stock (other than shares held by Idera, BioCryst, Holdco, Merger Sub A or Merger Sub B, or any subsidiary of each of the aforementioned entities, or by Idera as treasury shares, which will be canceled and retired and cease to exist) will be converted into the right to receive a number of fully paid and nonassessable shares of Holdco common stock equal to (i) \$1.00 divided by the 20-trading-day average trading price of Idera common stock, ending with the trading day prior to the third day prior to the closing date, multiplied by (ii) 0.20 (plus cash in lieu of fractional shares).

At the effective time of the mergers, each issued and outstanding share of BioCryst common stock (other than shares held by Idera, BioCryst, Holdco, Merger Sub A or Merger Sub B, or any subsidiary of each of the aforementioned entities, or by BioCryst as treasury shares, which will be canceled and retired and cease to exist) will be converted into the right to receive 0.50 fully paid and nonassessable shares of Holdco common stock (plus cash paid in lieu of fractional shares).

We expect a total of approximately 52.8 million Holdco shares to be issued to Idera stockholders and 56.5 million Holdco shares to be issued to BioCryst stockholders in connection with the mergers. We expect that, immediately following the consummation of the mergers, there will be a total of approximately 109.3 million Holdco shares outstanding.

Description of Holdco Common Stock

Dividends

Subject to the rights, if any, of holders of any series of outstanding preferred stock, if and when issued and subject to applicable law, holders of Holdco common stock will be entitled to receive dividends when, as and if declared by the Holdco board out of funds legally available for payment.

Table of Contents

Voting Rights

Subject to the rights, if any, of the holders of any series of preferred stock if and when issued and subject to applicable law, each holder of Holdco common stock will be entitled to one vote per share and all voting rights will be vested in the Holdco common stock. Holders of shares of Holdco common stock will have noncumulative voting rights, which means that the holders of more than 50% of the shares voting for the election of directors can elect 100% of the directors and the holders of the remaining shares will not be able to elect any directors.

Liquidation Rights

Subject to the rights, if any, of holders of any series of outstanding preferred stock, if and when issued and subject to applicable law, in the event of any liquidation, dissolution or winding up of the affairs of Holdco, whether voluntary or involuntary, the holders of shares of Holdco common stock shall have equal rights to receive the assets and funds of Holdco available for distribution to Holdco stockholders and such assets or funds shall be distributed pro rata in accordance with the number of shares of Holdco common stock held by each such holder.

Preemptive Rights

The shares of Holdco common stock to be issued at the effective time of the mergers will be validly issued, fully paid and nonassessable. Holders of shares of Holdco common stock will not be entitled to preemptive rights. Shares of Holdco common stock will not be convertible into shares of any other class of capital stock.

Description of Holdco Blank Check Preferred Stock

Under the Holdco charter, without further Holdco stockholder action, the Holdco board is authorized to provide for the issuance of preferred stock in one or more series, the shares of each of which series may have such voting powers, full or limited, or no voting powers, and such designations, preferences and relative, participating, optional or other special rights, and qualifications, limitations or restrictions thereof, as shall be permitted under the DGCL. Without limiting the generality of the foregoing, such shares may provide that such series shall be superior or rank equally or be junior to the preferred stock of other series to the extent permitted by law.

Anti-takeover Provisions

General

Some provisions of DGCL, the Holdco charter and the Holdco bylaws could discourage or make it more difficult to acquire control of the combined company by means of a tender offer, open market purchases, a proxy contest or otherwise. A description of these provisions is set forth below.

Meetings and Elections of Directors

The Holdco bylaws provide that special meetings of stockholders may be called only by the Holdco board or secretary of Holdco pursuant to a resolution adopted by the board of directors. The Holdco charter prohibits stockholders from calling special meetings.

The Holdco charter provides that Holdco stockholders cannot act by written consent in lieu of a meeting; provided, however, that any action required or permitted to be taken by the holders of any series of preferred stock, voting separately as a series or separately as a class with one or more other such series, may be taken without a meeting, without prior notice and without a vote, to the extent expressly so provided by the applicable certificate of designation for such series of preferred stock.

Table of Contents

The Holdco charter and the Holdco bylaws provide that any vacancies in the Holdco board, however occurring, shall be filled by an affirmative vote of a majority of the remaining directors then in office, even if less than a quorum, or by the sole remaining director, and shall not be filled by Holdco stockholders.

No Cumulative Voting

The Holdco charter does not grant stockholders the right to vote cumulatively.

Advance Notice Procedure

The Holdco bylaws provide an advance notice procedure for stockholders to nominate director candidates for election or to bring business before an annual meeting of stockholders, including proposed nominations of persons for election to the board of directors.

For nominations or other business to be properly brought before an annual meeting, such stockholder must give written notice to the secretary of Holdco in writing not later than the close of business on the 90th day or earlier than the close of business on the 120th day prior to the anniversary date on which Holdco first distributed its proxy materials for the prior year's annual meeting, provided that if the annual meeting is called for a date that is within 30 days before or more than 60 days after of the first anniversary of the prior year's annual meeting, notice must be provided not earlier than the close of business on the 120th day prior to such annual meeting and not later than the close of business on the later of (i) the 90th day prior to such annual meeting and (ii) the 10th day following the date on which public disclosure of the date of the annual meeting is first made by Holdco.

The notice must contain specific information concerning the person to be nominated or the matters to be brought before the meeting, as well as specific information concerning the stockholder submitting the proposal or making the nomination.

COMPARISON OF RIGHTS OF IDERA STOCKHOLDERS, BIOCRYST STOCKHOLDERS AND HOLDCO STOCKHOLDERS

Both Idera and BioCryst and are incorporated under the laws of the State of Delaware and, accordingly, the rights of the Idera stockholders and BioCryst stockholders are currently governed by the DGCL. Holdco will continue to be a Delaware corporation following completion of the mergers and will be governed by the DGCL.

Upon completion of the mergers, the Idera stockholders and the BioCryst stockholders will become Holdco stockholders, and the current certificate of incorporation of Holdco will be amended and restated as of the effective time of the mergers in substantially the form attached as Annex D to this joint proxy statement/prospectus. The current bylaws of Holdco will be amended and restated at the effective time of the mergers in substantially the form of the Holdco bylaws attached as Annex E to this joint proxy statement/prospectus. Following the mergers, the rights of the former Idera stockholders and the former BioCryst stockholders will thereafter be governed by the DGCL and by the Holdco charter and the Holdco bylaws.

The following description summarizes the expected material differences between the rights of the Idera stockholders, BioCryst stockholders and Holdco stockholders, but is not a complete statement of all those differences, or a complete description of the specific provisions referred to in this summary. Stockholders should read carefully the relevant provisions of the DGCL and the respective certificates of incorporation and bylaws of Idera, BioCryst and Holdco. For more information on how to obtain the documents that are not attached to this joint proxy statement/prospectus, see "Where You Can Find More Information" beginning on page 169 of this joint proxy statement/prospectus.

Authorized Capital

Rights of Idera StockholdersThe authorized capital stock of Idera

consists of 280,000,000 shares of

share, and 5,000,000 shares of

share

common stock, \$0.001 par value per

preferred stock, \$0.01 par value per

The authorized capital stock of BioCryst consists of 200,000,000 shares of common stock, \$0.01 par value per share, and 5,000,000 shares of preferred stock, \$0.01 par value per share, of which 200,000 shares

are designated series B junior

participating preferred.

Rights of BioCryst Stockholders

Rights of Holdco Stockholders

Upon completion of the mergers, the authorized capital stock of Holdco will consist of 380,000,000 shares of common stock, \$0.01 par value per share, and 20,000,000 shares of preferred stock, \$0.01 par value per share.

Outstanding Capital Stock

As of the Idera record date, Idera had 216,005,142 shares of common stock issued and outstanding, and 655 shares of preferred stock issued and outstanding.

As of March 22, 2018, the most recent practicable date before this joint proxy statement/prospectus was mailed to BioCryst stockholders, BioCryst had 98,701,856 shares of common stock issued and outstanding, and no shares of preferred stock issued and outstanding.

Upon completion of the mergers, Holdco will have 109.3 million shares of common stock issued and outstanding, and no shares of preferred stock issued and outstanding.

Number of Directors

The amended and restated certificate of incorporation of Idera (the "Idera charter") provides that the number of directors shall be fixed in accordance with the amended and restated bylaws of Idera (the "Idera bylaws"), and that there shall be not less than three directors. The Idera bylaws provide that the number of directors shall be set by resolution of the Idera board and that the Idera

The amended and restated certificate of incorporation of BioCryst (the "BioCryst charter") provides that the board of directors shall consist of not less than six nor more than 12 directors and that the exact number of directors will be fixed by resolution of the BioCryst board.

There are currently eight positions authorized by the BioCryst board and eight

The Holdco charter provides that the board of directors shall consist of not less than six nor more than 12 directors and that the exact number of directors will be fixed by resolution of the Holdco board.

Upon completion of the mergers, there will be nine positions authorized and nine directors serving on the Holdco board.

Rights of Idera Stockholders

board shall be divided into three classes, with no class having more than one director more than any other class.

There are currently eight positions authorized by the Idera board and seven directors serving on the Idera board.

Rights of BioCryst Stockholders

directors serving on the BioCryst board.

Rights of Holdco Stockholders

Election of Directors

The Idera bylaws provide that directors are elected by a plurality of the votes cast by the stockholders entitled to vote thereon.

Pursuant to the Idera bylaws and the Idera charter, each director holds office for three years following the annual meeting at which such director is elected and until such director's successor shall have been duly elected and qualified, or until such director's earlier death, resignation or removal.

The amended and restated bylaws of BioCryst (the "BioCryst bylaws") provide that directors are elected by a plurality of the votes cast by the stockholders entitled to vote thereon.

Pursuant to the BioCryst bylaws and the BioCryst charter, each director holds office for three years following the annual meeting at which such director is elected and until such director's successor shall have been duly elected and qualified, or until such director's earlier death, resignation or removal.

The Holdco bylaws provide that directors are elected by a majority of the votes cast at a meeting where there is a quorum, provided, however, that directors may be elected by a plurality of the votes cast at a meeting where there is a quorum if as of the record date for such meeting the number of nominees exceeds the number of directors to be elected.

Pursuant to the Holdco bylaws, each director holds office until the next annual meeting for the election of directors and until the director's successor is duly elected and qualified.

Removal of **Directors**

The Idera charter and the Idera bylaws provide that any director may be removed only for cause by the affirmative vote of holders of at least two-thirds of the shares of capital stock of Idera then issued and outstanding and entitled to vote generally for the election of directors.

The BioCryst charter and bylaws provide that any director may be removed from office at any time, but only for cause, by the affirmative vote of holders of at least 75% of the total number of votes entitled to be cast by holders of all of the shares of capital stock of BioCryst then entitled to vote generally in the election of directors.

The Holdco charter provides that any director may be removed from office at any time, with or without cause, only by the affirmative vote of the holders of a majority of the voting power of all of the shares of capital stock of Holdco then entitled to vote generally in the election of directors, voting as a single class.

Vacancies on the **Board**

The Idera charter and the Idera bylaws provides that vacancies in the Idera board, however occurring, may be filled for the applicable unexpired term exclusively by a majority vote of the directors then in office (even though they constitute less than a quorum), or by a sole remaining director.

The BioCryst charter and the BioCryst bylaws provide that vacancies in the BioCryst board, however occurring, shall be filled by an affirmative vote of a majority of the directors then in office.

The Holdco charter and the Holdco bylaws provide that vacancies in the Holdco board, however occurring, shall be filled by an affirmative vote of a majority of the directors then in office.

Advance Notice Requirements for Stockholder Nominations and Other Proposals

The Idera bylaws provide that nominations of persons for election to the Idera board and the proposal of business to be considered by

The BioCryst bylaws provide that nominations of persons for election to the BioCryst board and the proposal of business to be considered by

The Holdco bylaws provide that nomination of persons for election to the Holdco board and the proposal of business to be considered by the

Rights of Idera Stockholders

stockholders may be made only
(i) by or at the direction of the Idera
board (ii) by a stockholder who is
entitled to vote and who has
complied with the advance notice
procedures set forth in the Idera
bylaws.

In general, an Idera stockholder wishing to nominate a director or raise another proposal at an annual meeting of stockholders must notify the Secretary of Idera in writing not later than 60 days nor more than 90 days prior to the scheduled annual meeting, unless less than 70 days' notice or prior public disclosure of the date of the scheduled annual meeting is given or made, in which case notice by the stockholder must be delivered or received not later than the close of business on the 10th day following the day on which such notice of the date of the scheduled annual meeting was mailed or the day on which such public disclosure was made.

The notice must contain specific information concerning the person to be nominated or the matters to be brought before the meeting, as well as specific information concerning the stockholder submitting the proposal or making the nomination.

Notice of Special Meeting

The Idera bylaws generally provide that notice of a stockholder meeting must be given to each stockholder permitted to take any action at such meeting not less than 10 days nor more than 60 days before the date of the meeting.

Rights of BioCryst Stockholders

the stockholders may be made only (i) pursuant to BioCryst's notice of meeting, (ii) by or at the direction of the BioCryst board or (iii) by any stockholder who is entitled to vote and who has complied with the advance notice procedures set forth in the BioCryst bylaws.

For nominations or other business to be properly brought before an annual meeting, such stockholder must give written notice to the Secretary of BioCryst in writing not later than the close of business on the 90th day or earlier than the close of business on the 120th day prior to the anniversary date of the preceding year's annual meeting, provided that if the annual meeting is called for a date that is not within 30 days of the first anniversary of the prior year's annual meeting, notice must be provided not earlier than the close of business on the 120th day prior to such annual meeting and not later than the close of business on the 90th day prior to such annual meeting or the 10th day following the date on which public disclosure of the date of the annual meeting is first made by BioCryst.

The notice must contain specific information concerning the person to be nominated or the matters to be brought before the meeting, as well as specific information concerning the stockholder submitting the proposal or making the nomination.

The BioCryst bylaws provide that written notice of each meeting of stockholders will be given not less than 10 nor more than 60 days before the date of the meeting to each stockholder entitled to vote at such meeting.

164

Rights of Holdco Stockholders

stockholders may be made only
(i) by or at the direction of the
Holdco board or (ii) by any
stockholder who is entitled to vote
and who has complied with the
advance notice procedures set forth
in the Holdco bylaws.

For nominations or other business to be properly brought before an annual meeting, such stockholder must give written notice to the Secretary of Holdco in writing not later than the close of business on the 90th day or earlier than the close of business on the 120th day prior to the anniversary date on which Holdco first distributed its proxy materials for the prior year's annual meeting, provided that if the annual meeting is called for a date that is not within 30 days of the first anniversary of the prior year's annual meeting, notice must be provided not earlier than the close of business on the 120th day prior to such annual meeting and not later than the close of business on the later of (i) the 90th day prior to such annual meeting and (ii) the 10th day following the date on which public disclosure of the date of the annual meeting is first made by Holdco.

The notice must contain specific information concerning the person to be nominated or the matters to be brought before the meeting, as well as specific information concerning the stockholder submitting the proposal or making the nomination.

The Holdco charter and the Holdco bylaws provide that no less than 10 nor more than 60 days' prior written notice to every stockholder entitled to vote thereat is required for a special meeting of the stockholders.

Rights of Idera Stockholders

Any notice of a special meeting must include the place, date, hour and purpose for which the meeting is called.

Amendments to the Charter

The Idera charter provides that the provisions in the charter regarding the Idera board, voting requirements, special meeting requirements and the transaction of business thereat may be amended only by a vote of holders of at least 75% of the voting power of all shares of capital stock of Idera then issued and outstanding and entitled to vote generally for the election of directors.

Other amendments to the Idera charter that are not inconsistent with this restriction are to be made in accordance with the DGCL.

Amendments to Bylaws

The Idera charter provides that the Idera bylaws may be amended or repealed by the Idera board.

The Idera bylaws state that the bylaws may be altered, amended or repealed or new bylaws may be adopted by (i) the affirmative vote of a majority of the directors present at any regulator or special meeting of the Idera board at which a quorum is present or (ii) holders of a majority of shares of Idera's capital stock then issued and outstanding and entitled to vote generally for the election of directors (so long as appropriate notice was stated in the notice of special meeting); provided, however, that any stockholder

Rights of BioCryst Stockholders

Notices of a special meeting must include the place, date, hour and purpose for which the meeting is called.

The BioCryst charter provides that the provision regarding amendment of the charter may be amended, altered, changed or repealed only by an affirmative vote of holders of at least 75% of the total number of votes cast by the holders of the shares of capital stock of BioCryst that are entitled to vote generally in the election of directors.

The BioCryst charter cannot be amended, including any amendment through consolidation, merger, combination or other transaction, in any manner which would materially alter or change the powers, preferences or special rights of the holders of series B preferred stock so as to affect them adversely without the affirmative vote of a majority of the outstanding shares of series B preferred stock voting together as a single class.

The BioCryst charter provides that the BioCryst board is authorized to adopt, amend or repeal the BioCryst bylaws.

The BioCryst bylaws state that the bylaws may be altered, amended or repealed by (i) the affirmative vote of a majority of the directors present at any regular or special meeting of the BioCryst board at which a quorum is present or (ii) by the affirmative vote of the holders of a majority of the issued and outstanding shares of the capital stock of BioCryst entitled to vote at any regular meeting of stockholders or at any special meeting of stockholders, provided notice of such alteration, amendment or repeal shall have been stated in the notice of such special meeting.

Rights of Holdco Stockholders

Notice of a special meeting must include the date, time, place and purpose for which the meeting is called.

The Holdco charter provides that the provisions in the charter regarding the capital stock of Holdco, the Holdco board, limitation of director liability and indemnification, stockholder action, amendment of the bylaws, forum selection and amendment of the charter may be amended only by an affirmative vote of holders of at least 75% of the voting power of the outstanding shares entitled to vote for the election of directors.

Other amendments to the Holdco charter that are not inconsistent with this restriction are to be made in accordance with the DGCL.

The Holdco charter provides that the Holdco bylaws may be amended or repealed by the Holdco board or by the affirmative vote of the holders of a majority of all of the shares of capital stock of Holdco, voting as a single class.

The Holdco bylaws state that the bylaws may be amended, altered, changed, adopted or repealed by (i) the affirmative vote of a majority of the directors present at any regular or special meeting of the Holdco board at which a quorum is present or (ii) by the affirmative vote of the holders of a majority of the voting power of all of the shares of capital stock of Holdco then entitled to vote.

Rights of Idera Stockholders

amendment to the Idera bylaws regarding special meetings, director nominations, annual meetings notice, action without a meeting, organization, directors or amendments of the bylaws requires a vote of holders of at least 75% of the voting power of all shares of capital stock of Idera then issued and outstanding and entitled to vote generally for the election of directors.

Rights of BioCryst Stockholders

Rights of Holdco Stockholders

Special Meeting of Stockholders

The Idera charter and the Idera bylaws provide that special meetings may be called only (i) by the chief executive officer of Idera (or if no chief executive officer, by the president of Idera) or (ii) by the Idera board.

The BioCryst charter and the BioCryst bylaws provide that special meetings may only be called by the BioCryst board by a resolution approved by a majority of the entire BioCryst board.

The Holdco charter and the Holdco bylaws provide that special meetings may only be called by the Holdco board by a resolution approved by a majority of the entire Holdco board.

Forum Selection

The Idera bylaws, to the fullest extent permitted by applicable law, designate the Court of Chancery of the State of Delaware as the sole and exclusive forum for certain legal actions unless Idera consents in writing to the selection of an alternative forum, subject to certain exceptions.

The BioCryst bylaws designate the Court of Chancery of the State of Delaware as the sole and exclusive forum for certain legal actions unless BioCryst consents in writing to the selection of an alternative forum, subject to certain exceptions.

The Holdco bylaws designate the Court of Chancery of the State of Delaware as the sole and exclusive forum for certain legal actions unless Holdco consents in writing to the selection of an alternative forum, subject to certain exceptions.

Any person or entity purchasing or otherwise acquiring any interest in shares of capital stock of Holdco shall be deemed to have notice of and consented to the forum selection provision.

NO APPRAISAL RIGHTS

Holders of Idera common stock and BioCryst common stock who dissent to the mergers will not have rights to an appraisal of the fair value of their shares. Under the DGCL, appraisal rights are not available for the shares of any class or series if the shares of the class or series are listed on a national securities exchange or held of record by more than 2,000 holders on the record date, unless the stockholders receive in exchange for their shares anything other than shares of stock of the surviving or resulting corporation or of any other corporation that is publicly listed or held by more than 2,000 holders of record, cash in lieu of fractional shares or fractional depositary receipts or any combination of the foregoing. Idera's and BioCryst's common stock is listed on the Nasdaq and Idera stockholders and BioCryst stockholders will receive a combination of shares of stock of Holdco, which will be listed on the Nasdaq, and cash in lieu of fractional shares.

LEGAL MATTERS

The validity of the shares of Holdco common stock to be issued pursuant to the mergers will be passed upon by Skadden. The U.S. federal income tax consequences of the mergers will be passed upon for BioCryst by Skadden and for Idera by Latham, respectively.

EXPERTS

Idera

The financial statements of Idera Pharmaceuticals, Inc. incorporated in this joint proxy statement / prospectus by reference to Idera Pharmaceuticals, Inc. Annual Report (Form 10-K) for the year ended December 31, 2017 have been audited by Ernst & Young LLP, independent registered public accounting firm, as set forth in their report thereon, included therein, and incorporated herein by reference. Such financial statements are incorporated herein by reference in reliance upon such report given on the authority of such firm as experts in accounting and auditing.

BioCryst

The consolidated financial statements of BioCryst Pharmaceuticals, Inc. incorporated in this joint proxy statement / prospectus by reference to BioCryst Pharmaceuticals, Inc. Annual Report (Form 10-K) for the year ended December 31, 2017, and any amendments thereto, have been audited by Ernst & Young LLP, independent registered public accounting firm, as set forth in their report thereon, included therein, and incorporated herein by reference. Such consolidated financial statements are incorporated herein by reference in reliance upon such report given on the authority of such firm as experts in accounting and auditing.

FUTURE STOCKHOLDER PROPOSALS

Idera

If the mergers are completed on the expected timetable, Idera will not hold a regular annual meeting in 2018. If, however, the mergers are not completed and Idera holds a regular annual meeting in 2018, in order to be considered for inclusion in the proxy statement and form of proxy for the 2018 annual meeting of stockholders, stockholder proposals must have been submitted in writing and received no later than January 8, 2018 in accordance with the requirements of 14a-8 of the Exchange Act. Stockholders desiring to bring business before the 2018 annual meeting of stockholders in a form other than a stockholder proposal in accordance with the preceding paragraph must give written notice to Idera's Secretary at Idera's principal office received at least 60 days, but not more than 90 days prior to the date of such meeting; provided, however, that if less than 70 days' notice or prior public disclosure of the date of such meeting is given to stockholders or made, such notice must be so

Table of Contents

received not later than the close of business on the 10th day following the day on which notice or public disclosure of the date of such meeting is given or made. The written notice must comply with the provisions of the Idera bylaws.

BioCryst

If the mergers are completed on the expected timetable, BioCryst will not hold a regular annual meeting in 2018. If, however, the mergers are not completed and BioCryst holds a regular annual meeting in 2018, proposals of stockholders intended to be presented at BioCryst's 2018 annual meeting of stockholders must have been received by BioCryst no later than December 13, 2017 to be considered for inclusion in BioCryst's proxy statement relating to such meeting. Proposals for inclusion in the proxy statement must comply with the Exchange Act, including Rule 14a-8.

A stockholder must notify BioCryst no earlier than January 24, 2018 and no later than February 23, 2018 of a proposal for the 2018 annual meeting which the stockholder intends to present other than by inclusion in the proxy material (including director nominations) and must include with the notification the information required by the BioCryst bylaws, in order to be eligible for consideration at the 2018 annual meeting. In accordance with the BioCryst bylaws, any stockholder entitled to vote for directors at an annual meeting may nominate persons for election as directors. Any such notice must also include the information regarding the stockholder making the nomination and the nominee required by the BioCryst bylaws.

OTHER MATTERS

Other Matters Presented at the Special Meetings

As of the date of this joint proxy statement/prospectus, neither the Idera board nor the BioCryst board knows of any matters that will be presented for consideration at either the Idera special meeting or the BioCryst special meeting other than as described in this joint proxy statement/prospectus. Under the Idera bylaws, Idera will transact no other business at the special meeting except such business as stated in the notice of special meeting contained herein. If any other matters come before the BioCryst special meeting or any adjournments or postponements thereof and shall be voted upon, the proposed proxy will be deemed to confer authority to the individuals named as authorized therein to vote the shares represented by the proxy as to any matters that fall within the purposes set forth in the notice of special meeting. It is intended that the persons named in the enclosed proxy and acting thereunder will vote in accordance with their best judgment on such matters.

WHERE YOU CAN FIND MORE INFORMATION

Idera and BioCryst each file annual, quarterly and current reports, proxy statements and other information with the SEC under the Exchange Act. You may read and copy any of this information at the SEC's Public Reference Room at 100 F Street, N.E., Washington, D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information on the Public Reference Room. The SEC also maintains an Internet website that contains reports, proxy and information statements, and other information regarding issuers, including Idera and BioCryst, who file electronically with the SEC. The address of that site is www.sec.gov.

Investors may also consult Idera or BioCryst's website for more information about Idera or BioCryst, respectively. Idera's website is www.iderapharma.com. BioCryst's website is www.biocryst.com.

Holdco has filed with the SEC a registration statement of which this joint proxy statement/prospectus forms a part. The registration statement registers the shares of Holdco common stock to be issued to Idera stockholders and BioCryst stockholders pursuant to the mergers. The registration statement, including the attached exhibits, contains additional relevant information about Holdco and Holdco common stock. The rules and regulations of the SEC allow Idera and BioCryst to omit certain information included in the registration statement from this joint proxy statement/prospectus.

In addition, the SEC allows Idera and BioCryst to disclose important information to you by referring you to other documents filed separately with the SEC. This information is considered to be a part of this joint proxy statement/prospectus.

This joint proxy statement/prospectus incorporates by reference the documents listed below that Idera has previously filed with the SEC (other than information furnished pursuant to Item 2.02 or Item 7.01 of a Current Report on Form 8-K). These documents contain important information about Idera, its financial condition or other matters.

Annual Report on Form 10-K for the fiscal year ended December 31, 2017.

Current Reports on Form 8-K filed on January 5, 2018 (of which there were two), January 17, 2018 and January 22, 2018 (other than the portions of those documents not deemed to be filed pursuant to the rules promulgated under the Exchange Act).

In addition, Idera incorporates by reference any future filings it makes with the SEC under Section 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date of this joint proxy statement/prospectus and prior to the date of the Idera special meeting (other than information furnished pursuant to Item 2.02 or Item 7.01 of any Current Report on Form 8-K, unless expressly stated otherwise therein). Such documents are considered to be a part of this joint proxy statement/prospectus, effective as of the date such documents are filed.

You can obtain any of these documents from the SEC, through the SEC's website at the address described above, or Idera will provide you with copies of these documents, without charge, upon written or oral request to:

Idera Pharmaceuticals, Inc. 167 Sidney Street Cambridge, Massachusetts 02139 Attention: Investor Relations 877-888-6550 ir@iderapharma.com

Table of Contents

This joint proxy statement/prospectus also incorporates by reference the documents listed below that BioCryst has previously filed with the SEC (other than information furnished pursuant to Item 2.02 or Item 7.01 of a Current Report on Form 8-K). These documents contain important information about BioCryst, its financial condition or other matters.

Annual Report on Form 10-K for the fiscal year ended December 31, 2017, and any amendments thereto.

Proxy Statement on Schedule 14A filed April 12, 2017.

Current Reports on Form 8-K filed on January 5, 2018, January 22, 2018 and March 15, 2018 (other than the portions of those documents not deemed to be filed pursuant to the rules promulgated under the Exchange Act).

In addition, BioCryst incorporates by reference any future filings it makes with the SEC under Section 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date of this joint proxy statement/prospectus and prior to the date of the BioCryst special meeting (other than information furnished pursuant to Item 2.02 or Item 7.01 of any Current Report on Form 8-K, unless expressly stated otherwise therein). Such documents are considered to be a part of this joint proxy statement/prospectus, effective as of the date such documents are filed.

You can obtain any of these documents from the SEC, through the SEC's website at the address described above, or BioCryst will provide you with copies of these documents, without charge, upon written or oral request to:

BioCryst Pharmaceuticals, Inc. 4505 Emperor Blvd., Suite 200 Durham, North Carolina 27703 Attention: Investor Relations

In the event of conflicting information in this joint proxy statement/prospectus in comparison to any document incorporated by reference into this joint proxy statement/prospectus, or among documents incorporated by reference, the information in the latest filed document controls.

You should rely only on the information contained or incorporated by reference into this joint proxy statement/prospectus. No one has been authorized to provide you with information that is different from that contained in, or incorporated by reference into, this joint proxy statement/prospectus. This joint proxy statement/prospectus is dated March 29, 2018. You should not assume that the information contained in this joint proxy statement/prospectus is accurate as of any date other than that date. You should not assume that the information incorporated by reference into this joint proxy statement/prospectus is accurate as of any date other than the date of such incorporated document. Neither our mailing of this joint proxy statement/prospectus to Idera stockholders or BioCryst stockholders nor the issuance by Holdco of shares of common stock pursuant to the mergers will create any implication to the contrary.

This joint proxy statement/prospectus contains a description of the representations and warranties that each of Idera and BioCryst made to the other in the merger agreement. Representations and warranties made by Idera, BioCryst and other applicable parties are also set forth in contracts and other documents (including the merger agreement) that are attached or filed as exhibits to this joint proxy statement/prospectus or are incorporated by reference into this joint proxy statement/prospectus. These materials are included or incorporated by reference only to provide you with information regarding the terms and conditions of the agreements, and not to provide any other factual information regarding Idera, BioCryst or their businesses. Accordingly, the representations and warranties and other provisions of the merger agreement should not be read alone, but instead should be read only in conjunction with the other information provided elsewhere in this joint proxy statement/prospectus or incorporated by reference into this joint proxy statement/prospectus.

Annex A Agreement and Plan of Merger, dated January 21, 2018

Table of Contents

AGREEMENT AND PLAN OF MERGER

among

BIOCRYST PHARMACEUTICALS, INC., IDERA PHARMACEUTICALS, INC., NAUTILUS HOLDCO, INC., ISLAND MERGER SUB, INC.

and

BOAT MERGER SUB, INC.

Dated as of January 21, 2018

TABLE OF CONTENTS

		Page
	<u>LE 1 THE MERGERS</u>	<u>A-1</u>
<u>1.1</u>		
	The Mergers	<u>A-1</u>
<u>1.2</u>	Closing	<u>A-2</u>
<u>1.3</u>	Effective Time	<u>A-2</u>
<u>1.4</u>	<u>Charters and Bylaws</u>	<u>A-2</u>
1.5	Directors and Officers of the Surviving Corporations	A-3
1.6	Directors and Officers of Holdco	A-3
1.7	Headquarters and Primary Research and Development Location	<u>A-4</u>
1.8	Tax Treatment	<u>A-4</u>
	LE 2 CONVERSION OF SECURITIES IN THE MERGERS	<u></u>
<u> </u>	DE 2 CONVERSION OF SECONDIES IN THE MERCERS	<u>A-4</u>
2.1		<u>A-4</u>
<u>2.1</u>	Effect on Conital Stock of Island and Doot	A 1
2.2	Effect on Capital Stock of Island and Boat	<u>A-4</u>
<u>2.2</u>	Effect on Holdco Common Stock	<u>A-5</u>
2.3	<u>Certain Adjustments</u>	<u>A-6</u>
<u>2.4</u>	<u>Fractional Shares</u>	<u>A-6</u>
<u>2.5</u>	Exchange of Certificates and Book-Entry Shares	<u>A-7</u>
<u>2.6</u>	<u>Island Warrants</u>	<u>A-9</u>
<u>2.7</u>	Stock Options and Stock-Based Awards	<u>A-10</u>
2.8	Withholding Rights	A-12
	LE 3 REPRESENTATIONS AND WARRANTIES OF ISLAND	
		<u>A-12</u>
3.1		<u> </u>
<u>J.1</u>	Corporate Organization	۸ 12
2.2		<u>A-12</u>
<u>3.2</u>	Island Capitalization	<u>A-13</u>
3.3	Authority; Execution and Delivery; Enforceability	<u>A-14</u>
<u>3.4</u>	No Conflicts	<u>A-15</u>
3.5	SEC Documents; Financial Statements; Undisclosed Liabilities	<u>A-15</u>
<u>3.6</u>	Absence of Certain Changes or Events	<u>A-17</u>
<u>3.7</u>	Information Supplied	<u>A-17</u>
<u>3.8</u>	<u>Legal Proceedings</u>	<u>A-17</u>
3.9	Compliance with Laws	<u>A-17</u>
3.10	<u>Permits</u>	<u>A-18</u>
3.11	Employee Benefit Plans	A-19
3.12	Employee and Labor Matters	A-20
3.13	Environmental Matters	A-21
3.14	Real Property	A-22
3.15	Tax Matters	A-22 A-22
3.16	Material Contracts	<u>A-24</u>
3.17	Intellectual Property	<u>A-25</u>
<u>3.18</u>	Healthcare Compliance Matters	<u>A-27</u>
3.19	Broker's Fees	<u>A-28</u>
<u>3.20</u>	Opinion of Financial Advisor	<u>A-28</u>
<u>3.21</u>	Ownership of Boat Common Stock	<u>A-29</u>
3.22	Insurance	<u>A-29</u>
3.23	No Other Representations or Warranties	A-29
	LE 4 REPRESENTATIONS AND WARRANTIES OF BOAT	
		<u>A-29</u>
<u>4.1</u>		
	Corporate Organization	<u>A-30</u>
	A-i	<u> 11-30</u>
	A-1	

		Page
<u>4.2</u>	Boat Capitalization	<u>A-30</u>
<u>4.3</u>	Authority; Execution and Delivery; Enforceability	<u>A-31</u>
<u>4.4</u>	No Conflicts	<u>A-32</u>
<u>4.5</u>	SEC Documents; Financial Statements; Undisclosed Liabilities	<u>A-32</u>
<u>4.6</u>	Absence of Certain Changes or Events	<u>A-34</u>
<u>4.7</u>	<u>Information Supplied</u>	<u>A-34</u>
4.8	<u>Legal Proceedings</u>	<u>A-34</u>
<u>4.9</u>	<u>Compliance with Laws</u>	<u>A-35</u>
4.10	<u>Permits</u>	<u>A-35</u>
4.11	Employee Benefit Plans	<u>A-36</u>
4.12	Employee and Labor Matters	<u>A-38</u>
4.13	Environmental Matters	<u>A-39</u>
4.14	Real Property To Marie 1	<u>A-39</u>
4.15	Tax Matters Maria I G	<u>A-40</u>
4.16	Material Contracts Let U. d.	<u>A-42</u>
4.17	Intellectual Property Hardele and Caraclinea Matters	<u>A-43</u>
4.18	Healthcare Compliance Matters Productive Faces	<u>A-45</u>
<u>4.19</u>	Broker's Fees Opinion of Financial Advisor	<u>A-46</u>
<u>4.20</u> <u>4.21</u>	Ownership of Island Common Stock	<u>A-47</u> <u>A-47</u>
4.21 4.22	Insurance	<u>A-47</u> <u>A-47</u>
<u>4.22</u> <u>4.23</u>	Holdco, Merger Sub A and Merger Sub B	A-47 A-47
	No Other Representations or Warranties	A-47 A-47
	E 5 COVENANTS	<u> 11-47</u>
MATICE	ES COVERNIATE	<u>A-48</u>
<u>5.1</u>		
	Conduct of Business by Island Pending the Effective Time	<u>A-48</u>
<u>5.2</u>	Conduct of Business by Boat and its Subsidiaries Pending the Effective Time	<u>A-50</u>
5.3	Preparation of the Form S-4 and the Joint Proxy Statement; Stockholders Meetings	<u>A-53</u>
5.4	No Solicitation of Transactions	A-56
<u>5.5</u>	Access to Information; Confidentiality	<u>A-60</u>
<u>5.6</u>	Appropriate Action	<u>A-60</u>
<u>5.7</u>	<u>Certain Notices</u>	<u>A-62</u>
<u>5.8</u>	<u>Public Announcements</u>	<u>A-62</u>
<u>5.9</u>	<u>Director and Officer Indemnification</u>	<u>A-62</u>
<u>5.10</u>	Stock Exchange Listing	<u>A-63</u>
<u>5.11</u>	Section 16 Matters	<u>A-64</u>
<u>5.12</u>	Stockholder Litigation	<u>A-64</u>
<u>5.13</u>	<u>Tax Matters</u>	<u>A-64</u>
<u>5.14</u>	Employee Matters	<u>A-65</u>
	Cooperation	<u>A-66</u>
<u>5.16</u>	Financing Matters	<u>A-66</u>
<u>5.17</u>	Stock Exchange Delisting and Deregistration	<u>A-67</u>
ARTICLI	E 6 CONDITIONS TO CONSUMMATION OF THE MERGERS	A 65
6 1		<u>A-67</u>
<u>6.1</u>	Conditions to Obligations of Each Party Under This Agreement	۸ 67
6.2	Conditions to Obligations of Boat and Holdco Under This Agreement	<u>A-67</u> A-67
6.3	Conditions to Obligations of Island Under This Agreement	A-68
	E 7 TERMINATION, AMENDMENT AND WAIVER	11-00
		<u>A-69</u>
<u>7.1</u>		
	Termination	<u>A-69</u>
	A-ii	

Table of Contents

		Page
<u>7.2</u>	Effect of Termination	<u>A-70</u>
<u>7.3</u>	<u>Termination Fee; Expenses</u>	<u>A-70</u>
<u>7.4</u>	Amendment or Supplement	<u>A-72</u>
<u>7.5</u>	Extension of Time; Waiver	<u>A-72</u>
ARTICL	E 8 GENERAL PROVISIONS	
		<u>A-73</u>
<u>8.1</u>		
	Non-Survival of Representations and Warranties	<u>A-73</u>
<u>8.2</u>	Notices	<u>A-73</u>
<u>8.3</u>	<u>Certain Definitions</u>	<u>A-74</u>
<u>8.4</u>	<u>Headings</u>	<u>A-80</u>
<u>8.5</u>	Severability	<u>A-81</u>
8.5 8.6	Entire Agreement	<u>A-81</u>
<u>8.7</u>	Assignment	<u>A-81</u>
<u>8.8</u>	<u>Parties in Interest</u>	<u>A-81</u>
<u>8.9</u>	Mutual Drafting: Interpretation	<u>A-81</u>
<u>8.10</u>	Governing Law; Consent to Jurisdiction; Waiver of Trial by Jury	<u>A-81</u>
<u>8.11</u>	<u>Counterparts</u>	<u>A-82</u>
<u>8.12</u>	Delivery by Facsimile or Email	<u>A-82</u>
<u>8.13</u>	Specific Performance	<u>A-82</u>
<u>8.14</u>	<u>Disclosure Schedules</u>	<u>A-83</u>
Exhibit A	Form of Certificate of Incorporation of the Island Surviving Corporation	
Exhibit B	Form of Certificate of Incorporation of the Boat Surviving Corporation	
Exhibit C	Form of Amended and Restated Certificate of Incorporation of Holdco	

Exhibit D Form of Boat Tax Representation Letter Form of Island Tax Representation Letter Exhibit E

A-iii

GLOSSARY OF DEFINED TERMS

Acceptable Confidentiality Agreement Section 8.3 affiliate Section 8.3 Agreement Preamble Amended and Restated Holdco Charter Section 1.4(c) Anti-corruption Laws Section 8.3 Section 8.3 Section 8.3 Antitrust Restraint Section 5.6(c) associate Section 8.3 Section 8.3 beneficial ownership Section 9.3(c)(ii) Boat Benefit Plan Section 9.3(c)(iii) Boat Benefit Plan Section 9.3(c)(iii) Boat Boat Board Section 9.3(c)(iii) Section 9.3(c)(iii) Boat Boat Board Section 9.3(c)(iii) Section 9.3(c)(c)(iii) Section 9.3(c)(c)(c)(c)(c) Section 9.3(c)(c)(c) Section 9.3(c)(c) Section 9.3(c)(c) Section 9.3(c)(c) Section 9.3(c)(c) Section 9.3(c)	Defined Term	Section
Agreement Preamble Amneded and Restated Holdco Charter Section 1.4(c) Anti-corruption Laws Section 5.6(c) associate Section 8.3 beneficial ownership Section 8.3 Boat Preamble Boat Adverse Recommendation Change Section 5.3(c)(ii) Boat Benefit Plan Section 4.11(a) Boat Board Section 4.11(a) Boat Board Section 4.12(a) Boat Business Personnel Section 4.12(a) Boat Bysies Section 4.12(a) Boat Certificate of Merger Section 2.1(b)(iii) Boat Certificate of Merger Section 2.1(b)(iii) Boat Common Stock Section 2.1(b) Boat Disclosure Schedule Section 2.1(b) Boat ExpP Section 2.1(b)(ii) Boat ExpP Section 2.1(b)(ii) Boat ExpP (Section 2.1(b)(iii) Section 2.1(b)(iii) Boat Hacrial Contracts Section 2.1(b)(iii) Boat Material Contracts Section 4.17(d) Boat Material Intellectual Property Section 4.17(d) Boat Merger Section 4.10(c)	Acceptable Confidentiality Agreement	Section 8.3
Amended and Restated Holdco Charter Section 1.4(c) Anti-corruption Laws Section 5.6(c) Antitrust Restraint Section 5.6(c) seneficial ownership Section 8.3 beneficial ownership Section 5.3(c)(ii) Boat Preamble Boat Adverse Recommendation Change Section 5.3(c)(ii) Boat Board Section 4.11(a) Boat Board Section 4.10(bii) Boat Board Board Section 2.1(b)(iii) Section 4.12(a) Boat Board Board Section 4.12(a) Section 4.12(a) Boat Board Splaws Section 4.10(iii) Boat Entir Splaws Section 4.10(iii) Boat Certificate of Merger Section 1.3(a) Boat Charter Section 4.1 Boat Charter Section 4.1 Boat Charter Section 4.1 Boat Charter Section 4.1 Section 4.1 Boat Disclosure Schedule Section 4.1	affiliate	Section 8.3
Anti-corruption Laws Section 5.6(c) Antitust Restraint Section 5.6(c) associate Section 8.3 beneficial ownership Section 8.3 Boat Preamble Boat Adverse Recommendation Change Section 5.3(c)(ii) Boat Benefit Plan Section 4.1(la) Boat Board Section 4.3(b) Boat Book-Entry Shares Section 4.12(a) Boat Bylaws Section 4.12(a) Boat Certificate Section 4.1 Boat Certificate of Merger Section 4.1 Boat Common Stock Section 4.1 Boat Disclosure Schedule Section 2.7(d)(ii) Boat ESPP Section 2.7(d)(ii) Boat Eschange Ratio Section 2.7(d)(ii) Boat Foreign Benefit Plan Section 4.16(c) Boat Material Adverse Effect Section 8.3 Boat Material Contracts Section 4.16(c) Boat Material Intellectual Property Section 8.3 Boat Merger Section 4.16(c) Boat Merger Consideration Section 4.16(c) Boat Recommendation Section 8.3 B	Ü	Preamble
Antitrust Restraint Section 5.6(c) associate associate Section 8.3 Boat Boat Preamble Boat Adverse Recommendation Change Section 5.3(c)(ii) Boat Benefit Plan Section 4.11(a) Boat Board Section 4.2(b) Boat Board Section 2.1(b)(iii) Boat Bosk-Entry Shares Section 2.1(b)(iii) Boat Business Personnel Section 4.12(a) Boat Bylaws Section 4.1 Boat Certificate of Merger Section 1.3(a) Boat Certificate of Merger Section 1.3(a) Boat Charter Section 2.1(b)(iii) Boat Common Stock Section 2.1(b) Boat Disclosure Schedule Section 2.1(b) Boat ESPP Section 2.7(d)(ii) Boat Exchange Ratio Section 2.7(d)(ii) Boat Esper Section 2.1(b)(ii) Boat Adverse Effect Section 2.1(b)(ii) Boat Adverse Retio Section 2.1(b)(ii) Boat Adverse Retio Section 2.1(b)(ii) Boat Material Adverse Effect Section 4.1(i) Boat Material Adverse Effect Section 2.1(b)(ii)	Amended and Restated Holdco Charter	Section 1.4(c)
associate Section 8.3 beneficial ownership Section 8.3 Boat Preamble Boat Adverse Recommendation Change Section 5.3(c)(ii) Boat Benefit Plan Section 4.11(a) Boat Board Section 4.11(b)(iii) Boat Board Section 4.12(a) Section 2.1(b)(iii) Boat Business Personnel Section 4.12(a) Boat Bylaws Section 4.12(a) Boat Certificate Section 2.1(b)(iii) Boat Certificate of Merger Section 1.3(a) Boat Charter Section 2.1(b)(iii) Boat Common Stock Section 2.1(b) Boat Common Stock Section 2.1(b) Boat Eschange Ratio Section 2.1(b)(ii) Boat Eschange Ratio Section 2.1(b)(ii) Boat Eschange Ratio Section 2.1(b)(ii) Boat Leased Real Property Section 2.1(b)(ii) Boat Material Adverse Effect Section 2.1(b)(ii) Boat Material Contracts Section 4.1(a) Boat Material Contracts Section 4.1(a) Boat Merger Consideration Section 4.1(a) Boat Merger Consideration Sec	Anti-corruption Laws	Section 8.3
beneficial ownership Section 8.3 Boat Preamble Boat Adverse Recommendation Change Section 5.3(c)(ii) Boat Benefit Plan Section 4.11(a) Boat Board Section 4.1(b)(iii) Boat Bosh-Entry Shares Section 2.1(b)(iii) Boat Business Personnel Section 4.12(a) Boat Bylaws Section 4.1 Boat Certificate Section 2.1(b)(iii) Boat Certificate of Merger Section 1.3(a) Boat Common Stock Section 2.1(b) Boat Disclosure Schedule Section 2.1(b) Boat ESPP Section 2.7(d)(ii) Boat EsPP Section 2.7(d)(ii) Boat Foreign Benefit Plan Section 4.11(a) Boat Material Adverse Effect Section 8.3 Boat Material Adverse Effect Section 8.3 Boat Material Intellectual Property Section 4.17(d) Boat Merger Section 4.17(d) Boat Registered Intellectual Property Section 4.17(d) Boat Registered Intellectual Property Section 4.17(d) Boat SEC Documents Section 8.3 Boat SSC Documents	Antitrust Restraint	Section 5.6(c)
Boat Adverse Recommendation Change Section 5.3(c)(ii) are section 5.3(c)(iii) boat Benefit Plan Section 4.1(a) Section 4.3(b) Section 4.3(b) Boat Board Section 4.3(b) Section 4.3(b) Boat Book-Entry Shares Section 2.1(b)(iii) Boat Business Personnel Section 2.1(b)(iii) Boat Business Personnel Section 4.12(a) Section 4.12(a) Section 4.1 Boat Certificate of Merger Section 4.1 Section 4.1 Boat Certificate of Merger Section 1.3(a) Section 1.3(a) Boat Charter Section 1.3(a) Section 1.3(a) Section 2.1(b) Se	associate	Section 8.3
Boat Adverse Recommendation Change Section 5.3(c)(ii) Boat Benefit Plan Section 4.11(a) Boat Board Section 4.3(b) Boat Board Section 4.3(b) Boat Board Section 2.1(b)(iii) Boat Board Section 4.12(a) Boat Board Section 4.12(a) Boat Board Section 4.12(a) Boat Certificate Section 4.16(b)(iii) Boat Common Stock Section 1.3(a) Boat Common Stock Section 2.1(b) Boat Disclosure Schedule Section 2.1(b) Boat Eschange Ratio Section 2.1(b)(ii) Boat Foreign Benefit Plan Section 2.1(b)(ii) Boat Foreign Benefit Plan Section 4.16(c) Boat Material Adverse Effect Section 8.3 Boat Material Contracts Section 4.16(c) Boat Material Intellectual Property Section 4.17(d) Boat Merger Section 4.17(d) Boat Merger Consideration Section 4.16(c) Boat Merger Consideration Section 4.16(c) Boat Registered Intellectual Property Section 8.3 Boat SEC Documents Sect	beneficial ownership	Section 8.3
Boat Benefit Plan Section 4.1(a) Boat Board Section 4.3(b) Boat Book-Entry Shares Section 2.1(b)(iii) Boat Business Personnel Section 4.12(a) Boat Bylaws Section 4.1 Boat Certificate Section 2.1(b)(iii) Boat Certificate of Merger Section 1.3(a) Boat Common Stock Section 2.1(b) Boat Disclosure Schedule Section 2.1(b)(ii) Boat ESPP Section 2.1(b)(ii) Boat Esparage Ratio Section 2.1(b)(ii) Boat Foreign Benefit Plan Section 2.1(b)(ii) Boat Foreign Benefit Plan Section 8.3 Boat Material Adverse Effect Section 8.3 Boat Material Contracts Section 4.16(c) Boat Material Intellectual Property Section 4.17(d) Boat Merger Section 1.1(a) Boat Merger Consideration Section 2.1(b)(ii) Boat Registered Intellectual Property Section 4.3(b) Boat Registered Intellectual Property Section 4.3(b) Boat SEC Financial Statements Section 4.5(c) Boat SEC Financial Statements Section 5.3(c)(i)	Boat	Preamble
Boat Board Section 4.1(b) Boat Book-Entry Shares Section 2.1(b)(iii) Boat Business Personnel Section 4.12(a) Boat Cartificate Section 2.1(b)(iii) Boat Certificate of Merger Section 1.3(a) Boat Charter Section 1.3(a) Boat Common Stock Section 2.1(b) Boat Disclosure Schedule Section 2.1(b)(ii) Boat ESPP Section 2.7(d)(ii) Boat Foreign Benefit Plan Section 2.1(b)(ii) Boat Foreign Benefit Plan Section 4.11(j) Boat Material Adverse Effect Section 8.3 Boat Material Contracts Section 4.16(c) Boat Material Intellectual Property Section 4.17(d) Boat Merger Section 4.17(d) Boat Merger Consideration Section 2.1(b)(ii) Boat Recommendation Section 2.1(b)(ii) Boat Registered Intellectual Property Section 8.3 Boat RSU Section 9.2(c)(i) Boat SEC Documents Section 4.5(a) Boat SEC Documents Section 4.5(a) Boat Stock Option Section 8.3 Boat Stock Plan	Boat Adverse Recommendation Change	Section 5.3(c)(ii)
Boat Book-Entry SharesSection 2.1(b)(iii)Boat Business PersonnelSection 4.12(a)Boat BylawsSection 4.1Boat CertificateSection 2.1(b)(iii)Boat Certificate of MergerSection 1.3(a)Boat CharterSection 4.1Boat Common StockSection 2.1(b)Boat Disclosure ScheduleSection 2.7(d)(ii)Boat ESPPSection 2.7(d)(ii)Boat Eschange RatioSection 2.1(b)(ii)Boat Foreign Benefit PlanSection 4.11(j)Boat Leased Real PropertySection 8.3Boat Material Adverse EffectSection 8.3Boat Material ContractsSection 4.16(c)Boat Material Intellectual PropertySection 4.17(d)Boat MergerSection 1.1(a)Boat Merger ConsiderationSection 2.1(b)(ii)Boat Registered Intellectual PropertySection 2.1(b)(ii)Boat Registered Intellectual PropertySection 8.3Boat RSUSection 2.7(c)Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(a)Boat Stock OptionSection 4.5(a)Boat Stock PlanSection 5.3(c)(i)Boat Stockholder AprovalSection 7.3(b)Boat Surviving CorporationSection 7.3(b)Boat Surviving CorporationSection 7.3(b)Boat Surviving CorporationSection 8.3Coat Stock OptionSection 7.3(b)Boat Surviving CorporationSection 8.3Coat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization D	Boat Benefit Plan	Section 4.11(a)
Boat Business PersonnelSection 4.12(a)Boat BylawsSection 2.1(b)(iii)Boat CertificateSection 1.3(a)Boat CharterSection 1.3(a)Boat Common StockSection 2.1(b)Boat Disclosure ScheduleSection 2.1(b)Boat ESPPSection 2.7(d)(ii)Boat ESPPSection 2.1(b)(ii)Boat Foreign Benefit PlanSection 2.1(b)(ii)Boat Leased Real PropertySection 8.3Boat Material Adverse EffectSection 8.3Boat Material Adverse EffectSection 4.16(c)Boat Material Intellectual PropertySection 4.16(d)Boat MergerSection 4.17(d)Boat Merger ConsiderationSection 2.1(b)(ii)Boat Registered Intellectual PropertySection 4.3(b)Boat Registered Intellectual PropertySection 8.3Boat SEC DocumentsSection 9.3(b)Boat SEC Financial StatementsSection 4.5(a)Boat Stock OptionSection 9.2(b)Boat Stock OptionSection 8.3Boat Stockholder ApprovalSection 8.3Boat Stockholder MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 5.3(c)(i)Boat Surviving CorporationSection 1.1(a)Boat Tax OpinionSection 7.3(b)Boat Tax OpinionSection 9.3(d)Certificates of MergerSection 1.3(a)ChosingSection 1.3(a)ChosingSection 1.3(a)Chosing Bid PriceSection 1.2Closing Bid PriceSection 1.2Closing Bid PriceSection 1.2 <td>Boat Board</td> <td>Section 4.3(b)</td>	Boat Board	Section 4.3(b)
Boat Bylaws Section 4.1 Boat Certificate Section 2.1(b)(iii) Boat Certificate of Merger Section 1.3(a) Boat Charter Section 4.1 Boat Common Stock Section 2.1(b) Boat Disclosure Schedule Section 2.7(d)(ii) Boat ESPP Section 2.7(d)(ii) Boat Exchange Ratio Section 2.1(b)(ii) Boat Exchange Ratio Section 2.1(b)(ii) Boat Foreign Benefit Plan Section 2.1(b)(ii) Boat Leased Real Property Section 8.3 Boat Material Adverse Effect Section 8.3 Boat Material Intellectual Property Section 4.17(d) Boat Merger Section 4.17(d) Boat Merger Consideration Section 4.17(d) Boat Recommendation Section 2.7(b)(ii) Boat Registered Intellectual Property Section 8.3 Boat Registered Intellectual Property Section 8.3 Boat SEC Documents Section 4.5(c) Boat SEC Documents Section 4.5(c) Boat Stock Option Section 8.3 Boat Stock Plan Section 8.3 Boat Stockholder Aproval	Boat Book-Entry Shares	Section 2.1(b)(iii)
Boat Certificate Section 2.1(b)(iii) Boat Certificate of Merger Section 1.3(a) Boat Charter Section 4.1 Boat Common Stock Section 8.3 Boat ESPP Section 2.7(d)(ii) Boat ESPP Section 2.7(d)(iii) Boat Exchange Ratio Section 2.1(b)(ii) Boat Foreign Benefit Plan Section 4.11(j) Boat Leased Real Property Section 8.3 Boat Material Adverse Effect Section 8.3 Boat Material Contracts Section 4.16(c) Boat Material Intellectual Property Section 4.17(d) Boat Merger Section 4.17(d) Boat Merger Consideration Section 4.17(d) Boat Recommendation Section 4.17(d) Boat Recommendation Section 4.3(b) Boat REGU Section 8.3 Boat SEC Documents Section 8.3 Boat SEC Financial Statements Section 4.5(c) Boat Stock Option Section 9.3 Boat Stock Plan Section 9.3 Boat Stockholder Approval Section 9.3 Boat Stockholders Meeting Section 9.3 <tr< td=""><td>Boat Business Personnel</td><td>Section 4.12(a)</td></tr<>	Boat Business Personnel	Section 4.12(a)
Boat Certificate of MergerSection 1.3(a)Boat CharterSection 4.1Boat Common StockSection 2.1(b)Boat Disclosure ScheduleSection 8.3Boat ESPPSection 2.7(d)(ii)Boat ESPPSection 2.7(b)(ii)Boat ESPPSection 2.1(b)(ii)Boat Foreign Benefit PlanSection 4.11(j)Boat Leased Real PropertySection 8.3Boat Material Adverse EffectSection 8.3Boat Material Intellectual PropertySection 4.17(d)Boat Material Intellectual PropertySection 4.17(d)Boat MergerSection 4.17(d)Boat RecommendationSection 2.1(b)(ii)Boat Registered Intellectual PropertySection 8.3Boat RSUSection 2.7(c)Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 9.3Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 5.3(c)(i)Boat Surviving CorporationSection 5.3(c)(i)Boat Surviving CorporationSection 1.1(a)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 1.3(a)Chosen CourtsSection 1.3(a)Chosen CourtsSection 1.2ClosingSection 1.2Closing Bid PriceSection 1.2Closing Bid PriceSection 1.2	Boat Bylaws	Section 4.1
Boat Charter Section 4.1 Boat Common Stock Section 2.1(b) Boat Disclosure Schedule Section 8.3 Boat ESPP Section 2.7(d)(ii) Boat ESPP Section 2.1(b)(ii) Boat Foreign Benefit Plan Section 4.11(j) Boat Leased Real Property Section 8.3 Boat Material Adverse Effect Section 8.3 Boat Material Contracts Section 4.16(c) Boat Material Intellectual Property Section 4.17(d) Boat Merger Section 4.17(d) Boat Merger Consideration Section 2.1(b)(ii) Boat Recommendation Section 2.1(b)(ii) Boat Registered Intellectual Property Section 2.7(c) Boat SEC Documents Section 2.7(c) Boat SEC Financial Statements Section 4.5(c) Boat SEC Financial Statements Section 4.5(c) Boat Stock Option Section 9.3 Boat Stock Plan Section 9.3 Boat Stock Plan Section 9.3 Boat Stockholder Approval Section 1.1(a) Boat Surviving Corporation Section 1.2 Boat Surviving Corporation	Boat Certificate	Section 2.1(b)(iii)
Boat Common StockSection 2.1(b)Boat Disclosure ScheduleSection 8.3Boat ESPPSection 2.7(d)(ii)Boat Exchange RatioSection 2.1(b)(ii)Boat Foreign Benefit PlanSection 4.11(j)Boat Leased Real PropertySection 8.3Boat Material Adverse EffectSection 8.3Boat Material Intellectual PropertySection 4.16(c)Boat Material Intellectual PropertySection 4.17(d)Boat MergerSection 4.17(d)Boat Merger ConsiderationSection 2.1(b)(ii)Boat RecommendationSection 4.3(b)Boat Registered Intellectual PropertySection 8.3Boat SEC DocumentsSection 2.7(c)Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 4.5(c)Boat Stock OptionSection 8.3Boat Stockholder ApprovalSection 8.3Boat Stockholders MeetingSection 4.3(b)Boat Tax OpinionSection 1.1(a)Boat Termination FeeSection 1.1(a)Boat Termination FeeSection 8.3Capitalization DateSection 8.3Capitalization DateSection 8.3Chosen CourtsSection 8.3ClosingSection 1.12Closing Bid PriceSection 8.3Closing DateSection 8.3	Boat Certificate of Merger	Section 1.3(a)
Boat Disclosure Schedule Section 8.3 Boat ESPP Section 2.7(d)(ii) Boat Exchange Ratio Section 2.1(b)(ii) Boat Foreign Benefit Plan Section 4.11(j) Boat Leased Real Property Section 8.3 Boat Material Adverse Effect Section 8.3 Boat Material Contracts Section 4.16(c) Boat Material Intellectual Property Section 4.17(d) Boat Merger Section 4.17(d) Boat Merger Consideration Section 2.1(b)(ii) Boat Recommendation Section 4.3(b) Boat Registered Intellectual Property Section 8.3 Boat SEU Section 9.3(c) Boat SEC Documents Section 4.5(a) Boat SEC Financial Statements Section 4.5(c) Boat Stock Option Section 4.5(c) Boat Stock Plan Section 8.3 Boat Stockholder Approval Section 9.3(c) Boat Stockholders Meeting Section 9.3(c) Boat Tax Opinion Section 1.1(a) Boat Termination Fee Section 9.3(c) Boat Termination Fee Section 8.3 Boat Termination Fee	Boat Charter	Section 4.1
Boat ESPPSection 2.7(d)(ii)Boat Exchange RatioSection 2.1(b)(ii)Boat Foreign Benefit PlanSection 4.11(j)Boat Leased Real PropertySection 8.3Boat Material Adverse EffectSection 8.3Boat Material ContractsSection 4.16(c)Boat Material Intellectual PropertySection 4.17(d)Boat MergerSection 1.1(a)Boat Merger ConsiderationSection 2.1(b)(ii)Boat RecommendationSection 4.3(b)Boat Registered Intellectual PropertySection 8.3Boat SEC DocumentsSection 2.7(c)Boat SEC Financial StatementsSection 4.5(a)Boat Stock OptionSection 4.5(c)Boat Stock PlanSection 7.3(b)Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 4.3(c)Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 5.3(c)(i)Boat Tax OpinionSection 5.3(c)(i)Boat Tax OpinionSection 8.3Capitalization DateSection 7.3(b)Capitalization DateSection 8.3Capitalization DateSection 8.10(b)ClosingSection 8.10(b)Closing Bid PriceSection 8.3Closing Bid PriceSection 8.3Closing DateSection 8.3	Boat Common Stock	Section 2.1(b)
Boat Exchange RatioSection 2.1(b)(ii)Boat Foreign Benefit PlanSection 4.11(j)Boat Leased Real PropertySection 8.3Boat Material Adverse EffectSection 8.3Boat Material ContractsSection 4.16(c)Boat Material Intellectual PropertySection 4.17(d)Boat MergerSection 1.1(a)Boat Merger ConsiderationSection 2.1(b)(ii)Boat Registered Intellectual PropertySection 4.3(b)Boat Registered Intellectual PropertySection 8.3Boat RSUSection 2.7(c)Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 2.7(b)Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 8.3Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 1.1(a)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 8.3Certificates of MergerSection 8.10(b)ClosingSection 8.10(b)Closing Bid PriceSection 8.3Closing DateSection 8.3	Boat Disclosure Schedule	Section 8.3
Boat Foreign Benefit PlanSection 4.11(j)Boat Leased Real PropertySection 8.3Boat Material Adverse EffectSection 8.3Boat Material ContractsSection 4.16(c)Boat Material Intellectual PropertySection 4.17(d)Boat MergerSection 1.1(a)Boat Merger ConsiderationSection 2.1(b)(ii)Boat RecommendationSection 2.1(b)(ii)Boat Registered Intellectual PropertySection 8.3Boat RSUSection 2.7(c)Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 2.7(b)Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 8.3Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 5.3(c)(i)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 8.3Capitalization DateSection 3.2(a)Chosen CourtsSection 8.10(b)ClosingSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 8.3	Boat ESPP	Section 2.7(d)(ii)
Boat Foreign Benefit PlanSection 4.11(j)Boat Leased Real PropertySection 8.3Boat Material Adverse EffectSection 8.3Boat Material ContractsSection 4.16(c)Boat Material Intellectual PropertySection 4.17(d)Boat MergerSection 1.1(a)Boat Merger ConsiderationSection 2.1(b)(ii)Boat Registered Intellectual PropertySection 4.3(b)Boat Registered Intellectual PropertySection 8.3Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 2.7(b)Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 8.3Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 5.3(c)(i)Boat Tax OpinionSection 1.1(a)Boat Tax Opinion PeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 8.3	Boat Exchange Ratio	Section 2.1(b)(ii)
Boat Leased Real PropertySection 8.3Boat Material Adverse EffectSection 8.3Boat Material ContractsSection 4.16(c)Boat Material Intellectual PropertySection 4.17(d)Boat MergerSection 1.1(a)Boat Merger ConsiderationSection 2.1(b)(ii)Boat RecommendationSection 4.3(b)Boat Registered Intellectual PropertySection 8.3Boat RSUSection 2.7(c)Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 2.7(b)Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 8.3Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 5.3(c)(i)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 8.3Capitalization DateSection 8.10(b)ClosingSection 8.10(b)ClosingSection 8.10(b)Closing Bid PriceSection 8.3Closing DateSection 8.3		
Boat Material ContractsSection 4.16(c)Boat Material Intellectual PropertySection 4.17(d)Boat MergerSection 1.1(a)Boat Merger ConsiderationSection 2.1(b)(ii)Boat RecommendationSection 4.3(b)Boat Registered Intellectual PropertySection 8.3Boat RSUSection 2.7(c)Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 2.7(b)Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 4.3(c)Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 1.1(a)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 8.3	-	
Boat Material Intellectual PropertySection 4.17(d)Boat MergerSection 1.1(a)Boat Merger ConsiderationSection 2.1(b)(ii)Boat RecommendationSection 4.3(b)Boat Registered Intellectual PropertySection 8.3Boat RSUSection 2.7(c)Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 2.7(b)Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 4.3(c)Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 1.1(a)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 8.3Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 8.3	Boat Material Adverse Effect	Section 8.3
Boat MergerSection 1.1(a)Boat Merger ConsiderationSection 2.1(b)(ii)Boat RecommendationSection 4.3(b)Boat Registered Intellectual PropertySection 8.3Boat RSUSection 2.7(c)Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 2.7(b)Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 4.3(c)Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 1.1(a)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 8.3Closing Bid PriceSection 8.3Closing DateSection 8.3	Boat Material Contracts	Section 4.16(c)
Boat Merger ConsiderationSection 2.1(b)(ii)Boat RecommendationSection 4.3(b)Boat Registered Intellectual PropertySection 8.3Boat RSUSection 2.7(c)Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 2.7(b)Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 8.3Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 1.1(a)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 8.3	Boat Material Intellectual Property	Section 4.17(d)
Boat RecommendationSection 4.3(b)Boat Registered Intellectual PropertySection 8.3Boat RSUSection 2.7(c)Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 2.7(b)Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 4.3(c)Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 1.1(a)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 8.3	Boat Merger	Section 1.1(a)
Boat Registered Intellectual PropertySection 8.3Boat RSUSection 2.7(c)Boat SEC DocumentsSection 4.5(a)Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 2.7(b)Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 4.3(c)Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 1.1(a)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 8.3	Boat Merger Consideration	Section 2.1(b)(ii)
Boat RSU Section 2.7(c) Boat SEC Documents Section 4.5(a) Boat SEC Financial Statements Section 4.5(c) Boat Stock Option Section 2.7(b) Boat Stock Plan Section 8.3 Boat Stockholder Approval Section 4.3(c) Boat Stockholders Meeting Section 5.3(c)(i) Boat Surviving Corporation Section 1.1(a) Boat Tax Opinion Section 6.2(d) Boat Termination Fee Section 7.3(b) Business Day Section 8.3 Capitalization Date Section 3.2(a) Certificates of Merger Section 1.3(a) Chosen Courts Section 8.10(b) Closing Section 1.2 Closing Bid Price Section 8.3 Closing Date Section 1.2	Boat Recommendation	Section 4.3(b)
Boat RSU Section 2.7(c) Boat SEC Documents Section 4.5(a) Boat SEC Financial Statements Section 4.5(c) Boat Stock Option Section 2.7(b) Boat Stock Plan Section 8.3 Boat Stockholder Approval Section 4.3(c) Boat Stockholders Meeting Section 5.3(c)(i) Boat Surviving Corporation Section 1.1(a) Boat Tax Opinion Section 6.2(d) Boat Termination Fee Section 7.3(b) Business Day Section 8.3 Capitalization Date Section 3.2(a) Certificates of Merger Section 1.3(a) Chosen Courts Section 8.10(b) Closing Section 1.2 Closing Bid Price Section 8.3 Closing Date Section 1.2	Boat Registered Intellectual Property	Section 8.3
Boat SEC Financial StatementsSection 4.5(c)Boat Stock OptionSection 2.7(b)Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 4.3(c)Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 1.1(a)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 1.2		Section 2.7(c)
Boat Stock Option Section 2.7(b) Boat Stock Plan Section 8.3 Boat Stockholder Approval Section 4.3(c) Boat Stockholders Meeting Section 5.3(c)(i) Boat Surviving Corporation Section 1.1(a) Boat Tax Opinion Section 6.2(d) Boat Termination Fee Section 7.3(b) Business Day Section 8.3 Capitalization Date Section 3.2(a) Certificates of Merger Section 1.3(a) Chosen Courts Section 8.10(b) Closing Section 1.2 Closing Bid Price Section 8.3 Closing Date Section 1.2	Boat SEC Documents	Section 4.5(a)
Boat Stock PlanSection 8.3Boat Stockholder ApprovalSection 4.3(c)Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 1.1(a)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 1.2	Boat SEC Financial Statements	Section 4.5(c)
Boat Stockholder ApprovalSection 4.3(c)Boat Stockholders MeetingSection 5.3(c)(i)Boat Surviving CorporationSection 1.1(a)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 1.2	Boat Stock Option	Section 2.7(b)
Boat Stockholders Meeting Section 5.3(c)(i) Boat Surviving Corporation Section 1.1(a) Boat Tax Opinion Section 6.2(d) Boat Termination Fee Section 7.3(b) Business Day Section 8.3 Capitalization Date Section 3.2(a) Certificates of Merger Section 1.3(a) Chosen Courts Section 8.10(b) Closing Section 1.2 Closing Bid Price Section 8.3 Closing Date Section 1.2	Boat Stock Plan	Section 8.3
Boat Surviving Corporation Section 1.1(a) Boat Tax Opinion Section 6.2(d) Boat Termination Fee Section 7.3(b) Business Day Section 8.3 Capitalization Date Section 3.2(a) Certificates of Merger Section 1.3(a) Chosen Courts Section 8.10(b) Closing Section 1.2 Closing Bid Price Section 8.3 Closing Date Section 1.2	Boat Stockholder Approval	Section 4.3(c)
Boat Surviving CorporationSection 1.1(a)Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 1.2	Boat Stockholders Meeting	Section 5.3(c)(i)
Boat Tax OpinionSection 6.2(d)Boat Termination FeeSection 7.3(b)Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 1.2	Boat Surviving Corporation	Section 1.1(a)
Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 1.2		Section 6.2(d)
Business DaySection 8.3Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 1.2	Boat Termination Fee	Section 7.3(b)
Capitalization DateSection 3.2(a)Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 1.2	Business Day	
Certificates of MergerSection 1.3(a)Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 1.2		Section 3.2(a)
Chosen CourtsSection 8.10(b)ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 1.2		
ClosingSection 1.2Closing Bid PriceSection 8.3Closing DateSection 1.2	-	
Closing Bid Price Section 8.3 Closing Date Section 1.2	Closing	
Closing Date Section 1.2		
		Section 1.2
		A-iv

Defined Term	Section
COBRA	Section 3.11(g)
Code	Section 8.3
Common Shares Trust	Section 2.4(b)
Competing Proposal	Section 5.4(f)(i)
Competing Proposal Notice	Section 5.4(1)(1)
Confidentiality Agreement	Section 5.5(a)
	Section 5.14(b)
Continuing Employee Contract	Section 3.14(b)
control	Section 8.3
D&O Insurance	
	Section 5.9(c)
DEA	Section 3.18(b)
Designated Chairman	Section 1.6(a)
DGCL	Section 8.3
Disclosure Schedules	Section 8.3
Effective Time	Section 1.3(b)
Environmental Claim	Section 8.3
Environmental Laws	Section 8.3
Environmental Permits	Section 8.3
Equity Interest	Section 8.3
ERISA	Section 8.3
ERISA Affiliate	Section 8.3
Excess Shares	Section 2.4(a)
Exchange Act	Section 8.3
Exchange Agent	Section 2.5(a)
Exchange Fund	Section 2.5(a)
Existing Credit Facility	Section 8.3
Expenses	Section 8.3
FCPA	Section 3.9(b)
FDA	Section 3.18(b)
Form S-4	Section 3.7
Fractional Shares Cash Amount	Section 2.4(b)
GAAP	Section 8.3
Government Official	Section 8.3
Governmental Entity	Section 8.3
group	Section 8.3
Hazardous Materials	Section 8.3
Health Care Laws	Section 8.3
Health Care Permits	Section 3.18(b)
HIPAA	Section 8.3
Holdco	Preamble
Holdco Board	Section 1.6(a)
Holdco Common Stock	Section 2.1(a)(ii)
Holdco RSU	Section 2.7(c)
Holdco Stock Option	Section 2.7(a)
Holdco Warrant	Section 2.6(a)
HSR Act	Section 8.3
Indemnitee	Section 5.9(a)
Indemnitees	Section 5.9(a)
Intellectual Property	Section 8.3
Intended Tax Treatment	Section 5.13(a)
monded fun fromhom	A-v
	2 i v

Defined Term	Section
Intervening Event	Section 5.4(f)(ii)
IRS	Section 8.3
Island	Preamble
Island Adverse Recommendation Change	Section 5.3(b)(ii)
Island Benefit Plan	Section 3.11(a)
Island Board	Section 3.3(b)
Island Book-Entry Shares	Section 2.1(a)(iii)
Island Business Personnel	Section 3.12(a)
Island Bylaws	Section 3.12(a)
Island Certificate	Section 3.1
Island Certificate of Merger	Section 2.1(a)(iii)
Island Charter	Section 3.1
Island Common Stock	
	Section 2.1(a)
Island Common Stock Merger Consideration	Section 2.1(a)(ii)
Island Disclosure Schedule	Section 8.3
Island ESPP	Section 2.7(d)(i)
Island Exchange Ratio	Section 2.1(a)(ii)
Island Leased Real Property	Section 8.3
Island Material Adverse Effect	Section 8.3
Island Material Contracts	Section 3.16(c)
Island Material Intellectual Property	Section 3.17(d)
Island Merger	Section 1.1(a)
Island Merger Consideration	Section 2.1(a)(ii)
Island Preferred Stock	Section 2.1(a)
Island Pre-Funded Warrants	Section 3.2(a)
Island Recommendation	Section 3.3(b)
Island Registered Intellectual Property	Section 8.3
Island SEC Documents	Section 3.5(a)
Island SEC Financial Statements	Section 3.5(c)
Island Stock Option	Section 2.7(a)
Island Stock Plans	Section 8.3
Island Stockholder Approval	Section 3.3(c)
Island Stockholders Meeting	Section 5.3(b)(i)
Island Surviving Corporation	Section 1.1(a)
Island Tax Opinion	Section 6.3(d)
Island Termination Fee	Section 7.3(d)
Island Warrants	Section 3.2(a)
IT Assets	Section 8.3
Joint Proxy Statement	Section 8.3
Knowledge	Section 8.3
known	Section 8.3
Law	Section 8.3
Licensed Boat Registered Intellectual Property	Section 4.17(a)
Licensed Island Registered Intellectual Property	Section 3.17(a)
Lien	Section 8.3
Market Price	Section 8.3
	Section 8.3
Merger Consideration Value	
Merger Sub A	Preamble
Merger Sub B	Preamble Section 1.1(a)
Mergers	Section 1.1(a)
	A-vi

Defined Term	Section
Multiemployer Plan	Section 3.11(e)
NASDAQ	Section 8.3
NASDAQ Listing Application	Section 5.10
Negotiation Period	Section 5.4(b)
Net Boat Share	Section 8.3
Net Island Share	Section 8.3
OFAC	Section 3.9(e)
Order	Section 8.3
Other Covered Party	Section 8.3
Outside Date	Section 7.1(b)(ii)
Payoff Letter	Section 5.16(a)
Permits	Section 3.10
Permitted Liens	Section 8.3
Person	Section 8.3
Post-Closing Plans	Section 5.14(c)
Preferred Stock Liquidation Preference	Section 8.3
Proceeding	Section 8.3
Release	Section 8.3
Representatives	Section 8.3
Trading Day	Section 8.3
Sarbanes-Oxley Act	Section 3.5(d)
SEC	Section 8.3
Securities Act	Section 8.3
Solely Owned Boat Registered Intellectual Property	Section 4.17(a)
Solely Owned Island Registered Intellectual Property	Section 3.17(a)
Subsidiary	Section 8.3
Superior Proposal	Section 5.4(f)(iii)
Tax Return	Section 8.3
Taxes	Section 8.3
Taxing Authority	Section 8.3
Transactions	Section 1.1(a)
VWAP of Boat Common Stock	Section 8.3
Willful Breach	Section 8.3
	A-vii

AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER (this "Agreement") is dated as of January 21, 2018, by and among BioCryst Pharmaceuticals, Inc., a Delaware corporation ("Boat"), Idera Pharmaceuticals, Inc., a Delaware corporation ("Island"), and Nautilus Holdco, Inc., a Delaware corporation and a wholly owned subsidiary of Boat ("Holdco"), Island Merger Sub, Inc., a Delaware corporation and a wholly owned subsidiary of Holdco ("Merger Sub A"), Boat Merger Sub, Inc., a Delaware corporation and a wholly owned subsidiary of Holdco ("Merger Sub B").

RECITALS

WHEREAS, Boat has organized Holdco, Merger Sub A and Merger Sub B for the purpose of effecting the transactions contemplated by this Agreement;

WHEREAS, each of Boat, Island and Holdco desire, following the satisfaction or waiver of the conditions set forth in *Article 6*, to effect the Mergers upon the terms and subject to the conditions set forth in this Agreement, pursuant to which Merger Sub A shall be merged with and into Island, with Island as the surviving entity in the Island Merger and continuing as a wholly owned subsidiary of Holdco, and Merger Sub B shall be merged with and into Boat, with Boat as the surviving entity in the Boat Merger and continuing as a wholly owned subsidiary of Holdco;

WHEREAS, the Boards of Directors of each of Boat, Island, Merger Sub A, Merger Sub B and Holdco have each unanimously approved and declared advisable this Agreement and the Transactions, including the Mergers, and determined that it is advisable and in the best interests of their respective companies and stockholders to consummate the Mergers and the Transactions on the terms and conditions set forth in this Agreement;

WHEREAS, the Board of Directors of each of Island and Boat has, subject to *Section 5.4*, resolved to recommend the adoption of this Agreement to their respective stockholders; and

WHEREAS, Boat, Island, Merger Sub A, Merger Sub B and Holdco desire to make certain representations, warranties, covenants and agreements in connection with the Mergers and also to prescribe various conditions to the Mergers.

AGREEMENT

NOW, THEREFORE, in consideration of the mutual covenants and premises contained in this Agreement and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties to this Agreement agree as follows:

ARTICLE 1 THE MERGERS

1.1 The Mergers.

(a) At the Effective Time, (a) Merger Sub A shall be merged with and into Island (the "Island Merger") in accordance with the DGCL, and upon the terms and conditions set forth in this Agreement, whereupon the separate existence of Merger Sub A shall cease and Island shall continue as the surviving corporation (the "Island Surviving Corporation"), and (b) Merger Sub B shall be merged with and into Boat (the "Boat Merger", and, together with the Island Merger, the "Mergers") in accordance with the DGCL, and upon the terms and conditions set forth in this Agreement, whereupon the separate existence of Merger Sub B shall cease and Boat shall continue as the surviving corporation (the "Boat Surviving Corporation"). As a result of the Mergers, the Island Surviving Corporation and the Boat Surviving Corporation shall become wholly owned subsidiaries of Holdco. The Mergers and other transactions contemplated by this Agreement are

A-1

Table of Contents

referred to herein as the "*Transactions*". References herein to "Island" and "Boat" with respect to the period from and after the Effective Time shall be deemed to be references to the Island Surviving Corporation and the Boat Surviving Corporation, respectively.

- (b) The Mergers shall have the effects specified in the DGCL. From and after the Effective Time, (a) the Island Surviving Corporation shall possess all the rights, powers, privileges and franchises and be subject to all of the obligations, liabilities and duties of Island and Merger Sub A, and (b) the Boat Surviving Corporation shall possess all the rights, powers, privileges and franchises and be subject to all of the obligations, liabilities and duties of Boat and Merger Sub B, all as provided under the DGCL.
- 1.2 Closing. The closing of the Mergers (the "Closing") shall take place at the offices of Skadden, Arps, Slate, Meagher & Flom LLP, Four Times Square, New York, New York, 10036 at 9:00 a.m. local time, as soon as practicable (and, in any event, within three (3) Business Days) after satisfaction or, to the extent permitted hereunder, waiver of all applicable conditions set forth in Article 6 (except for any conditions that by their nature can only be satisfied at the Closing, but subject to the satisfaction of such conditions or, to the extent permitted hereunder, waiver by the party entitled to waive such conditions) or at such other time and place as Boat and Island shall agree. The date and time at which the Closing occurs is referred to herein as the "Closing Date".

1.3 Effective Time.

- (a) On the Closing Date, (a) Island and Merger Sub A shall file a certificate of merger relating to the Island Merger (the "Island Certificate of Merger") with the Secretary of State of the State of Delaware in accordance with the relevant provisions of the DGCL and shall make all other filings or recordings required under the DGCL, and (b) Boat and Merger Sub B shall file a certificate of merger relating to the Boat Merger (the "Boat Certificate of Merger" and, together with the Island Certificate of Merger, the "Certificates of Merger") with the Secretary of State of the State of Delaware in accordance with the relevant provisions of the DGCL and shall make all other filings or recordings required under the DGCL.
- (b) The Mergers shall become effective concurrently upon the due filing of the Certificates of Merger with the Secretary of State of the State of Delaware, or such later time as Island and Boat shall agree and specify in the Certificates of Merger (such time as the Mergers become effective being the "Effective Time").

1.4 Charters and Bylaws.

- (a) Island Surviving Corporation Certificate of Incorporation and Bylaws. At the Effective Time, the certificate of incorporation of Island in effect immediately prior to the Effective Time shall be amended and restated pursuant to the Island Merger in its entirety as set forth as Exhibit A, and as so amended and restated shall be the certificate of incorporation of the Island Surviving Corporation until thereafter changed or amended as provided therein or by applicable Law. At the Effective Time, Holdco shall take such action necessary to cause the bylaws of Island in effect immediately prior to the Effective Time to be amended and restated in their entirety so as to be identical to the bylaws of Merger Sub A as in effect immediately prior to the Effective Time (except that the name of the Island Surviving Corporation shall be changed to "Idera Pharmaceuticals, Inc."), and as so amended and restated shall be the bylaws of the Island Surviving Corporation until thereafter changed or amended as provided therein or by applicable Law.
- (b) Boat Surviving Corporation Certificate of Incorporation and Bylaws. At the Effective Time, the certificate of incorporation of Boat in effect immediately prior to the Effective Time shall be amended and restated pursuant to the Boat Merger in its entirety as set forth as Exhibit B, and as so amended and restated shall be the certificate of incorporation of the Boat Surviving

Table of Contents

Corporation until thereafter changed or amended as provided therein or by applicable Law. At the Effective Time, Holdco shall take such action necessary to cause the bylaws of Boat in effect immediately prior to the Effective Time to be amended and restated in their entirety so as to be identical to the bylaws of Merger Sub B as in effect immediately prior to the Effective Time (except that the name of the Boat Surviving Corporation shall be changed to "BioCryst Pharmaceuticals, Inc."), and as so amended and restated shall be the bylaws of the Boat Surviving Corporation until thereafter changed or amended as provided therein or by applicable Law.

- (c) Holdco Certificate of Incorporation and Bylaws; Name. Boat shall take and shall cause Holdco to take all requisite action to cause the certificate of incorporation of Holdco (as in effect immediately prior to the Effective Time) to be amended and restated at the Effective Time in its entirety as set forth on Exhibit C (the "Amended and Restated Holdco Charter"), until thereafter amended as provided therein or by applicable Law. The Amended and Restated Holdco Charter will reflect the change in the corporate name of Holdco to a name to be mutually agreed by Boat and Island. Boat shall take and shall cause Holdco to take all requisite action to cause the Holdco Bylaws (as in effect immediately prior to the Effective Time) to be amended and restated as mutually agreed by the parties.
- 1.5 Directors and Officers of the Surviving Corporations.
 - (a) The parties shall take all necessary action such that, from and after the Effective Time, until successors are duly elected or appointed and qualified in accordance with applicable Law, (i) the directors of the Island Surviving Corporation shall be those individuals who were directors of Merger Sub A as of immediately prior to the Effective Time and (ii) the officers of the Island Surviving Corporation shall be those individuals who were officers of Island as of immediately prior to the Effective Time.
 - (b) The parties shall take all necessary action such that, from and after the Effective Time, until successors are duly elected or appointed and qualified in accordance with applicable Law, (i) the directors of the Boat Surviving Corporation shall be those individuals who were directors of Merger Sub B as of immediately prior to the Effective Time and (ii) the officers of the Boat Surviving Corporation shall be those individuals who were officers of Boat as of immediately prior to the Effective Time.
- 1.6 Directors and Officers of Holdco.
 - (a) Effective at the Effective Time, unless otherwise agreed by Island and Boat in writing, the parties shall take all necessary action to cause the size of the Board of Directors of Holdco (the "Holdco Board") to be comprised of nine (9) members. Four (4) directors shall be designated by the Board of Directors of Island, each of whom shall be a director of Island immediately prior to the Effective Time and be reasonably acceptable to Boat. Four (4) directors shall be designated by the Board of Directors of Boat, each of whom shall be a director of Boat immediately prior to the Effective Time and be reasonably acceptable to Island. One director shall be a person mutually agreed to by the Board of Directors of Island and the Board of Directors of Boat, who shall not be a director, officer or affiliate of either Boat or Island. Robert A. Ingram shall serve as chairman of the Holdco Board (the "Designated Chairman"). One of Island's director designees shall be Vincent J. Milano if he is serving as chief executive officer and a director of Boat immediately prior to the Effective Time, and one of Boat's director designees shall be Jon T. Stonehouse if he is serving as chief executive officer and a director of Boat immediately prior to the Effective Time. Each director designee (other than the chief executive officer of Boat or Island immediately prior to the Effective Time) shall qualify as an "independent director", as such term is defined in NASDAQ Equity Rule 5605(a)(2); provided, that at least two (2) of the Island designees and two (2) of the Boat designees shall meet the minimum requirements to serve on Boat's audit committee under the NASDAQ Marketplace Rules. Effective at the Effective Time, unless

A-3

Table of Contents

otherwise agreed by Island and Boat in writing, the parties shall take all necessary action to cause (i) the Designated Chairman to be appointed to act as chairman of the Holdco Board, (ii) the directors designated by Island and Boat respectively to be appointed to the Holdco Board and (iii) at least one Island designee to be appointed to each committee of the Holdco Board.

- (b) Effective at the Effective Time, unless otherwise approved by Island and Board in writing, the parties shall take all necessary action to cause Vincent J. Milano to be appointed as chief executive officer of Holdco. In the event that Vincent J. Milano is not the chief executive officer of Island immediately prior to the Effective Time, Boat and Island will agree on another individual to be appointed as chief executive officer of Holdco.
- 1.7 Headquarters and Primary Research and Development Location. The corporate headquarters for Holdco and its Subsidiaries will be located in Exton, Pennsylvania. The primary location for research and development for Holdco and its Subsidiaries will be located in Birmingham, Alabama.
- 1.8 *Tax Treatment.* For U.S. federal income Tax purposes, it is intended that (a) the Mergers together be treated as an "exchange" described in Section 351 of the Code, and the regulations promulgated thereunder, (b) the Boat Merger qualify as a "reorganization" within the meaning of Section 368(a) of the Code, and the regulations promulgated thereunder, and (c) this Agreement will constitute a "plan of reorganization" for purposes of Sections 354 and 361 of the Code.

ARTICLE 2 CONVERSION OF SECURITIES IN THE MERGERS

- 2.1 Effect on Capital Stock of Island and Boat.
 - (a) Conversion of Island Common Stock, Island Preferred Stock and Merger Sub A Common Stock. At the Effective Time, by virtue of the Island Merger and without any action on the part of Holdco, Island, Merger Sub A or any holder of any shares of Island common stock, \$0.001 par value per share ("Island Common Stock") or Island Series A convertible preferred stock, \$0.01 par value per share ("Island Preferred Stock"):
 - (i) All shares of Island Common Stock or Island Preferred Stock that are owned by Boat, Island, Holdco, Merger Sub A or Merger Sub B or any wholly owned Subsidiary of Boat, Island, Holdco, Merger Sub A or Merger Sub B (or are held in treasury by Island) immediately prior to the Effective Time shall cease to be outstanding and shall be cancelled and retired and shall cease to exist, and no consideration shall be delivered in exchange therefor.
 - (ii) Subject to Section 2.1(a)(i), Section 2.3, Section 2.4 and Section 2.7, (i) each share of Island Common Stock issued and outstanding immediately prior to the Effective Time shall be converted into and thereafter represent the right to receive that number of validly issued, fully paid and non-assessable shares of Holdco common stock, \$0.01 par value per share ("Holdco Common Stock"), equal to the Island Exchange Ratio and cash in lieu of fractional shares of Holdco Common Stock as contemplated by Section 2.4 (the "Island Common Stock Merger Consideration"), and (ii) each share of Island Preferred Stock issued and outstanding immediately prior to the Effective Time shall be converted into and thereafter represent the right to receive the Preferred Stock Liquidation Preference. The shares of Holdco Common Stock to be issued upon the conversion of shares of Island Common Stock and Island Preferred Stock pursuant to this Section 2.1(a) and the Fractional Shares Cash Amount to be paid in lieu of fractional shares pursuant to Section 2.4 are referred to, as applicable, as the "Island Merger Consideration". For purposes of this Agreement, "Island Exchange Ratio" means 0.20.

A-4

- (iii) All of the outstanding shares of Island Common Stock and Island Preferred Stock shall be converted into the right to receive the applicable Island Merger Consideration pursuant to this *Section 2.1(a)* and all such shares of Island Common Stock and Island Preferred Stock shall cease to be outstanding and shall cease to exist and, as of the Effective Time, each holder of a certificate representing any such shares of Island Common Stock or Island Preferred Stock (an "*Island Certificate*") or shares of Island Common Stock or Island Preferred Stock held in book entry form ("*Island Book-Entry Shares*") shall cease to have any rights with respect thereto, except the right to receive, in accordance with *Section 2.1(a)(ii)*, the applicable Island Merger Consideration, upon surrender of such Island Certificate or Island Book-Entry Share in accordance with *Section 2.5*.
- (iv) Each share of Merger Sub A Common Stock issued and outstanding immediately prior to the Effective Time shall be converted into one share of common stock, par value \$0.001 per share, of the Island Surviving Corporation.
- (b) Conversion of Boat Common Stock and Merger Sub B Common Stock. At the Effective Time, by virtue of the Boat Merger and without any action on the part of Holdco, Boat, Merger Sub B or any holder of any shares of Boat common stock, \$0.01 par value per share ("Boat Common Stock"):
 - (i) All shares of Boat Common Stock that are owned by Boat, Island, Holdco, Merger Sub A or Merger Sub B or any wholly owned Subsidiary of Boat, Island, Holdco, Merger Sub A or Merger Sub B (or are held in treasury by Boat) immediately prior to the Effective Time shall cease to be outstanding and shall be cancelled and retired and shall cease to exist, and no consideration shall be delivered in exchange therefor.
 - (ii) Subject to Section 2.1(b)(i), Section 2.3, Section 2.4 and Section 2.7, each share of Boat Common Stock issued and outstanding immediately prior to the Effective Time shall be converted into and thereafter represent the right to receive that number of validly issued, fully paid and non-assessable share of Holdco Common Stock equal to the Boat Exchange Ratio and cash in lieu of fractional shares of Holdco Common Stock as contemplated by Section 2.4. The shares of Holdco Common Stock to be issued upon the conversion of shares of Boat Common Stock pursuant to this Section 2.1(b) and the Fractional Shares Cash Amount to be paid in lieu of fractional shares pursuant to Section 2.4 are referred to as the "Boat Merger Consideration". For purposes of this Agreement, "Boat Exchange Ratio" means 0.50.
 - (iii) All of the outstanding shares of Boat Common Stock shall be converted into the right to receive the Boat Merger Consideration pursuant to this *Section 2.1(b)* and all such shares of Boat Common Stock shall cease to be outstanding and shall cease to exist and, as of the Effective Time, each holder of a certificate representing any such shares of Boat Common Stock (a "*Boat Certificate*") or shares of Boat Common Stock held in book entry form ("*Boat Book-Entry Shares*") shall cease to have any rights with respect thereto, except the right to receive, in accordance with *Section 2.1(b)(ii)*, the Boat Merger Consideration, upon surrender of such Boat Certificate or Boat Book-Entry Share in accordance with *Section 2.5*.
 - (iv) Each share of common stock, par value \$0.01 per share, of Merger Sub B issued and outstanding immediately prior to the Effective Time shall be converted into one share of common stock, par value \$0.001 per share, of the Boat Surviving Corporation.
- 2.2 Effect on Holdco Common Stock. At the Effective Time, each share of Holdco Common Stock issued and outstanding immediately prior to the Effective Time shall remain outstanding. Immediately following the Effective Time, shares of capital stock of Holdco owned by the Island Surviving Corporation or the Boat Surviving Corporation shall be surrendered to Holdco without payment therefor.

Table of Contents

2.3 Certain Adjustments. Notwithstanding anything in this Agreement to the contrary, if, from the date of this Agreement until the Effective Time, the outstanding shares of Boat Common Stock, Island Common Stock or Island Preferred Stock shall have been changed into a different number of shares or a different class by reason of any reclassification, stock split (including a reverse stock split), recapitalization, split-up, combination, exchange of shares, readjustment, or other similar transaction, or a stock dividend or stock distribution thereon shall be declared with a record date within said period, the applicable Island Merger Consideration, the Island Exchange Ratio and the Boat Exchange Ratio and the Boat Merger Consideration and any other similarly dependent items, as the case may be, shall be equitably adjusted to provide the holders of Island Common Stock, Island Preferred Stock or Boat Common Stock, as applicable, the same economic effect as contemplated by this Agreement prior to such event.

2.4 Fractional Shares.

- (a) No fractional shares of Holdco Common Stock or shall be issued in the Mergers, but in lieu thereof each holder of Island Common Stock, Island Preferred Stock or Boat Common Stock otherwise entitled to a fractional share of Holdco Common Stock will be entitled to receive, from the Exchange Agent in accordance with the provisions of this *Section 2.4*, a cash payment in lieu of such fractional share of Holdco Common Stock representing such holder's proportionate interest, if any, in the proceeds from the sale by the Exchange Agent (reduced by any fees of the Exchange Agent attributable to such sale) in one or more transactions of shares of Holdco Common Stock equal to the excess of (A) the aggregate number of shares of Holdco Common Stock to be delivered to the Exchange Agent by Holdco pursuant to *Section 2.5(a)* over (B) the aggregate number of whole shares of Holdco Common Stock to be distributed to the holders of shares of Island Common Stock, Island Preferred Stock or Boat Common Stock pursuant to *Section 2.5(b)* (such excess, the "*Excess Shares*"). Island, Boat and Holdco acknowledge that payment of the cash consideration in lieu of issuing fractional shares of Holdco Common Stock was not separately bargained-for consideration but merely represents a mechanical rounding off for purposes of avoiding the expense and inconvenience to Holdco that would otherwise be caused by the issuance of fractional shares of Holdco Common Stock. As soon as practicable after the Effective Time, the Exchange Agent, as agent for the holders of Island Common Stock, Island Preferred Stock and Boat Common Stock that would otherwise receive fractional shares of Holdco Common Stock, shall sell the Excess Shares at then prevailing prices on the NASDAQ in the manner provided in the following paragraph.
- (b) The sale of the Excess Shares by the Exchange Agent, as agent for the holders of Island Common Stock, Island Preferred Stock and Boat Common Stock that would otherwise receive fractional shares of Holdco Common Stock, shall be executed on the NASDAQ and shall be executed in round lots to the extent practicable. Until the proceeds of such sale or sales have been distributed to the holders of Island Common Stock, Island Preferred Stock and Boat Common Stock, the Exchange Agent shall hold such proceeds in trust for the holders of Island Common Stock, Island Preferred Stock and Boat Common Stock that would otherwise receive fractional shares of Holdco Common Stock (the "Common Shares Trust"). The Exchange Agent shall determine the portion of the Common Shares Trust to which each holder of Island Common Stock, Island Preferred Stock or Boat Common Stock shall be entitled, if any, by multiplying the amount of the aggregate proceeds comprising the Common Shares Trust by a fraction, the numerator of which is the amount of the fractional share interest to which such holder of Island Common Stock, Island Preferred Stock or Boat Common Stock would otherwise be entitled and the denominator of which is the aggregate amount of fractional share interests to which all holders of Island Common Stock, Island Preferred Stock and Boat Common Stock would otherwise be entitled (the "Fractional Shares Cash Amount").

A-6

Table of Contents

(c) As soon as practicable after the determination of the amount of cash, if any, to be paid to holders of Island Common Stock in lieu of any fractional shares of Holdco Common Stock, the Exchange Agent shall make available such amounts to such holders of shares of Island Common Stock without interest, subject to and in accordance with *Section 2.5*.

2.5 Exchange of Certificates and Book-Entry Shares.

- (a) Prior to the dissemination of the Joint Proxy Statement to the stockholders of Island and the stockholders of Boat, Boat and Island shall appoint a nationally recognized financial institution or trust company to act as exchange agent (the "Exchange Agent") for the payment and delivery of the Island Merger Consideration and the Boat Merger Consideration. At or prior to the Effective Time, Holdco shall (and Boat and Island shall cause Holdco to) deposit, in trust for the benefit of the holders of Island Common Stock, the holders of Island Preferred Stock and the holders of Boat Common Stock, with the Exchange Agent for exchange in accordance with this Article 2, a number of shares of Holdco Common Stock equal to the total shares of Holdco Common Stock issuable pursuant to Section 2.1 (together with the Fractional Shares Cash Amount, the "Exchange Fund"). The Exchange Agent shall, pursuant to irrevocable instructions, deliver the whole shares of Holdco Common Stock contemplated to be issued pursuant to Section 2.1, the Fractional Shares Cash Amount out of the Exchange Fund. The Exchange Fund shall not be used for any other purpose.
- (b) Exchange Procedures. As promptly as practicable after the Effective Time (but in no event later than three (3) Business Days thereafter), Holdco shall cause the Exchange Agent to mail to each holder of record of an Island Certificate, Island Book-Entry Share, Boat Certificate or Boat Book-Entry Share, in each case which shares were converted into the right to receive the applicable Island Merger Consideration or Boat Merger Consideration at the Effective Time pursuant to this Agreement: (i) a letter of transmittal, which shall specify that delivery shall be effected, and risk of loss and title to the Island Certificates and Boat Certificates shall pass, only upon delivery of the Island Certificates and Boat Certificates to the Exchange Agent, and shall otherwise be in such form and have such other provisions as Boat, Island and the Exchange Agent may reasonably specify, and (ii) instructions for effecting the surrender of the Island Certificates, Island Book-Entry Shares, Boat Certificates and Boat Book-Entry Shares in exchange for the applicable Island Merger Consideration or Boat Merger Consideration, as applicable, and the method of payment of such holder's Fractional Shares Cash Amount. Upon surrender of Island Certificates, Island Book-Entry Shares, Boat Certificates or Boat Book-Entry Shares for cancellation to the Exchange Agent or to such other agent or agents as may be appointed by Island and Boat, and upon delivery of a letter of transmittal, duly executed and in proper form with all required enclosures and attachments, with respect to such Island Certificates, Island Book-Entry Shares, Boat Certificates or Boat Book-Entry Shares, the holder of such Island Certificates, Island Book-Entry Shares, Boat Certificates or Boat Book-Entry Shares shall be entitled to receive the whole shares of Holdco Common Stock that such holder is entitled to receive pursuant to Section 2.1 and, if applicable, the Fractional Shares Cash Amount for each share of Island Common Stock, Island Preferred Stock or Boat Common Stock formerly represented by such Island Certificates or Island Book-Entry Share or Boat Certificates or Boat Book Entry Share. Any Island Certificates, Island Book-Entry Shares, Boat Certificates or Boat Book-Entry Shares so surrendered shall forthwith be cancelled. If payment of the applicable Island Merger Consideration or Boat Merger Consideration is to be made to a Person other than the Person in whose name any surrendered Island Certificate or Boat Certificate, as applicable is registered, it shall be a condition precedent to payment that the Island Certificate or Boat Certificate so surrendered shall be properly endorsed or shall be otherwise in proper form for transfer, and the Person requesting such payment shall have paid any transfer and other similar Taxes required by reason of the delivery of the applicable Island Merger Consideration or Boat

Table of Contents

Merger Consideration, as applicable, to a Person other than the registered holder of the Island Certificate or Boat Certificate so surrendered and shall have established to the satisfaction of Holdco that such Taxes either have been paid or are not required to be paid. Delivery of the applicable Island Merger Consideration or Boat Merger Consideration, as applicable, with respect to Island Book-Entry Shares and Boat Book-Entry Shares shall only be made to the Person in whose name such shares are registered. Until surrendered as contemplated hereby, each Island Certificate, Island Book-Entry Share, Boat Certificate or Boat Book-Entry Share shall be deemed at any time after the Effective Time to represent only the right to receive the applicable Island Merger Consideration or Boat Merger Considerations provided in this Agreement.

- (c) *Transfer Books.* At the Effective Time, the stock transfer books of Island and Boat shall be closed and thereafter there shall be no further registration of transfers of shares of Island Common Stock or Island Preferred Stock on the records of Island or shares of Boat Common Stock on the records of Boat. From and after the Effective Time, the holders of Island Certificates, Island Book-Entry Shares, Boat Certificates and Boat Book-Entry Shares representing shares of Island Common Stock, Island Preferred Stock and Boat Common Stock, as applicable, outstanding immediately prior to the Effective Time shall cease to have any rights with respect to such shares except as otherwise provided for herein or by applicable Law. If, after the Effective Time, Island Certificates or Boat Certificates representing shares of Island Common Stock, Island Preferred Stock or Boat Common Stock are presented to the Island Surviving Corporation or the Boat Surviving Corporation for any reason, they shall be cancelled and exchanged for the whole shares of Holdco Common Stock that such holder is entitled to receive pursuant to *Section 2.1* and, if applicable, the Fractional Shares Cash Amount as provided in this Agreement subject, in each case, to any applicable abandoned property, escheat or similar Laws.
- (d) Termination of Fund; Abandoned Property. At any time following six months after the Closing Date, Holdco shall be entitled to require the Exchange Agent to deliver to it any shares of Holdco Common Stock or cash remaining in the Exchange Fund made available to the Exchange Agent and not delivered to holders of Island Certificates, Island Book-Entry Shares, Boat Certificates or Boat Book-Entry Shares and thereafter such holders shall be entitled to look only to Holdco (subject to abandoned property, escheat or other similar Laws) only as general creditors thereof with respect to the applicable Island Merger Consideration or Boat Merger Consideration, as applicable, payable upon due surrender of their Island Certificates, Island Book-Entry Shares, Boat Certificates or Boat Book-Entry Shares and compliance with the procedures in Section 2.5. If, prior to six years after the Closing Date (or otherwise immediately prior to such time on which any payment in respect hereof would escheat to or become the property of any Governmental Entity pursuant to any applicable abandoned property, escheat or similar Laws), any holder of Island Certificates, Island Book-Entry Shares, Boat Certificates or Boat Book-Entry Shares has not complied with the procedures in Section 2.5 to receive the applicable Island Merger Consideration or Boat Merger Consideration to which such holder would otherwise be entitled, the applicable Island Merger Consideration or Boat Merger Consideration to which such holder would otherwise be entitled in respect of such Island Certificates, Island Book-Entry Shares, Boat Certificates or Boat Book-Entry Shares shall, to the extent permitted by applicable Law, become the property of Holdco, free and clear of all claims or interest of any Person previously entitled thereto. Notwithstanding the foregoing, none of Boat, the Island Surviving Corporation, the Boat Surviving Corporation or the Exchange Agent, or any Representative or affiliate thereof, shall be liable to any holder of an Island Certificate, Island Book-Entry Shares, Boat Certificate or Boat Book-Entry Shares for Island Merger Consideration or Boat Merger Consideration delivered to a public official pursuant to any applicable abandoned property, escheat or similar Law.
- (e) Lost, Stolen or Destroyed Certificates. In the event that any Island Certificates or Boat Certificates shall have been lost, stolen or destroyed, the Exchange Agent shall issue in exchange

A-8

Table of Contents

for such lost, stolen or destroyed Island Certificates or Boat Certificates, upon the making of an affidavit of that fact by the holder thereof, the applicable Island Merger Consideration or Boat Merger Consideration payable in respect thereof pursuant to *Sections 2.1(a)(ii)* and *2.1(b)(ii)*; *provided, however*, that Holdco or the Exchange Agent may, in its reasonable discretion and as a condition precedent to the payment of the applicable Island Merger Consideration or Boat Merger Consideration, require the owners of such lost, stolen or destroyed Island Certificates or Boat Certificates to deliver a customary indemnity against any claim that may be made against Holdco, the Island Surviving Corporation, the Boat Surviving Corporation or the Exchange Agent with respect to the Island Certificates or Boat Certificates alleged to have been lost, stolen or destroyed.

(f) Distributions with Respect to Unexchanged Holdco Common Stock. No dividends or other distributions declared or made after the Effective Time with respect to Holdco Common Stock with a record date after the Effective Time shall be paid to the holder of any unsurrendered Island Certificate, Island Book-Entry Share, Boat Certificate or Boat Book-Entry Share with respect to any shares of Holdco Common Stock represented thereby, and, if applicable, no portion of the Fractional Shares Cash Amount shall be paid to any such holder pursuant to Section 2.4, unless and until the holder of such Island Certificate, Island Book-Entry Share, Boat Certificate or Boat Book-Entry Share shall surrender such Island Certificate, Island Book-Entry Share, Boat Certificate or Boat Book-Entry Share in accordance with this Section 2.5. Subject to the effect of escheat, Tax or other applicable Laws, following surrender of any such Island Certificate, Island Book-Entry Share, Boat Certificate or Boat Book-Entry Share in accordance with this Section 2.5, there shall be paid by Holdco to the holder of the certificates representing whole shares of Holdco Common Stock issued in exchange therefor, without interest, (i) promptly, the portion of the Fractional Shares Cash Amount, if applicable, payable with respect to a fractional share of Holdco Common Stock to which such holder is entitled pursuant to Section 2.4 and the amount of dividends or other distributions with a record date after the Effective Time theretofore paid with respect to such whole shares of Holdco Common Stock and (ii) at the appropriate payment date, the amount of dividends or other distributions, with a record date after the Effective Time but prior to surrender and a payment date occurring after surrender, payable with respect to such whole shares of Holdco Common Stock.

2.6 Island Warrants.

- (a) As of the Effective Time, each Island Warrant and each Island Pre-Funded Warrant that is outstanding and unexercised immediately prior to the Effective Time, whether or not then exercisable, shall be assumed by Holdco and shall be converted into a warrant (a "Holdco Warrant") to acquire Holdco Common Stock. Each such Holdco Warrant as so assumed and converted shall continue to have, and shall be subject to, the same terms and conditions as applied to the Island Warrant or Island Pre-Funded Warrant, as applicable, immediately prior to the Effective Time except that, as of the Effective Time, each such Holdco Warrant as so assumed and converted shall be a warrant to acquire that number of whole shares of Holdco Common Stock (rounded down to the nearest whole share) equal to the product of (i) the number of shares of Island Common Stock subject to such Island Warrant or Island Pre-Funded Warrant, as applicable, and (ii) the Island Exchange Ratio, at an exercise price per share of Holdco Common Stock (rounded up to the nearest whole cent) equal to the quotient obtained by dividing (x) the exercise price per share of Island Common Stock of such Island Warrant or Island Pre-Funded Warrant, as applicable, by (y) the Island Exchange Ratio.
- (b) Not later than the Closing Date, Holdco shall deliver to the holders of Island Warrants and to the holders of Island Pre-Funded Warrants any notices required under the terms of the Island Warrants or Island Pre-Funded Warrants, as applicable, in connection with this Agreement and the Transactions.

A-9

Table of Contents

- 2.7 Stock Options and Stock-Based Awards.
 - (a) Treatment of Island Stock Options. As of the Effective Time, by virtue of the Mergers and without any action on the part of the holders thereof, each option to acquire shares of Island Common Stock (an "Island Stock Option") that is outstanding and unexercised immediately prior to the Effective Time, whether or not then vested or exercisable, shall (i) be assumed by Holdco and shall be converted into an option (a "Holdco Stock Option") to acquire shares of Holdco Common Stock in accordance with this Section 2.7(a) and (ii) if the Island Stock Option is held by an individual who is not a party to Island Severance and Change of Control Agreement, such Island Stock Option shall accelerate and vest in full. Each such Holdco Stock Option as so assumed and converted shall continue to have, and shall be subject to, the same terms and conditions (other than, with respect to the options described in subclause (ii), vesting) as applied to the Holdco Stock Option immediately prior to the Effective Time except that, as of the Effective Time, each such Holdco Stock Option as so assumed and converted shall be an option to acquire that number of whole shares of Holdco Common Stock (rounded down to the nearest whole share) equal to the product of (A) the number of shares of Island Common Stock subject to such Island Stock Option and (B) the Island Exchange Ratio, at an exercise price per share of Holdco Common Stock (rounded up to the nearest whole cent) equal to the quotient obtained by dividing (x) the exercise price per share of Island Common Stock of such Island Stock Option by (y) the Island Exchange Ratio (provided, that the exercise price and the number of shares of Holdco Common Stock subject to the Holdco Stock Option shall be determined in a manner consistent with the requirements of Section 409A of the Code, and, in the case of Island Stock Options that are intended to qualify as incentive stock options within the meaning of Section 422 of the Code, consistent with the requirements of Section 424 of the Code).
 - (b) *Treatment of Boat Stock Options.* As of the Effective Time, by virtue of the Mergers and without any action on the part of the holders thereof, each option to acquire shares of Boat Common Stock (a "*Boat Stock Option*") that is outstanding and unexercised immediately prior to the Effective Time, whether or not then vested or exercisable, shall (i) accelerate and vest in full and (ii) be assumed by Holdco and shall be converted into a Holdco Stock Option to acquire shares of Holdco Common Stock in accordance with this *Section 2.7(b)*. Each such Holdco Stock Option as so assumed and converted shall continue to have, and shall be subject to, the same terms and conditions (other than vesting) as applied to the Holdco Stock Option immediately prior to the Effective Time, except that, as of the Effective Time, each such Boat Stock Option as so assumed and converted shall be an option to acquire that number of whole shares of Holdco Common Stock (rounded down to the nearest whole share) equal to the product of (A) the number of shares of Boat Common Stock subject to such Boat Stock Option and (B) the Boat Exchange Ratio, at an exercise price per share of Boat Common Stock of such Boat Stock Option by (y) the Boat Exchange Ratio (*provided*, that the exercise price and the number of shares of Holdco Common Stock subject to the Holdco Stock Option shall be determined in a manner consistent with the requirements of Section 409A of the Code, and, in the case of Boat Stock Options that are intended to qualify as incentive stock options within the meaning of Section 422 of the Code, consistent with the requirements of Section 424 of the Code).
 - (c) As of the Effective Time, by virtue of the Mergers and without any action on the part of the holders thereof, each restricted stock unit covering Boat Common Stock (each, a "Boat RSU") that is outstanding immediately prior to the Effective Time shall be assumed by Holdco and shall (i) accelerate and vest in full and (ii) be converted into a restricted stock unit (a "Holdco RSU") to acquire Holdco Common Stock in accordance with this Section 2.7(c). Each such Holdco RSU as so assumed and converted shall continue to have, and shall be subject to, the same terms and

Table of Contents

conditions (other than vesting) as applied to the Boat RSU immediately prior to the Effective Time, except that, as of the Effective Time, each such Boat RSU as so assumed and converted shall be a restricted stock unit covering that number of whole shares of Holdco Common Stock (rounded down to the nearest whole share) equal to the product of (A) the number of shares of Boat Common Stock subject to such Boat RSU and (B) the Boat Exchange Ratio.

- (d) Treatment of Employee Stock Purchase Plans.
 - (i) As soon as practicable following the date of this Agreement, the Island Board shall take all actions with respect to the Island 2017 Employee Stock Purchase Plan, as amended (the "Island ESPP") that are necessary to provide that with respect to the offering period that would otherwise be in effect on the Closing Date, such offering period shall terminate and any options then-outstanding under the Island ESPP shall be exercised, on the date immediately preceding the Closing Date or such earlier date as mutually agreed in writing by Boat and Island.
 - (ii) As soon as practicable following the date of this Agreement, the Boat Board shall take all actions with respect to the Boat Amended and Restated Employee Stock Purchase Plan (the "Boat ESPP") that are necessary to provide that with respect to the offering period that would otherwise be in effect on the Closing Date, such offering period shall terminate and any options then-outstanding under the Boat ESPP shall be exercised, on the date immediately preceding the Closing Date or such earlier date as mutually agreed in writing by Boat and Island.
- (e) Subject to Section 5.14(f), as of the Effective Time, Holdco shall, by virtue of the Mergers, assume the Boat Stock Plan, each Island Stock Plan, the Boat ESPP and the Island ESPP. Following the Effective Time, Holdco shall continue to maintain any such assumed plans (each, as may be amended or terminated from time to time in accordance therewith), and shall be able to grant "incentive stock options" (within the meaning of Section 422 of the Code and the Treasury Regulations promulgated thereunder) under each such plan. As of immediately prior to the Effective Time: (i) the maximum number of shares of Boat Common Stock that may be issued under the Boat Stock Plan and the Boat ESPP upon the exercise of incentive stock options (including outstanding grants) is 24,190,000 and 1,475,000, respectively (subject to certain adjustments in the applicable plan in accordance with the Treasury Regulations promulgated thereunder), (ii) employees of Boat are eligible to receive incentive stock options under each of the Boat Stock Plan and the Boat ESPP, (iii) the maximum number of shares of Island Common Stock that may be issued under the Island 2005 Stock Incentive Plan, the 2008 Stock Incentive Plan, the 2013 Stock Incentive Plan and the Island ESPP upon the exercise of incentive stock options (including outstanding grants) is 1,125,000, 6,000,000, 32,171,438 and 500,000, respectively (subject to certain adjustments in the applicable plan in accordance with the Treasury Regulations promulgated thereunder) and (iv) employees of Island are eligible to receive incentive stock options under each such Island Stock Plan and the Island ESPP. As of immediately following the Effective Time, by virtue of the Mergers, (i) the maximum number of shares of Holdco Common Stock that may be issued under the Boat Stock Plan and the Boat ESPP as incentive stock options (including outstanding grants) shall be 24,190,000 and 1,475,000, respectively (subject to certain adjustments (x) in the applicable plan in accordance with the Treasury Regulations promulgated thereunder and (y) to reflect the Boat Exchange Ratio), (ii) employees of Holdco and its subsidiaries shall be eligible to receive incentive stock options under the Boat Stock Plan and the Boat ESPP, (iii) the maximum number of shares of Holdco Common Stock that may be issued under the Island 2005 Stock Incentive Plan, the 2008 Stock Incentive Plan, the 2013 Stock Incentive Plan and the Island ESPP as incentive stock options (including outstanding grants) shall be 1,125,000, 6,000,000, 32,171,438 and 500,000, respectively (subject to certain adjustments (x) in the applicable plan in accordance with the Treasury Regulations promulgated thereunder and

Table of Contents

- (y) to reflect the Island Exchange Ratio) and (iv) the employees of Holdco and its subsidiaries shall be eligible to receive incentive stock options under the Island Stock Plans and the Island ESPP.
- (f) At the Effective Time, Holdco shall assume all the obligations of Island and Boat under the Island Stock Plans, the Boat Stock Plan, each outstanding Island Stock Option, Boat Stock Option, Boat RSU and the agreements evidencing the grants thereof, and the number and kind of shares available for issuance under each Island Stock Plan and the Boat Stock Plan shall be adjusted to reflect shares of Holdco Common Stock in accordance with the provisions of the applicable Island Stock Plan and the Boat Stock Plan. Prior to the Effective Time, Island and Boat shall take all necessary action for the adjustment and assumption of equity awards under this *Section 2.7*. Holdco shall reserve for future issuance a number of shares of Holdco Common Stock at least equal to the number of shares of Holdco Common Stock that will be subject to Holdco Stock Options and Holdco RSUs as a result of the actions contemplated by this *Section 2.7*. Not later than the Closing Date, Holdco shall file an effective registration statement on Form S-8 (or other applicable form) with respect to the Holdco Common Stock subject to such Holdco Stock Options and Holdco RSUs and shall distribute a prospectus relating to such Form S-8, and Holdco shall use reasonable best efforts to maintain the effectiveness of such registration statement or registration statements (and maintain the current status of the prospectus or prospectuses contained therein) for so long as such Holdco Stock Options and Holdco RSUs remain outstanding.
- 2.8 Withholding Rights. Each of Holdco, Boat, the Boat Surviving Corporation, Island, the Island Surviving Corporation, Merger Sub A, Merger Sub B and the Exchange Agent shall be entitled to deduct and withhold from any amount payable pursuant to this Agreement to any Person, such amounts as are required to be deducted and withheld under the Code, Treasury Regulations promulgated under the Code or any applicable provisions of state, local or non-U.S. Tax Law. Boat, Merger Sub or the Exchange Agent, as appropriate, shall provide commercially reasonable notice to Island upon becoming aware of any such withholding obligation and shall cooperate with Island to the extent reasonable to obtain reduction of or relief from such withholding. To the extent that amounts are so deducted and withheld and timely paid over to the applicable Governmental Entity, such deducted and withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person(s) in respect of which such deduction and withholding was made.

ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF ISLAND

Except (a) as set forth in the Island Disclosure Schedule (subject to *Section 8.14*) and (b) as otherwise disclosed or identified in the Island SEC Documents publicly filed or furnished after December 31, 2015 and prior to the date hereof (other than (i) any forward-looking disclosures contained in the "Forward Looking Statements" and "Risk Factors" sections of the Island SEC Documents and any other disclosures included therein to the extent they are primarily predictive, cautionary or forward looking in nature and (ii) information included in, or incorporated by reference as, exhibits and schedules to any Island SEC Document, and provided that the exception provided for in this clause (b) shall be applied if, and only if, the nature and content of the applicable disclosure in any such Island SEC Document publicly filed or furnished prior to the date hereof is reasonably specific as to matters and items such that the subject matter of such disclosure is reasonably apparent on the face of the text of such disclosure to be applicable to the representation set forth herein), Island hereby represents and warrants to Boat as follows:

3.1 Corporate Organization. Island is a corporation duly organized, validly existing and in good standing under the laws of Delaware and has the requisite corporate power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted. Island is

Table of Contents

duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed or qualified, has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect. Island has provided Boat true, complete and correct copies of the Restated Certificate of Incorporation, as amended (the "Island Charter"), and By-Laws of Island (the "Island Bylaws") as in effect as of the date of this Agreement. Island is not in violation of any of the provisions of the Island Charter or the Island Bylaws.

3.2 Island Capitalization.

(a) The authorized capital stock of Island consists of two hundred eighty million (280,000,000) shares of Island Common Stock and five million (5,000,000) shares of Island Preferred Stock, As of January 3, 2018 (the "Capitalization Date"), (i) 195,635,196 shares of Island Common Stock (other than treasury shares) were issued and outstanding, all of which were validly issued and fully paid, nonassessable and free of preemptive rights, (ii) no shares of Island Common Stock were held in the treasury of Island, (iii) 25,225,500 shares of Island Common Stock were issuable (and such number was reserved for issuance) upon exercise of Island Stock Options outstanding as of such date, (iv) 456,961 shares of Island Common Stock were authorized for issuance pursuant to the Island ESPP, (v) 1,926 shares of Island Common Stock were issuable (and such number was reserved for issuance) upon conversion of Island Preferred Stock, at a conversion rate of 2.94118 per share, (vi) 21,606,327 shares of Island Common Stock were issuable (and such number was reserved for issuance) upon exercise of warrants (the "Island Warrants"), at a weighted average exercise price of \$0.47 per share, (vii) 22,151,052 shares of Island Common Stock were issuable (and such number was reserved for issuance) upon exercise of pre-funded warrants (the "Island Pre-Funded Warrants"), at an exercise price of \$0.01 per share and (viii) 655 shares of Island Preferred Stock were issued and outstanding, all of which were validly issued. Except for Island Stock Options to purchase not more than 25,225,500 shares of Island Common Stock, 456,961 shares of Island Common Stock authorized for issuance under the Island ESPP, Island Preferred Stock convertible into not more than 1.926 shares of Island Common Stock, Island Warrants exercisable into not more than 21,606,327 shares of Island Common Stock and Island Pre-Funded Warrants exercisable into not more than 22,151,052 shares of Island Common Stock, there are no options, warrants or other rights, agreements, arrangements or commitments of any character to which Island is a party or by which Island is bound relating to the issued or unissued capital stock or other Equity Interests of Island, or securities convertible into or exchangeable for such capital stock or other Equity Interests, or obligating Island to issue or sell any shares of its capital stock or other Equity Interests, or securities convertible into or exchangeable for such capital stock of, or other Equity Interests in, Island. Since the Capitalization Date, Island has not issued any shares of its capital stock or other Equity Interests, or securities convertible into or exchangeable for such capital stock or other Equity Interests, other than those shares of capital stock reserved for issuance described in this Section 3.2(a). Island has previously provided Boat with a true and complete list, by holder and as of the date hereof, of (i) the prices at which each outstanding Island Stock Option may be exercised, the number of shares of Island Common Stock underlying such Island Stock Option, the vesting schedule of the Island Stock Options and whether such option is intended to be an "incentive stock option" within the meaning of Section 422 of the Code, (ii) the number of shares of Island Preferred Stock outstanding, (iii) the prices at which each Island Warrant may be exercised under the terms of such Island Warrant and the number of shares of Island Common Stock subject to each such Island Warrant, and (iv) the prices at which each Island Pre-Funded Warrant may be exercised under the terms of such Island Pre-Funded Warrant and the number of shares of Island Common Stock subject to each such Island Pre-Funded Warrant. All shares of Island Common Stock subject to issuance under an Island Stock Plan, or subject to issuance upon conversion of any Island Preferred Stock or exercise of any

Table of Contents

Island Warrant or Island Pre-Funded Warrant, upon issuance prior to the Effective Time on the terms and conditions specified in the instruments pursuant to which they are issuable, will be duly authorized, validly issued, fully paid, nonassessable and free of preemptive rights. There are no bonds, debentures, notes or other indebtedness of Island having the right to vote on any matters on which stockholders of Island may vote. Island is not a party to any voting agreement with respect to the voting of any Equity Interests of Island.

- (b) There are no outstanding contractual obligations of Island (i) restricting the transfer of, (ii) affecting the voting rights of, (iii) requiring the repurchase, redemption or disposition of, or containing any right of first refusal with respect to, (iv) requiring the registration for sale of, or (v) granting any preemptive or antidilutive right with respect to, any shares of Island Common Stock or Island Preferred Stock or any capital stock of, or other Equity Interests in, Island.
- (c) Island has no Subsidiaries. There are no outstanding contractual obligations of Island to provide funds to, or make any investment (in the form of a loan, capital contribution or otherwise) in, any other Person.
- 3.3 Authority; Execution and Delivery; Enforceability.
 - (a) Island has all necessary power and authority to execute and deliver this Agreement, to perform and comply with each of its obligations under this Agreement and, subject to the receipt of the Island Stockholder Approval, to consummate the Transactions. The execution and delivery by Island of this Agreement, the performance and compliance by Island with each of its obligations herein and the consummation by Island of the Transactions have been duly authorized by all necessary corporate action on the part of Island, subject to receipt of the Island Stockholder Approval, and no other corporate proceedings on the part of Island and no stockholder votes are necessary to authorize this Agreement or the consummation by Island of the Transactions. Island has duly and validly executed and delivered this Agreement and, assuming the due authorization, execution and delivery by Boat, Merger Sub A, Merger Sub B and Holdco of this Agreement, this Agreement constitutes its legal, valid and binding obligation, enforceable against it in accordance with its terms, except as limited by Laws affecting the enforcement of creditors' rights generally, by general equitable principles or by the discretion of any Governmental Entity before which any Proceeding seeking enforcement may be brought.
 - (b) The Board of Directors of Island (the "Island Board"), at a meeting duly called and held, unanimously adopted resolutions (i) adopting and approving this Agreement and the consummation of the Transactions upon the terms and subject to the conditions set forth in this Agreement, (ii) determining that the terms of this Agreement, the Mergers and the other Transactions are fair to, and in the best interests of, Island and its stockholders, (iii) directing that this Agreement be submitted to the stockholders of Island for approval and adoption, (iv) recommending that its stockholders adopt this Agreement and (v) declaring that this Agreement is advisable (the "Island Recommendation").
 - (c) Assuming the accuracy of the representations and warranties in *Section 4.21*, to the Knowledge of Island, no takeover, anti-takeover, business combination, control share acquisition or similar Law applies to this Agreement, the Mergers or the other Transactions. The only vote of holders of any class or series of Equity Interests of Island necessary to adopt this Agreement is the adoption of this Agreement by the holders of a majority of the shares of Island Common Stock outstanding and entitled to vote thereon at the Island Stockholders Meeting (the "*Island Stockholder Approval*"). No other vote of the holders of Island Common Stock, Island Preferred Stock or any other Equity Interests of Island is necessary to consummate the Transactions.

Table of Contents

3.4 No Conflicts.

- (a) The execution and delivery of this Agreement by Island does not and will not, and the performance of this Agreement by Island and the consummation of the Transactions will not, (i) assuming the Island Stockholder Approval is obtained, conflict with or violate any provision of the Island Charter or the Island Bylaws, (ii) assuming that all consents, approvals, authorizations and permits described in *Section 3.4(b)* have been obtained and all filings and notifications described in *Section 3.4(b)* have been made and any waiting periods thereunder have terminated or expired, conflict with or violate any Law applicable to Island or by which any property or asset of Island is bound or affected or (iii) require any consent or approval under, result in any breach of or any loss or impairment of any benefit under, constitute a change of control or default (or an event which with notice or lapse of time or both would become a default) under or give to others any right of termination, vesting, amendment, acceleration or cancellation of, or result in the creation of a Lien on any property or asset of Island pursuant to, any Contract or Permit (other than an Island Benefit Plan), except, with respect to clauses (ii) and (iii), for any such conflicts, violations, breaches, defaults or other occurrences which would not, individually or in the aggregate, reasonably be expected to (A) prevent or materially delay consummation of the Mergers, (B) otherwise prevent or materially delay performance by Island of any of its material obligations under this Agreement or (C) have an Island Material Adverse Effect.
- (b) The execution and delivery of this Agreement by Island does not and will not, and the consummation by Island of the Transactions and compliance by Island with any of the terms or provisions hereof or thereof will not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Entity or any other Person, except (i) under the Exchange Act, Securities Act, any applicable blue sky Law, and the rules and regulations of the NASDAQ, (ii) under the HSR Act or other applicable antitrust Laws, (iii) the filing of the Certificates of Merger as required by the DGCL and (iv) where failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications to a Person other than a Governmental Entity, would not, individually or in the aggregate, reasonably be expected to (A) prevent or materially delay consummation of the Mergers or other Transactions, (B) otherwise prevent or materially delay performance by Island of any of its material obligations under this Agreement or (C) have an Island Material Adverse Effect.
- 3.5 SEC Documents; Financial Statements; Undisclosed Liabilities.
 - (a) Island has filed or furnished on a timely basis all reports, schedules, forms, statements, registration statements, prospectuses and other documents required to be filed or furnished by Island with the SEC under the Securities Act or the Exchange Act since January 1, 2016 (the "Island SEC Documents").
 - (b) As of its respective filing date, and, if amended, as of the date of the last amendment prior to the date of this Agreement, each Island SEC Document complied in all material respects with the requirements of the Exchange Act or the Securities Act, as the case may be, and the rules and regulations of the SEC promulgated thereunder applicable to such Island SEC Document and did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.
 - (c) The consolidated financial statements of Island included in the Island SEC Documents (including, in each case, any notes or schedules thereto) (the "Island SEC Financial Statements"), comply as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto. The Island SEC Financial Statements fairly present, in all material respects, the financial condition and the results of operations, cash flows and changes in stockholders' equity of Island (on a consolidated basis) as of

Table of Contents

the respective dates of and for the periods referred to in the Island SEC Financial Statements, and were prepared in accordance with GAAP applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto), subject, in the case of interim Island SEC Financial Statements, to normal year-end adjustments (which are not material in significance or amount) and the absence of notes.

- (d) Island is in compliance in all material respects with (i) the applicable provisions of the Sarbanes-Oxley Act of 2002 and the related rules and regulations promulgated thereunder or under the Exchange Act (the "Sarbanes-Oxley Act") and (ii) the applicable listing and corporate governance rules and regulations of the NASDAQ.
- (e) To the Knowledge of Island, there are no outstanding or unresolved comments in comment letters received from the SEC staff with respect to any Island SEC Documents and none of the Island SEC Documents is the subject of ongoing SEC review. To the Knowledge of Island, there are no SEC inquiries or investigations pending or threatened, in each case regarding any accounting practices of Island.
- (f) Island has established and maintains disclosure controls and procedures and internal control over financial reporting (as such terms are defined in paragraphs (e) and (f), respectively, of Rule 13a-15 and paragraph (e) of Rule 15d-15 under the Exchange Act) as required by Rules 13a-15 and 15d-15 under the Exchange Act. Island's disclosure controls and procedures are designed to ensure that all information (both financial and non-financial) required to be disclosed by Island in the reports that it files or furnishes under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC, and that all such information is accumulated and communicated to Island's management as appropriate to allow timely decisions regarding required disclosure and to make the certifications required pursuant to Sections 302 and 906 of the Sarbanes-Oxley Act. Since January 1, 2016, Island has had no significant deficiencies or material weaknesses in the design or operation of its internal control over financial reporting that would reasonably be expected to adversely affect Island's ability to record, process, summarize and report financial information and Island does not have Knowledge of any fraud, whether or not material, that involves management or other employees who have a significant role in Island's internal control over financial reporting.
- (g) Island does not have any liabilities or obligations of any nature (whether absolute or contingent, asserted or unasserted, known or unknown, primary or secondary, direct or indirect, and whether or not accrued or required to be reflected in Island's financial statements in accordance with GAAP), except (i) as disclosed, reflected or reserved against in the most recent audited balance sheet included in the Island SEC Financial Statements or the notes thereto, (ii) for liabilities and obligations incurred in the ordinary course of business since the date of the most recent audited balance sheet included in the Island SEC Financial Statements, (iii) for liabilities and obligations arising out of or in connection with this Agreement, the Mergers or the Transactions and (iv) for liabilities and obligations that, individually or in the aggregate, have not had, and would not reasonably be expected to have, an Island Material Adverse Effect.
- (h) Island is not a party to, or has any commitment to become a party to, any joint venture, off-balance sheet partnership or any similar Contract (including any Contract or arrangement relating to any transaction or relationship between or among Island, on the one hand, and any unconsolidated affiliate, including any structured finance, special purpose or limited purpose entity or Person, on the other hand, or any "off-balance-sheet arrangements" (as defined in Item 303(a) of Regulation S-K under the Exchange Act)), where the result, purpose or intended effect of such Contract is to avoid disclosure of any material transaction involving, or material liabilities of, Island in Island's published financial statements or other Island SEC Documents.

Table of Contents

- 3.6 Absence of Certain Changes or Events. Since October 1, 2017 and through the date of this Agreement, (a) Island has conducted its business in all material respects only in the ordinary course and in a manner consistent with past practice and (b) there has not been any change, event, development, condition or occurrence that, individually or in the aggregate, has had or would reasonably be expected to have an Island Material Adverse Effect. Since October 1, 2017 and through the date of this Agreement, Island has not taken any action that would have constituted a breach of, or required Boat's consent pursuant to, Section 5.1 had the covenants therein applied since October 1, 2017.
- 3.7 Information Supplied. None of the information supplied or to be supplied by Island for inclusion or incorporation by reference in (i) the registration statement on Form S-4 to be filed with the SEC by Boat in connection with the Mergers (the "Form S-4") will, at the time the Form S-4 is filed with the SEC, and at any time it is amended or supplemented or at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, and (ii) the Joint Proxy Statement will, at the date it or any amendment or supplement is mailed to each of the holders of Boat Common Stock and Island Common Stock and at the time of each of the Boat Stockholders Meeting and Island Stockholders Meeting, contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in light of the circumstances in which they are made, not misleading (except that no representation or warranty is made by Island to such portions thereof that relate expressly to Boat or any of its Subsidiaries, including Holdco, Merger Sub A or Merger Sub B, or to statements made therein based on information supplied by or on behalf of Boat for inclusion or incorporation by reference therein). The Form S-4 and the Joint Proxy Statement will comply as to form in all material respects with the requirements of the Securities Act or Exchange Act, as applicable, and other applicable Law.
- 3.8 Legal Proceedings. As of the date hereof, there are no Proceedings pending, or to the Knowledge of Island, threatened against Island or any of its assets, rights or properties or any of the officers or directors of Island, that would, in each case, individually or in the aggregate, reasonably be expected to prevent or materially delay the consummation of the Transactions. There are no Proceedings pending, or to the Knowledge of Island, threatened against Island or any of its assets, rights or properties or any of the officers or directors of Island, except, in each case, for those that, individually or in the aggregate, have not had, and would not reasonably be expected to have, an Island Material Adverse Effect. None of Island or any of its properties, rights or assets is or are subject to any Order, except for those that, individually or in the aggregate, have not had, and would not reasonably be expected to have, an Island Material Adverse Effect.
 - 3.9 Compliance with Laws.
 - (a) (i) Island is in compliance, and for the past five (5) years has been in compliance, with all Laws and Orders applicable to Island or any assets owned or used by it (except for such past noncompliance as has been remedied and imposes no continuing obligations or costs on Island), and (ii) Island has not received any written communication during the past five (5) years from a Governmental Entity that alleges that Island is not in compliance with any such Law or Order, except in the case of clauses (i) and (ii) where any non-compliance, individually or in the aggregate, has not had and would not reasonably be expected to have, an Island Material Adverse Effect.
 - (b) Island (i) is in compliance, and for the past five (5) years has been in compliance, in all material respects with the United States Foreign Corrupt Practices Act of 1977 (the "FCPA") and any other applicable Anti-corruption Laws; (ii) during the past five (5) years has not been investigated, to the Knowledge of Island, by any Governmental Entity with respect to, or been given notice by a Governmental Entity or any other Person of, any actual or alleged violation by

Table of Contents

Island of the FCPA or any other Anti-corruption Laws; and (iii) during the past five (5) years has had an operational and effective FCPA and anti-corruption compliance program that includes, at a minimum, policies, procedures and training intended to enhance awareness of and compliance by Island with the FCPA and any other applicable Anti-corruption Laws.

- (c) To the Knowledge of Island, Island has not, directly or indirectly through its Representatives or any Person authorized to act on its behalf (including any distributor, agent, sales intermediary or other third party), offered, promised, paid, authorized or given money or anything of value to any Person for the purpose of: (i) influencing any act or decision of any Government Official or Other Covered Party; (ii) inducing any Government Official or Other Covered Party to do or omit to do an act in violation of a lawful duty; (iii) securing any improper advantage; or (iv) inducing any Government Official or Other Covered Party to influence the act or decision of a government or government instrumentality, in order to obtain or retain business, or direct business to, any Person or entity, in any way.
- (d) To the Knowledge of Island, during the past five (5) years, Island has maintained complete and accurate books and records, including records of payments to any agents, consultants, representatives, third parties and Government Officials, in accordance with GAAP, in all material respects. There have been no false or fictitious entries made in the books and records of Island relating to any unlawful offer, payment, promise to pay, or authorization of the payment of any money, or unlawful offer, gift, promise to give, or authorization of the giving of anything of value, including any bribe, kickback or other illegal or improper payment, and Island has not established or maintained a secret or unrecorded fund.
- (e) To the Knowledge of Island, during the past five (5) years, Island has not had a customer or supplier or other business relationship with, is not a party to any Contract with, or has not engaged in any transaction with, any Person (i) that is organized or domiciled in or that is a citizen of Crimea, Cuba, Iran, North Korea or Syria (including any Governmental Entity within such country) or (ii) that is the subject of any international economic or trade sanction administered or enforced by the Office of Foreign Assets Control of the United States Department of the Treasury ("*OFAC*"), the United Nations Security Council, the European Union, Her Majesty's Treasury, the United Kingdom Export Control Organization or other relevant sanctions authority (including but not limited to being listed on the Specially Designated Nationals and Blocked Persons List administered by OFAC).
- 3.10 *Permits*. Island has all required governmental permits, licenses, franchises, certificates, registrations, approvals, exemptions, clearances, billings and authorizations and similar rights ("*Permits*") necessary for the conduct of its business and the use of its properties and assets, as presently conducted and used, and each of the Permits is valid, subsisting and in full force and effect, except where the failure to have or maintain such Permit, individually or in the aggregate, has not had and would not reasonably be expected to have an Island Material Adverse Effect. Except as would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, (i) the operation of the business of Island as currently conducted is not, and has not been since January 1, 2016, in violation of, nor is Island in default or violation under, any Permit (except for such past violation or default as has been remedied and imposes no continuing obligations or costs on Island), and (ii) to the Knowledge of Island, no event has occurred which, with notice or the lapse of time or both, would constitute a default or violation of any term, condition or provision of any Permit. There are no actions pending or, to the Knowledge of Island, threatened, that seek the revocation, cancellation or modification of any Permit, except where such revocation, cancellation or modification, individually or in the aggregate, has not had and would not reasonably be expected to have, an Island Material Adverse Effect. Since January 1, 2016, Island has not received written notice of any charge, claim or assertion alleging any violations of or noncompliance with any Permit, nor to the Knowledge of Island, has any charge, claim or assertion been threatened, except where such notice, charge, claim

Table of Contents

or assertion, individually or in the aggregate, has not had and would not reasonably be expected to have, an Island Material Adverse Effect.

3.11 Employee Benefit Plans.

- (a) Section 3.11(a) of the Island Disclosure Schedule sets forth a true and complete list of each material "employee benefit plan" as defined in Section 3(3) of ERISA and any other plan, policy, program, Contract, or arrangement providing compensation or other benefits to any current or former director, officer or employee (or to any dependent or beneficiary thereof) of Island, in each case that is maintained, sponsored or contributed to by Island, or under which Island has any material obligation or material liability, whether actual or contingent, including each employment, incentive, bonus, deferred compensation, profit-sharing, pension, retirement, vacation, holiday, sick pay, cafeteria, fringe benefit, medical, disability, retention, severance, termination, change in control, stock purchase, stock option, stock appreciation, phantom stock, restricted stock, restricted stock unit or other equity-based compensation plans, policies, programs, practices or arrangements, in each case, which (x) is not sponsored and administered by a Governmental Entity and (y) is not required by Law to be provided (each, an "Island Benefit Plan"). Neither Island, nor to the Knowledge of Island, any other Person, has any express or implied commitment, whether legally enforceable or not, to (i) modify, change or terminate any Island Benefit Plan, other than with respect to a modification, change or termination required by ERISA or the Code or (ii) adopt any new Island Benefit Plan.
- (b) With respect to each Island Benefit Plan, Island has made available to Boat a current written copy thereof (if any) and, to the extent applicable: (i) any amendments and any related trust agreement; (ii) the most recent IRS determination letter; (iii) the most recent summary plan description and summary of material modifications, and (iv) for the most recent plan year (A) the Form 5500 and attached schedules and (B) audited financial statements.
- (c) Except as has not had or would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, each Island Benefit Plan has been administered in accordance with its terms and all applicable Laws, including ERISA and the Code, and contributions required to be made under the terms of any of the Island Benefit Plans have been timely made or, if not yet due, have been properly reflected on the most recent consolidated balance sheet filed or incorporated by reference in the Island SEC Documents prior to the date of this Agreement.
- (d) Except as has not resulted or would not reasonably be expected to result, individually or in the aggregate, in an Island Material Adverse Effect: (i) each Island Benefit Plan which is intended to qualify under Section 401(a) of the Code has either received a favorable determination letter or opinion letter from the IRS as to its qualified status, and each trust established in connection with any Island Benefit Plan which is intended to be exempt from federal income taxation under Section 501(a) of the Code is so exempt, and to Island's Knowledge no fact or event has occurred that could adversely affect the qualified status of any such Island Benefit Plan or the exempt status of any such trust, (ii) there has been no prohibited transaction (within the meaning of Section 406 of ERISA or Section 4975 of the Code), other than a transaction that is exempt under a statutory or administrative exemption, with respect to any Island Benefit Plan, and (iii) no Proceeding has been brought, or to the Knowledge of Island is threatened, against or with respect to any Island Benefit Plan, including any audit or inquiry by the IRS or United States Department of Labor (other than for routine benefits claims).
- (e) No Island Benefit Plan is, and neither Island nor any of its ERISA Affiliates contributes to, has during the preceding six (6) years contributed to or has any liability or obligation, whether fixed or contingent, with respect to a multiemployer plan (as defined in Section 3(37) or Section 4001(a)(3) of ERISA) ("Multiemployer Plan") or other plan subject to Title IV of ERISA

Table of Contents

or the minimum funding requirements of Section 302 of ERISA or Section 412 of the Code. No material liability under Title IV of ERISA has been incurred by Island or any ERISA Affiliate thereof that has not been satisfied in full, and no condition exists that would reasonably be expected to have an Island Material Adverse Effect (whether primarily, jointly or secondarily) under Title IV of ERISA.

- (f) Except as has not had or would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, no amount that could be received (whether in cash or property or the vesting of property), as a result of the consummation of the Mergers or other Transactions (either alone or in combination with another event), by any employee, officer or director of Island who is a "disqualified individual" (within the meaning of Section 280G of the Code) could be characterized as an "excess parachute payment" (within the meaning of Section 280G(b)(1) of the Code).
- (g) Except as required by Law, no Island Benefit Plan provides post-employment medical, disability or life insurance benefits to any former director, employee or their respective dependents (other than as required by health care continuation coverage as required by Section 4980B of the Code or any similar state law ("COBRA") or ERISA or coverage through the end of the calendar month in which a termination of employment occurs.
- (h) Except for the adjustment and assumption of the Island Stock Options in accordance with *Section 2.7*, neither the execution of this Agreement nor the consummation of the Mergers or other Transactions (either alone or in combination with another event) will (i) entitle any employee or director of Island to a bonus, severance or change in control payment, (ii) accelerate the time of payment or vesting or trigger any payment or funding (through a grantor trust or otherwise) of compensation or benefits, (iii) increase the amount payable or trigger any other material obligation pursuant to any of the Island Benefit Plans or (iv) result in any breach or violation of, or default under any Island Benefit Plan.
- (i) Except as has not had or would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, each Island Benefit Plan that constitutes in any part a nonqualified deferred compensation plan within the meaning of Section 409A of the Code has been operated and maintained in operational and documentary compliance with Section 409A of the Code and all IRS guidance promulgated thereunder, to the extent such section and such guidance have been applicable to such Island Benefit Plan. There is no agreement, plan, Contract or other arrangement to which Island is a party or by which it is otherwise bound to compensate any Person in respect of Taxes pursuant to Sections 409A or 4999 of the Code.
- (j) No Island Benefit Plan is governed by the laws of any jurisdiction outside of the United States or provides compensation or benefits to any former director, employee or their respective dependents, who resides outside of the United States.

3.12 Employee and Labor Matters.

(a) Island is not a party to or bound by any collective bargaining agreement, agreement with any works council, or labor contract, no labor union, labor organization, works council, or group of employees of Island has made a pending demand for recognition or certification, and there are no representation or certification proceedings or petitions seeking a representation proceeding presently pending or, to the Knowledge of Island, threatened in writing to be brought or filed with the National Labor Relations Board or any other labor relations tribunal or authority. Island has not engaged in any unfair labor practice with respect to any individuals employed by or otherwise performing services for Island (the "Island Business Personnel"), and there is no unfair labor practice complaint or grievance or other administrative or judicial complaint, action or investigation pending or, to the Knowledge of Island, threatened in writing against Island by the

Table of Contents

National Labor Relations Board or any other Governmental Entity with respect to any present or former Island Business Personnel which would reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect. Except as would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, there is no labor strike, dispute, lockout, slowdown or stoppage pending or, to the Knowledge of Island, threatened against or affecting Island.

- (b) Island is not required to provide notice to any work council or similar representative body prior to the execution of this Agreement or the consummation of the Transactions, except where the failure to provide such notice would not, individually or in the aggregate, reasonably be expected to (i) result in material liability to Island or (ii) materially delay or prevent the consummation of the Transactions.
- (c) To the Knowledge of Island, Island is and has for the past five (5) years been in compliance with all applicable Laws respecting employment and employment practices, including, without limitation, all Laws respecting terms and conditions of employment, health and safety, wage payment, wages and hours, child labor, collective bargaining, immigration and work authorizations, employment discrimination, retaliation, civil rights, veterans' rights, disability rights or benefits, equal opportunity, plant closures and layoffs, affirmative action, workers' compensation, labor relations, social welfare obligations, proper classification of employees as exempt and non-exempt and as employees and independent contractors, unemployment insurance and the collection and payment of withholding and/or social security Taxes and any similar Tax, except for noncompliance as individually or in the aggregate, has not had and would not reasonably be expected to have, an Island Material Adverse Effect.
- (d) To the Knowledge of Island, no Island Business Personnel is in any respect in violation of any term of any employment agreement, nondisclosure agreement, common law nondisclosure obligation, fiduciary duty, noncompetition agreement, restrictive covenant or other obligation to Island or to a former employer of any such employee relating (i) to the right of any such individual to be employed by or provide services for Island or (ii) to the knowledge or use of trade secrets or proprietary information, in each case except as, individually or in the aggregate, currently does not have and would not reasonably be expected to have, an Island Material Adverse Effect.
- (e) To the Knowledge of Island, Island is not delinquent in payments to any current or former Island Business Personnel for any services or amounts required to be reimbursed or otherwise paid except as currently does not have and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect.
- 3.13 Environmental Matters. Except as, individually or in the aggregate, has not had and would not reasonably be expected to have, an Island Material Adverse Effect:
 - (a) Island (i) has been within the last five (5) years, and is, in compliance with all, and is not subject to any liability with respect to noncompliance with any, applicable Environmental Laws, (ii) has and holds, or has applied for, all Environmental Permits necessary for the conduct of its business and the use of its properties and assets, as currently conducted and used, and (iii) is in compliance with its Environmental Permits.
 - (b) There are no Environmental Claims pending, nor to the Knowledge of Island, threatened against Island, and Island has not received any notification of any allegation of actual or potential responsibility for any Release or threatened Release of any Hazardous Materials with respect to any location.

Table of Contents

- (c) There have been no Releases of Hazardous Materials at any properties that are owned, operated, leased or used by Island, or to the Knowledge of Island, at properties that were formerly owned, operated, leased or used by Island, that are reasonably likely to cause Island to incur liability pursuant to applicable Environmental Law.
- (d) Island (i) has not entered into or agreed to any consent decree or consent order or is otherwise subject to any judgment, decree, or judicial or administrative Order relating to compliance with Environmental Laws or Environmental Permits, the investigation, sampling, monitoring, treatment, remediation, response, removal or cleanup of Hazardous Materials, and no Proceeding is pending, or to the Knowledge of Island is threatened, with respect thereto, and (ii) is not an indemnitor by contract or otherwise in connection with any claim, demand, suit or action threatened or asserted by any third-party for any liability under any Environmental Law or otherwise relating to any Hazardous Materials.

3.14 Real Property.

- (a) Island owns no Real Property.
- (b) Island has a valid and subsisting leasehold or subleasehold interest in the Island Leased Real Property free and clear of all Liens, except for Permitted Liens.
- (c) The Island Leased Real Property constitutes all real property necessary for the conduct of the business of Island as currently conducted. Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, Island has not received written notice of any Proceedings in eminent domain, condemnation or other similar Proceedings that are pending, and to Island's Knowledge there are no such Proceedings threatened, affecting any portion of the Island Leased Real Property and Island has not received written notice of the existence of any Order or of any pending Proceeding relating to the ownership, lease, use, occupancy or operation by any Person of the Island Leased Real Property. Island has not leased, subleased, licensed or otherwise granted any Person a material right to use or occupy any of the Island Leased Real Property or any material portion thereof. Island has not granted any option or other right to any third party to purchase any of the Island Leased Real Property or any material portion thereof. Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, each Island Leased Real Property and all buildings and improvements located on the Island Leased Real Property are in a state of good operating condition, subject to reasonable wear and tear.

3.15 Tax Matters.

- (a) Except as would not, individually or in the aggregate, reasonably be expected to have an Island Material Adverse Effect:
 - (i) (A) All Tax Returns that are required to be filed by or with respect to Island have been timely filed (taking into account any extension of time within which to file), (B) all such Tax Returns are or will be true, complete and accurate, (C) Island has paid all Taxes due and owing by it (whether or not shown on any Tax Return) or appropriate reserves have been recorded in the books and records of Island, (D) Island has established adequate accruals and reserves, in accordance with GAAP, on the financial statements included in the Island SEC Documents for all Taxes payable by Island for all taxable periods and portions thereof through the date of such financial statements and (E) Island has withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, stockholder or other third party; such withheld amounts were either timely paid to the appropriate Taxing Authority or set aside in accounts for such purpose and were reported to the appropriate Taxing Authority and to each such employee, independent contractor, creditor, stockholder or other third party, as required under Law.

Table of Contents

- (ii) No written agreement or other written document waiving or extending, or having the effect of waiving or extending, the statute of limitations or the period of assessment or collection of any Taxes relating to Island has been filed or entered into with any Taxing Authority, and no power of attorney with respect to any such Taxes has been granted to any Person that is in force.
- (iii) (A) Island is not the subject of any currently ongoing Tax audit or other proceeding with respect to any Taxes or Tax Return nor has any audit or other proceeding with respect to Taxes been proposed against it in writing and any deficiencies asserted or assessments made as a result of any audit or other proceeding with respect to Taxes have been resolved, are being contested in good faith, or adequate accruals or reserves for such deficiencies or assessments have been established and (B) Island has not received written notice of any deficiencies for any Tax of Island from any Taxing Authority for which there are not adequate reserves on the financial statements included in the Island SEC Documents.
- (iv) No claim has ever been made in writing by a Taxing Authority of a jurisdiction where Island has not filed Tax Returns that Island is or may be subject to taxation by that jurisdiction.
- (v) Island has not constituted, in the two (2) years prior to the date of this Agreement, a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of stock intended to qualify for tax-free treatment under Section 355 of the Code (or any similar provision of state, local, or non-U.S. Law) or otherwise as part of a plan (or series of related transactions), within the meaning of Section 355(e) of the Code, that includes the Merger.
- (vi) Island (A) is not a party to or bound by or has any obligation under any Tax indemnification, separation, sharing or similar agreement or arrangement (other than such an agreement or arrangement entered into in the ordinary course of business which does not relate primarily to Taxes), (B) is not and has not been a member of any consolidated, combined, unitary or similar group for purposes of filing Tax Returns or paying Taxes, (C) has not entered into a closing agreement pursuant to Section 7121 of the Code or any other binding agreement with a Taxing Authority that would have an effect on the determination of Island's liability to Tax in a tax year ending after the Effective Time, or any predecessor provision or any similar provision of foreign, state or local Tax law and (D) has no liability for Taxes of any Person (other than Island) under U.S. Treasury Regulation Section 1.1502-6 (or any similar provision of state, local, or non-U.S. Law) or as transferee or successor.
 - (vii) There are no Liens for Taxes upon any property or assets of Island, except for Taxes not yet due and payable.
- (viii) Island has not entered into any "listed transaction" within the meaning of Section 6707A(c)(2) of the Code and U.S. Treasury Regulation Section 1.6011-4(b)(2) (or any similar provision of state, local or non-U.S. Law).
- (ix) Island has made available to Boat or its legal or accounting representative copies of all U.S. federal and state income Tax Returns for Island filed for all periods including and after the period ended December 31, 2014.
- (x) Island is not a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code.
- (xi) Island (A) has not filed any extension of time within which to file any Tax Returns that have not been filed, except in the ordinary course of business and (B) will not be required to include any item of income or gain in, or be required to exclude any item of

Table of Contents

deduction of loss from, any period ending after the Closing Date as a result of any (i) installment sale or open transaction made on or prior to the Closing Date, (ii) election under Section 108(i) of the Code or (iii) prepaid amount received prior to the Closing Date.

- (xii) Island has not agreed to, requested, and is not required to include any adjustment under Section 481 of the Code (or any corresponding provision of applicable state, local or non-U.S. Tax Law) by reason of a change in accounting method or otherwise.
- (b) Island has no knowledge of any facts, agreements, plans or other circumstances or has taken or agreed to take any action that would reasonably be expected to prevent or impede (i) the Mergers from together being treated as an "exchange" described in Section 351 of the Code or (ii) the Boat Merger from qualifying as a "reorganization" within the meaning of Section 368(a) of the Code.
- (c) It is agreed and understood that no representation or warranty is made by Island in respect of Tax matters in any Section of this Agreement other than Section 3.5, Section 3.6, Section 3.7, Section 3.11 and this Section 3.15.

3.16 Material Contracts.

- (a) All Contracts required to be filed as exhibits to the Island SEC Documents have been so filed in a timely manner. *Section 3.16(a)* of the Island Disclosure Schedule sets forth a true and complete list, as of the date hereof, of each of the following Contracts, excluding any Island Benefit Plans, to which Island is a party or by which Island or any of its assets or businesses is bound (and any amendments, supplements and modifications thereto):
 - (i) any Contract that is a "material contract" (as such term is defined in Item 601(b)(10) of Regulation S-K of the Exchange Act);
 - (ii) any Contract that materially limits the ability of Island or any of its affiliates (including, following the consummation of the Transactions, the Surviving Corporation and its affiliates) to engage or compete in any line of business or in any geographic area or from developing or commercializing compounds, any therapeutic area, class of drugs or mechanism of action;
 - (iii) any Contract required to be disclosed pursuant to Item 404 of Regulation S-K of the Exchange Act;
 - (iv) any Contract or series of related Contracts relating to indebtedness for borrowed money (A) in excess of \$2,000,000 or (B) that becomes due and payable as a result of the Transactions;
 - (v) any license, sublicense, option, development or collaboration agreement or other Contract relating to Island Material Intellectual Property reasonably expected to result in payments in excess of \$2,000,000 in any twelve (12) month period after the Closing Date (excluding license agreements for "shrink-wrap," "click-wrap" or other commercially available off-the-shelf software that is not the subject of a negotiated agreement, and excluding agreements the primary purpose of which is to purchase tangible goods or procure services unrelated to Intellectual Property and in the ordinary course of business);
 - (vi) any Contract reasonably expected to result in payments in excess of \$2,000,000 in any twelve (12) month period after the Closing Date;
 - (vii) any purchase, sale or supply contract that contains volume requirements or commitments, exclusive or preferred purchasing arrangements, most favored nation provisions or promotional requirements;

Table of Contents

- (viii) any lease, sublease, occupancy agreement or other Contract with respect to the Island Leased Real Property reasonably expected to result in payments in excess of \$2,000,000 in any twelve (12) month period after the Closing Date;
- (ix) any agreement that grants any right of first refusal or right of first offer or similar right or that limits or purports to limit the ability of Island to own, operate, sell, transfer, pledge or otherwise dispose of any material amount of assets or businesses (in any case in excess of \$2,000,000);
- (x) any acquisition or divestiture agreement (A) entered into since January 1, 2015 with a purchase price in excess of \$2,000,000 or (B) that contains "earn-out" provisions or other contingent payment obligations that could reasonably be expected to exceed \$2,000,000 (including indemnification obligations) that have not been satisfied in full;
 - (xi) any agreement that by its terms limits the payment of dividends or other distributions by Island;
- (xii) any Contract for any joint venture, partnership or similar arrangement, or any Contract involving a sharing of revenues, profits, losses, costs, or liabilities by Island with any other Person;
- (xiii) any "single source" supply Contract pursuant to which goods or materials that are material to Island are supplied to Island from an exclusive source; or
 - (xiv) any Contract with any Governmental Entity.
- (b) Island has heretofore made available to Boat true, correct and complete copies of the Contracts set forth in Section 3.16(a).
- (c) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, (i) all Contracts set forth or required to be set forth in *Section 3.16(a)* of the Island Disclosure Schedule or filed or required to be filed as exhibits to the Island SEC Documents (the "*Island Material Contracts*") are valid, binding and in full force and effect and are enforceable by Island in accordance with their terms, except as limited by Laws affecting the enforcement of creditors' rights generally, by general equitable principles or by the discretion of any Governmental Entity before which any Proceeding seeking enforcement may be brought, (ii) Island has performed all obligations required to be performed by it under the Island Material Contracts, and it is not (with or without notice or lapse of time, or both) in breach or default thereunder and, to the Knowledge of Island, no other party to any Island Material Contract is (with or without notice or lapse of time, or both) in breach or default thereunder, (iii) since January 1, 2017, Island has not received written notice of any actual, alleged, possible or potential violation of, or failure to comply with, any term or requirement of any Island Material Contract, and (iv) Island has not received any written notice of the intention of any party to cancel, terminate, materially change the scope of rights under or fail to renew any Island Material Contract

3.17 *Intellectual Property*.

(a) Section 3.17(a)(1) of the Island Disclosure Schedule sets forth a complete and accurate list of all (i) issued patents and pending patent applications, (ii) trademark and service mark registrations and applications, (iii) copyright registrations and applications, and (iv) internet domain name registrations, in each case that are owned by Island (collectively, the "Solely Owned Island Registered Intellectual Property"). Section 3.17(a)(2) of the Island Disclosure Schedule sets forth a complete and accurate list of all (i) issued patents and pending patent applications, (ii) trademark and service mark registrations and applications, (iii) copyright registrations and applications, and (iv) internet domain name registrations, in each case that are licensed to Island

Table of Contents

(collectively, the "Licensed Island Registered Intellectual Property"). Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, Island owns all right, title and interest in the Solely Owned Island Registered Intellectual Property free and clear of all Liens other than Permitted Liens or Liens created by Boat. To the Knowledge of Island, each item of material Solely Owned Island Registered Intellectual Property that is shown as registered, filed, issued or applied for in Section 3.17(a) of the Island Disclosure Schedule has been duly registered in, filed in or issued by an official governmental register and/or issuer (or officially recognized register or issuer) for such Intellectual Property and each such registration, filing, issuance and/or application, to the Knowledge of Island, (x) has not been abandoned or cancelled, (y) has been maintained effective by all requisite filings, renewals and payments, and (z) remains in full force and effect.

- (b) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, to the Knowledge of Island, Island owns and possesses all right, title and interest in and to (or has the right pursuant to a valid and enforceable license or otherwise possesses legally enforceable rights to use) all Intellectual Property that is necessary for or used or held for use in the conduct of the business of Island in substantially the same manner as presently conducted. Neither the execution and delivery of this Agreement by Island, nor the performance of this Agreement by Island, will result in the loss, forfeiture, termination, or impairment of, or give rise to a right of any Person to limit, terminate, or consent to the continued use of, any rights of Island in any Island Registered Intellectual Property except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect.
- (c) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, to the Knowledge of Island neither Island nor the conduct of its business is infringing, misappropriating, diluting, or otherwise violating the Intellectual Property rights of any Person. Island has not received any written charge, complaint, claim, demand, or notice since January 1, 2015 (or earlier, if presently not resolved) alleging any such infringement, misappropriation, dilution, or violation (including any claim that Island must license or refrain from using any Intellectual Property rights of any Person) which alleged infringement, misappropriation, dilution, or violation, if true, would reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect. To the Knowledge of Island, no Person is infringing, misappropriating, diluting or otherwise violating any Intellectual Property owned by Island except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect. Island has not made or asserted any charge, complaint, claim, demand or notice since January 1, 2015 (or earlier, if presently not resolved) alleging any such infringement, misappropriation, dilution, or violation which alleged infringement, misappropriation, dilution, or violation, if true, would reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect.
- (d) Island has taken commercially reasonable steps to maintain, police and protect the Intellectual Property that is material to its business ("Island Material Intellectual Property"). Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, all Island Material Intellectual Property that derives actual or potential economic value from not being generally known to the public or to other persons who can obtain economic value from its disclosure or use has been maintained in confidence in accordance with protection procedures that are in accordance with procedures customarily used in the industry to protect rights of like importance and, to the Knowledge of Island, adequate for protection against unauthorized disclosure or use. To the Knowledge of Island, there has been no unauthorized disclosure of any Island Material Intellectual Property. Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material

Table of Contents

Adverse Effect, to the Knowledge of Island, all former and current officers, directors, employees, personnel, consultants, advisors, agents, and independent contractors of Island, and each of its predecessors, who have contributed to or participated in the conception and development of Intellectual Property for such entities have entered into valid and binding proprietary rights agreements with Island or one of its predecessors, vesting ownership of such Intellectual Property in Island. No such Person has asserted, and to the Knowledge of Island, no such Person has, any right, title, interest or other claim in, or the right to receive any royalties or other consideration with respect to, any Island Material Intellectual Property.

(e) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, the IT Assets of Island operate in all material respects in accordance with their documentation and functional specifications and as required by Island to operate its business as presently conducted and have not, since January 1, 2015, materially malfunctioned or failed. Island has implemented commercially reasonable measures to protect the confidentiality and security of such IT Assets and information stored or contained therein against any unauthorized use, access, interruption or corruption. Island has implemented commercially reasonable data backup, data storage, system redundancy and disaster avoidance procedures with respect to their IT Assets.

3.18 Healthcare Compliance Matters.

- (a) (i) Island is in compliance and since January 1, 2015 has been in compliance with all Health Care Laws applicable to Island or any assets owned or used by it and (ii) Island has not received any written communication or has been subject to any Proceeding (other than routine FDA inspections) since January 1, 2015 from a Governmental Entity that alleges that Island is not in compliance with any Health Care Law, except in the case of clauses (i) and (ii) where any non-compliance, individually or in the aggregate, has not had and would not reasonably be expected to have, an Island Material Adverse Effect. Island is not a party to and does not have any ongoing obligations pursuant to or under any corporate integrity agreements, deferred prosecution agreements, monitoring agreements, consent decrees, settlement orders, plans of correction or similar agreements with or imposed by any Governmental Entity. Additionally, none of Island or any of its employees, officers or directors has been excluded, suspended or debarred from participation in any U.S. state or federal health care program or, to the Knowledge of Island, has been convicted of any crime or is subject to a governmental inquiry, investigation, Proceeding, or other similar action, or has engaged in any conduct, that could reasonably be expected to result in debarment, suspension, or exclusion.
- (b) Island has, maintains and is operating in material compliance with all Permits of the United States Food and Drug Administration ("FDA"), Drug Enforcement Administration ("DEA") and comparable Governmental Entities which are required for the conduct of its business as currently conducted (collectively, the "Health Care Permits"), and all such Health Care Permits are valid, subsisting, and in full force and effect, except where the failure to have, maintain or operate in compliance with the Health Care Permits has not had, and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect. Island has fulfilled and performed all of its obligations with respect to the Health Care Permits, and no event has occurred which allows, or with notice or lapse of time or both, would allow revocation or termination thereof or results in any other material impairment of the rights of the holder of any Health Care Permit, except where the failure to so fulfill or perform, or the occurrence of such event, would not result in an Island Material Adverse Effect. There is no Proceeding pending or, to the Knowledge of Island, threatened in writing that could result in the suspension, termination, revocation, cancellation, limitation or impairment of any such Health Care Permit other than those that have not had and would not reasonably expected to have, individually or in the aggregate, an Island Material Adverse Effect.

Table of Contents

- (c) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, all applications, notifications, submissions, information, claims, reports and statistics, and other data and conclusions derived therefrom, utilized as the basis for or submitted in connection with any and all requests for a Health Care Permit relating to Island, its business and product candidates, when submitted to the FDA, DEA or other Governmental Entity were true, complete and correct as of the date of submission and any necessary or required updates, changes, corrections or modification to such applications, submissions, information and data have been submitted to the FDA, DEA or other Governmental Entity.
- (d) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, since January 1, 2015, Island has not had any product, product candidate or manufacturing site (whether owned by Island or a contract manufacturer for its product candidates) subject to a Governmental Entity (including FDA or DEA) shutdown or import or export prohibition, and has not received any FDA Form 483 or other Governmental Entity notice of inspectional observations, "warning letters," "untitled letters" or written requests or requirements to make changes to a product candidate, or similar correspondence or written notice from the FDA, DEA or other Governmental Entity alleging or asserting noncompliance with any applicable Health Care Law, Health Care Permit or such requests or requirements of a Governmental Entity.
- (e) To the Knowledge of Island, there are no facts or circumstances that would be reasonably likely to result in (A) a change in the marketing classification or a material change in the labeling of any product candidate or (B) a termination or suspension of any Health Care Permits for such product candidates.
- (f) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, (i) the clinical, pre-clinical and other studies and tests conducted by or on behalf of or sponsored by Island or in which Island, or any of its product candidates have participated were, and if still pending are, being conducted in accordance with standard medical and scientific research procedures and all applicable Laws, including, but not limited to, the Federal Food, Drug and Cosmetic Act and its applicable implementing regulations and (ii) no investigational new drug application filed by or on behalf of Island with the FDA has been terminated or suspended by the FDA, and neither the FDA nor any applicable foreign Governmental Entity has commenced, or, to the Knowledge of Island, threatened to commence, any action to place a clinical hold order on, or otherwise terminate, delay or suspend, any proposed or ongoing clinical investigation conducted or proposed to be conducted by or on behalf of Island.
- (g) Island is not the subject of any pending or, to the Knowledge of Island, threatened investigation in respect of Island or its product candidates, by the FDA pursuant to its "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities" Final Policy set forth in 56 Fed. Reg. 46191 (September 10, 1991) and any amendments thereto. Island has provided Boat accurate and complete copies of all Health Care Permits and correspondence with any Governmental Entity related to all product candidates of Island.
- 3.19 *Broker's Fees.* Except for the financial advisors' fees set forth in *Section 3.19* of the Island Disclosure Schedule, neither Island nor any of its officers or directors on behalf of Island has employed any financial advisor, broker or finder or incurred any liability for any financial advisory fee, broker's fees, commissions or finder's fees in connection with any of the Transactions.
- 3.20 Opinion of Financial Advisor. Goldman Sachs & Co. LLC, Island's financial advisor, has delivered to the Island Board its opinion in writing or orally, in which case, such opinion will be subsequently confirmed in writing, to the effect that, as of the date hereof and based upon and subject

Table of Contents

to the factors and assumptions set forth therein, the Island Exchange Ratio pursuant to this Agreement is fair from a financial point of view to the holders of the shares of Island Common Stock.

- 3.21 Ownership of Boat Common Stock. Neither Island nor any of its affiliates or associates is, nor at any time during the last three (3) years has Island or any of its affiliates or associates been, an "interested stockholder" of Boat as defined in Section 203 of the DGCL. Island and its affiliates and associates do not beneficially own any shares of Boat Common Stock or other securities of Boat or any options, warrants or other rights to acquire Boat Common Stock or other securities of, or any other economic interest (through derivative securities or otherwise) in, Boat.
- 3.22 *Insurance*. All insurance policies (including policies providing casualty, liability, and workers compensation coverage, but excluding any Island Benefit Plans) to which Island is currently a party are in full force and effect, and, to the Knowledge of Island, have been issued by licensed insurers, all premiums with respect thereto covering all periods up to and including the Closing Date have been paid, and no notice of cancellation or termination has been received with respect to any such policies, except for such cancellations or terminations which would have not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect. Except as has not had and would not reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect, Island is, and since January 1, 2015 has been, insured with respect to its assets and properties and the conduct of its business in such amounts and against such risks as are in its reasonable judgment sufficient for compliance with Law and as are adequate to protect its assets and properties and the conduct of its business.
- 3.23 No Other Representations or Warranties. Island hereby acknowledges and agrees that, except for the representations and warranties set forth in Article 4, (a) none of Boat, or any of its affiliates, stockholders or representatives, or any other Person, has made or is making any express or implied representation or warranty with respect to Boat or its business or operations, including with respect to any information provided or made available to Island or any of its affiliates, stockholders or representatives, or any other Person, or, except as otherwise expressly set forth in this Agreement, had or has any duty or obligation to provide any information to Island or any of its affiliates, stockholders or Representatives, or any other Person, in connection with this Agreement, the transactions contemplated hereby or otherwise, and (b) to the fullest extent permitted by Law, none of Boat, nor any of its affiliates, stockholders or representatives, or any other Person, will have or be subject to any liability or indemnification or other obligation of any kind or nature to Island, or any of its affiliates, stockholders or representatives, or any other Person, resulting from the delivery, dissemination or any other distribution to Island or any of its affiliates, stockholders or representatives, or any other Person, of any such information provided or made available to any of them by Boat, or any of its affiliates, stockholders or representatives, or any other Person, and (subject to the express representations and warranties of Boat set forth in Article 4) Island expressly disclaims reliance on any such information (including the accuracy or completeness thereof) or any representations and warranties or other statements or omissions that may have been made by Boat or any Person with respect to Boat other than the representations and warranties set forth in this Agreement.

ARTICLE 4 REPRESENTATIONS AND WARRANTIES OF BOAT

Except (a) as set forth in the Boat Disclosure Schedule (subject to *Section 8.14*) and (b) as otherwise disclosed or identified in the Boat SEC Documents publicly filed or furnished after December 31, 2015 and prior to the date hereof (other than (i) any forward-looking disclosures contained in the "Forward Looking Statements" and "Risk Factors" sections of the Boat SEC Documents and any other disclosures included therein to the extent they are primarily predictive, cautionary or forward looking in nature and (ii) information included in, or incorporated by reference

Table of Contents

as, exhibits and schedules to any Boat SEC Document, and provided that the exception provided for in this clause (b) shall be applied if, and only if, the nature and content of the applicable disclosure in any such Boat SEC Document publicly filed or furnished prior to the date hereof is reasonably specific as to matters and items such that the subject matter of such disclosure is reasonably apparent on the face of the text of such disclosure to be applicable to the representation set forth herein), Boat hereby represents and warrants to Island as follows:

4.1 Corporate Organization. Each of Boat and, except where the failure to be in good standing has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, its Subsidiaries, including Holdco, Merger Sub A and Merger Sub B, is a corporation or other entity duly organized, validly existing and, to the extent applicable, in good standing under the laws of the jurisdiction of its organization and has the requisite corporate or other entity power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted. Each of Boat and its Subsidiaries, including Holdco, Merger Sub A and Merger Sub B, is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed or qualified, has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect. Boat has provided Island true, complete and correct copies of the Restated Certificate of Incorporation, as amended (the "Boat Charter") and Amended and Restated By-Laws (the "Boat Bylaws") of Boat, as in effect as of the date of this Agreement. Boat is not in violation of any of the provisions of the Boat Charter or the Boat Bylaws.

4.2 Boat Capitalization.

(a) The authorized capital stock of Boat consists of two hundred million (200,000,000) shares of Boat Common Stock. As of the Capitalization Date, (i) 98,432,868 shares of Boat Common Stock (other than treasury shares) were issued and outstanding, all of which were validly issued and fully paid, nonassessable and free of preemptive rights, (ii) no shares of Boat Common Stock were held in the treasury of Boat or by its Subsidiaries, (iii) 14,647,465 shares of Boat Common Stock were issuable (and such number was reserved for issuance) upon exercise of Boat Stock Options and Boat RSUs outstanding as of such date, (iv) 326,270 shares of Boat Common Stock were authorized for issuance pursuant to the Boat ESPP and (v) no shares of Boat Preferred Stock were issued and outstanding. Except for Boat Stock Options and Boat RSUs covering not more than 14,647,465 shares of Boat Common Stock (at maximum performance levels) and 326,270 shares of Boat Common Stock authorized for issuance pursuant to the Boat ESPP, there are no options, warrants or other rights, agreements, arrangements or commitments of any character to which Boat or any of its Subsidiaries is a party or by which Boat or any of its Subsidiaries is bound relating to the issued or unissued capital stock or other Equity Interests of Boat or any of its Subsidiaries, or securities convertible into or exchangeable for such capital stock or other Equity Interests, or obligating Boat or any of its Subsidiaries to issue or sell any shares of its capital stock or other Equity Interests, or securities convertible into or exchangeable for such capital stock of, or other Equity Interests in, Boat or any of its Subsidiaries. Since the Capitalization Date, Boat has not issued any shares of its capital stock or other Equity Interests, or securities convertible into or exchangeable for such capital stock or other Equity Interests, other than those shares of capital stock reserved for issuance described in this Section 4.2(a). Boat has previously provided Island with a true and complete list, by holder and as of the date hereof, of the prices at which each outstanding Boat Stock Option may be exercised, the number of shares of Boat Common Stock underlying such Boat Stock Options and Boat RSUs and the vesting schedule of the Boat Stock Options and Boat RSUs and whether the Boat Stock Option is intended to be an "incentive stock option" within the meaning of Section 422 of the Code. All shares of Boat Common Stock subject to issuance under a Boat Stock Plan, upon issuance prior to the Effective Time on the terms and

Table of Contents

conditions specified in the instruments pursuant to which they are issuable, will be duly authorized, validly issued, fully paid, nonassessable and free of preemptive rights. There are no bonds, debentures, notes or other indebtedness of Boat having the right to vote on any matters on which stockholders of Boat may vote. Neither Boat nor any of its Subsidiaries is a party to any voting agreement with respect to the voting of any Equity Interests of Boat or any of its Subsidiaries.

- (b) There are no outstanding contractual obligations of Boat or any of its Subsidiaries (i) restricting the transfer of, (ii) affecting the voting rights of, (iii) requiring the repurchase, redemption or disposition of, or containing any right of first refusal with respect to, (iv) requiring the registration for sale of, or (v) granting any preemptive or antidilutive right with respect to, any shares of Boat Common Stock or any capital stock of, or other Equity Interests in, Boat or any of its Subsidiaries.
- (c) Section 4.2(c) of the Boat Disclosure Schedule sets forth a true and complete list of all of the Subsidiaries of Boat and the jurisdiction of organization of each such Subsidiary. Except for Holdco, Merger Sub A and Merger Sub B and as set forth in Section 4.2(c) of the Boat Disclosure Schedule, none of Boat or any of its Subsidiaries holds an Equity Interest in any other Person. Each outstanding share of capital stock or other Equity Interest in each Subsidiary of Boat is duly authorized, validly issued, fully paid, nonassessable and free of preemptive rights and is owned, beneficially and of record, by Boat or one or more of its wholly owned Subsidiaries free and clear of all Liens. There are no outstanding contractual obligations of Boat or any of its Subsidiaries to provide funds to, or make any investment (in the form of a loan, capital contribution or otherwise) in, any Subsidiary of Boat (in excess of \$2,000,000) or any other Person, other than guarantees by Boat of any indebtedness or other obligations of any wholly owned Subsidiary of Boat.
- 4.3 Authority; Execution and Delivery; Enforceability.
 - (a) Boat and Holdco have all necessary power and authority to execute and deliver this Agreement, to perform and comply with their obligations under this Agreement and, subject to the receipt of the Boat Stockholder Approval and to the adoption of this Agreement by Holdco as the sole stockholder of Merger Sub A and Merger Sub B, to consummate the Transactions applicable to such party. The execution and delivery by Boat and Holdco of this Agreement, the performance and compliance by Boat and Holdco with each of its obligations herein and the consummation by Boat and Holdco of the Transactions have been duly authorized by all necessary corporate action on the part of Boat and Holdco, subject to the receipt of the Boat Stockholder Approval and to the adoption of this Agreement by Boat as the sole stockholder of Holdco, and no other corporate proceedings on the part of Boat and Holdco and no other stockholder votes are necessary to authorize this Agreement or the consummation by Boat and Holdco of the Transactions to which it is a party. Each of Boat and Holdco has duly and validly executed and delivered this Agreement and, assuming the due authorization, execution and delivery by Island of this Agreement, this Agreement constitutes its legal, valid and binding obligation, enforceable against it in accordance with its terms, except as limited by Laws affecting the enforcement of creditors' rights generally, by general equitable principles or by the discretion of any Governmental Entity before which any Proceeding seeking enforcement may be brought.
 - (b) The Board of Directors of Boat (the "Boat Board"), at a meeting duly called and held, unanimously adopted resolutions (i) adopting and approving this Agreement and the consummation of the Transactions upon the terms and subject to the conditions set forth in this Agreement, (ii) determining that the terms of this Agreement, the Mergers and the other Transactions are fair to, and in the best interests of, Boat and its stockholders, (iii) directing that this Agreement be submitted to the stockholders of Boat for approval and adoption, (iv) recommending that its stockholders adopt this agreement and (v) declaring that this Agreement is advisable (the "Boat Recommendation").

Table of Contents

(c) To the Knowledge of Boat, no takeover, anti-takeover, business combination, control share acquisition or similar Law applies to this Agreement, the Mergers or the other Transactions. The only vote of holders of any class or series of Equity Interests of Boat necessary to approve the Transactions is the adoption of this Agreement by the holders of a majority of the Boat Common Stock outstanding and entitled to vote thereon at the Boat Stockholders Meeting (together, the "Boat Stockholder Approval"). No other vote of the holders of Boat Common Stock or any other Equity Interests of Boat is necessary to consummate the Transactions.

4.4 No Conflicts.

- (a) The execution and delivery of this Agreement by Boat and Holdco does not and will not, and the performance of this Agreement by Boat and Holdco and the consummation of the Transactions will not, (i) assuming the Boat Stockholder Approval is obtained and this Agreement is adopted by Holdco as the sole stockholder of Merger Sub A and Merger Sub B and this Agreement is adopted by Boat as the sole stockholder of Holdco, conflict with or violate any provision of the Boat Charter or the Boat Bylaws or any equivalent organizational documents of any Subsidiary of Boat, including Holdco, Merger Sub A and Merger Sub B, (ii) assuming that all consents, approvals, authorizations and permits described in Section 4.4(b) have been obtained and all filings and notifications described in Section 4.4(b) have been made and any waiting periods thereunder have terminated or expired, conflict with or violate any Law applicable to Boat or any of its Subsidiaries, including Holdco, Merger Sub A and Merger Sub B, or by which any property or asset of Boat or any of its Subsidiaries, including Holdco, Merger Sub A and Merger Sub B, is bound or affected or (iii) require any consent or approval under, result in any breach of or any loss or impairment of any benefit under, constitute a change of control or default (or an event which with notice or lapse of time or both would become a default) under or give to others any right of termination, vesting, amendment, acceleration or cancellation of, or result in the creation of a Lien on any property or asset of Boat or any of its Subsidiaries, including Holdco, Merger Sub A and Merger Sub B, pursuant to, any Contract or Permit (other than a Boat Benefit Plan), except, with respect to clauses (ii) and (iii), for any such conflicts, violations, breaches, defaults or other occurrences which would not, individually or in the aggregate, reasonably be expected to (A) prevent or materially delay consummation of the Mergers, (B) otherwise prevent or materially delay performance by Boat of any of its material obligations under this Agreement or (C) have a Boat Material Adverse Effect.
- (b) The execution and delivery of this Agreement by Boat and Holdco does not and will not, and the consummation by Boat and Holdco of the Transactions to which it is a party and compliance by Boat and Holdco with any of the terms or provisions hereof or thereof will not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Entity or any other Person, except (i) under the Exchange Act, Securities Act, any applicable blue sky Law and the rules and regulations of the NASDAQ (ii) under the HSR Act or other applicable antitrust Laws, (iii) the filing of the Certificates of Merger as required by the DGCL and (iv) where failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications to a Person other than a Governmental Entity, would not, individually or in the aggregate, reasonably be expected to (A) prevent or materially delay consummation of the Mergers or other Transactions, (B) otherwise prevent or materially delay performance by Boat or Holdco of any of their respective material obligations under this Agreement or (C) have a Boat Material Adverse Effect.
- 4.5 SEC Documents; Financial Statements; Undisclosed Liabilities.
 - (a) Boat has filed or furnished on a timely basis all reports, schedules, forms, statements, registration statements, prospectuses and other documents required to be filed or furnished by Boat with the SEC under the Securities Act or the Exchange Act since January 1, 2016 (the "*Boat*")

Table of Contents

SEC Documents"). None of the Subsidiaries of Boat is required to make or makes any filings with the SEC.

- (b) As of its respective filing date, and, if amended, as of the date of the last amendment prior to the date of this Agreement, each Boat SEC Document complied in all material respects with the requirements of the Exchange Act or the Securities Act, as the case may be, and the rules and regulations of the SEC promulgated thereunder applicable to such Boat SEC Document and did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.
- (c) The consolidated financial statements of Boat included in the Boat SEC Documents (including, in each case, any notes or schedules thereto) (the "Boat SEC Financial Statements"), comply as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto. The Boat SEC Financial Statements fairly present, in all material respects, the financial condition and the results of operations, cash flows and changes in stockholders' equity of Boat (on a consolidated basis) as of the respective dates of and for the periods referred to in the Boat SEC Financial Statements, and were prepared in accordance with GAAP applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto), subject, in the case of interim Boat SEC Financial Statements, to normal year-end adjustments (which are not material in significance or amount) and the absence of notes.
- (d) Boat is in compliance in all material respects with (i) the applicable provisions of the Sarbanes-Oxley Act and (ii) the applicable listing and corporate governance rules and regulations of the NASDAQ.
- (e) To the Knowledge of Boat, there are no outstanding or unresolved comments in comment letters received from the SEC staff with respect to any Boat SEC Documents and none of the Boat SEC Documents is the subject of ongoing SEC review. To the Knowledge of Boat, there are no SEC inquiries or investigations in each case pending or threatened regarding any accounting practices of Boat.
- (f) Boat has established and maintains disclosure controls and procedures and internal control over financial reporting (as such terms are defined in paragraphs (e) and (f), respectively, of Rule 13a-15 and paragraph (e) of Rule 15d-15 under the Exchange Act) as required by Rules 13a-15 and 15d-15 under the Exchange Act. Boat's disclosure controls and procedures are designed to ensure that all information (both financial and non-financial) required to be disclosed by Boat in the reports that it files or furnishes under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC, and that all such information is accumulated and communicated to Boat's management as appropriate to allow timely decisions regarding required disclosure and to make the certifications required pursuant to Sections 302 and 906 of the Sarbanes-Oxley Act. Since January 1, 2016, Boat has had no significant deficiencies or material weaknesses in the design or operation of its internal control over financial reporting that would reasonably be expected to adversely affect Boat's ability to record, process, summarize and report financial information and Boat does not have Knowledge of any fraud, whether or not material, that involves management or other employees who have a significant role in Boat's internal control over financial reporting.
- (g) Boat and its Subsidiaries do not have any liabilities or obligations of any nature (whether absolute or contingent, asserted or unasserted, known or unknown, primary or secondary, direct or indirect, and whether or not accrued or required to be reflected in Boat's financial statements in accordance with GAAP), except (i) as disclosed, reflected or reserved against in the most recent audited balance sheet included in the Boat SEC Financial Statements or the notes thereto, (ii) for

Table of Contents

liabilities and obligations incurred in the ordinary course of business since the date of the most recent audited balance sheet included in the Boat SEC Financial Statements, (iii) for liabilities and obligations arising out of or in connection with this Agreement, the Mergers or the Transactions and (iv) for liabilities and obligations that, individually or in the aggregate, have not had, and would not reasonably be expected to have, a Boat Material Adverse Effect.

- (h) Neither Boat nor any of its Subsidiaries is a party to, or has any commitment to become a party to, any joint venture, off-balance sheet partnership or any similar Contract (including any Contract or arrangement relating to any transaction or relationship between or among Boat and any of its Subsidiaries, on the one hand, and any unconsolidated affiliate, including any structured finance, special purpose or limited purpose entity or Person, on the other hand, or any "off-balance-sheet arrangements" (as defined in Item 303(a) of Regulation S-K under the Exchange Act)), where the result, purpose or intended effect of such Contract is to avoid disclosure of any material transaction involving, or material liabilities of, Boat or any of its Subsidiaries in Boat's or such Subsidiary's published financial statements or other Boat SEC Documents.
- 4.6 Absence of Certain Changes or Events. Since October 1, 2017 and through the date of this Agreement, (a) Boat and its Subsidiaries have conducted their businesses in all material respects only in the ordinary course and in a manner consistent with past practice and (b) there has not been any change, event, development, condition or occurrence that, individually or in the aggregate, has had or would reasonably be expected to have a Boat Material Adverse Effect. Since October 1, 2017 and through the date of this Agreement, neither Boat nor any of its Subsidiaries has taken any action that would have constituted a breach of, or required Island's consent pursuant to, Section 5.2 had the covenants therein applied since October 1, 2017.
- 4.7 Information Supplied. None of the information supplied or to be supplied by Boat or Holdco for inclusion or incorporation by reference in (i) the Form S-4 will, at the time the Form S-4 is filed with the SEC, and at any time it is amended or supplemented or at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, and (ii) the Joint Proxy Statement will, at the date it or any amendment or supplement is mailed to each of the holders of Boat Common Stock, Island Common Stock and Island Preferred Stock and at the time of each of the Boat Stockholders Meeting and Island Stockholders Meeting, contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in light of the circumstances in which they are made, not misleading (except that no representation or warranty is made by Boat or Holdco to such portions thereof that relate expressly to Island or to statements made therein based on information supplied by or on behalf of Island for inclusion or incorporation by reference therein). The Form S-4 and the Joint Proxy Statement will comply as to form in all material respects with the requirements of the Securities Act or Exchange Act, as applicable, and other applicable Law.
- 4.8 Legal Proceedings. As of the date hereof, there are no Proceedings pending, or to the Knowledge of Boat, threatened against Boat or any of its Subsidiaries or any of their respective assets, rights or properties or any of the officers or directors of Boat, that would, in each case, individually or in the aggregate, reasonably be expected to prevent or materially delay the consummation of the Transactions. There are no Proceedings pending, or to the Knowledge of Boat, threatened against Boat or any of its Subsidiaries or any of their respective assets, rights or properties or any of the officers or directors of Boat, except, in each case, for those that, individually or in the aggregate, have not had, and would not reasonably be expected to have, a Boat Material Adverse Effect. Neither Boat nor any of its Subsidiaries nor any of their respective properties, rights or assets is or are subject to any Order, except for those that, individually or in the aggregate, have not had, and would not reasonably be expected to have, a Boat Material Adverse Effect.

Table of Contents

- 4.9 Compliance with Laws.
 - (a) (i) Boat and its Subsidiaries are in compliance, and for the past five (5) years have been in compliance, with all Laws and Orders applicable to Boat or any of its Subsidiaries or any assets owned or used by any of them (except for such past noncompliance as has been remedied and imposes no continuing obligations or costs on Boat or its Subsidiaries), and (ii) neither Boat nor any of its Subsidiaries has received any written communication during the past five (5) years from a Governmental Entity that alleges that Boat or any of its Subsidiaries is not in compliance with any such Law or Order, except in the case of clauses (i) and (ii) where any non-compliance, individually or in the aggregate, has not had and would not reasonably be expected to have, a Boat Material Adverse Effect.
 - (b) Boat and each of its Subsidiaries (i) are in compliance, and for the past five (5) years have been in compliance, in all material respects with the FCPA and any other applicable Anti-corruption Laws; (ii) during the past five (5) years have not been investigated, to the Knowledge of Boat, by any Governmental Entity with respect to, or been given notice by a Governmental Entity or any other Person of, any actual or alleged violation by Boat or any of its Subsidiaries of the FCPA or any other Anti-corruption Laws; and (iii) during the past five (5) years have had an operational and effective FCPA and anti-corruption compliance program that includes, at a minimum, policies, procedures and training intended to enhance awareness of and compliance by Boat and its Subsidiaries with the FCPA and any other applicable Anti-corruption Laws.
 - (c) To the Knowledge of Boat, none of Boat or its Subsidiaries has directly or indirectly through its Representatives or any Person authorized to act on its behalf (including any distributor, agent, sales intermediary or other third party), offered, promised, paid, authorized or given money or anything of value to any Person for the purpose of: (i) influencing any act or decision of any Government Official or Other Covered Party; (ii) inducing any Government Official or Other Covered Party to do or omit to do an act in violation of a lawful duty; (iii) securing any improper advantage; or (iv) inducing any Government Official or Other Covered Party to influence the act or decision of a government or government instrumentality, in order to obtain or retain business, or direct business to, any Person or entity, in any way.
 - (d) To the Knowledge of Boat, during the past five (5) years, Boat and its Subsidiaries have maintained complete and accurate books and records, including records of payments to any agents, consultants, representatives, third parties and Government Officials, in accordance with GAAP, in all material respects. There have been no false or fictitious entries made in the books and records of Boat or its Subsidiaries relating to any unlawful offer, payment, promise to pay, or authorization of the payment of any money, or unlawful offer, gift, promise to give, or authorization of the giving of anything of value, including any bribe, kickback or other illegal or improper payment, and Boat and its Subsidiaries have not established or maintained a secret or unrecorded fund.
 - (e) To the Knowledge of Boat, during the past five (5) years, none of Boat or its Subsidiaries has had a customer or supplier or other business relationship with, is a party to any Contract with, or has engaged in any transaction with, any Person (i) that is organized or domiciled in or that is a citizen of Crimea, Cuba, Iran, North Korea or Syria (including any Governmental Entity within such country) or (ii) that is the subject of any international economic or trade sanction administered or enforced by OFAC, the United Nations Security Council, the European Union, Her Majesty's Treasury, the United Kingdom Export Control Organization or other relevant sanctions authority (including but not limited to being listed on the Specially Designated Nationals and Blocked Persons List administered by OFAC).
- 4.10 *Permits*. Boat and each of its Subsidiaries have all Permits necessary for the conduct of their business and the use of their properties and assets, as presently conducted and used, and each of

Table of Contents

the Permits is valid, subsisting and in full force and effect, except where the failure to have or maintain such Permit, individually or in the aggregate, has not had and would not reasonably be expected to have, a Boat Material Adverse Effect. Except as would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, (i) the operation of the business of Boat and its Subsidiaries as currently conducted is not, and has not been since January 1, 2016, in violation of, nor is Boat or its Subsidiaries in default or violation under, any Permit (except for such past violation or default as has been remedied and imposes no continuing obligations or costs on Boat or its Subsidiaries), and (ii) to the Knowledge of Boat, no event has occurred which, with notice or the lapse of time or both, would constitute a default or violation of any term, condition or provision of any Permit. There are no actions pending or, to the Knowledge of Boat, threatened, that seek the revocation, cancellation or modification of any Permit, except where such revocation, cancellation or modification, individually or in the aggregate, has not had and would not reasonably be expected to have, a Boat Material Adverse Effect. Since January 1, 2016, neither Boat nor to the Knowledge of Boat, has any charge, claim or assertion alleging any violations of or noncompliance with any Permit, nor to the Knowledge of Boat, has any charge, claim or assertion been threatened, except where such notice, charge, claim or assertion, individually or in the aggregate, has not had and would not reasonably be expected to have, a Boat Material Adverse Effect.

4.11 Employee Benefit Plans

- (a) Section 4.11(a) of the Boat Disclosure Schedule sets forth a true and complete list of each material "employee benefit plan" as defined in Section 3(3) of ERISA and any other plan, policy, program, Contract, or arrangement providing compensation or other benefits to any current or former director, officer or employee (or to any dependent or beneficiary thereof) of Boat or any of its Subsidiaries, in each case that is maintained, sponsored or contributed to by Boat or any of its Subsidiaries, or under which Boat or any of its Subsidiaries has any material obligation or material liability, whether actual or contingent, including each employment, incentive, bonus, deferred compensation, profit-sharing, pension, retirement, vacation, holiday, sick pay, cafeteria, fringe benefit, medical, disability, retention, severance, termination, change in control, stock purchase, stock option, stock appreciation, phantom stock, restricted stock, restricted stock unit or other equity-based compensation plans, policies, programs, practices or arrangements, in each case, which (x) is not sponsored and administered by a Governmental Entity and (y) is not required by Law to be provided (each, a "Boat Benefit Plan"). Neither Boat, nor to the Knowledge of Boat, any other Person, has any express or implied commitment, whether legally enforceable or not, to (i) modify, change or terminate any Boat Benefit Plan, other than with respect to a modification, change or termination required by ERISA or the Code or (ii) adopt any new Boat Benefit Plan.
- (b) With respect to each Boat Benefit Plan, Boat has made available to Island a current written copy thereof (if any) and, to the extent applicable: (i) any amendments and any related trust agreement; (ii) the most recent IRS determination letter; (iii) the most recent summary plan description and summary of material modifications, and (iv) for the most recent plan year (A) the Form 5500 and attached schedules and (B) audited financial statements.
- (c) Except as has not had or would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, each Boat Benefit Plan has been administered in accordance with its terms and all applicable Laws, including ERISA and the Code, and contributions required to be made under the terms of any of the Boat Benefit Plans have been timely made or, if not yet due, have been properly reflected on the most recent consolidated balance sheet filed or incorporated by reference in the Boat SEC Documents prior to the date of this Agreement.
- (d) Except as has not resulted or would not reasonably be expected to result, individually or in the aggregate, in a Boat Material Adverse Effect: (i) each Boat Benefit Plan which is intended

Table of Contents

to qualify under Section 401(a) of the Code has either received a favorable determination letter or opinion letter from the IRS as to its qualified status, and each trust established in connection with any Boat Benefit Plan which is intended to be exempt from federal income taxation under Section 501(a) of the Code is so exempt, and to Boat's Knowledge no fact or event has occurred that could adversely affect the qualified status of any such Boat Benefit Plan or the exempt status of any such trust, (ii) there has been no prohibited transaction (within the meaning of Section 406 of ERISA or Section 4975 of the Code), other than a transaction that is exempt under a statutory or administrative exemption, with respect to any Boat Benefit Plan, and (iii) no Proceeding has been brought, or to the Knowledge of Boat is threatened, against or with respect to any Boat Benefit Plan, including any audit or inquiry by the IRS or United States Department of Labor (other than for routine benefits claims).

- (e) No Boat Benefit Plan is, and neither Boat nor any of its ERISA Affiliates contributes to, has during the preceding six (6) years contributed to or has any liability or obligation, whether fixed or contingent, with respect to a Multiemployer Plan or other plan subject to Title IV of ERISA or the minimum funding requirements of Section 302 of ERISA or Section 412 of the Code. No material liability under Title IV of ERISA has been incurred by Boat or any ERISA Affiliate thereof that has not been satisfied in full, and no condition exists that would reasonably be expected to have a Boat Material Adverse Effect (whether primarily, jointly or secondarily) under Title IV of ERISA.
- (f) Except as has not had or would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, no amount that could be received (whether in cash or property or the vesting of property), as a result of the consummation of the Mergers or other Transactions (either alone or in combination with another event), by any employee, officer or director of Boat or any of its Subsidiaries who is a "disqualified individual" (within the meaning of Section 280G of the Code) could be characterized as an "excess parachute payment" (within the meaning of Section 280G(b)(1) of the Code).
- (g) Except as required by Law, no Boat Benefit Plan provides post-employment medical, disability or life insurance benefits to any former director, employee or their respective dependents (other than as required by health care continuation coverage as required by COBRA or ERISA or coverage through the end of the calendar month in which a termination of employment occurs.
- (h) Except for the adjustment and assumption of the Boat Stock Options in accordance with *Section 2.7*, neither the execution of this Agreement nor the consummation of the Mergers or other Transactions (either alone or in combination with another event) will (i) entitle any employee or director of Boat or its Subsidiaries to a bonus, severance or change in control payment, (ii) accelerate the time of payment or vesting or trigger any payment or funding (through a grantor trust or otherwise) of compensation or benefits, (iii) increase the amount payable or trigger any other material obligation pursuant to any of the Boat Benefit Plans or (iv) result in any breach or violation of, or default under any Boat Benefit Plan.
- (i) Except as has not had or would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, each Boat Benefit Plan that constitutes in any part a nonqualified deferred compensation plan within the meaning of Section 409A of the Code has been operated and maintained in operational and documentary compliance with Section 409A of the Code and all IRS guidance promulgated thereunder, to the extent such section and such guidance have been applicable to such Boat Benefit Plan. There is no agreement, plan, Contract or other arrangement to which Boat or any of its Subsidiaries is a party or by which any of them is otherwise bound to compensate any Person in respect of Taxes pursuant to Sections 409A or 4999 of the Code.

Table of Contents

Material Adverse Effect, with respect to each Boat Benefit Plan established or maintained outside of the United States of America primarily for the benefit of employees of Boat residing outside the United States of America (an "Boat Foreign Benefit Plan"): (i) all employer and employee contributions to each Boat Foreign Benefit Plan required by law or by the terms of any Boat Foreign Benefit Plan have been made or, if applicable, accrued, in accordance with normal accounting practices; (ii) the fair market value of the assets of each funded Boat Foreign Benefit Plan, the liability of each insurer for any Boat Foreign Benefit Plan funded through insurance or the book reserve established for any Boat Foreign Benefit Plan, together with any accrued contributions, is sufficient to procure or provide for the accrued benefit obligations with respect to all current and former participants in such Boat Foreign Benefit Plan according to the actuarial assumptions and valuations most recently used to determine employer contributions to such Boat Foreign Benefit Plan, and neither the execution of this Agreement nor the consummation of the Mergers or other Transactions will cause such assets or insurance obligations to be less than such benefit obligations; and (iii) to the Knowledge of Boat, each Boat Foreign Benefit Plan required to be registered has been registered and has been maintained in good standing with applicable regulatory authorities.

4.12 Employee and Labor Matters.

- (a) Neither Boat nor any of its Subsidiaries is a party to or bound by any collective bargaining agreement, agreement with any works council, or labor contract, no labor union, labor organization, works council, or group of employees of Boat or any of its Subsidiaries has made a pending demand for recognition or certification, and there are no representation or certification proceedings or petitions seeking a representation proceeding presently pending or, to the Knowledge of Boat, threatened in writing to be brought or filed with the National Labor Relations Board or any other labor relations tribunal or authority. Neither Boat nor any Subsidiary has engaged in any unfair labor practice with respect to any individuals employed by or otherwise performing services for Boat or any of its Subsidiaries (the "Boat Business Personnel"), and there is no unfair labor practice complaint or grievance or other administrative or judicial complaint, action or investigation pending or, to the Knowledge of Boat, threatened in writing against Boat or any of its Subsidiaries by the National Labor Relations Board or any other Governmental Entity with respect to any present or former Boat Business Personnel which would reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect. Except as would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, there is no labor strike, dispute, lockout, slowdown or stoppage pending or, to the Knowledge of Boat, threatened against or affecting Boat or any Subsidiary.
- (b) Neither Boat nor any of its Subsidiaries are required to provide notice to any work council or similar representative body prior to the execution of this Agreement or the consummation of the Transactions, except where the failure to provide such notice would not, individually or in the aggregate, reasonably be expected to (i) result in material liability to Boat and its Subsidiaries, taken as a whole, or (ii) materially delay or prevent the consummation of the Transactions.
- (c) To the Knowledge of Boat, Boat and its Subsidiaries are and have for the past five (5) years been in compliance with all applicable Laws respecting employment and employment practices, including, without limitation, all Laws respecting terms and conditions of employment, health and safety, wage payment, wages and hours, child labor, collective bargaining, immigration and work authorizations, employment discrimination, retaliation, civil rights, veterans' rights, disability rights or benefits, equal opportunity, plant closures and layoffs, affirmative action, workers' compensation, labor relations, social welfare obligations, proper classification of employees as exempt and non-exempt and as employees and independent contractors,

Table of Contents

unemployment insurance and the collection and payment of withholding and/or social security Taxes and any similar Tax, except for noncompliance as individually or in the aggregate, has not had and would not reasonably be expected to have, a Boat Material Adverse Effect.

- (d) To the Knowledge of Boat, no Boat Business Personnel is in any respect in violation of any term of any employment agreement, nondisclosure agreement, common law nondisclosure obligation, fiduciary duty, noncompetition agreement, restrictive covenant or other obligation to Boat or any of its Subsidiaries or to a former employer of any such employee relating (i) to the right of any such individual to be employed by or provide services for Boat or its Subsidiaries or (ii) to the knowledge or use of trade secrets or proprietary information, in each case except as, individually or in the aggregate, currently does not have and would not reasonably be expected to have a Boat Material Adverse Effect.
- (e) To the Knowledge of Boat, Boat and its Subsidiaries are not delinquent in payments to any current or former Boat Business Personnel for any services or amounts required to be reimbursed or otherwise paid except as currently does not have and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect.
- 4.13 Environmental Matters. Except as, individually or in the aggregate, has not had and would not reasonably be expected to have, a Boat Material Adverse Effect:
 - (a) Boat and each of its Subsidiaries (i) have been within the last five (5) years, and are, in compliance with all, and are not subject to any liability with respect to noncompliance with any, applicable Environmental Laws, (ii) have and hold, or have applied for, all Environmental Permits necessary for the conduct of their business and the use of their properties and assets, as currently conducted and used, and (iii) are in compliance with their respective Environmental Permits.
 - (b) There are no Environmental Claims pending, nor to the Knowledge of Boat, threatened against Boat or any of its Subsidiaries, and none of Boat or any of its Subsidiaries has received any notification of any allegation of actual or potential responsibility for any Release or threatened Release of any Hazardous Materials with respect to any location.
 - (c) There have been no Releases of Hazardous Materials at any properties that are owned, operated, leased or used by Boat or any of its Subsidiaries, or to the Knowledge of Boat, at properties that were formerly owned, operated, leased or used by Boat or any of its Subsidiaries, that are reasonably likely to cause Boat or any of its Subsidiaries to incur liability pursuant to applicable Environmental Law.
 - (d) None of Boat or any of its Subsidiaries (i) has entered into or agreed to any consent decree or consent order or is otherwise subject to any judgment, decree, or judicial or administrative Order relating to compliance with Environmental Laws or Environmental Permits, the investigation, sampling, monitoring, treatment, remediation, response, removal or cleanup of Hazardous Materials, and no Proceeding is pending, or to the Knowledge of Boat is threatened, with respect thereto, or (ii) is an indemnitor by contract or otherwise in connection with any claim, demand, suit or action threatened or asserted by any third-party for any liability under any Environmental Law or otherwise relating to any Hazardous Materials.

4.14 Real Property.

- (a) Boat owns no Real Property.
- (b) Boat or its Subsidiaries, as the case may be, has a valid and subsisting leasehold or subleasehold interest in the Boat Leased Real Property free and clear of all Liens, except for Permitted Liens.

Table of Contents

(c) The Boat Leased Property constitutes all real property necessary for the conduct of the business of Boat and its Subsidiaries, taken as a whole, as currently conducted. Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, neither Boat nor any of its Subsidiaries has received written notice of any Proceedings in eminent domain, condemnation or other similar Proceedings that are pending, and to Boat's Knowledge there are no such Proceedings threatened, affecting any portion of the Boat Real Property and neither Boat nor any of its Subsidiaries has received written notice of the existence of any Order or of any pending Proceeding relating to the ownership, lease, use, occupancy or operation by any Person of the Boat Real Property. Neither Boat nor any of its Subsidiaries has leased, subleased, licensed or otherwise granted any Person a material right to use or occupy any of the material Boat Real Property or any material portion thereof. Neither Boat nor any of its Subsidiaries has granted any option or other right to any third party to purchase any of the material Boat Real Property or any material portion thereof. Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, each Boat Real Property and all buildings and improvements located on the Boat Real Property are in a state of good operating condition, subject to reasonable wear and tear.

4.15 Tax Matters.

- (a) Except as would not, individually or in the aggregate, reasonably be expected to have a Boat Material Adverse Effect:
 - (i) (A) All Tax Returns that are required to be filed by or with respect to Boat or any of its Subsidiaries have been timely filed (taking into account any extension of time within which to file), (B) all such Tax Returns are or will be true, complete and accurate, (C) Boat and its Subsidiaries have paid all Taxes due and owing by any of them (whether or not shown on any Tax Return) or appropriate reserves have been recorded in their books and records, (D) Boat and its Subsidiaries have established adequate accruals and reserves, in accordance with GAAP, on the financial statements included in the Boat SEC Documents for all Taxes payable by Boat and its Subsidiaries for all taxable periods and portions thereof through the date of such financial statements and (E) Boat and its Subsidiaries have withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, stockholder or other third party; such withheld amounts were either timely paid to the appropriate Taxing Authority or set aside in accounts for such purpose and were reported to the appropriate Taxing Authority and to each such employee, independent contractor, creditor, stockholder or other third party, as required under Law.
 - (ii) No written agreement or other written document waiving or extending, or having the effect of waiving or extending, the statute of limitations or the period of assessment or collection of any Taxes relating to Boat or any of its Subsidiaries has been filed or entered into with any Taxing Authority, and no power of attorney with respect to any such Taxes has been granted to any Person that is in force.
 - (iii) (A) Neither Boat nor any of its Subsidiaries is the subject of any currently ongoing Tax audit or other proceeding with respect to any Taxes or Tax Return nor has any audit or other proceeding with respect to Taxes been proposed against any of them in writing and any deficiencies asserted or assessments made as a result of any audit or other proceeding with respect to Taxes have been resolved, are being contested in good faith, or adequate accruals or reserves for such deficiencies or assessments have been established and (B) neither Boat nor any of its Subsidiaries have received written notice of any deficiencies for any Tax of Boat or any of its Subsidiaries from any Taxing Authority for which there are not adequate reserves on the financial statements included in the Boat SEC Documents.

Table of Contents

- (iv) No claim has ever been made in writing by a Taxing Authority of a jurisdiction where Boat or any of its Subsidiaries has not filed Tax Returns that Boat or such a Subsidiary is or may be subject to taxation by that jurisdiction.
- (v) Neither Boat nor any of its Subsidiaries has constituted, in the two (2) years prior to the date of this Agreement, a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of stock intended to qualify for tax-free treatment under Section 355 of the Code (or any similar provision of state, local, or non-U.S. Law) or otherwise as part of a plan (or series of related transactions), within the meaning of Section 355(e) of the Code, that includes the Merger.
- (vi) Neither Boat nor any of its Subsidiaries (A) is a party to or bound by or has any obligation under any Tax indemnification, separation, sharing or similar agreement or arrangement (other than such an agreement or arrangement exclusively between or among Boat and its Subsidiaries or an agreement entered into in the ordinary course of business which does not relate primarily to Taxes), (B) is or has been a member of any consolidated, combined, unitary or similar group for purposes of filing Tax Returns or paying Taxes (other than a group comprised solely of (I) Subsidiaries of Boat or (II) Boat and any of its Subsidiaries), (C) has entered into a closing agreement pursuant to Section 7121 of the Code or any other binding agreement with a Taxing Authority that would have an effect on the determination of Boat's or any of its Subsidiaries' liability to Tax in a tax year ending after the Effective Time, or any predecessor provision or any similar provision of foreign, state or local Tax law or (D) has any liability for Taxes of any Person (other than Boat or any of its Subsidiaries) under U.S. Treasury Regulation Section 1.1502-6 (or any similar provision of state, local, or non-U.S. Law) or as transferee or successor.
- (vii) There are no Liens for Taxes upon any property or assets of Boat or any of its Subsidiaries, except for Taxes not yet due and payable.
- (viii) Neither Boat nor any of its Subsidiaries has entered into any "listed transaction" within the meaning of Section 6707A(c)(2) of the Code and U.S. Treasury Regulation Section 1.6011-4(b)(2) (or any similar provision of state, local or non-U.S. Law).
- (ix) Boat has made available to Island or its legal or accounting representative copies of all U.S. federal and state income Tax Returns for Boat and each of its Subsidiaries filed for all periods including and after the period ended December 31, 2014.
 - (x) Boat is not a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code.
- (xi) Neither Boat nor any of its Subsidiaries (A) has filed any extension of time within which to file any Tax Returns that have not been filed, except in the ordinary course of business and (B) will be required to include any item of income or gain in, or be required to exclude any item of deduction of loss from, any period ending after the Closing Date as a result of any (i) installment sale or open transaction made on or prior to the Closing Date, (ii) election under Section 108(i) of the Code or (iii) prepaid amount received prior to the Closing Date.
- (xii) Neither Boat nor any of its Subsidiaries has agreed to, requested, or is required to include any adjustment under Section 481 of the Code (or any corresponding provision of applicable state, local or non-U.S. Tax Law) by reason of a change in accounting method or otherwise.
- (b) Neither Boat nor any of its Subsidiaries has knowledge of any facts, agreements, plans or other circumstances or has taken or agreed to take any action that would reasonably be expected

Table of Contents

to prevent or impede (i) the Mergers from together being treated as an "exchange" described in Section 351 of the Code or (ii) the Boat Merger from qualifying as a "reorganization" within the meaning of Section 368(a) of the Code.

(c) It is agreed and understood that no representation or warranty is made by Boat in respect of Tax matters in any Section of this Agreement other than Section 4.5, Section 4.6, Section 4.7, Section 4.11 and this Section 4.15.

4.16 Material Contracts.

- (a) All Contracts required to be filed as exhibits to the Boat SEC Documents have been so filed in a timely manner. *Section 4.16(a)* of the Boat Disclosure Schedule sets forth a true and complete list, as of the date hereof, of each of the following Contracts, excluding any Boat Benefit Plans, to which Boat or any of its Subsidiaries is a party or by which Boat or any of its Subsidiaries or any of their assets or businesses are bound (and any amendments, supplements and modifications thereto):
 - (i) any Contract that is a "material contract" (as such term is defined in Item 601(b)(10) of Regulation S-K of the Exchange Act);
 - (ii) any Contract that materially limits the ability of Boat or any of its affiliates (including, following the consummation of the Transactions, the Surviving Corporation and its affiliates) to engage or compete in any line of business or in any geographic area or from developing or commercializing compounds, any therapeutic area, class of drugs or mechanism of action;
 - (iii) any Contract required to be disclosed pursuant to Item 404 of Regulation S-K of the Exchange Act;
 - (iv) any Contract or series of related Contracts relating to indebtedness for borrowed money (A) in excess of \$2,000,000 or (B) that becomes due and payable as a result of the Transactions;
 - (v) any license, sublicense, option, development or collaboration agreement or other Contract relating to Boat Material Intellectual Property reasonably expected to result in payments in excess of \$2,000,000 in any twelve (12) month period after the Closing Date (excluding license agreements for "shrink-wrap," "click-wrap" or other commercially available off-the-shelf software that is not the subject of a negotiated agreement, and excluding agreements the primary purpose of which is to purchase tangible goods or procure services unrelated to Intellectual Property and in the ordinary course of business);
 - (vi) any Contract reasonably expected to result in payments in excess of \$2,000,000 in any twelve (12) month period after the Closing Date;
 - (vii) any purchase, sale or supply contract that contains volume requirements or commitments, exclusive or preferred purchasing arrangements, most favored nation provisions or promotional requirements;
 - (viii) any lease, sublease, occupancy agreement or other Contract with respect to the Boat Leased Real Property reasonably expected to result in payments in excess of \$2,000,000 in any twelve (12) month period after the Closing Date;

Table of Contents

- (ix) any agreement that grants any right of first refusal or right of first offer or similar right or that limits or purports to limit the ability of Boat or any of its Subsidiaries to own, operate, sell, transfer, pledge or otherwise dispose of any material amount of assets or businesses (in any case in excess of \$2,000,000);
- (x) any acquisition or divestiture agreement (A) entered into since January 1, 2015 with a purchase price in excess of \$2,000,000 or (B) that contains "earn-out" provisions or other contingent payment obligations that could reasonably be expected to exceed \$2,000,000 (including indemnification obligations) that have not been satisfied in full;
- (xi) any agreement that by its terms limits the payment of dividends or other distributions by Boat or any of its Subsidiaries;
- (xii) any Contract for any joint venture, partnership or similar arrangement, or any Contract involving a sharing of revenues, profits, losses, costs, or liabilities by Boat or any of its Subsidiaries with any other Person;
- (xiii) any "single source" supply Contract pursuant to which goods or materials that are material to Boat or any of its Subsidiaries are supplied to Boat or such Subsidiary from an exclusive source; or
 - (xiv) any Contract with any Governmental Entity.
- (b) Boat has heretofore made available to Island true, correct and complete copies of the Contracts set forth in Section 4.16(a).
- (c) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, (i) all Contracts set forth or required to be set forth in *Section 4.16(a)* of the Boat Disclosure Schedule or filed or required to be filed as exhibits to the Boat SEC Documents (the "*Boat Material Contracts*") are valid, binding and in full force and effect and are enforceable by Boat or its applicable Subsidiary in accordance with their terms, except as limited by Laws affecting the enforcement of creditors' rights generally, by general equitable principles or by the discretion of any Governmental Entity before which any Proceeding seeking enforcement may be brought, (ii) Boat, or its applicable Subsidiary, has performed all obligations required to be performed by it under the Boat Material Contracts, and it is not (with or without notice or lapse of time, or both) in breach or default thereunder and, to the Knowledge of Boat, no other party to any Boat Material Contract is (with or without notice or lapse of time, or both) in breach or default thereunder, (iii) since January 1, 2017, neither Boat nor any of its Subsidiaries has received written notice of any actual, alleged, possible or potential violation of, or failure to comply with, any term or requirement of any Boat Material Contract, and (iv) neither Boat nor any of its Subsidiaries has received any written notice of the intention of any party to cancel, terminate, materially change the scope of rights under or fail to renew any Boat Material Contract.

4.17 Intellectual Property.

(a) Section 4.17(a)(1) of the Boat Disclosure Schedule sets for a complete and accurate list of all (i) issued patents and pending patent applications, (ii) trademark and service mark registrations and applications, (iii) copyright registrations and applications, and (iv) internet domain name registrations, in each case that are owned by Boat or any of its Subsidiaries (collectively, the "Solely Owned Boat Registered Intellectual Property"). Section 4.17(a)(2) of the Boat Disclosure Schedule sets forth a complete and accurate list of all (i) issued patents and pending patent applications, (ii) trademark and service mark registrations and applications, (iii) copyright registrations and applications, and (iv) internet domain name registrations, in each case that are licensed to Boat (collectively, the "Licensed Boat Registered Intellectual Property"). Except as has

Table of Contents

not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, Boat owns all right, title and interest in the Solely Owned Boat Registered Intellectual Property free and clear of all Liens other than Permitted Liens or Liens created by Island. To the Knowledge of Boat, each item of material Solely Owned Boat Registered Intellectual Property that is shown as registered, filed, issued or applied for in *Section 4.17(a)* of the Boat Disclosure Schedule has been duly registered in, filed in or issued by an official governmental register and/or issuer (or officially recognized register or issuer) for such Intellectual Property and each such registration, filing, issuance and/or application, to the Knowledge of Boat, (x) has not been abandoned or cancelled, (y) has been maintained effective by all requisite filings, renewals and payments, and (z) remains in full force and effect.

- (b) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, to the Knowledge of Boat, each of Boat and its Subsidiaries owns and possesses all right, title and interest in and to (or has the right pursuant to a valid and enforceable license or otherwise possesses legally enforceable rights to use) all Intellectual Property that is necessary for or used or held for use in the conduct of the business of Boat and its Subsidiaries in substantially the same manner as presently conducted. Neither the execution and delivery of this Agreement by Boat, nor the performance of this Agreement by Boat, will result in the loss, forfeiture, termination, or impairment of, or give rise to a right of any Person to limit, terminate, or consent to the continued use of, any rights of Boat or any of its Subsidiaries in any Intellectual Property except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect.
- (c) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, to the Knowledge of Boat, neither Boat nor any of its Subsidiaries nor the conduct of their businesses is infringing, misappropriating, diluting, or otherwise violating the Intellectual Property rights of any Person. Neither Boat nor any of its Subsidiaries has received any written charge, complaint, claim, demand, or notice since January 1, 2015 (or earlier, if presently not resolved) alleging any such infringement, misappropriation, dilution, or violation (including any claim that Boat or any of its Subsidiaries must license or refrain from using any Intellectual Property rights of any Person) which alleged infringement, misappropriation, dilution, or violation, if true, would reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect. To the Knowledge of Boat, no Person is infringing, misappropriating, diluting or otherwise violating any Intellectual Property owned by Boat or any of its Subsidiaries except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect. Neither Boat nor any of its Subsidiaries has made or asserted any charge, complaint, claim, demand or notice since January 1, 2015 (or earlier, if presently not resolved) alleging any such infringement, misappropriation, dilution, or violation which alleged infringement, misappropriation, dilution, or violation, if true, would reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect.
- (d) Boat and its Subsidiaries have taken commercially reasonable steps to maintain, police and protect the Intellectual Property that is material to their business ("Boat Material Intellectual Property"). Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, all Boat Material Intellectual Property that derives actual or potential economic value from not being generally known to the public or to other persons who can obtain economic value from its disclosure or use has been maintained in confidence in accordance with protection procedures that are in accordance with procedures customarily used in the industry to protect rights of like importance and, to the Knowledge of Boat, adequate for protection against unauthorized disclosure or use. To the Knowledge of Boat, there has been no unauthorized disclosure of any Boat Material Intellectual Property. Except as

Table of Contents

has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, to the Knowledge of Boat, all former and current officers, directors, employees, personnel, consultants, advisors, agents, and independent contractors of Boat and its Subsidiaries, and each of their predecessors, who have contributed to or participated in the conception and development of Intellectual Property for such entities have entered into valid and binding proprietary rights agreements with Boat or one of its Subsidiaries or predecessors, vesting ownership of such Intellectual Property in Boat or one of its Subsidiaries. No such Person has asserted, and to the Knowledge of Boat, no such Person has, any right, title, interest or other claim in, or the right to receive any royalties or other consideration with respect to, any Boat Material Intellectual Property.

(e) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, the IT Assets of Boat and its Subsidiaries operate in all material respects in accordance with their documentation and functional specifications and as required by Boat and its Subsidiaries to operate the business of Boat and its Subsidiaries as presently conducted and have not, since January 1, 2015, materially malfunctioned or failed. Boat and its Subsidiaries have implemented commercially reasonable measures to protect the confidentiality and security of such IT Assets and information stored or contained therein against any unauthorized use, access, interruption or corruption. Boat and its Subsidiaries have implemented commercially reasonable data backup, data storage, system redundancy and disaster avoidance procedures with respect to their IT Assets.

4.18 Healthcare Compliance Matters.

- (a) (i) Boat and each of its Subsidiaries are in compliance and since January 1, 2015 have been in compliance with all Health Care Laws applicable to Boat or any of its Subsidiaries or any assets owned or used by any of them and (ii) neither Boat nor any of its Subsidiaries has received any written communication or has been subject to any Proceeding (other than routine FDA inspections) since January 1, 2015 from a Governmental Entity that alleges that Boat or any of its Subsidiaries is not in compliance with any Health Care Law, except in the case of clauses (i) and (ii) where any non-compliance, individually or in the aggregate, has not had and would not reasonably be expected to have, a Boat Material Adverse Effect. Neither Boat nor any of its Subsidiaries is a party to or has any ongoing obligations pursuant to or under any corporate integrity agreements, deferred prosecution agreements, monitoring agreements, consent decrees, settlement orders, plans of correction or similar agreements with or imposed by any Governmental Entity.

 Additionally, none of Boat, its Subsidiaries or any of its respective employees, officers or directors has been excluded, suspended or debarred from participation in any U.S. state or federal health care program or, to the Knowledge of Boat, has been convicted of any crime or is subject to a governmental inquiry, investigation, Proceeding, or other similar action, or has engaged in any conduct, that could reasonably be expected to result in debarment, suspension, or exclusion.
- (b) Boat and each of its Subsidiaries have, maintain and are operating in material compliance with all Health Care Permits, and all such Health Care Permits are valid, subsisting, and in full force and effect, except where the failure to have, maintain or operate in compliance with the Health Care Permits has not had, and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect. Boat and each of its Subsidiaries have fulfilled and performed all of their obligations with respect to the Health Care Permits, and no event has occurred which allows, or with notice or lapse of time or both, would allow revocation or termination thereof or results in any other material impairment of the rights of the holder of any Health Care Permit, except where the failure to so fulfill or perform, or the occurrence of such event, would not result in a Boat Material Adverse Effect. There is no Proceeding pending or, to the Knowledge of Boat, threatened in writing that could result in the suspension, termination, revocation, cancellation, limitation or impairment of any such Health Care Permit other than those

Table of Contents

that have not had and would not reasonably expected to have, individually or in the aggregate, a Boat Material Adverse Effect.

- (c) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, all applications, notifications, submissions, information, claims, reports and statistics, and other data and conclusions derived therefrom, utilized as the basis for or submitted in connection with any and all requests for a Health Care Permit relating to Boat and each of its Subsidiaries, their business and products and product candidates, when submitted to the FDA, DEA or other Governmental Entity were true, complete and correct as of the date of submission and any necessary or required updates, changes, corrections or modification to such applications, submissions, information and data have been submitted to the FDA, DEA or other Governmental Entity.
- (d) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, since January 1, 2015, Boat and each of its Subsidiaries have not had any product, product candidate or manufacturing site (whether owned by Boat or its Subsidiary(ies), or a contract manufacturer for their products and product candidates) subject to a Governmental Entity (including FDA or DEA) shutdown or import or export prohibition, nor received any FDA Form 483 or other Governmental Entity notice of inspectional observations, "warning letters," "untitled letters" or written requests or requirements to make changes to a product or product candidate, or similar correspondence or written notice from the FDA, DEA or other Governmental Entity alleging or asserting noncompliance with any applicable Health Care Law, Health Care Permit or such requests or requirements of a Governmental Entity.
- (e) To the Knowledge of Boat, there are no facts or circumstances that would be reasonably likely to result in (A) a change in the marketing classification or a material change in the labeling of any product or product candidate or (B) a termination or suspension of any Health Care Permits for such products or product candidates.
- (f) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, (i) the clinical, pre-clinical and other studies and tests conducted by or on behalf of or sponsored by Boat or its Subsidiaries or in which Boat or its Subsidiaries, or any of their products and product candidates have participated were, and if still pending are, being conducted in accordance with standard medical and scientific research procedures and all applicable Laws, including, but not limited to, the Federal Food, Drug and Cosmetic Act and its applicable implementing regulations and (ii) no investigational new drug application filed by or on behalf of Boat or its Subsidiaries with the FDA has been terminated or suspended by the FDA, and neither the FDA nor any applicable foreign Governmental Entity has commenced, or, to the Knowledge of Boat, threatened to commence, any action to place a clinical hold order on, or otherwise terminate, delay or suspend, any proposed or ongoing clinical investigation conducted or proposed to be conducted by or on behalf of Boat and its Subsidiaries.
- (g) Neither Boat nor any of its Subsidiaries is the subject of any pending or, to the Knowledge of Boat, threatened investigation in respect of Boat, its Subsidiaries or their products and product candidates, by the FDA pursuant to its "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities" Final Policy set forth in 56 Fed. Reg. 46191 (September 10, 1991) and any amendments thereto. Boat has provided Island accurate and complete copies of all Health Care Permits and correspondence with any Governmental Entity related to all products and product candidates of Boat and its Subsidiaries.
- 4.19 Broker's Fees. Except for the financial advisors' fees set forth in Section 4.19 of the Boat Disclosure Schedule, neither Boat nor any of its Subsidiaries nor any of their respective officers or directors on behalf of Boat or such Subsidiaries has employed any financial advisor, broker or finder or

Table of Contents

incurred any liability for any financial advisory, broker's fees, commissions or finder's fees in connection with any of the Transactions.

- 4.20 *Opinion of Financial Advisor*. J.P. Morgan Securities LLC, Boat's financial advisor, has delivered to the Boat Board its opinion in writing or orally, in which case, such opinion will be subsequently confirmed in writing, to the effect that, as of the date thereof and based upon and subject to the factors and assumptions set forth therein, the Boat Exchange Ratio is fair from a financial point of view to the holders of the shares of Boat Common Stock.
- 4.21 Ownership of Island Common Stock. Neither Boat nor any of its affiliates or associates is, nor at any time during the last three (3) years has Boat or any of its affiliates or associates been, an "interested stockholder" of Island as defined in Section 203 of the DGCL. Boat and its Subsidiaries, affiliates and associates do not beneficially own any shares of Island Common Stock, Island Preferred Stock or other securities of Island or any options, warrants or other rights to acquire Island Common Stock or other securities of, or any other economic interest (through derivative securities or otherwise) in, Island.
- 4.22 *Insurance*. All insurance policies (including policies providing casualty, liability and workers compensation coverage, but excluding any Boat Benefit Plan) to which Boat or any of its Subsidiaries is currently a party are in full force and effect, and, to the Knowledge of Boat, have been issued by licensed insurers, all premiums with respect thereto covering all periods up to and including the Closing Date have been paid, and no notice of cancellation or termination has been received with respect to any such policies, except for such cancellations or terminations which would have not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect. Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect, each of Boat and its Subsidiaries is, and since January 1, 2015 has been, insured with respect to its assets and properties and the conduct of its business in such amounts and against such risks as are in its reasonable judgment sufficient for compliance with Law and as are adequate to protect its assets and properties and the conduct of its business.
- 4.23 Holdco, Merger Sub A and Merger Sub B. Since its date of incorporation, none of Holdco, Merger Sub A or Merger Sub B has carried on any business or conducted any operations other than the execution of this Agreement, the performance of its obligations hereunder and matters ancillary thereto. The authorized capital stock of each of Merger Sub A and Merger Sub B consists of 1,000 shares of common stock, par value \$0.01 per share and the authorized capital stock of Holdco consists of 200,000,000 shares of common stock, par value \$0.01 per share, all of which have been validly issued, are fully paid and nonassessable and are owned directly or indirectly by Boat free and clear of any Liens.
- 4.24 No Other Representations or Warranties. Boat hereby acknowledges and agrees that, except for the representations and warranties set forth in Article 3, (a) none of Island, or any of its stockholders or representatives, or any other Person, has made or is making any express or implied representation or warranty with respect to Island or its business or operations, including with respect to any information provided or made available to Boat or any of its affiliates, stockholders or representatives, or any other Person, or, except as otherwise expressly set forth in this Agreement, had or has any duty or obligation to provide any information to Boat or any of its affiliates, stockholders or Representatives, or any other Person, in connection with this Agreement, the transactions contemplated hereby or otherwise, and (b) to the fullest extent permitted by Law, none of Island, nor any of its affiliates, stockholders or representatives, or any other Person, will have or be subject to any liability or indemnification or other obligation of any kind or nature to Boat, or any of its affiliates, stockholders or representatives, or any other Person, resulting from the delivery, dissemination or any other distribution to Boat or any of its affiliates, stockholders or representatives, or any other Person, or the use by Boat or any of its affiliates, stockholders or representatives, or any other Person, of any such

Table of Contents

information provided or made available to any of them by Island, or any of its affiliates, stockholders or representatives, or any other Person, and (subject to the express representations and warranties of Island set forth in *Article 3*) Boat expressly disclaims reliance on any such information (including the accuracy or completeness thereof) or any representations or warranties or other statements or omissions that may have been made by Island or any Person with respect to Island other than the representations and warranties set forth in this Agreement.

ARTICLE 5 COVENANTS

- 5.1 Conduct of Business by Island Pending the Effective Time. Island agrees that, between the date of this Agreement and the Effective Time, except as set forth in Section 5.1 of the Island Disclosure Schedule, as required by Law or as otherwise expressly contemplated or required by this Agreement, unless Boat shall otherwise consent in writing (such consent not to be unreasonably withheld, conditioned or delayed), Island will, (i) conduct its operations only in the ordinary course of business consistent with past practice, (ii) use its commercially reasonable efforts to keep available the services of the current officers, employees and consultants of Island and to preserve the goodwill and current relationships of Island with Persons with which Island has business relations, and (iii) use its commercially reasonable efforts to preserve intact its business organization and comply with all applicable Law. Without limiting the foregoing, except as set forth in Section 5.1 of the Island Disclosure Schedule, as required by Law or as otherwise expressly contemplated or required by this Agreement, Island shall not between the date of this Agreement and the Effective Time, directly or indirectly, take any of the following actions without the prior written consent of Boat (such consent not to be unreasonably withheld, conditioned or delayed):
 - (a) amend or otherwise change its certificate of incorporation or bylaws or equivalent organizational documents (including by merger consolidation or otherwise);
 - (b) issue, sell, pledge, dispose of, grant, transfer or encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer, or encumbrance of, any shares of capital stock of, or other Equity Interests in, Island of any class, or securities convertible into, or exchangeable or exercisable for, any shares of such capital stock or other Equity Interests, or any options, warrants or other rights of any kind to acquire any shares of such capital stock or other Equity Interests or such convertible or exchangeable securities, or any other ownership interest (including any such interest represented by Contract right), or any "phantom" stock, "phantom" stock rights, stock appreciation rights or stock based performance rights, of Island, other than (i) the issuance of Island Common Stock in connection with the exercise of Island Stock Options or the conversion of Island Preferred Stock, in each case, outstanding as of the date hereof in accordance with their terms, (ii) the issuance of any Island Stock Options or Island Common Stock (in lieu of cash compensation otherwise payable to members of the Island Board), in either case, in the ordinary course of business consistent with past practice up to a maximum of 200,000 shares of Island Common Stock subject thereto or (iii) the issuance of any shares of Island Common Stock in accordance with the Island ESPP;
 - (c) sell, pledge, dispose of, transfer, lease, license, guarantee or encumber any property or assets of Island with value in excess of \$2,000,000, except pursuant to existing Contracts or the sale or purchase of goods in the ordinary course of business consistent with past practice;
 - (d) (i) sell, assign, pledge, grant or acquire, agree to grant to or acquire from any Person, or otherwise encumber, transfer, license, abandon, place in the public domain, permit to lapse, disclose or agree to disclose or otherwise dispose of any Island Material Intellectual Property, except pursuant to the terms of existing Contracts or the licensing of any such Intellectual Property in the ordinary course of business consistent with past practice or (ii) compromise, settle or agree

Table of Contents

to settle, or consent to judgment in, any one or more actions or institute any action concerning any Island Material Intellectual Property except in the ordinary course of business consistent with past practice;

- (e) declare, set aside, make or pay any dividend or other distribution (whether payable in cash, shares, property or a combination thereof) with respect to any of its capital stock or other Equity Interests or enter into any agreement with respect to the voting or registration of its capital stock or other Equity Interests;
- (f) reclassify, combine, split, subdivide or amend the terms of, or redeem, purchase or otherwise acquire, directly or indirectly, any of its capital stock or other Equity Interests;
- (g) form any new Subsidiary, or merge or consolidate Island with any Person or adopt a plan of complete or partial liquidation or resolutions providing for a complete or partial liquidation, dissolution, restructuring, recapitalization or other reorganization of Island:
- (h) acquire (including by merger, consolidation, or acquisition of shares or assets) any interest in any Person or any assets thereof in each case with value in excess of \$2,000,000, other than in the ordinary course of business consistent with past practice or pursuant to the Contracts set forth on *Section 5.1(h)* of the Island Disclosure Schedule;
- (i) repurchase, repay, refinance or incur any indebtedness for borrowed money, except as required by the terms of any such indebtedness as of the date hereof, or issue any debt securities or assume, guarantee or endorse, or otherwise as an accommodation become responsible for (whether directly, contingently or otherwise), the obligations of any Person for borrowed money, except for borrowings under Island's existing credit facilities or issuances of commercial paper for working capital and general corporate purposes in the ordinary course of business consistent with past practice;
- (j) make any loans, advances or capital contributions to, or investments in, any other Person in excess of \$2,000,000 in the aggregate;
- (k) terminate, cancel, renew, or request or agree to any material change in or waiver under any Island Material Contract, or enter into or amend any Contract that, if existing on the date hereof, would be an Island Material Contract, in each case other than in the ordinary course of business consistent with past practice;
- (l) make or authorize any capital expenditure in excess of Island's capital expenditure budget as disclosed to Boat prior to the date hereof, other than capital expenditures that are not, in the aggregate, in excess of \$2,000,000;
- (m) except (i) in the ordinary course of business consistent with past practice with respect to individuals with a title below Vice President, or (ii) to the extent required by applicable Law or the existing terms of any Island Benefit Plan disclosed in *Section 3.11(a)* of the Island Disclosure Schedule: (A) materially increase the compensation or benefits payable or to become payable to directors, officers or employees of Island; (B) grant any rights to severance or termination pay or transaction or retention bonus payments to, or enter into any severance agreement with, any director, officer or employee of Island; (C) establish, adopt, enter into or materially amend any collective bargaining agreement or other Contract with any labor union or labor organization, Island Benefit Plan or similar plan, agreement, trust, fund, policy or arrangement for the benefit of any director, officer or employee of Island; (D) take any action to amend or waive any performance or vesting criteria or accelerate vesting, exercisability or funding under any Island Benefit Plan; or (E) hire or terminate the employment of any executive officer of Island other than terminations for cause, death or disability;
 - (n) forgive any loans to directors, officers or employees of Island;

Table of Contents

- (o) waive, release, pay, discharge or satisfy any claims, liabilities or obligations (absolute, accrued, contingent or otherwise) with value in excess of \$2,000,000, except in the ordinary course of business consistent with past practice and in accordance with their terms;
- (p) make any change in accounting policies, practices, principles, methods or procedures, other than as required by GAAP or by a Governmental Entity;
 - (q) waive, release, assign, settle or compromise any claims or rights with value in excess of \$2,000,000 held by Island;
- (r) compromise, settle or agree to settle any Proceeding or investigation (including any Proceeding or investigation relating to this Agreement or the Transactions) other than compromises, settlements or agreements in the ordinary course of business consistent with past practice that involve only the payment of monetary damages not in excess of \$2,000,000 individually or in the aggregate, in any case without the imposition of equitable relief on, or the admission of wrongdoing by, Island;
- (s) except as required by applicable Law, make, change or revoke any material Tax election, change any Tax accounting period for purposes of a material Tax or material method of Tax accounting, file any material amended Tax Return, settle or compromise any audit or proceeding relating to a material amount of Taxes, agree to an extension or waiver of the statute of limitations with respect to a material amount of Taxes, enter into any "closing agreement" within the meaning of Section 7121 of the Code (or any similar provision of state, local, or non-U.S. Law) with respect to any material Tax, or surrender any right to claim a material Tax refund;
- (t) write up, write down or write off the book value of any assets, in the aggregate, in excess of \$2,000,000, except for depreciation and amortization in accordance with GAAP consistently applied;
- (u) convene any annual or special meeting (or any adjournment thereof) of the stockholders of Island, other than the Island Stockholders Meeting and the 2018 annual meeting of stockholders (only if such 2018 annual meeting is not otherwise combined with the Island Stockholders Meeting);
- (v) fail to use reasonable efforts to maintain existing material insurance policies or comparable replacement policies to the extent available for a reasonable cost; or
- (w) authorize or enter into any Contract to do any of the foregoing or otherwise make any commitment to do any of the foregoing.
- 5.2 Conduct of Business by Boat and its Subsidiaries Pending the Effective Time. Boat agrees that, between the date of this Agreement and the Effective Time, except as set forth in Section 5.2 of the Boat Disclosure Schedule, as required by Law or as otherwise expressly contemplated or required by this Agreement, unless Island shall otherwise consent in writing (such consent not to be unreasonably withheld, conditioned or delayed), Boat will, and will cause each of its Subsidiaries to, (i) conduct its operations only in the ordinary course of business consistent with past practice, (ii) use its commercially reasonable efforts to keep available the services of the current officers, employees and consultants of Boat and each of its Subsidiaries and to preserve the goodwill and current relationships of Boat and each of its Subsidiaries with Persons with which Boat or any of its Subsidiaries has business relations, and (iii) use its commercially reasonable efforts to preserve intact its business organization and comply with all applicable Laws. Without limiting the foregoing, except as set forth in Section 5.2 of the Boat Disclosure Schedule, as required by Law or as otherwise expressly contemplated or required by this Agreement, Boat shall not, and shall not permit any of its Subsidiaries to, between the date of this

Table of Contents

Agreement and the Effective Time, directly or indirectly, take any of the following actions without the prior written consent of Island (such consent not to be unreasonably withheld, conditioned or delayed):

- (a) amend or otherwise change its certificate of incorporation or bylaws or equivalent organizational documents (including by merger, consolidation or otherwise);
- (b) issue, sell, pledge, dispose of, grant, transfer or encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer, or encumbrance of, any shares of capital stock of, or other Equity Interests in, Boat or any of its Subsidiaries of any class, or securities convertible into, or exchangeable or exercisable for, any shares of such capital stock or other Equity Interests, or any options, warrants or other rights of any kind to acquire any shares of such capital stock or other Equity Interests or such convertible or exchangeable securities, or any other ownership interest (including any such interest represented by Contract right), or any "phantom" stock, "phantom" stock rights, stock appreciation rights or stock based performance rights, of Boat or any of its Subsidiaries, other than (i) the issuance of Boat Common Stock in connection with the exercise, vesting or payment of Boat Stock Options or Boat RSUs outstanding as of the date hereof in accordance with their terms, (ii) the issuance of any Boat Stock Options or Boat RSUs or Boat Common Stock (in lieu of cash compensation otherwise payable to members of the Boat Board), in either case, in the ordinary course of business consistent with past practice up to a maximum of 200,000 shares of Boat Common Stock subject thereto, (iii) in connection with the assumption and conversion of Island Stock Options and Island RSUs in accordance with the terms of this Agreement or (iv) the issuance of any shares of Boat Common Stock in accordance with the Boat ESPP;
- (c) sell, pledge, dispose of, transfer, lease, license, guarantee or encumber any property or assets of Boat or any of its Subsidiaries with value in excess of \$2,000,000, except pursuant to existing Contracts or the sale or purchase of goods in the ordinary course of business consistent with past practice;
- (d) (i) sell, assign, pledge, grant or acquire, agree to grant to or acquire from any Person, or otherwise encumber, transfer, license, abandon, place in the public domain, permit to lapse, disclose or agree to disclose or otherwise dispose of any Boat Material Intellectual Property, except pursuant to the terms of existing Contracts or the licensing of any such Intellectual Property in the ordinary course of business consistent with past practice or (ii) compromise, settle or agree to settle, or consent to judgment in, any one or more actions or institute any action concerning any Boat Material Intellectual Property except in the ordinary course of business consistent with past practice;
- (e) declare, set aside, make or pay any dividend or other distribution (whether payable in cash, shares, property or a combination thereof) with respect to any of its capital stock or other Equity Interests (other than dividends paid by a wholly owned Subsidiary of Boat to Boat or another wholly owned Subsidiary of Boat) or enter into any agreement with respect to the voting or registration of its capital stock or other Equity Interests;
- (f) reclassify, combine, split, subdivide or amend the terms of, or redeem, purchase or otherwise acquire, directly or indirectly, any of its capital stock or other Equity Interests;
- (g) form any new Subsidiary, or merge or consolidate Boat or any of its Subsidiaries with any Person or adopt a plan of complete or partial liquidation or resolutions providing for a complete or partial liquidation, dissolution, restructuring, recapitalization or other reorganization of Boat or any of its Subsidiaries, other than internal reorganizations in the ordinary course of business that would not have a material and adverse impact on Boat and its Subsidiaries or the Transaction;
- (h) acquire (including by merger, consolidation, or acquisition of shares or assets) any interest in any Person or any assets thereof in each case with value in excess of \$2,000,000, other than in the ordinary course of business consistent with past practice or pursuant to the Contracts set forth on *Section 5.2(h)* of the Boat Disclosure Schedule;

Table of Contents

- (i) repurchase, repay, refinance or incur any indebtedness for borrowed money, except as required by the terms of any such indebtedness as of the date hereof, or issue any debt securities or assume, guarantee or endorse, or otherwise as an accommodation become responsible for (whether directly, contingently or otherwise), the obligations of any Person (other than a wholly owned Subsidiary of Boat) for borrowed money, except for borrowings under Boat's existing credit facilities or issuances of commercial paper for working capital and general corporate purposes in the ordinary course of business consistent with past practice;
- (j) make any loans, advances or capital contributions to, or investments in, any other Person (other than any wholly owned Subsidiary of Boat) in excess of \$2,000,000 in the aggregate;
- (k) terminate, cancel, renew, or request or agree to any material change in or waiver under any Boat Material Contract, or enter into or amend any Contract that, if existing on the date hereof, would be a Boat Material Contract, in each case other than in the ordinary course of business consistent with past practice;
- (l) make or authorize any capital expenditure in excess of Boat's capital expenditure budget as disclosed to Island prior to the date hereof, other than capital expenditures that are not, in the aggregate, in excess of \$2,000,000;
- (m) except (i) in the ordinary course of business consistent with past practice with respect to individuals with a title below Vice President, or (ii) to the extent required by applicable Law or the existing terms of any Boat Benefit Plan disclosed in *Section 4.11(a)* of the Boat Disclosure Schedule: (A) materially increase the compensation or benefits payable or to become payable to directors, officers or employees of Boat or its Subsidiaries; (B) grant any rights to severance or termination pay or transaction or retention bonus payments to, or enter into any severance agreement with, any director, officer or employee of Boat or its Subsidiaries, (C) establish, adopt, enter into or materially amend any collective bargaining agreement or other Contract with any labor union or labor organization, Boat Benefit Plan or similar plan, agreement, trust, fund, policy or arrangement for the benefit of any director, officer or employee of Boat or its Subsidiaries; (D) take any action to amend or waive any performance or vesting criteria or accelerate vesting, exercisability or funding under any Boat Benefit Plan; or (E) hire or terminate the employment of any executive officer of Boat other than terminations for cause, death or disability;
 - (n) forgive any loans to directors, officers or employees of Boat or its Subsidiaries;
- (o) waive, release, pay, discharge or satisfy any claims, liabilities or obligations (absolute, accrued, contingent or otherwise) with value in excess of \$2,000,000, except in the ordinary course of business consistent with past practice and in accordance with their terms:
- (p) make any change in accounting policies, practices, principles, methods or procedures, other than as required by GAAP or by a Governmental Entity;
- (q) waive, release, assign, settle or compromise any claims or rights with value in excess of \$2,000,000 held by Boat or any of its Subsidiaries;
- (r) compromise, settle or agree to settle any Proceeding or investigation (including any Proceeding or investigation relating to this Agreement or the Transactions) other than compromises, settlements or agreements in the ordinary course of business consistent with past practice that involve only the payment of monetary damages not in excess of \$2,000,000 individually or in the aggregate, in any case without the imposition of equitable relief on, or the admission of wrongdoing by, Boat or any of its Subsidiaries;
- (s) except as required by applicable Law, make, change or revoke any material Tax election, change any Tax accounting period for purposes of a material Tax or material method of Tax accounting, file any material amended Tax Return, settle or compromise any audit or proceeding

Table of Contents

relating to a material amount of Taxes, agree to an extension or waiver of the statute of limitations with respect to a material amount of Taxes, enter into any "closing agreement" within the meaning of Section 7121 of the Code (or any similar provision of state, local, or non-U.S. Law) with respect to any material Tax, or surrender any right to claim a material Tax refund, or undertake any transaction with any Subsidiary of Boat that would reasonably be expected to have any significant U.S. federal income tax consequences to Boat or its Subsidiaries;

- (t) write up, write down or write off the book value of any assets, in the aggregate, in excess of \$2,000,000, except for depreciation and amortization in accordance with GAAP consistently applied;
- (u) convene any annual or special meeting (or any adjournment thereof) of the stockholders of Boat, other than the Boat Stockholders Meeting and the 2018 annual meeting of stockholders (only if such 2018 annual meeting is not otherwise combined with the Boat Stockholders Meeting);
- (v) fail to use reasonable efforts to maintain existing material insurance policies or comparable replacement policies to the extent available for a reasonable cost; or
- (w) authorize or enter into any Contract to do any of the foregoing or otherwise make any commitment to do any of the foregoing.
- 5.3 Preparation of the Form S-4 and the Joint Proxy Statement; Stockholders Meetings.
 - (a) As promptly as practicable after the execution of this Agreement, (i) Boat and Island shall jointly prepare and cause to be filed with the SEC, the Joint Proxy Statement to be sent to the stockholders of Boat and the stockholders of Island, as applicable, relating to the Boat Stockholders Meeting and the Island Stockholders Meeting and (ii) Boat and Island shall jointly prepare and Holdco shall file with the SEC the Form S-4, in which the Joint Proxy Statement will be included as a prospectus, in connection with the registration under the Securities Act of the shares of Holdco Common Stock to be issued in the Mergers, Each of Boat, Island and Holdco shall use its reasonable best efforts to have the Form S-4 declared effective as promptly as practicable after such filing (including by responding to comments of the SEC) and, prior to the effective date of the Form S-4, each of Boat, Island and Holdco shall take all action reasonably required to be taken under any applicable securities Laws in connection with the issuance of Boat Common Stock. Each of Boat and Island shall furnish all information as may be reasonably requested by the other party in connection with any such action and the preparation, filing and distribution of the Form S-4 and the Joint Proxy Statement. As promptly as practicable after the Form S-4 shall have become effective, each of Boat and Island shall use its reasonable best efforts to cause the Joint Proxy Statement to be mailed to its respective stockholders. No filing of, or amendment or supplement to, the Form S-4 will be made by Holdco, and no filing of, or amendment or supplement to, the Joint Proxy Statement will be made by Boat or Island, in each case without providing the other party with a reasonable opportunity to review and comment thereon. If at any time prior to the Effective Time any information relating to Boat or Island, or any of their respective affiliates, directors or officers, should be discovered by Boat or Island which should be set forth in an amendment or supplement to either the Form S-4 or the Joint Proxy Statement, so that either such document would not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, the party that discovers such information shall promptly notify the other party and an appropriate amendment or supplement describing such information shall be promptly filed with the SEC and, to the extent required by Law, disseminated to the stockholders of Boat and the stockholders of Island. Each party shall notify the other party promptly of the time when the Form S-4 has become effective, and of the issuance of any stop order or suspension of the qualification of the shares of Holdco Common Stock issuable in connection with the Mergers for offering or sale in any jurisdiction. In addition,

Table of Contents

each party agrees to provide the other party and its legal counsel with copies of any written comments, and shall inform the other party of any oral comments, that such party or its counsel may receive from time to time from the SEC or its staff with respect to the Form S-4 or the Joint Proxy Statement promptly after receipt of such comments, and any written or oral responses thereto. Each party and their respective counsel shall be given a reasonable opportunity to review any such written responses and each party shall give due consideration to the additions, deletions or changes suggested thereto by the other party and their respective counsel.

(b) Island Stockholders Meeting.

- (i) Island shall, promptly following the date on which the Form S-4 has been filed with the SEC and the parties otherwise mutually determine to be appropriate, establish a record date for, and, as soon as practicable following the effectiveness of the Form S-4, duly call and give notice of and convene and hold a meeting of its stockholders (the "Island Stockholders Meeting") for the purpose of seeking the Island Stockholder Approval; provided, however, that Island may postpone or adjourn the Island Stockholders Meeting (A) with the prior written consent of Boat, which consent shall not be unreasonably withheld, conditioned or delayed; (B) if a quorum has not been established; (C) to allow reasonable additional time for the filing and mailing of any supplemental or amended disclosure which the Island Board has determined in good faith after consultation with outside counsel is necessary under applicable Law and for such supplemental or amended disclosure to be disseminated and reviewed by Island's stockholders prior to the Island Stockholders Meeting; (D) to allow reasonable additional time to solicit additional proxies, if and to the extent the requisite Island Stockholder Approval would not otherwise be obtained; or (E) if required by Law; provided, however, that in the case of clauses (B), (C), (D) and (E), the Island Stockholders Meeting shall not be postponed or adjourned for more than ten (10) Business Days in the aggregate from the originally scheduled date of the Island Stockholders Meeting without the prior written consent of Boat; provided, further, that in the event that Boat postpones or adjourns the Boat Stockholders Meeting in accordance with the terms of this Agreement, Island may postpone or adjourn the Island Stockholders Meeting for an equal number of days. Island shall, upon the reasonable request of Boat, advise Boat at least on a daily basis on each of the last seven (7) Business Days prior to the date of the Island Stockholders Meeting as to the aggregate tally of proxies received by Island with respect to the Island Stockholder Approval.
- (ii) Island shall, through the Island Board, make the Island Recommendation and include such Island Recommendation in the Joint Proxy Statement (subject to *Section 5.4*) and use its reasonable best efforts to (A) solicit from its stockholders proxies in favor of the adoption of this Agreement and the Transactions, including the Mergers, and (B) take all other action necessary or advisable to secure the Island Stockholder Approval. Except as expressly permitted in *Section 5.4(b)* and *Section 5.4(d)*, neither the Island Board nor any committee thereof shall (x) withhold, withdraw, modify or qualify, or propose publicly to withhold, withdraw, modify or qualify, in a manner adverse to Boat, the approval, determination of advisability, or recommendation by the Island Board of, this Agreement, the Mergers and the other Transactions, (y) make, or permit any director or executive officer to make, any public statement in connection with the Island Stockholders Meeting by or on behalf of the Island Board or such committee that would reasonably be expected to have the same effect or (z) approve, determine to be advisable, or recommend, or propose publicly to approve, determine to be advisable, or recommend, any Competing Proposal (the actions specified in the foregoing clauses (x), (y) and (z) being referred to as an "*Island Adverse Recommendation Change*").
- (iii) Notwithstanding any Island Adverse Recommendation Change, unless this Agreement is terminated in accordance with its terms, the obligations of the parties hereunder

Table of Contents

shall continue in full force and effect. Without limiting the generality of the foregoing, unless this Agreement is terminated in accordance with its terms, this Agreement and the Island Merger shall be submitted to the stockholders of Island for approval at the Island Stockholders Meeting whether or not (x) the Island Board shall have effected an Island Adverse Recommendation Change or (y) any Competing Proposal shall have been publicly proposed or announced or otherwise submitted to Island or any of its Representatives.

(c) Boat Stockholders Meeting.

- (i) Boat shall, as soon as practicable following the date on which the Form S-4 has been filed with the SEC and the parties otherwise mutually determine to be appropriate, establish a record date for, and, as soon as practicable following the effectiveness of the Form S-4, duly call and give notice of and convene and hold a meeting of its stockholders (the "Boat Stockholders Meeting") for the purpose of seeking the Boat Stockholder Approval; provided, however, that Boat may postpone or adjourn the Boat Stockholders Meeting (A) with the prior written consent of Island, which consent shall not be unreasonably withheld, conditioned or delayed; (B) if a quorum has not been established; (C) to allow reasonable additional time for the filing and mailing of any supplemental or amended disclosure which the Boat Board has determined in good faith after consultation with outside counsel is necessary under applicable Law and for such supplemental or amended disclosure to be disseminated and reviewed by Boat's stockholders prior to the Boat Stockholders Meeting; (D) to allow reasonable additional time to solicit additional proxies, if and to the extent the requisite Boat Stockholder Approval would not otherwise be obtained; or (E) if required by Law; provided, however, that in the case of clauses (B), (C), (D) and (E), the Boat Stockholders Meeting shall not be postponed or adjourned for more than ten (10) Business Days in the aggregate from the originally scheduled date of the Boat Stockholders Meeting without the prior written consent of Island; provided, further, that in the event that Island postpones or adjourns the Island Stockholders Meeting in accordance with the terms of this Agreement, Boat may postpone or adjourn the Boat Stockholders Meeting for an equal number of days. Boat shall, upon the reasonable request of Island, advise Island at least on a daily basis on each of the last seven (7) Business Days prior to the date of the Boat Stockholders Meeting as to the aggregate tally of proxies received by Boat with respect to the Boat Stockholder Approval.
- (ii) Boat shall, through the Boat Board, make the Boat Recommendation and include such Boat Recommendation in the Joint Proxy Statement (subject to *Section 5.4*) and use its reasonable best efforts to (A) solicit from its stockholders proxies in favor of the adoption of this Agreement and the Transactions, including the Mergers, and (B) take all other action necessary or advisable to secure the Boat Stockholder Approval. Except as expressly permitted in *Section 5.4(b)* and *Section 5.4(d)*, neither the Boat Board nor any committee thereof shall (x) withhold, withdraw, modify or qualify, or propose publicly to withhold, withdraw, modify or qualify, in a manner adverse to Island, the approval, determination of advisability, or recommendation by the Boat Board of, this Agreement, the Mergers and the other Transactions, (y) make, or permit any director or executive officer to make, any public statement in connection with the Boat Stockholders Meeting by or on behalf of the Boat Board or such committee that would reasonably be expected to have the same effect or (z) approve, determine to be advisable, or recommend, or propose publicly to approve, determine to be advisable, or recommend, any Competing Proposal (the actions specified in the foregoing clauses (x), (y) and (z) being referred to as a "*Boat Adverse Recommendation Change*").
- (iii) Notwithstanding any Boat Adverse Recommendation Change, unless this Agreement is terminated in accordance with its terms, the obligations of the parties hereunder shall continue in full force and effect. Without limiting the generality of the foregoing, unless this

Table of Contents

Agreement is terminated in accordance with its terms, this Agreement and the Boat Merger shall be submitted to the stockholders of Boat for approval at the Boat Stockholders Meeting whether or not (x) the Boat Board shall have effected a Boat Adverse Recommendation Change or (y) any Competing Proposal shall have been publicly proposed or announced or otherwise submitted to Boat or any of its Representatives.

- (d) Island and Boat will use their respective reasonable best efforts to hold the Island Stockholders Meeting and the Boat Stockholders Meeting on the same date.
- (e) Promptly following the execution of this Agreement, Boat shall cause Holdco to execute and deliver, in accordance with Section 228 of the DGCL, written consents adopting this Agreement in its capacity as the sole stockholder of each of Merger Sub A and Merger Sub B.

5.4 No Solicitation of Transactions.

(a) Each of Island and Boat shall immediately cease, and shall cause its respective Subsidiaries and Representatives to immediately cease, any discussions or negotiations with any Person that may be ongoing with respect to a Competing Proposal, or any proposal that would reasonably be expected to lead to a Competing Proposal, and shall request to have returned promptly to Island or Boat, as applicable, any confidential information that has been provided in any such discussions or negotiations. From the date hereof until the earlier of the Effective Time or the date of termination of this Agreement in accordance with Article 7, each of Island and Boat shall not, and shall cause its respective Subsidiaries and Representatives not to, directly or indirectly, (i) solicit, initiate or knowingly encourage or induce (including by way of furnishing information), or take any other action designed to facilitate, any inquiries or the making of any proposal which constitutes, or would reasonably be expected to lead to, any Competing Proposal, or (ii) engage in any discussions or negotiations regarding any Competing Proposal; provided, however, that (x) such party may ascertain facts from the Person making an unsolicited Competing Proposal for the sole purpose of the Island Board or the Boat Board, as applicable, informing itself about the terms of such Competing Proposal and the Person that made it and (y) if, prior to obtaining the Island Stockholder Approval (in the case of Island) or the Boat Stockholder Approval (in the case of Boat) and following the receipt of a bona fide written Competing Proposal made after the date hereof that the Island Board or Boat Board, as applicable, determines in good faith (after receiving advice of its financial advisor and of its outside legal counsel) is or would reasonably be expected to lead to a Superior Proposal and that was not, directly or indirectly, solicited, initiated or knowingly encouraged in violation of this Section 5.4, the Island Board or the Boat Board, as applicable, determines in good faith, after consultation with outside legal counsel, that a failure to take action with respect to such Competing Proposal, as applicable, would constitute a breach of its fiduciary duties under applicable Law, Island or Boat may, in response to such Competing Proposal, as applicable, and subject to compliance with Section 5.4(c), (A) furnish information with respect to Island or Boat, as applicable, to the Person making such Competing Proposal pursuant to an Acceptable Confidentiality Agreement, and (B) engage in discussions or negotiations with such Person regarding such Competing Proposal. Except as expressly permitted by this Section 5.4, each of Boat and Island shall not, and shall cause their respective Subsidiaries and Representatives not to, from and after the date of this Agreement until the earlier of the Effective Time or the date, if any, on which this Agreement is terminated pursuant to Article 7, directly or indirectly (1) approve, endorse, recommend or enter into, or publicly propose to approve, endorse, recommend or enter into, any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement or similar definitive agreement (other than an Acceptable Confidentiality Agreement) with respect to any Competing Proposal; (2) take any action to make the provisions of any takeover statute inapplicable to any transactions contemplated by a Competing Proposal; or (3) terminate, amend, release, modify or knowingly fail to enforce any provision of, or grant any permission, waiver or

Table of Contents

request under, any standstill, confidentiality or similar agreement entered into by the applicable party in respect of or in contemplation of a Competing Proposal (other than to the extent the Boat Board or the Island Board, as applicable, determines in good faith after consultation with its outside legal counsel, that failure to take any of such actions under clause (3) would be constitute a breach of its fiduciary duties under applicable Law), or (4) propose to do any of the foregoing. For the avoidance of doubt, nothing in this Section 5.4(a) shall relieve any party from its obligations under Section 5.6.

- (b) Notwithstanding any other provision of this Agreement, including Section 5.3 but subject to compliance with this Section 5.4, prior to receipt of the Island Stockholder Approval, the Island Board may, or, prior to receipt of the Boat Stockholder Approval, the Boat Board may, in response to any bona fide written Competing Proposal that was not, directly or indirectly, solicited, initiated or knowingly encouraged in violation of this Section 5.4, effect an Island Adverse Recommendation Change or a Boat Adverse Recommendation Change, as applicable, if and only if (i) the Island Board or the Board, as applicable, concludes in good faith, after consultation with Island's or Boat's outside financial advisors and outside legal counsel, that such Competing Proposal constitutes a Superior Proposal; (ii) the Boat Board or the Island Board, as applicable, provides the other party four (4) Business Days prior written notice of its intention to take such action (a "Competing Proposal Notice"), which notice shall include the information with respect to such Competing Proposal that is specified in Section 5.4(c), as well as a copy of such Competing Proposal (it being agreed that neither the delivery of such notice by a party nor any public announcement thereof that such party determines it is required to make under applicable Law shall constitute an Island Adverse Recommendation Change or a Boat Adverse Recommendation Change, as applicable, unless and until such party shall have failed at or prior to the end of the period referred to in clause (iii) below (and, upon the occurrence of such failure, such notice and such public announcement shall constitute an Island Adverse Recommendation Change or a Boat Adverse Recommendation Change, as applicable) to publicly announce that it (A) is recommending the Transactions and (B) has determined that such other Competing Proposal (taking into account (x) any modifications or adjustments made to the Transactions agreed to by the other party in writing and (y) any modifications or adjustments made to such other Competing Proposal) is not a Superior Proposal and has publicly rejected such Competing Proposal); (iii) during the four (4) Business Days following such written notice (the "Negotiation Period"), if requested by the other party, the Board of Directors effecting the recommendation change and its Representatives have negotiated in good faith with the other party regarding any revisions to the terms of the Transactions proposed by the other party in response to such Competing Proposal; and (iv) at the end of the four (4) Business Day period described in the foregoing clause (iii), the Island Board or Boat Board, as applicable, concludes in good faith, after consultation with Island's or Boat's outside legal counsel and financial advisors (and taking into account any adjustment or modification of the terms of this Agreement to which the other party has agreed in writing to make to the terms of the Transactions), that the Competing Proposal continues to be a Superior Proposal and, after consultation with Island's or Boat's outside legal counsel, that the failure to make an Island Adverse Recommendation Change or Boat Adverse Recommendation Change, as applicable, would constitute a breach of the fiduciary duties of the Island Board or Boat Board under applicable Law. Any material amendment or modification to any Competing Proposal shall require a new Competing Proposal Notice and the Negotiation Period shall be extended by an additional two (2) Business Days from the date of receipt of such new Competing Proposal Notice.
- (c) In addition to the obligations of Boat and Island set forth in *Section 5.4(a)* and *Section 5.4(b)*, Boat or Island shall promptly, and in any event no later than 24 hours, after it receives (i) any Competing Proposal or indication by any Person that is considering making a Competing Proposal, (ii) any request for non-public information relating to Boat or Island or their respective Subsidiaries, other than requests for information that would not reasonably be expected

Table of Contents

to result in a Competing Proposal, or (iii) any inquiry or request for discussions or negotiations regarding any Competing Proposal, notify the other party orally and in writing of any of the foregoing occurrences, the identity of the person making such request, inquiry or Competing Proposal (or where no such copy is available, a reasonably detailed description of such request, inquiry or Competing Proposal). Each party shall keep the other party reasonably informed (orally and in writing) on a current basis (and in any event at the other party's request and otherwise no later than 24 hours after the occurrence of any material changes, developments, discussions or negotiations) of the status of any request, inquiry or Competing Proposal (including the terms and conditions thereof and of any modification thereto), and any material developments, discussions and negotiations, including furnishing copies of any written inquiries, material correspondence and draft documentation, and written summaries of any material oral inquiries or discussions. Without limiting the foregoing, each party shall promptly (and in any event within 24 hours) notify the other party orally and in writing if it determines to begin providing information or to engage in discussions or negotiations concerning a Competing Proposal pursuant to this *Section 5.4*. Each of Boat and Island agrees that, subject to applicable restrictions under applicable Law, it shall, prior to or concurrent with the time it is provided to any third parties, provide to the other party any non-public information concerning Boat or Island and their respective Subsidiaries that Boat or Island provided to any third party in connection with any Competing Proposal which was not previously provided to the other party.

- (d) Notwithstanding anything in this *Section 5.4* to the contrary, at any time prior to obtaining the Boat Stockholder Approval or the Island Stockholder Approval, the Boat Board or the Island Board, as applicable, may make a Boat Adverse Recommendation Change or an Island Adverse Recommendation Change, as applicable, if (i) such board determines that an Intervening Event has occurred and is continuing and (ii) such board determines in good faith (after consultation with outside counsel) that the failure to make a Boat Adverse Recommendation Change or an Island Adverse Recommendation Change, as applicable, in response to such Intervening Event would constitute a breach of its fiduciary duties under applicable Law; *provided* that (x) the Boat Board or the Island Board has given the other party at least four (4) Business Days prior written notice of its intention to take such action and specifying in reasonable detail the circumstances related to such determination and (y) prior to effecting a Boat Adverse Recommendation Change or an Island Adverse Recommendation Change, the applicable party has negotiated, and has caused its Representatives to negotiate, in good faith with the other party during such notice period to the extent such other party wishes to negotiate, to enable such party to revise the terms of this Agreement, such that the failure to make a Boat Adverse Recommendation Change or an Island Adverse Recommendation Change, as applicable, would not constitute a breach of its fiduciary duties under applicable Law.
- (e) Nothing contained in this Agreement shall prohibit the Boat Board or the Island Board from (i) taking and disclosing to their stockholders, as applicable, a position contemplated by Rule 14d-9 or Rule 14e-2(a) promulgated under the Exchange Act regarding a Company Proposal, or from issuing a "stop, look and listen" statement pending disclosure of its position thereunder, or (ii) making any disclosure to their stockholders regarding a Company Proposal if the Island Board or Boat Board, as applicable, determines in good faith, after consultation with its outside counsel, that the failure to make such disclosure would be inconsistent with its duties to stockholders under, or would violate, applicable Law, *provided*, that if any such disclosure pursuant to this clause (ii) has the substantive effect of withdrawing or adversely modifying the Boat Recommendation or the Island Recommendation, such disclosure shall be deemed to be Boat Adverse Recommendation Change or an Island Adverse Recommendation Change, as applicable for purposes of this Agreement (it being understood that any "stop, look or listen" communication that contains only the information set forth in Rule 14d-9(f) shall not be deemed, in and of itself to be a Boat Adverse Recommendation Change or an Island Adverse Recommendation Change).

Table of Contents

- (f) For purposes of this Agreement:
 - (i) "Competing Proposal" shall mean, with respect to Boat or Island, as applicable, other than the Transactions, any proposal or offer from a third party relating to (A) a merger, reorganization, sale of assets, out license, share exchange, consolidation, business combination, recapitalization, dissolution, liquidation, joint venture or similar transaction involving Boat or Island, as applicable, or any of their respective Subsidiaries; (B) the acquisition (whether by merger, consolidation, equity investment, joint venture, out license or otherwise) by any Person of twenty percent (20%) or more of the consolidated assets of Boat or Island, and their respective Subsidiaries, as determined on a book-value or fair-market-value basis; (C) the purchase or acquisition, in any manner, directly or indirectly, by any Person of twenty percent (20%) or more of the issued and outstanding shares of the Boat Common Stock or the Island Common Stock, as applicable, or any other Equity Interests in Boat or Island, as applicable, (D) any purchase, acquisition, tender offer or exchange offer that, if consummated, would result in any Person beneficially owning twenty percent (20%) or more of the shares of Boat Common Stock or Island Common Stock, as applicable, or any other Equity Interests of Boat, Island or any of their respective Subsidiaries, as applicable, or (E) any combination of the foregoing.
 - (ii) "Intervening Event" means any material event, development or change in circumstances that first occurs, arises or becomes known to Boat or Island or its respective board, as applicable, after the date of this Agreement, to the extent that such event, development or change in circumstances was not reasonably foreseeable as of the date of this Agreement; provided, however, that in no event shall the following events, developments or changes in circumstances constitute an Intervening Event: (1) the receipt, existence or terms of a Competing Proposal or any matter relating thereto or consequence thereof; (2) any change in the price, or change in trading volume, of Boat Common Stock or Island Common Stock (provided, however, that the exception to this clause (2) shall not apply to the underlying causes giving rise to or contributing to such change or prevent any of such underlying causes from being taken into account in determining whether an Intervening Event has occurred); and (3) meeting or exceeding internal or analysts' expectations, projections or results of operations (provided, however, that the exception to this clause (3) shall not apply to the underlying causes giving rise to or contributing to such circumstances or prevent any of such underlying causes from being taken into account in determining whether an Intervening Event has occurred).
 - (iii) "Superior Proposal" means a bona fide written Competing Proposal (except the references therein to "20%" shall be replaced by "80%") made by a third party which was not solicited by Boat or Island or any of their respective Representatives and which, in the good faith judgment of the Boat Board or the Island Board, as applicable, and after consultation with its outside financial and legal advisors, taking into account the various legal, financial and regulatory aspects of the Competing Proposal, including the financing terms thereof, and the third party making such Competing Proposal (A) if accepted, is reasonably likely to be consummated, (B) if consummated, would result in a transaction that is more favorable to Boat's stockholders or Island's stockholders, as applicable, from a financial point of view, than the Mergers and the other Transactions contemplated hereby (after giving effect to all adjustments or modifications to the terms thereof which may be agreed in writing to be made by the other party (including pursuant to Section 5.4(b))) and (C) if a cash transaction (in whole or in part), financing for which is then fully committed or reasonably determined to be available.

Table of Contents

- 5.5 Access to Information; Confidentiality.
 - (a) Upon reasonable notice, each of Boat and Island shall (and shall cause their respective Subsidiaries and Representatives to) afford to the other party and its Representatives reasonable access during normal business hours, during the period prior to the Effective Time, to all its properties, books, Contracts and records and its officers, employees and Representatives and, during such period, each of Boat and Island shall (and shall cause its Subsidiaries and Representatives to) furnish promptly to the other party (i) a copy of each report, schedule, registration statement and other document filed, published, announced or received by it during such period pursuant to the requirements of applicable securities Laws (other than reports or documents which such party is not permitted to disclose under applicable Law) and (ii) consistent with its obligations under applicable Law, all other information concerning its business, properties and personnel as the other party may reasonably request; provided, however, none of Boat or Island or any of their respective Subsidiaries or Representatives shall be required to provide access to or disclose information where such information or access would, in the reasonable judgment of such party, (x) breach any agreement with any third party, (y) constitute a waiver of the attorney-client or other privilege held by such party or (z) otherwise violate any applicable Law. In the event any of the restrictions in clauses (x) through (z) of the foregoing sentence shall apply, each party shall advise the other party of the subject matter of any such information that cannot be disclosed and the parties shall use their reasonable best efforts to make appropriate alternate disclosure arrangements, including adopting additional specific procedures to protect the confidentiality of sensitive material and to ensure compliance with applicable Laws. Any such information provided pursuant to this Section 5.5 shall be held in confidence to the extent required by, and in accordance with, the provisions of that certain Confidentiality Agreement, dated October 18, 2017, as such agreement may be amended from time to time in accordance with its terms (the "Confidentiality Agreement"), between Boat and Island, which Confidentiality Agreement shall remain in full force and effect.
 - (b) Each of the parties hereby agree that no investigation by any of the parties or their respective Representatives or information provided, made available or delivered pursuant to this Agreement shall affect the representations, warranties, covenants or agreements of any other party set forth herein.
- 5.6 Appropriate Action.
 - (a) Island, Merger Sub A, Merger Sub B, Holdco and Boat shall use their reasonable best efforts to take or cause to be taken all appropriate action, and to do, or cause to be done, all things necessary to consummate and make effective the transactions contemplated under this Agreement, including using their reasonable best efforts to obtain, or cause to be obtained, all waivers, permits, consents, approvals, authorizations, qualifications and orders of all Governmental Entities and parties to Contracts with Island, Boat or any of their respective Subsidiaries that may be or become necessary for the performance of obligations pursuant to this Agreement and the consummation of the transactions contemplated hereby. The parties shall cooperate and assist one another in good faith (i) in connection with all actions to be taken pursuant to this *Section 5.6(a)*, including the preparation and making of the filings referred to herein and, if requested, amending or furnishing additional information thereunder, and (ii) in seeking, as promptly as reasonably practicable, to obtain all such waivers, permits, consents, approvals, authorizations, qualifications and orders. Upon the terms and subject to the conditions set forth in this Agreement, each party agrees to make any filings required to be made pursuant to the HSR Act or other applicable Antitrust Laws with respect to the Transactions as promptly as reasonably practicable (and, in the case of the required Notification and Report Forms pursuant to the HSR Act, in any event within ten (10) Business Days after the date hereof) and to supply as promptly as reasonably practicable to the appropriate Governmental Entities any additional information and documentary material

Table of Contents

that may be requested by such Governmental Entities pursuant to the HSR Act or such other applicable Antitrust Laws. All such antitrust filings to be made shall be made in substantial compliance with the requirements of the HSR Act and such other applicable Antitrust Laws, as applicable.

- (b) In furtherance, and without limiting the generality, of the foregoing, the parties shall use their reasonable best efforts to cooperate with and assist each other in good faith to (A) determine, as promptly as reasonably practicable, which filings are required to be made pursuant to the HSR Act or other applicable Antitrust Laws with respect to the Transactions, (B) provide or cause to be provided as promptly as reasonably practicable to the other party all necessary information and assistance as any Governmental Entity may from time to time require of such party in connection with obtaining the relevant waivers, permits, consents, approvals, authorizations, qualifications, Orders or expiration of waiting periods in relation to such filings or in connection with any other review or investigation of the Transactions by a Governmental Entity pursuant to the HSR Act or other applicable Antitrust Laws and (C) provide or cause to be provided as promptly as reasonably practicable all assistance and cooperation to allow the other party to prepare and submit any such filings or submissions required to be submitted under the HSR Act or other applicable Antitrust Laws, including providing to the other party any information that the other party may from time to time require for the purpose of any filing with, notification to, application with, or request for further information made by, any Governmental Entity in respect of any such filing. To the extent permitted by applicable Law and not prohibited by any Governmental Entity, each party shall (A) keep the other apprised of the content and status of any communications with, and communications from, any Governmental Entity with respect to the Transactions, including promptly notifying the other party of any communication it receives from any Governmental Entity relating to any review or investigation of the Transactions under the HSR Act or other applicable Antitrust Laws, and (B) provide advance notice of and permit representatives of the other party to be present and participate at each meeting or teleconference relating to any review or investigation of the Transactions under the HSR Act or other applicable Antitrust Laws; provided, further, that each party shall be consulted in advance in connection with any argument, opinion or proposal to be made or submitted to any Governmental Entity in such meetings or teleconferences. To the extent permitted by applicable Law, the parties shall, and shall use their reasonable best efforts to cause their respective affiliates to use their reasonable best efforts to, provide each other with copies of all correspondence, filings or communications between them or any of their respective Representatives, on the one hand, and any Governmental Entity or members of its staff, on the other hand, with respect to this Agreement and the Transactions; provided, however, that materials may be redacted (1) to remove references concerning the valuation of Boat and its Subsidiaries or Island, as applicable; (2) as necessary to comply with any Contract or Laws; and (3) as necessary to address reasonable attorney-client or other privilege or confidentiality concerns.
- (c) Subject to the penultimate sentence of this *Section 5.6(c)*, Island and Boat shall, and shall cause each of their respective Subsidiaries to, take any and all steps necessary to obtain approval of the consummation of the Transactions by any antitrust or competition Governmental Entity, including taking all steps necessary to avoid or eliminate each and every legal impediment under any applicable Antitrust Law that may be asserted by any antitrust or competition Governmental Entity so as to enable the parties hereto to close the Transactions as promptly as reasonably practicable, and in any event prior to the Outside Date. Notwithstanding the foregoing or any other provision of this Agreement, nothing in this *Section 5.6* shall require Boat nor any of its Subsidiaries and affiliates nor Island nor any of its Subsidiaries and affiliates to agree to any sale, transfer, license, separate holding, divestiture or other disposition of, or to any prohibition of or any limitation on the acquisition, ownership, operation, effective control or exercise of full rights of ownership, or other modification of rights in respect of, any material business, product line or

Table of Contents

asset, or otherwise modify any material business practice or contractual relationship (each, an "Antitrust Restraint"). In the event of any conflict between subsections (a), (b) or (c) of this Section 5.6, the provisions of this Section 5.6(c) shall, with respect to the matters addressed in this Section 5.6(c), supersede the provisions of subsection (a) and (b) of this Section 5.6.

- (d) Each party shall, and shall cause its respective Subsidiaries to, respond as promptly as reasonably practicable to any inquiries or requests for information and documentary material received from any Governmental Entity in connection with any antitrust or competition matters related to this Agreement and the transactions contemplated hereby.
- (e) Without the prior written approval of the other party, no party shall effect or agree to any business combination (whether structured as a merger, business combination, tender offer, exchange offer or similar transaction) or acquisition of any assets, licenses, rights, product lines, operations or businesses of any person that would reasonably be expected to prevent or materially delay consummation of the Transactions.
- (f) Boat will take all actions necessary to cause Holdco, Merger Sub A and Merger Sub B to perform their respective obligations under this Agreement and to consummate the Mergers on the terms and subject to the conditions set forth in this Agreement.
- 5.7 Certain Notices. Each party shall give prompt notice to the other parties if any of the following occur after the date of this Agreement: (a) receipt of any notice or other communication in writing from any Person alleging that the consent or approval of such Person is or may be required in connection with the Transactions; (b) receipt of any notice or other communication from any Governmental Entity or the NASDAQ (or any other securities market) in connection with the Transactions; or (c) such party becoming aware of the occurrence of an event that could prevent or delay beyond the Outside Date the consummation of the Transactions or that would reasonably be expected to result in any of the conditions to the Mergers set forth in Article 6 not being satisfied; provided, however, that the delivery of any notice pursuant to this Section 5.7 shall not cure any breach of any representation, warranty, covenant or agreement contained in this Agreement or otherwise limit or affect the remedies available hereunder to the party receiving such notice. Island and Boat shall promptly provide, to the extent not publicly available, the other party with copies of all filings made by such party with any Governmental Entity in connection with the Transactions.
- 5.8 Public Announcements. Each party agrees that no public release or announcement concerning the Transactions shall be issued by any party without the prior written consent of Boat and Island (which consent shall not be unreasonably withheld, conditioned or delayed), except to the extent such release or announcement may be required by applicable Law or the rules or regulations of any applicable securities exchange or regulatory or governmental body to which the relevant party is subject, in which case the party required to make the release or announcement shall use its reasonable best efforts to allow each other party reasonable time to comment on such release or announcement in advance of such issuance. Boat and Island agree that the press release announcing the execution and delivery of this Agreement shall be a joint release in the form heretofore agreed by the parties.
 - 5.9 Director and Officer Indemnification.
 - (a) From and after the Effective Time, Holdco shall, and shall cause each of the Island Surviving Corporation and the Boat Surviving Corporation to, indemnify, defend and hold harmless, and shall advance expenses as incurred, to the fullest extent permitted under (i) applicable Law, (ii) Island's and Boat's organizational documents in effect as of the date of this Agreement, and (iii) any Contract of Island or Boat in effect as of the date of this Agreement, each present and former director and officer of Island or Boat (each, an "Indemnitee" and, collectively, the "Indemnitees") against any costs or expenses (including reasonable attorneys' fees), judgments, settlements, fines, losses, claims, damages or liabilities incurred in connection with any

Table of Contents

Proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to matters existing or occurring at or prior to the Effective Time, including in connection with this Agreement or the Transactions.

- (b) Holdco agrees that all rights to exculpation, indemnification or advancement of expenses arising from, relating to, or otherwise in respect of, acts or omissions occurring prior to the Effective Time (including in connection with this Agreement or the Transactions) now existing in favor of an Indemnitee as provided in its certificate of incorporation, bylaws or other organizational documents shall survive the Mergers and shall continue in full force and effect in accordance with their terms. For a period of no less than six (6) years from the Effective Time, Holdco shall cause each of the Island Surviving Corporation and the Boat Surviving Corporation to, and each of the Island Surviving Corporation and the Boat Surviving Corporation shall maintain in effect the exculpation, indemnification and advancement of expenses provisions of the organizational documents of Island and Boat, respectively, in effect as of the date of this Agreement, and shall not amend, repeal or otherwise modify any such provisions in any manner that would adversely affect the rights thereunder of any individuals who immediately before the Effective Time were current or former directors or officers of Island or Boat, *provided*, *however*, that all rights to exculpation, indemnification and expenses in respect of any Proceeding pending or asserted or any claim within such period shall continue until the final disposition of such proceeding.
- (c) For six (6) years from and after the Effective Time, Holdco shall maintain for the benefit of the Indemnitees, D&O insurance policies that provide coverage for events occurring prior to the Closing Date (the "D&O Insurance") that is substantially equivalent to and in any event not less favorable in the aggregate than the existing policies of Island or Boat, or, if substantially equivalent insurance coverage is unavailable, the best available coverage; provided, however, that Holdco shall not be required to pay an annual premium for the D&O Insurance in excess of 300% of the last annual premium paid by Island or Boat, as applicable, prior to the date of this Agreement (it being understood and agreed that, in the event that the requisite coverage is not available for an annual premium less than or equal to 300% of such last annual premium, Holdco shall nevertheless be obligated to provide such coverage as may be obtained for 300% of such last annual premium). The provisions of the immediately preceding sentence shall be deemed to have been satisfied if prepaid "tail" policies have been obtained by Island or Boat prior to the Effective Time, which policies provide such directors and officers with coverage for an aggregate period of six (6) years with respect to claims arising from facts or events that occurred on or before the Effective Time, including in respect of the Transactions.
- 5.10 Stock Exchange Listing. Holdco, Boat and Island shall use their reasonable best efforts to cause the shares of Holdco Common Stock to be issued in connection with the Mergers (including shares of Holdco Common Stock to be reserved upon exercise of Holdco Stock Options or Holdco Warrants) to be listed on the NASDAQ under a ticker symbol to be be agreed upon in writing by Boat and Island, subject to official notice of issuance, prior to the Effective Time. Prior to the Closing, the parties shall cause Holdco to submit an initial listing application with NASDAQ (the "NASDAQ Listing Application") with respect to such shares of Holdco Common Stock. Each of Boat, Island and Holdco shall use its reasonable best efforts to have the NASDAQ Listing Application approved (subject to official notice of issuance) as promptly as practicable after such submission (including by responding to comments of NASDAQ). Each of Boat and Island shall furnish all information as may be reasonably requested by the other party in connection with any such action and the preparation and submission of the NASDAQ Listing Application. No submission of, or amendment or supplement to, the NASDAQ Listing Application will be made by any party without providing the other parties hereto with a reasonable opportunity to review and comment thereon. In addition, each party agrees to provide the other party and its legal counsel with copies of any written comments, and shall inform the other party

Table of Contents

of any oral comments, that such party or its counsel may receive from time to time from NASDAQ or its staff with respect to the NASDAQ Listing Application promptly after receipt of such comments, and any written or oral responses thereto. Each party and their respective counsel shall be given a reasonable opportunity to review any such written responses and each party shall give due consideration to the additions, deletions or changes suggested thereto by the other party and their respective counsel.

- 5.11 Section 16 Matters. Prior to the Effective Time, Boat and Island shall take all such steps as may be required to cause any dispositions of Island Common Stock (including derivative securities with respect to Island Common Stock) or Boat Common Stock (including derivative securities with respect to Boat Common Stock) or acquisitions of Holdco Common Stock (including derivative securities with respect to Holdco Common Stock) resulting from the Transactions by each individual who is subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to Boat or Island, to be exempt under Rule 16b-3 promulgated under the Exchange Act.
- 5.12 Stockholder Litigation. Each of Boat and Island shall provide the other party the opportunity to participate in the defense of any litigation brought by stockholders of Boat or stockholders of Island or in the name of Boat or Island against Boat or Island, as applicable, and/or their respective directors relating to the Transactions contemplated by this Agreement, including the Mergers; provided, however, that no party shall compromise, settle, come to an arrangement regarding or agree to compromise, settle or come to an arrangement regarding any litigation arising or resulting from the Transactions contemplated by this Agreement, or consent to the same, without the prior written consent of the other party (not to be unreasonably withheld, conditioned or delayed).

5.13 Tax Matters.

- (a) Notwithstanding anything herein to the contrary, none of Island, Boat or Holdco shall (and shall cause their respective Subsidiaries not to) take, or omit to take, any action that would, or could reasonably be expected to, prevent or impede (i) the Mergers from together being treated as an "exchange" described in Section 351 of the Code or (ii) the Boat Merger from qualifying as a "reorganization" within the meaning of Section 368(a) of the Code. Both prior to and following the Effective Time, Island, Boat and Holdco shall use their reasonable best efforts, and shall cause their respective Subsidiaries to use their reasonable best efforts, to take or cause to be taken any action necessary for (A) the Mergers together to be treated as an "exchange" described in Section 351 of the Code and (B) the Boat Merger to qualify as a "reorganization" within the meaning of Section 368(a) of the Code (collectively, the "Intended Tax Treatment"), including by reasonably refraining from any action that such party knows, or is reasonably expected to know, is reasonably likely to prevent the Intended Tax Treatment.
- (b) Island and Boat shall use their reasonable best efforts to obtain any Tax opinions required to be filed with the SEC in connection with the filing of the Form S-4, the Boat Tax Opinion and the Island Tax Opinion, including by (i) delivering to Skadden, Arps, Slate, Meagher and Flom LLP and Latham & Watkins LLP Tax representation letters in substantially the forms set forth in *Exhibit D* and *Exhibit E*, respectively, dated and executed as of the date the Form S-4 shall have been declared effective by the SEC or such other date(s) as determined necessary by counsel in connection with the filing of the Form S-4 or its exhibits, and (ii) delivering to Skadden, Arps, Slate, Meagher and Flom LLP and Latham & Watkins LLP, dated and executed as of the dates of the Boat Tax Opinion and the Island Tax Opinion, tax representation letters in substantially the forms set forth in *Exhibit D* and *Exhibit E*, respectively.
- (c) This Agreement is intended to constitute, and the parties hereto hereby adopt this Agreement as, a "plan of reorganization" within the meaning of Treasury Regulation Section 1.368-2(g) and 1.368-3(a). Each of Boat and Island shall report the Mergers in accordance with the Intended Tax Treatment and shall not take any position inconsistent with such treatment unless otherwise required pursuant to a "determination" within the meaning of Section 1313(a) of the Code.

Table of Contents

5.14 Employee Matters.

- (a) At and following the Effective Time, Holdco will and will cause its Subsidiaries (including the Boat Surviving Corporation and the Island Surviving Corporation and their respective Subsidiaries) to honor all obligations under the Island Benefit Plans, Boat Benefit Plans and applicable collective bargaining agreements, agreements with a works council and labor contracts in accordance with their terms as in effect immediately before the Effective Time; *provided*, that nothing in this *Section 5.14(a)* is intended to prevent Holdco from causing them to be enforced in accordance with their terms, including any reserved right to amend, modify, suspend, revoke or terminate them.
- (b) Each employee of the Boat Surviving Corporation and the Island Surviving Corporation and their respective Subsidiaries as of the Effective Time (including any employee who is full-time, part-time, temporary, on vacation or on a medical or disability or any other paid or unpaid approved leave of absence) who continues employment with Holdco or its Subsidiaries (including the Boat Surviving Corporation and the Island Surviving Corporation and their respective Subsidiaries) following the Effective Time (each, a "Continuing Employee") shall continue to receive, through December 31, 2018 (i) base salary or wages and annual bonus opportunity that are not less than the base salary or wages and annual bonus opportunity received by such Continuing Employee immediately prior to the Effective Time and (ii) retirement, health and welfare, severance and other benefits (including equity awards but excluding defined benefit arrangements and retiree health and welfare arrangements), in each case, that are, in the aggregate, no less favorable than those provided to such Continuing Employee immediately prior to the Closing Date.
- (c) Subject to applicable Law, Holdco will and will cause its Subsidiaries (including the Boat Surviving Corporation and the Island Surviving Corporation and their respective Subsidiaries) to give credit under each of their respective employee benefit plans, programs and arrangements (the "Post-Closing Plans") to Continuing Employees for all service prior to the Effective Time with Island or Boat or their respective Subsidiaries, as applicable, or any predecessor employer (to the extent that such credit was given by Island or Boat or any of their respective Subsidiaries, as applicable) for all purposes for which such service was taken into account or recognized by Island or Boat or their respective Subsidiaries, as applicable, for purposes of eligibility for participation and vesting (but not accrual of benefits, other than for purposes of determining the level of vacation, travel and/or severance benefits), but not to the extent crediting such service would result in duplication of benefits and not for purposes of defined benefits plans or retiree welfare plans, except to the extent required by applicable Laws. In addition, Holdco, Boat and Island agree to the general terms set forth on Section 5.14(c) of each of the Island Disclosure Schedule and the Boat Disclosure Schedule.
- (d) In addition, and without limiting the generality of *Section 5.14(b)* or any other provisions herein, for purposes of each Post-Closing Plan providing medical, dental, pharmaceutical, vision and/or other health benefits to any Continuing Employee and his or her dependents, Holdco will use its commercially reasonable efforts to and to cause its Subsidiaries (including the Boat Surviving Corporation and the Island Surviving Corporation and their respective Subsidiaries) to, (i) cause all waiting periods, pre-existing condition exclusions, evidence of insurability requirements and actively-at-work requirements of such Post-Closing Plan to be waived for such Continuing Employee and his or her covered dependents, to the extent any such waiting periods, pre-existing condition exclusions, evidence of insurability requirements and actively-at-work requirements were waived or were inapplicable under the comparable Island Benefit Plan or Boat Benefit Plan, (ii) fully credit each Continuing Employee with all deductible payments, co-payments and other out-of-pocket expenses incurred by such Continuing Employee and his or her covered dependents under the medical, dental, pharmaceutical or vision benefit plans of Island or Board (or their

Table of Contents

respective Subsidiaries), as applicable, prior to the Effective Time during the plan year in which the Effective Time occurs for the purpose of determining the extent to which such Continuing Employee has satisfied the deductible, co-payments, or maximum out-of-pocket requirements applicable to such Continuing Employee and his or her covered dependents for such plan year under any Pre-Closing Plan providing medical, dental, pharmaceutical, vision or health benefits, as if such amounts had been paid in accordance with such plan, and (iii) credit the accounts of such Continuing Employees under any Post-Closing Plan that is a flexible spending plan with any unused balance in the account of such Continuing Employee under the applicable Island Benefit Plan or Boat Benefit Plan.

- (e) The parties to this Agreement hereby acknowledge and agree that the consummation of the Mergers shall constitute a change in control or change of control (or words of similar import), as the case may be, for all purposes under each of the Island Benefit Plans and Boat Benefit Plans for which such concept is relevant.
- (f) Prior to the Effective Time, unless as otherwise mutually agreed in writing by Boat and Island no later than ten (10) Business Days prior to the Effective Date, (i) the Island Board shall take all actions with respect to the Island ESPP that are necessary to provide that, subject to the consummation of the Mergers, the Island ESPP shall terminate, effective immediately prior to the Effective Time and (ii) the Boat Board shall take all actions with respect to the Boat ESPP that are necessary to provide that subject to the consummation of the Mergers, the Boat ESPP shall terminate, effective immediately prior to the Effective Time.
- (g) Without limiting the generality of *Section 8.8*, this *Section 5.14* shall be binding upon and inure solely to the benefit of the parties to this Agreement, and nothing in this *Section 5.14*, express or implied, shall confer upon any other Person any rights or remedies of any nature whatsoever under or by reason of this *Section 5.14*. Nothing in this Agreement, express or implied, is or shall be construed to establish, amend or modify any Island Benefit Plan, Boat Benefit Plan or other employee or director compensation or benefit plan. The parties to this Agreement acknowledge and agree that the terms set forth in this *Section 5.14* shall not create any right in any employee of Island, Boat, Holdco or their respective Subsidiaries or any other Person to any continued employment with Island, Boat, the Surviving Corporation or their respective Subsidiaries or any right to compensation or benefits of any nature or kind whatsoever.
- 5.15 *Cooperation.* Each of Boat and Island will, and will cause each of its respective Representatives to, use its reasonable best efforts, subject to applicable Laws, to cooperate with the other party in connection with planning the integration of the business operations of Boat and Island.
 - 5.16 Financing Matters. Except as otherwise agreed by the parties to the Existing Credit Agreement:
 - (a) at least three (3) Business Days prior to the Closing Date, Boat shall deliver to Island an executed copy of a customary payoff letter from the agents under the Existing Credit Agreement in form and substance reasonably satisfactory to Island relating to the repayment in full of all obligations thereunder or secured thereby, the termination of all commitments in connection therewith and the release of all Liens securing the obligations thereunder (the "Payoff Letter");
 - (b) Boat shall, and shall cause its Subsidiaries to, deliver to Island (or the agent under the Existing Credit Agreement, in the case of prepayment and termination notices) prior to the Closing, in form and substance reasonably satisfactory to Island, all the documents, filings and notices required for the termination of commitments under the Existing Credit Agreement and the release of all Liens securing the obligations thereunder, including the filing of UCC releases, termination of control agreements, and delivery of possessory collateral, which shall in each case

Table of Contents

be subject to the occurrence of the Closing and the repayment in full of all obligations then outstanding under the Existing Credit Agreement; and

- (c) at the Closing, Holdco shall pay or shall cause to be paid, in full and in immediately available funds, any and all amounts outstanding and then due and payable under the Existing Credit Agreement in accordance with the Payoff Letter.
- 5.17 Stock Exchange Delisting and Deregistration. Boat and Island will cooperate and use their respective reasonable best efforts to cause the delisting of shares of Island Common Stock and Boat Common Stock from NASDAQ and the deregistration of such shares under the Exchange Act as promptly as practicable following the Closing in compliance with applicable Law.

ARTICLE 6 CONDITIONS TO CONSUMMATION OF THE MERGERS

- 6.1 Conditions to Obligations of Each Party Under This Agreement. The obligations of Boat, Island, Merger Sub A, Merger Sub B and Holdco to consummate the Mergers are subject to the satisfaction or waiver (to the extent permitted by applicable Law), at or prior to the Closing, of the following conditions:
 - (a) Stockholder Approvals. Island shall have obtained the Island Stockholder Approval, and Boat shall have obtained the Boat Stockholder Approval.
 - (b) NASDAQ Listing. The shares of Holdco Common Stock issuable to the stockholders of Island and Boat pursuant to this Agreement shall have been approved for listing on the NASDAQ, subject to official notice of issuance.
 - (c) Statutes and Injunctions. No Law or Order shall have been promulgated, entered, enforced, enacted or issued by any Governmental Entity (whether temporary, preliminary or permanent) which prohibits, restrains or makes illegal the consummation of the Mergers and shall continue in effect.
 - (d) Form S-4. The Form S-4 shall have become effective under the Securities Act and shall not be the subject of any stop order.
 - (e) *HSR Act; Other Competition Authority Approvals.* (i) Any applicable waiting period, together with any extensions thereof, under the HSR Act shall have expired or been terminated; and (ii) any other antitrust, competition, investment, trade regulation or similar approvals that are required under Law to have been obtained prior to Closing shall have been obtained.
- 6.2 Conditions to Obligations of Boat and Holdco Under This Agreement. The obligations of Boat and Holdco to consummate the Mergers are subject to the satisfaction or waiver (to the extent permitted by applicable Law), at or prior to the Closing, of the following conditions:
 - (a) The representations and warranties of Island set forth in this Agreement (except those representations and warranties set forth in the proviso below) shall be true and correct in all respects (without giving effect to any materiality or Island Material Adverse Effect qualifiers therein) as of the Closing Date as though made on and as of such date (or, in the case of representations and warranties that address matters only as of a particular date, as of such date), except to the extent that any failures of such representations and warranties to be so true and correct, individually or in the aggregate, have not had and would not reasonably be expected to have an Island Material Adverse Effect; provided, that (i) the representations and warranties set forth in Section 3.1 (Corporate Organization), Section 3.3 (Authority; Execution and Delivery; Enforceability), Section. 3.19 (Broker's Fees) and Section 3.20 (Opinion of Financial Advisor) shall be true and correct in all material respects (without giving effect to any materiality or Island Material Adverse Effect qualifiers therein) as of the Closing Date as though made on and as of

Table of Contents

such date (or, in the case of representations and warranties that address matters only as of a particular date, as of such date) (ii) the representations and warranties set forth in *Section 3.2* (Island Capitalization) shall be true and correct in all respects on the Closing Date (or, in the case of representations and warranties that address matters only as of a particular date, as of such date), except for inaccuracies that are *de minimis* and (iii) the representation and warranty set forth in clause (b) of the first sentence of *Section 3.6* (Absence of Certain Changes or Events) shall be true and correct in all respects as of the date of this Agreement and as of the Closing Date as though made on and as of such date. Boat shall have received a certificate validly executed and signed on behalf of Island by an executive officer of Island certifying that this condition has been satisfied.

- (b) Island shall have performed or complied with, as applicable, all of the covenants and agreements required by this Agreement to be performed or complied with by it in all material respects and Boat shall have received a certificate validly executed and signed on behalf of Island by an executive officer of Island certifying that this condition has been satisfied.
- (c) No change, event, development, condition, circumstance or effect shall have occurred since the date of this Agreement that has had, or would reasonably be expected to have, individually or in the aggregate, an Island Material Adverse Effect.
- (d) Boat shall have received a written tax opinion from Skadden, Arps, Slate, Meagher & Flom, LLP, counsel to Boat, dated as of the Closing Date, in form and substance reasonably satisfactory to Boat and based on the facts, representations, assumptions and exclusions set forth or described in such opinion, to the effect that (i) the Boat Merger will qualify as a "reorganization" within the meaning of Section 368(a) of the Code or, alternatively, (ii) the Mergers together will be treated as an "exchange" described in Section 351 of the Code (the "Boat Tax Opinion"). Such counsel shall be entitled to rely upon customary assumptions, representations, warranties, and covenants from each of Boat, Island and Holdco (or any other relevant parties), in each case, in form and substance reasonably satisfactory to such counsel, including tax representation letters in substantially the forms set forth in Exhibit D and Exhibit E. Each such representation letter shall be dated as of the date of such opinion.
- 6.3 Conditions to Obligations of Island Under This Agreement. The obligation of Island to consummate the Island Merger is subject to the satisfaction or waiver (to the extent permitted by applicable Law), at or prior to the Closing, of the following conditions:
 - (a) The representations and warranties of Boat set forth in this Agreement (except those representations and warranties set forth in the proviso below) shall be true and correct in all respects (without giving effect to any materiality or Boat Material Adverse Effect qualifiers therein) as of the Closing Date as though made on and as of such date (or, in the case of representations and warranties that address matters only as of a particular date, as of such date), except to the extent that any failures of such representations and warranties to be so true and correct, individually or in the aggregate, have not had, and would not reasonably be expected to have, a Boat Material Adverse Effect; *provided*, that (i) the representations and warranties set forth in *Section 4.1* (Corporate Organization), *Section 4.3* (Authority; Execution and Delivery; Enforceability), *Section 4.19* (Broker's Fees), *Section 4.20* (Opinion of Financial Advisor) and *Section 4.23* (Holdco, Merger Sub A and Merger Sub B) shall be true and correct in all material respects (without giving effect to any materiality or Boat Material Adverse Effect qualifiers therein) as of the Closing Date as though made on and as of such date (or, in the case of representations and warranties that address matters only as of a particular date, as of such date), (ii) the representations and warranties set forth in *Section 4.2* (Boat Capitalization) shall be true and correct in all respects on the Closing Date (or, in the case of representations and warranties that address matters only as of a particular date, as of such date), except for inaccuracies that are

Table of Contents

de minimis, and (iii) the representation and warranty set forth in clause (b) of the first sentence of Section 4.6 (Absence of Certain Changes or Events) shall be true and correct in all respects as of the date of this Agreement and as of the Closing Date as though made on and as of such date. Island shall have received a certificate validly executed and signed on behalf of Boat by an executive officer of Boat certifying that this condition has been satisfied.

- (b) Boat and Holdco shall have performed or complied with, as applicable, all of the covenants and agreements required by this Agreement to be performed or complied with by it in all material respects and Island shall have received a certificate validly executed and signed on behalf of Boat by an executive officer of Boat certifying that this condition has been satisfied.
- (c) No change, event, development, condition, circumstance or effect shall have occurred since the date of this Agreement that has had, or would reasonably be expected to have, individually or in the aggregate, a Boat Material Adverse Effect.
- (d) Island shall have received a written tax opinion from Latham & Watkins LLP, counsel to Island, dated as of the Closing Date, in form and substance reasonably satisfactory to Island and based on the facts, representations, assumptions and exclusions set forth or described in such opinion, to the effect that the Mergers together will be treated as an "exchange" described in Section 351 of the Code (the "Island Tax Opinion"). Such counsel shall be entitled to rely upon customary assumptions, representations, warranties, and covenants from each of Boat, Island and Holdco (or any other relevant parties), in each case, in form and substance reasonably satisfactory to such counsel, including tax representation letters in substantially the forms set forth in Exhibit D and Exhibit E. Each such representation letter shall be dated as of the date of such opinion.

ARTICLE 7 TERMINATION, AMENDMENT AND WAIVER

- 7.1 *Termination.* This Agreement may be terminated, and the Mergers contemplated hereby may be abandoned by action taken or authorized by the Board of Directors of the terminating party or parties, whether before or after the Boat Stockholder Approval or the Island Stockholder Approval:
 - (a) By mutual written consent of Boat and Island;
 - (b) By either Boat or Island:
 - (i) if any Law or final and non-appealable Order shall have been promulgated, entered, enforced, enacted or issued or shall be deemed to be applicable to the Mergers by any Governmental Entity of competent jurisdiction which permanently prohibits, restrains or makes illegal the consummation of the Mergers; provided, that the right to terminate this Agreement pursuant to this Section 7.1(b)(i) shall not be available to any party whose failure to perform any of its obligations under this Agreement is the principal cause of the enactment or issuance of any such Law or Order;
 - (ii) if the Transactions shall not have been consummated by July 21, 2018 provided, that in the event that, as of July 21, 2018, all conditions to Closing set forth in Article 6 have been satisfied or waived (other than such conditions that by their terms are satisfied at the Closing, which shall be reasonably capable of being satisfied as of such date) other than the conditions set forth in Section 6.1(c) (solely with respect to the matters addressed in Section 6.1(e)) or Section 6.1(e), the termination date shall be automatically extended to October 21, 2018 (such date, including any such permitted extensions thereof, the "Outside Date"), provided, further, that the right to terminate this Agreement pursuant to this Section 7.1(b)(ii) shall not be available to any party whose failure to perform any of its obligations under this Agreement is the principal cause of the failure of the Transactions to be consummated by such time;

Table of Contents

- (iii) if the Boat Stockholder Approval shall not have been obtained upon a vote taken thereon at the Boat Stockholders Meeting duly convened therefor or at any adjournment or postponement thereof; *provided* that the right to terminate this Agreement pursuant to this *Section 7.1(b)(iii)* shall not be available to Boat if Boat's failure to perform any of its obligations under this Agreement is the principal cause of the failure to obtain the Boat Stockholder Approval; or
- (iv) if the Island Stockholder Approval shall not have been obtained upon a vote taken thereon at the Island Stockholders Meeting duly convened therefor or at any adjournment or postponement thereof; provided that the right to terminate this Agreement pursuant to this Section 7.1(b)(iv) shall not be available to Island if Island's failure to perform any of its obligations under this Agreement is the principal cause of the failure to obtain the Island Stockholder Approval; or

(c) By Island:

- (i) if Boat shall have breached or failed to perform any of its representations, warranties or covenants contained in this Agreement, which breach or failure to perform (A) is incapable of being cured by Boat prior to the Outside Date or otherwise is not cured by the earlier of (x) twenty (20) Business Days following written notice to Boat by Island of such breach or (y) the Business Day prior to the Outside Date and (B) would result in a failure of any condition set forth in *Section 6.3(a)* or *Section 6.3(b)*; or
 - (ii) if the Boat Board has effected a Boat Adverse Recommendation Change; or

(d) By Boat:

- (i) if Island shall have breached or failed to perform any of its representations, warranties or covenants contained in this Agreement, which breach or failure to perform (A) is incapable of being cured by Island prior to the Outside Date or otherwise is not cured by the earlier of (x) twenty (20) Business Days following written notice to Island by Boat of such breach or (y) the Business Day prior to the Outside Date and (B) would result in a failure of any condition set forth in Section 6.2(a) or Section 6.2(b); or
 - (ii) if the Island Board has effected an Island Adverse Recommendation Change.
- 7.2 Effect of Termination. In the event of the valid termination of this Agreement by either Boat or Island as provided in Section 7.1, written notice thereof shall forthwith be given by the terminating party to the other party specifying the provision hereof pursuant to which such termination is made. In the event of the valid termination of this Agreement pursuant to Section 7.1, this Agreement shall be terminated and this Agreement shall forthwith become void and have no effect, without any liability or obligation on the part of Boat, Island, Merger Sub A, Merger Sub B or Holdco, other than this Section 7.2, Section 7.3 and Article 8, which provisions shall survive such termination; provided, that nothing herein shall relieve any party for any liability for any Willful Breach or fraud. No termination of this Agreement shall affect the obligations of the parties contained in the Confidentiality Agreement, all of which obligations shall survive the termination of this Agreement in accordance with their terms.

7.3 Termination Fee; Expenses.

(a) Except as otherwise provided in this *Section 7.3* and except for (i) the expenses in connection with printing and mailing the Joint Proxy Statement and the Form S-4 required in connection with the actions specified in *Section 5.3*, (ii) all SEC filing fees relating to the Transactions and (iii) the fees in connection with the approvals required under *Section 6.1(e)* related to the Transactions (each of which fees and expenses shall be borne, in each case, equally by Boat and Island), all fees and expenses incurred by the parties shall be borne solely by the party that has incurred such fees and expenses.

Table of Contents

- (b) Boat shall pay to Island \$25,000,000 (the "Boat Termination Fee") if this Agreement is terminated as follows:
 - (i) if this Agreement is terminated by Island pursuant to Section 7.1(c)(ii), then Boat shall pay the Boat Termination Fee (to the extent not previously paid) on the second Business Day following such termination; and
 - (ii) (x) if this Agreement is terminated (A) pursuant to *Section 7.1(b)(iii)*, (B) pursuant to *Section 7.1(c)(i)* or (C) pursuant to *Section 7.1(b)(ii)* without the Boat Stockholders Meeting having occurred, and (y) in any such case a Competing Proposal shall have been publicly announced or otherwise communicated to the Boat Board (and not withdrawn) after the date of this Agreement and prior to the date of the Boat Stockholders Meeting, in the case of clause (A), or the date of termination, in the case of clauses (B) and (C), and (z) if within twelve (12) months after the date of such termination, a transaction in respect of a Competing Proposal is consummated or Boat enters into a definitive agreement in respect of a Competing Proposal that is ultimately consummated, then Boat shall pay the Boat Termination Fee (less any Expenses previously paid to Island pursuant to *Section 7.3(c)*) on the second Business Day following the date Boat consummates such transaction; *provided*, that, solely for purposes of this *Section 7.3(b)(ii)*, the term "Competing Proposal" shall have the meaning ascribed thereto in *Section 5.4(f)(i)*, except that all references to 20% shall be changed to 50%.

Any Boat Termination Fee due under this Section 7.3(b) shall be paid by wire transfer of immediately available funds.

- (c) Boat shall pay to Island a fixed expense reimbursement amount of \$6,000,000, if this Agreement is terminated (i) pursuant to Section 7.1(b)(iii) (except if the Island Stockholder Approval had not been obtained upon a vote taken thereon at the Island Stockholders Meeting duly convened therefor or at any adjournment or postponement thereof) or (ii) pursuant to Section 7.1(c)(i). Any Expenses of Island due under this Section 7.3(c) shall be paid by wire transfer of immediately available funds no later than two (2) Business Days after Boat's receipt from Island of an itemized statement identifying such Expenses.
 - (d) Island shall pay to Boat \$25,000,000 (the "Island Termination Fee") if this Agreement is terminated as follows:
 - (i) if this Agreement is terminated by Boat pursuant to Section 7.1(d)(ii), then Island shall pay the Island Termination Fee (to the extent not previously paid) on the second Business Day following such termination; and
 - (ii) (x) if this Agreement is terminated (A) pursuant to Section 7.1(b)(iv), (B) pursuant to Section 7.1(d)(i) or (C) pursuant to Section 7.1(b)(ii) without the Island Stockholders Meeting having occurred, and (y) in any such case a Competing Proposal shall have been publicly announced or otherwise communicated to the Island Board (and not withdrawn) after the date of this Agreement and prior to the date of the Island Stockholders Meeting, in the case of clause (A), or the date of termination, in the case of clauses (B) or (C), and (z) if within twelve (12) months after the date of such termination, a transaction in respect of a Competing Proposal is consummated or Boat enters into a definitive agreement in respect of a Competing Proposal that is ultimately consummated, then Island shall pay the Island Termination Fee (less any Expenses previously paid to Boat pursuant to Section 7.3(e)) on the second Business Day following the date Island consummates such transaction; provided, that, solely for purposes of this Section 7.3(d)(ii), the term "Competing Proposal" shall have the meaning ascribed thereto in Section 5.4(f)(i), except that all references to 20% shall be changed to 50%.

Table of Contents

Any Island Termination Fee due under this Section 7.3(d) shall be paid by wire transfer of immediately available funds.

- (e) Island shall pay to Boat a fixed expense reimbursement of \$6,000,000, if this Agreement is terminated (i) pursuant to Section 7.1(b)(iv) (except if the Boat Stockholder Approval had not been obtained upon a vote taken thereon at the Boat Stockholders Meeting duly convened therefor or at any adjournment or postponement thereof) or (ii) pursuant to Section 7.1(d)(i). Any Expenses of Boat due under this Section 7.3(e) shall be paid by wire transfer of immediately available funds no later than two (2) Business Days after Island's receipt from Boat of an itemized statement identifying such Expenses.
- (f) The parties each agree that the agreements contained in this Section 7.3 are an integral part of the Transactions, and that, without these agreements, the parties would not enter into this Agreement. Accordingly, if a party fails to pay any amounts due under this Section 7.3 and, in order to obtain such payment, Island or Boat, as the case may be, commences a suit that results in a judgment against such party for such amounts, such party shall pay interest on such amounts from the date payment of such amounts were due to the date of actual payment at the rate equal to the prime rate published in the Wall Street Journal for the relevant period, together with the costs and expenses of the other party (including reasonable legal fees and expenses) in connection with such suit. Notwithstanding anything to the contrary in this Agreement, in the event that the Boat Termination Fee or the Island Termination Fee is payable and actually paid to Boat or Island in accordance with this Section 7.3, payment of such Boat Termination Fee, as applicable, shall be the sole and exclusive remedy of the non-terminating party and its affiliates against any other party or such other party's stockholders, directors, officers, affiliates and other Representatives, for any loss or damage based upon, arising out of or relating to this Agreement or the negotiation, execution or performance hereof or the Transactions contemplated hereby, except in the case of Willful Breach or fraud. Solely for purposes of establishing the basis for the amount thereof, and without in any way increasing the amount of the Boat Termination Fee or the Island Termination Fee, expanding the circumstances in which the Boat Termination Fee or the Island Termination Fee, as applicable, is to be paid or restricting or modifying the other rights of any party hereunder, in the event of the valid termination of this Agreement under circumstances in which the Boat Termination Fee or the Island Termination Fee is payable pursuant to this Section 7.3, it is agreed that each of the Boat Termination Fee and the Island Termination Fee is liquidated damages, and not a penalty, and the payment thereof in such circumstances is supported by due and sufficient consideration; provided, that no payment of a Boat Termination Fee or an Island Termination Fee shall be considered in lieu of, or a replacement or substitution for, damages incurred in the event of Willful Breach or fraud.
- 7.4 Amendment or Supplement. This Agreement may be amended, modified or supplemented by the parties by action taken or authorized by their respective Boards of Directors at any time prior to the Effective Time, whether before or after the Boat Stockholder Approval or the Island Stockholder Approval has been obtained; provided, that after the Boat Stockholder Approval or the Island Stockholder Approval has been obtained, no amendment shall be made that pursuant to applicable Law requires further approval or adoption by the stockholders of Boat or stockholders of Island, as applicable, without such further approval or adoption. This Agreement may not be amended, modified or supplemented in any manner, whether by course of conduct or otherwise, except by an instrument in writing specifically designated as an amendment hereto, signed on behalf of each of the parties in interest at the time of the amendment.
- 7.5 Extension of Time; Waiver. At any time prior to the Effective Time, the parties may, by action taken or authorized by their respective Boards of Directors, to the extent permitted by applicable Law, (a) extend the time for the performance of any of the obligations or acts of the other parties, (b) waive any inaccuracies in the representations and warranties of the other parties set forth in

Table of Contents

this Agreement or any document delivered pursuant hereto, or (c) waive compliance with any of the agreements or conditions of the other parties contained herein; *provided*, that after the Boat Stockholder Approval or the Island Stockholder Approval, as applicable, has been obtained, no waiver may be made that pursuant to applicable Law requires further approval or adoption by the stockholders of Boat or stockholders of Island, as applicable, without such further approval or adoption. Any agreement on the part of a party to any such waiver shall be valid only if set forth in a written instrument executed and delivered by a duly authorized officer on behalf of such party. No failure or delay of any party in exercising any right or remedy hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such right or power, or any abandonment or discontinuance of steps to enforce such right or power, or any course of conduct, preclude any other or further exercise thereof or the exercise of any other right or power. The rights and remedies of the parties hereunder are cumulative and are not exclusive of any rights or remedies which they would otherwise have hereunder.

ARTICLE 8 GENERAL PROVISIONS

- 8.1 *Non-Survival of Representations and Warranties*. None of the representations, warranties, covenants or agreements in this Agreement or in any instrument delivered pursuant to this Agreement shall survive the Effective Time, other than those covenants or agreements of the parties which by their terms apply, or are to be performed in whole or in part, after the Effective Time.
- 8.2 *Notices*. Any notices or other communications required or permitted under, or otherwise given in connection with, this Agreement shall be in writing and shall be deemed to have been duly given (a) when delivered or sent if delivered in person or sent by facsimile transmission (provided confirmation of facsimile transmission is obtained), (b) on the next Business Day if transmitted by national overnight courier or (c) on the date delivered if sent by email (provided confirmation of email receipt is obtained), in each case as follows:

If to Island, addressed to it at:

Idera Pharmaceuticals, Inc. 167 Sidney Street Cambridge, Massachusetts 02139 Attention: Vincent J. Milano

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 885 Third Avenue New York, New York 10022-4834 Tel: (212) 906-1200 Fax: (212) 751-4864 Attention: Charles K. Ruck R. Scott Shean

Email: charles.ruck@lw.com scott.shean@lw.com

Table of Contents

If to Boat, Merger Sub A, Merger Sub B or Holdco, addressed to it at:

BioCryst Pharmaceuticals, Inc.
4505 Emperor Blvd., Suite 200
Durham, North Carolina 27703
Attention: Jon Stonehouse
Alane Barnes
Email: jstonehouse@biocryst.com
abarnes@biocryst.com

with a copy (which shall not constitute notice) to:

Skadden, Arps, Slate, Meagher & Flom LLP Four Times Square New York, NY 10036-6522 Tel: (212) 735-3000 Fax: (212) 735-2000 Attention: Stephen F. Arcano

Ann Beth Stebbins
Email: stephen.arcano@skadden.com
annbeth.stebbins@skadden.com

8.3 Certain Definitions. For purposes of this Agreement, the term:

"Acceptable Confidentiality Agreement" means a confidentiality agreement that contains confidentiality and standstill provisions that are no less favorable in the aggregate to Boat or Island, as applicable, than those contained in the Confidentiality Agreement; provided that such confidentiality agreement may not include any provision calling for an exclusive right to negotiate with any party to this Agreement or otherwise conflicting with the obligations of any party under this Agreement.

"affiliate" means a Person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, the first-mentioned Person.

"Anti-corruption Laws" means Laws relating to anti-bribery or anti-corruption (governmental or commercial) which apply to Island, Boat or their respective Subsidiaries, including Laws that prohibit the corrupt payment, offer, promise or authorization of the payment or transfer of anything of value (including gifts or entertainment), directly or indirectly, to any foreign Government Official or other Person to obtain a business advantage, including the FCPA, the U.K. Bribery Act of 2010 and all national and international Laws enacted to implement the OECD Convention on Combating Bribery of Foreign Officials in International Business Transactions.

"associate" has the meaning set forth in Section 203 of the DGCL.

"beneficial ownership" (and related terms such as "beneficially owned" or "beneficial owner") has the meaning set forth in Rule 13d-3 under the Exchange Act.

"Boat Disclosure Schedule" means the disclosure schedule delivered by Boat to Island prior to the execution of this Agreement.

"Boat Leased Real Property" means all real property leased, subleased, or otherwise occupied pursuant to an occupancy agreement by Boat or any of its Subsidiaries.

Table of Contents

"Boat Material Adverse Effect" means any change, event, development, condition, occurrence or effect that is, or would reasonably be expected to be, materially adverse to the business, condition (financial or otherwise), assets, liabilities or results of operations of Boat and its Subsidiaries, taken as a whole; provided, however, that none of the following shall be deemed in themselves, either alone or in combination, to constitute, and none of the following shall be taken into account in determining whether there has been, a Boat Material Adverse Effect: (i) any changes resulting from general market, economic, financial, capital markets or political or regulatory conditions, (ii) any changes or proposed changes of Law or GAAP, (iii) any changes resulting from any act of terrorism, war, national or international calamity, or any worsening thereof, (iv) any changes generally affecting the industries in which Boat and its Subsidiaries conduct their businesses, (v) any changes resulting from the execution of this Agreement or the announcement or the pendency of the Mergers, including any loss of employees or any disruption in or termination of (or loss of or other negative effect or change with respect to) supplier, distributor or similar business relationships or partnerships resulting from the Transactions (provided, that this clause (v) does not apply in the context of the representations and warranties set forth in Section 4.4), (vi) changes in Boat's stock price or the trading volume of Boat's stock or any change in the credit rating of Boat (but not, in each case, the underlying cause of any such changes, unless such underlying cause would otherwise be excepted from this definition), (vii) any changes or effects resulting from any action required to be taken by the terms of this Agreement, (viii) the failure to meet internal or analysts' expectations, projections or results of operations (but not, in each case, the underlying cause of any such changes, unless such underlying cause would otherwise be excepted from this definition), or (ix) any Proceeding arising from or relating to the Mergers or the other Transactions; provided, that in the case of clauses (i), (ii), (iii) and (iv), if and only to the extent such changes do not have a disproportionate impact on Boat and its Subsidiaries, taken as a whole, as compared to other participants in the industries in which Boat and its Subsidiaries conduct their businesses.

"Boat Registered Intellectual Property" means the Solely Owned Boat Registered Intellectual Property and the Licensed Boat Registered Intellectual Property.

"Boat Stock Plan" means the Boat Stock Incentive Plan.

"Business Day" means any date except Saturday or Sunday on which commercial banks are not required or authorized to close in New York, New York, United States.

"Closing Bid Price" means for any security for each Trading Day shall be the reported per share closing bid price of such security regular way on the NASDAQ on such Trading Day, or, if there were no transactions on such Trading Day, the average of the reported closing bid and asked prices, regular way, of such security on the NASDAQ on such Trading Day.

"Code" means the Internal Revenue Code of 1986, as amended.

"Contract" means any agreements, arrangements, commitments, understandings, contracts, leases (whether for real or personal property), powers of attorney, notes, bonds, mortgages, indentures, deeds of trust, loans, evidences of indebtedness, purchase orders, letters of credit, settlement agreements, franchise agreements, undertakings, covenants not to compete, employment agreements, licenses, instruments, obligations, commitments, understandings, policies, purchase and sales orders, quotations and other commitments to which a Person is a party or to which any of the assets of such Person or its Subsidiaries are subject, whether oral or written, express or implied, including all amendments or modifications thereto.

"control" (including the terms "controlled by" and "under common control with") means the possession, directly or indirectly or as trustee or executor, of the power to direct or cause the direction of the management or policies of a Person, whether through the ownership of stock or shares or as trustee or executor, by Contract or otherwise.

Table of Contents

"DGCL" means the General Corporation Law of the State of Delaware, as amended.

"Disclosure Schedules" means the Boat Disclosure Schedule and the Island Disclosure Schedule.

"Environmental Claim" means any Proceeding, investigation, order, demand, allegation, accusation or notice (written or oral) by any Person or entity alleging actual or potential liability (including actual or potential liability for investigatory costs, cleanup costs, response costs, natural resources damages, property damages, personal injuries, attorneys' fees or penalties) arising out of, based on, resulting from or relating to any Environmental Laws, Environmental Permits or the presence, or Release into the environment, of, or exposure to, any Hazardous Materials at any location, but shall not include any claims relating to products liability.

"Environmental Laws" means any and all applicable, federal, state, provincial, local or foreign Laws, statutes, ordinances, regulations, treaties, policies, guidance, rules, judgments, orders, writs, court decisions or rule of common law, stipulations, injunctions, consent decrees, permits, restrictions and licenses, regulating or relating to Hazardous Materials, pollution, protection of the environment (including ambient air, surface water, ground water, land surface, subsurface strata, wildlife, plants or other natural resources), and/or the protection of health and safety of persons from exposures to Hazardous Materials in the environment.

"Environmental Permits" means any permit, certificate, approval, identification number, license or other authorization required under, or issued pursuant to, an Environmental Law.

"Equity Interest" means any share, capital stock, partnership, member, membership or similar interest in any Person, and any option, warrant, right or security (including debt securities) convertible, exchangeable or exercisable thereto or therefor.

"ERISA" means the United States Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means any Person (whether or not incorporated) that, together with Boat or Island, as applicable, is considered under common control and treated as one employer under Section 414(b), (c), (m) or (o) of the Code.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Existing Credit Facility" means the that certain Credit and Security Agreement, dated September 23, 2016, by and among MidCap Financial Trust, the Lenders party thereto and Boat.

"Expenses" means, with respect to any Person, all reasonable and documented out-of-pocket fees and expenses (including all fees and expenses of counsel, accountants, financial advisors and investment bankers of such Person and its affiliates), incurred by such Person or on its behalf in connection with or related to the authorization, preparation, negotiation, execution and performance of this Agreement and the Transactions, any litigation with respect thereto, the preparation, printing, filing and mailing of the Joint Proxy Statement or Form S-4, the filing of any required notices under the HSR Act or foreign antitrust, competition, investment, trade regulation or similar Laws, or in connection with other regulatory approvals, and all other matters related to the Mergers or the other Transactions.

"GAAP" means generally accepted accounting principles as applied in the United States.

"Government Official" means (i) any official, officer, employee, representative or any person acting in an official capacity for or on behalf of any Governmental Entity; (ii) any political party or party official or candidate for political office; (iii) any public international organization or any department or agency thereof; or (iv) any Person or other entity owned in whole or in part, or controlled by any Person described in the foregoing clauses (i), (ii) or (iii) of this definition.

Table of Contents

"Governmental Entity" means any transnational, national, federal, state, county, municipal, local or foreign government, or other political subdivision thereof, any entity exercising executive, legislative, judicial, regulatory, taxing or administrative functions of or pertaining to government, and any arbitrator or arbitral body or panel of competent jurisdiction.

"group" has the meaning ascribed to in the Exchange Act, except where the context otherwise requires.

"Hazardous Materials" means any pollutants, chemicals, contaminants or wastes and any other toxic, infectious, carcinogenic, reactive, corrosive, ignitable, flammable or otherwise hazardous substance, whether solid, liquid or gas, that is subject to regulation, control or remediation under any Environmental Laws, including any quantity of asbestos in any form, urea formaldehyde, PCBs, radon gas, crude oil or any fraction thereof, all forms of natural gas, petroleum products or by-products or derivatives.

"Health Care Laws" means: (i) the Federal Food, Drug, and Cosmetic Act (21 U.S.C. § 301 et seq.); (ii) the Controlled Substances Act (21 U.S.C. § 801 et seq.); (iii) the Public Health Service Act (42 U.S.C. § 201 et seq.); (iv) all federal, state, local and all applicable foreign health care related fraud and abuse, false claims, and anti-kickback laws, including, without limitation, the U.S. Anti-Kickback Statute (42 U.S.C. § 1320a-7b(b)), the U.S. Physician Payment Sunshine Act (42 U.S.C. § 1320a-7h) and similar gift and disclosure Laws, the U.S. Civil False Claims Act (31 U.S.C. § 3729 et seq.), the criminal False Claims Law (42 U.S.C. § 1320a-7b(a)), all criminal laws relating to health care fraud and abuse, including but not limited to 18 U.S.C. § 286 and 287, and the health care fraud criminal provisions under the U.S. Health Insurance Portability and Accountability Act of 1996 ("HIPAA") (42 U.S.C. § 1320d et seq.), the exclusion laws (42 U.S.C. § 1320a-7), the civil monetary penalties law (42 U.S.C. § 1320a-7a), and Laws relating to price reporting requirements and the requirements relating to the processing of any applicable rebate, chargeback or adjustment, under applicable rules and regulations relating to the Medicaid Drug Rebate Program (42 U.S.C. § 1396r-8), any state supplemental rebate program, and Medicare average sales price reporting (42 U.S.C. § 1395w-3a); (v) state Laws relating to the manufacture, sale and distribution of pharmaceutical and medical products; (vi) Medicare (Title XVIII of the Social Security Act); (vii) Medicaid (Title XIX of the Social Security Act); and (viii) in each case, the regulations promulgated thereunder.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations thereunder.

"Intellectual Property" means all domestic and foreign intellectual property rights, including all (a) inventions (whether or not patentable and whether or not reduced to practice), all improvements thereto, and all patents, patent applications, and patent disclosures, together with all provisionals, reissues, continuations, continuations-in-part, divisions, revisions, extensions, and reexaminations thereof, (b) trademarks, service marks, trade names, trade dress, logos, corporate names, brand names and other source indicators, together with all translations, adaptations, derivations, and combinations thereof, and all applications, registrations, and renewals in connection therewith, (c) domain names, uniform resource locators and other names and locators associated with the Internet, and all registrations in connection therewith, (d) works of authorship (whether or not published), and all copyrights, designs and mask works, and all registrations, applications and renewals in connection therewith, (e) software and all website content (including text, graphics, images, audio, video and data) and (f) trade secrets, confidential business information, and other proprietary information (including ideas, know-how, formulas, compositions, processes and techniques, research and development information, data, designs, drawings, specifications, research records, records of inventions, test information, financial, marketing and business data, pricing and cost information, business and marketing plans and proposals and customer and supplier lists and information).

"IRS" means the United States Internal Revenue Service.

Table of Contents

"Island Disclosure Schedule" means the disclosure schedule delivered by Island to Boat prior to the execution of this Agreement.

"Island Leased Real Property" means all real property leased, subleased, or otherwise occupied pursuant to an occupancy agreement by Island.

"Island Material Adverse Effect" means any change, event, development, condition, occurrence or effect that is, or would reasonably be expected to be, materially adverse to the business, condition (financial or otherwise), assets, liabilities or results of operations of Island, taken as a whole; provided, however, that none of the following shall be deemed in themselves, either alone or in combination, to constitute, and none of the following shall be taken into account in determining whether there has been, an Island Material Adverse Effect: (i) any changes resulting from general market, economic, financial, capital markets or political or regulatory conditions, (ii) any changes or proposed changes of Law or GAAP, (iii) any changes resulting from any act of terrorism, war, national or international calamity, or any worsening thereof, (iv) any changes generally affecting the industries in which Island conducts its business, (v) any changes resulting from the execution of this Agreement or the announcement or the pendency of the Mergers, including any loss of employees or any disruption in or termination of (or loss of or other negative effect or change with respect to) supplier, distributor or similar business relationships or partnerships resulting from the Transactions (provided, that this clause (v) does not apply in the context of the representations and warranties set forth in Section 3.4), (vi) changes in Island's stock price or the trading volume of Island's stock or any change in the credit rating of Island (but not, in each case, the underlying cause of any such changes, unless such underlying cause would otherwise be excepted from this definition), (vii) any changes or effects resulting from any action required to be taken by the terms of this Agreement, (viii) the failure to meet internal or analysts' expectations, projections or results of operations (but not, in each case, the underlying cause of any such changes, unless such underlying cause would otherwise be excepted from this definition), or (ix) any Proceeding arising from or relating to the Mergers or the other Transactions; provided, that in the case of clauses (i), (ii), (iii) and (iv), if and only to the extent such changes do not have a disproportionate impact on Island as compared to other participants in the industries in which Island conducts its business.

"Island Registered Intellectual Property" means the Solely Owned Island Registered Intellectual Property and the Licensed Island Registered Intellectual Property.

"Island Stock Plans" means the Island 1995 Director Stock Option Plan, the 1997 Stock Incentive Plan, the Island 2005 Stock Incentive Plan, the 2008 Stock Incentive Plan and the 2013 Stock Incentive Plan.

"IT Assets" means computers, software, servers, workstations, routers, hubs, switches, circuits, networks, data communications lines and all other information technology equipment.

"Joint Proxy Statement" means a proxy statement or similar disclosure document or circular relating to the Boat Stockholders Meeting and the Island Stockholders Meeting.

"*Knowledge*" and "*known*" means the actual knowledge of the officers of Island set forth in *Section 8.3(a)* of the Island Disclosure Schedule or of the officers of Boat set forth in *Section 8.3(a)* of the Boat Disclosure Schedule, as the case may be.

"Law" means any federal, state, provincial, municipal, local or foreign law, statute, code, ordinance, rule, regulation, circular, order, judgment, writ, stipulation, award, injunction, decree or arbitration award or finding.

"Lien" means any lien, mortgage, pledge, conditional or installment sale agreement, encumbrance, defect in title, covenant, condition, restriction, charge, option, right of first refusal, easement, security interest, deed of trust, right-of-way, encroachment, community property interest or other claim or

Table of Contents

restriction of any nature, whether voluntarily incurred or arising by operation of Law (including any restriction on the voting of any security, any restriction on the transfer of any security or other asset, and any restriction on the possession, exercise or transfer of any other attribute of ownership of any asset).

"Market Price" means the average Closing Bid Price for twenty (20) consecutive Trading Days, ending with the Trading Day prior to the date as of which the Market Price is being determined (with appropriate adjustments for subdivisions or combinations of shares effected during such period), provided that if the prices referred to in the definition of Closing Bid Price cannot be determined on any Trading Day, the Closing Bid Price for such Trading Day will be deemed equal to Fair Market Value of such security on such Trading Day.

"Merger Consideration Value" means the product obtained by multiplying (i) the Common Stock Merger Consideration by (ii) the VWAP of Boat Common Stock.

"NASDAQ" means the NASDAQ Global Select Market.

"Net Boat Share" means, with respect to a Boat Stock Option, a number of whole and partial shares of Boat Common Stock (computed to the nearest five decimal places) equal to the quotient obtained by dividing (a) the product of (i) the number of shares of Boat Common Stock subject to such Boat Stock Option immediately prior to the Effective Time and (ii) the excess, if any, of the VWAP of Boat Common Stock over the exercise price per share of Boat Common Stock subject to such Boat Stock Option, by (b) the VWAP of Boat Common Stock.

"Net Island Share" means, with respect to an Island Stock Option, a number of whole and partial shares of Island Common Stock (computed to the nearest five decimal places) equal to the quotient obtained by dividing (a) the product of (i) the number of shares of Island Common Stock subject to such Island Stock Option immediately prior to the Effective Time, and (ii) the excess, if any, of the Island Merger Consideration Value over the exercise price per share of Island Common Stock subject to such Island Stock Option, by (b) the Island Merger Consideration Value.

"Order" means any order, writ, injunction, decree, circular, judgment, award, injunction, settlement or stipulation issued, promulgated, made, rendered, entered into or enforced by or with any Governmental Entity (in each case, whether temporary, preliminary or permanent).

"Other Covered Party" means any political party or party official, or any candidate for political office.

"Permitted Liens" means (i) Liens for current Taxes, or governmental assessments, charges or claims of payment not yet past due or the amount or validity of which is being contested in good faith by appropriate Proceedings and for which adequate reserves in accordance with GAAP have been established in the latest financial statements of Boat included in the Boat SEC Documents or Island included in the Island SEC Documents, as the case may be, (ii) mechanics', workmen's, repairmen's, warehousemen's and carriers' Liens arising in the ordinary course of business consistent with past practice for sums not yet due and payable, (iii) any such matters of record, Liens and other imperfections of title that do not and would not reasonably be expected to, individually or in the aggregate, materially impair the continued ownership, use and operation of the assets to which they relate and (iv) restrictions on transfers under applicable securities Laws.

"Person" means an individual, corporation, limited company, limited liability company, partnership, association, trust, unincorporated organization, Governmental Entity, other entity or group (as defined in Section 13(d) of the Exchange Act).

"Preferred Stock Liquidation Preference" means an amount of Holdco Common Stock equal to (a) (i) \$1.00 plus any accrued but unpaid dividends, divided by (ii) the Market Price of Island Common Stock as of the third day prior to the Closing Date multiplied by (b) the Island Exchange Ratio.

Table of Contents

"Proceeding" means any suit, action, proceeding, arbitration, mediation, audit, hearing, inquiry or, to the Knowledge of the Person in question, investigation (in each case, whether civil, criminal, administrative, investigative, formal or informal) commenced, brought, conducted or heard by or before, or otherwise involving, any Governmental Entity.

"Release" means disposing, discharging, injecting, spilling, leaking, pumping, pouring, leaching, dumping, emitting, escaping or emptying into or upon the indoor or outdoor environmental, including any soil, sediment, subsurface strata, surface water, groundwater, ambient air, the atmosphere or any other media.

"Representatives" means, with respect to any Person, such Person's officers, directors, employees, accountants, legal counsel, financial advisors, consultants, financing sources and other advisors and representatives.

"SEC" means the Securities and Exchange Commission.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"Subsidiary" means, with respect to any Person, any corporation, partnership, joint venture or other legal entity of which such Person (either alone or through or together with any other Subsidiary), owns, directly or indirectly, a majority of the stock or other Equity Interests, the holders of which are generally entitled to vote for the election of the board of directors or other governing body of such corporation, partnership, joint venture or other legal entity, or any Person that would otherwise be deemed a "subsidiary" under Rule 12b-2 promulgated under the Exchange Act.

"*Tax Return*" means any report, return (including information return), claim for refund, election, estimated tax filing, declaration or similar filing supplied or required to be supplied to any Governmental Entity with respect to Taxes, including any election, notification, appendix schedule or attachment thereto, and including any amendments thereof.

"Taxes" means any and all domestic or foreign, federal, state, local or other taxes, of any kind (together with any and all interest, penalties, additions to tax and additional amounts imposed with respect thereto) imposed by any Governmental Entity, including taxes on or with respect to income, franchise, windfall or other profits, gross receipts, occupation, severance, alternative minimum, disability, estimated, property, escheat or unclaimed property, sales, use, net worth, capital stock, payroll, employment, social security, workers' compensation, unemployment compensation, and taxes in the nature of excise, withholding, ad valorem, stamp, transfer, value-added, gains tax and license, registration and documentation fees, and other taxes, fees, levies, duties, tariffs, imposts, assessments, obligations and charges of the same or a similar nature to any of the foregoing.

"Taxing Authority" means any Governmental Entity responsible for the administration or the imposition of any Tax.

"Trading Day" means a day on which the NASDAQ is open for the transaction of business.

"VWAP of Boat Common Stock" means the volume weighted average price of a share of Boat Common Stock for a ten (10) Trading Day period, starting with the opening of trading on the eleventh (11th) Trading Day prior to the Closing Date to the closing of trading on the second to last Trading Day prior to the Closing Date, as reported by Bloomberg.

"Willful Breach" means a material breach of this Agreement that is the consequence of an act or omission by a party with the actual knowledge that the taking of such act or failure to take such action would be a material breach of this Agreement.

8.4 *Headings*. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.

Table of Contents

- 8.5 Severability. If any term or other provision (or part thereof) of this Agreement is invalid, illegal or incapable of being enforced by any rule of Law or public policy, all other conditions and provisions (or part thereof) of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the Transactions is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision (or part thereof) is invalid, illegal or incapable of being enforced, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner to the end that the Transactions are fulfilled to the extent possible.
- 8.6 Entire Agreement. This Agreement (together with the Exhibits, the Boat Disclosure Schedules, the Island Disclosure Schedules and the other documents delivered pursuant hereto) and the Confidentiality Agreement constitute the entire agreement of the parties and supersede all prior agreements and undertakings, both written and oral, among the parties, or any of them, with respect to the subject matter hereof.
- 8.7 Assignment. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any of the parties hereto, in whole or in part (whether by operation of law or otherwise), without the prior written consent of the other parties, and any attempt to make any such assignment without such consent shall be null and void. Subject to the preceding sentence, this Agreement will be binding upon, inure to the benefit of and be enforceable by the parties and their respective successors and assigns.
- 8.8 *Parties in Interest.* This Agreement shall be binding upon and inure solely to the benefit of the parties and their respective successors and assigns, and nothing in this Agreement, express or implied, other than *Section 5.9* (of which each Indemnitee and each of his or her heirs shall be an express third party beneficiary), is intended to or shall confer upon any other Person any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- 8.9 Mutual Drafting; Interpretation. Each party has participated in the drafting of this Agreement, which each party acknowledges is the result of extensive negotiations between the parties. If an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision. For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders. As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation." As used in this Agreement, references to a "party" or the "parties" are intended to refer to a party to this Agreement or the parties to this Agreement. Except as otherwise indicated, all references in this Agreement to "Sections," "Exhibits," "Annexes" and "Schedules" are intended to refer to Sections of this Agreement and Exhibits, Annexes and Schedules to this Agreement. All references in this Agreement to "\$" are intended to refer to U.S. dollars. Unless otherwise specifically provided for herein, the term "or" shall not be deemed to be exclusive. In this Agreement, when determining whether an item is "material" to Boat or its Subsidiary or Island, the term "material" shall be interpreted to mean, as applicable, "material to the business of Boat and its Subsidiaries, taken as a whole" or "material to the business of Island."
 - 8.10 Governing Law; Consent to Jurisdiction; Waiver of Trial by Jury.
- (a) This Agreement shall be governed by, and construed in accordance with, the Laws of the State of Delaware, without regard to laws that may be applicable under conflicts of laws principles (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Laws of any jurisdiction other than the State of Delaware.

Table of Contents

- (b) Each of the parties hereby irrevocably and unconditionally submits, for itself and its property, to the exclusive jurisdiction of the Court of Chancery of the State of Delaware or, in the event, but only in the event, that such court shall not have jurisdiction over such action or proceeding, the Superior Court of the State of Delaware (Complex Commercial Division) or, if jurisdiction over the action or proceeding is vested exclusively in the federal courts of the United States of America, the United States District Court for the District of Delaware, and any appellate court from any thereof (such courts, the "Chosen Courts"), in any Proceeding arising out of or relating to this Agreement or the transactions contemplated hereby or for recognition or enforcement of any judgment relating thereto, and each of the parties hereby irrevocably and unconditionally (i) agrees not to commence any such Proceeding except in such Chosen Courts, (ii) agrees that any claim in respect of any such Proceeding may be heard and determined in the Chosen Courts, (iii) waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any such Proceeding in any such court, and (iv) waives, to the fullest extent permitted by Law, the defense of an inconvenient forum to the maintenance of such Proceeding in any such Chosen Court. Each of the parties agrees that a final judgment in any such Proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law. Each party to this Agreement irrevocably consents to service of process in the manner provided for notices in Section 8.2. Nothing in this Agreement will affect the right of any party to this Agreement to serve process in any other manner permitted by Law.
- (c) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT AND ANY OF THE AGREEMENTS DELIVERED IN CONNECTION HEREWITH OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (I) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE EITHER OF SUCH WAIVERS, (II) IT UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF SUCH WAIVERS, (III) IT MAKES SUCH WAIVERS VOLUNTARILY, AND (IV) IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS. THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 8.10(c).
- 8.11 *Counterparts*. This Agreement may be executed in one or more counterparts, and by the different parties hereto in separate counterparts, each of which when executed shall be deemed to be an original but all of which taken together shall constitute one and the same agreement.
- 8.12 Delivery by Facsimile or Email. This Agreement, and any amendments hereto, waivers hereof or consents or notifications hereunder, to the extent signed and delivered by facsimile or by email with scan attachment, shall be treated in all manner and respects as an original contract and shall be considered to have the same binding legal effects as if it were the original signed version thereof delivered in person. At the request of any party, each other party shall re-execute original forms thereof and deliver them to all other parties. No party shall raise the use of facsimile or email to deliver a signature or the fact that any signature or Contract was transmitted or communicated by facsimile or email with scan attachment as a defense to the formation of a legally binding contract, and each such party forever waives any such defense.
- 8.13 *Specific Performance.* The parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an

A-82

Table of Contents

injunction or injunctions to prevent actual or threatened breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state or foreign jurisdiction having jurisdiction, this being in addition to any other remedy to which they are entitled at Law or in equity. The parties waive, in connection with any action for specific performance or injunctive relief, the defense of adequacy of remedies at Law and any requirement under Law to post a bond or other security as a prerequisite to obtaining equitable relief.

8.14 Disclosure Schedules. The parties acknowledge and agree that (a) the inclusion of any items or information in the Disclosure Schedules that are not required by this Agreement to be so included is solely for the convenience of parties, (b) the disclosure by the parties of any matter in the Disclosure Schedules shall not be deemed to constitute an acknowledgement by any party that the matter is required to be disclosed by the terms of this Agreement or that the matter is material or significant, (c) if any section of the Disclosure Schedules lists an item or information in such a way as to make its relevance to the disclosure required by or provided in another section of the Disclosure Schedules or the statements contained in any Section of this Agreement reasonably apparent, the matter shall be deemed to have been disclosed in or with respect to such other section, notwithstanding the omission of an appropriate cross-reference to such other section or the omission of a reference in the particular representation and warranty to such section of the Disclosure Schedules, (d) except as provided in clause (c) above, headings have been inserted in the Disclosure Schedules for convenience of reference only, (e) the Disclosure Schedules are qualified in their entirety by reference to specific provisions of this Agreement and (f) the Disclosure Schedules and the information and statements contained therein are not intended to constitute, and shall not be construed as constituting, representations or warranties of the parties except as and to the extent provided in this Agreement.

[Remainder of page intentionally left blank.]

A-83

Table of Contents

IN WITNESS WHEREOF, each of the parties hereto has caused this Agreement to be executed as of the date first written above by their respective officers thereunto duly authorized.

IDERA PHARMACEUTICALS, INC.

By: /s/ VINCENT J. MILANO

Name: Vincent J. Milano

Title: President and Chief Executive Officer

BIOCRYST PHARMACEUTICALS, INC.

By: /s/ JON STONEHOUSE

Name: Jon Stonehouse

Title: President and Chief Executive Officer

NAUTILUS HOLDCO, INC.

By: /s/ JON STONEHOUSE

Name: Jon Stonehouse Title: *President* **ISLAND MERGER SUB, INC.**

By: /s/ JON STONEHOUSE

Name: Jon Stonehouse
Title: President
BOAT MERGER SUB. INC.

By: /s/ JON STONEHOUSE

Name: Jon Stonehouse

Title: President

[Signature Page to Merger Agreement]

A-84

Table of Contents

EXHIBIT A

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF IDERA PHARMACEUTICALS, INC.

ARTICLE I

The name of the corporation is Idera Pharmaceuticals, Inc. (the "Corporation").

ARTICLE II

The address of the Corporation's registered office in the State of Delaware is 251 Little Falls Drive, Wilmington, County of New Castle, 19808. The name of its registered agent at such address is Corporation Service Company.

ARTICLE III

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of the State of Delaware (as amended or supplemented from time to time, the "DGCL").

ARTICLE IV

The aggregate number of shares which the Corporation shall have authority to issue is 1,000 shares of capital stock all of which shall be designated "Common Stock" and have a par value of \$0.01 per share.

ARTICLE V

In furtherance of and not in limitation of the powers conferred by the DGCL, the Board of Directors of the Corporation (the "*Board of Directors*") is expressly authorized to make, amend or repeal Bylaws of the Corporation (the "*Bylaws*").

ARTICLE VI

The business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors. The number of directors which shall constitute the whole Board of Directors shall be fixed by the Board of Directors in the manner provided in the Bylaws. Elections of directors need not be by written ballot unless otherwise provided in the Bylaws.

ARTICLE VII

Except to the extent that the DGCL prohibits the elimination or limitation of liability of directors for breaches of fiduciary duty, no director of the Corporation shall be personally liable to the Corporation or its stockholders for monetary damages for any breach of fiduciary duty as a director, notwithstanding any provision of law imposing such liability. No amendment to or repeal of this provision shall apply to or have any effect on the liability or alleged liability of any director of the Corporation for or with respect to any acts or omissions of such director occurring prior to such amendment.

ARTICLE VIII

1. Action, Suits And Proceedings Other than by or in the Right of the Corporation. The Corporation shall indemnify each person who was or is a party or is threatened to be made a party to

Table of Contents

any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the Corporation), by reason of the fact that he is or was, or has agreed to become, a director or officer of the Corporation, or is or was serving, or has agreed to serve, at the request of the Corporation, as a director, officer or trustee of, or in a similar capacity with, another corporation, partnership, joint venture, trust or other enterprise (including any employee benefit plan) (all such persons being referred to hereafter as an "*Indemnitee*"), or by reason of any action alleged to have been taken or omitted in such capacity, against all expenses (including attorneys' fees) judgment, fines and amounts paid in settlement actually and reasonably incurred by him or on his behalf in connection with such action, suit or proceeding and any appeal therefrom, if he acted in good faith and in a manner he reasonably believed to be in, or not opposed to, the best interests of the Corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his conduct was unlawful. The termination of any action, suit or proceeding by judgment, order, settlement, conviction or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that the person did not act in good faith and in a manner which he reasonably believed to be in, or not opposed to, the best interests of the Corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that his conduct was unlawful. Notwithstanding anything to the contrary in this *Article*, except as set forth in *Section 6* below, the Corporation shall not indemnify an Indemnitee seeking indemnification in connection with a proceeding (or part thereof) initiated by the Indemnitee unless the initiation thereof was approved by the Board of Directors.

- 2. Actions or Suits By or in the Right of the Corporation. The Corporation shall indemnify any Indemnitee who was or is a party or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the Corporation to procure a judgment in its favor by reason of the fact that he is or was, or has agreed to become, a director or officer of the Corporation, or is or was serving, or has agreed to serve, at the request of the Corporation, as a director, officer or trustee of, or in a similar capacity with, another corporation, partnership, joint venture, trust or other enterprise (including any employee benefit plan), or by reason of any action alleged to have been taken or omitted in such capacity, against all expenses (including attorneys' fees) and amounts paid in settlement actually and reasonably incurred by him or on his behalf in connection with such action, suit or proceeding and any appeal therefrom, if he acted in good faith and in a manner he reasonably believed to be in, or not opposed to, the best interests of the Corporation, except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable to the Corporation unless and only to the extent that the Court of Chancery of Delaware or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of such liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses (including attorneys' fees) which the Court of Chancery of Delaware or such other court shall deem proper.
- 3. Indemnification For Expenses Of Successful Party. Notwithstanding the other provisions of this Article, to the extent that an Indemnitee has been successful, on the merits or otherwise, in defense of any action, suit or proceeding referred to in Sections 1 and 2 of this Article, or in defense of any claim, issue or matter therein, or on appeal from any such action, suit or proceeding, he shall be indemnified against all expenses (including attorneys' fees) actually and reasonably incurred by him or on his behalf in connection therewith. Without limiting the foregoing, if any action, suit or proceeding is disposed of, on the merits or otherwise (including a disposition without prejudice), without (i) the disposition being adverse to the Indemnitee, (ii) an adjudication that the Indemnitee was liable to the Corporation, (iii) a plea of guilty or nolo contendere by the Indemnitee, (iv) an adjudication that the Indemnitee did not act in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the Corporation, and (v) with respect to any criminal proceeding, an adjudication that the Indemnitee had reasonable cause to believe his conduct was unlawful, the Indemnitee shall be considered for the purposes hereof to have been wholly successful with respect thereto.

Table of Contents

- 4. Notification and Defense of Claim. As a condition precedent to his right to be indemnified, the Indemnitee must notify the Corporation in writing as soon as practicable of any action, suit, proceeding or investigation involving him for which indemnity will or could be sought. With respect to any action, suit, proceeding or investigation of which the Corporation is so notified, the Corporation will be entitled to participate therein at its own expense and/or to assume the defense thereof at its own expense, with legal counsel reasonably acceptable to the Indemnitee. After notice from the Corporation to the Indemnitee of its election so to assume such defense, the Corporation shall not be liable to the Indemnitee for any legal or other expenses subsequently incurred by the Indemnitee in connection with such claim, other than as provided below in this Section 4. The Indemnitee shall have the right to employ his own counsel in connection with such claim, but the fees and expenses of such counsel incurred after notice from the Corporation of its assumption of the defense thereof shall be at the expense of the Indemnitee unless (i) the employment of counsel by the Indemnitee has been authorized by the Corporation, (ii) counsel to the Indemnitee shall have reasonably concluded that there may be a conflict of interest or position on any significant issue between the Corporation and the Indemnitee in the conduct of the defense of such action or (iii) the Corporation shall not in fact have employed counsel to assume the defense of such action, in each of which cases the fees and expenses of counsel for the Indemnitee shall be at the expense of the Corporation, except as otherwise expressly provided by this Article. The Corporation shall not be entitled, without the consent of the Indemnitee, to assume the defense of any claim brought by or in the right of the Corporation or as to which counsel for the Indemnitee shall have reasonably made the conclusion provided for in clause (ii) above.
- 5. Advance of Expenses. Subject to the provisions of Section 6 below, in the event that the Corporation does not assume the defense pursuant to Section 4 of this Article of any action, suit, proceeding or investigation of which the Corporation receives notice under this Article, any expenses (including attorneys' fees) incurred by an Indemnitee in defending a civil or criminal action, suit, proceeding or investigation or any appeal therefrom shall be paid by the Corporation in advance of the final disposition of such matter, provided, however, that the payment of such expense incurred by an Indemnitee in advance of the final disposition of such matter shall be made only upon receipt of an undertaking by or on behalf of the Indemnitee to repay all amounts so advanced in the event that it shall ultimately be determined that the Indemnitee is not entitled to be indemnified by the Corporation as authorized in this Article. Such undertaking may be accepted without reference to the financial ability of such person to make such repayment.
- 6. Procedure for Indemnification. In order to obtain indemnification or advancement of expenses pursuant to Section 1, 2, 3 or 5 of this Article, the Indemnitee shall submit to the Corporation a written request, including in such request such documentation and information as is reasonably available to the Indemnitee and is reasonably necessary to determine whether and to what extent the Indemnitee is entitled to indemnification or advancement of expenses. Any such indemnification or advancement of expenses shall be made promptly, and in any event within 60 days after receipt by the Corporation of the written request of the Indemnitee, unless with respect to requests under Section 1, 2 or 5 the Corporation determines, by clear and convincing evidence, within such 60-day period that the Indemnitee did not meet the applicable standard of conduct set forth in Section 1 or 2, as the case may be. Such determination shall be made in each instance by (a) a majority vote of a quorum of the directors of the Corporation consisting of persons who are not at that time parties to the action, suit or proceeding in question ("disinterested directors"), (b) if no such quorum is obtainable, a majority vote of a committee of two or more disinterested directors, (c) a majority vote of a quorum of the outstanding shares of stock of all classes entitled to vote for directors, voting as a single class, which quorum shall consist of stockholders who are not at that time parties to the action, suit or proceeding in question, (d) independent legal counsel (who may be regular legal counsel to the Corporation), or (e) a court of competent jurisdiction.

Table of Contents

- 7. Remedies. The right to indemnification or advances as granted by this Article shall be enforceable by the Indemnitee in any court of competent jurisdiction if the Corporation denies such request, in whole or in part, or if no disposition thereof is made within the 60-day period referred to above in Section 6. Unless otherwise provided by law, the burden of proving that the Indemnitee is not entitled to indemnification or advanced of expenses under this Article shall be on the Corporation. Neither the failure of the Corporation to have made a determination prior to the commencement of such action that indemnification is proper in the circumstances because the Indemnitee has met the applicable standard of conduct, nor an actual determination by the Corporation pursuant to Section 6 that the Indemnitee has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that the Indemnitee has not met the applicable standard of conduct. The Indemnitee's expenses (including attorneys' fees) incurred in connection with successfully establishing his right to indemnification, in whole or in part, in any such proceeding shall also be indemnified by the Corporation.
- 8. Subsequent Amendment. No amendment, termination or repeal of this Article or of the relevant provisions of the DGCL or any other applicable laws shall affect or diminish in any way the rights of any Indemnitee to indemnification under the provisions hereof with respect to any action, suit, proceeding or investigation arising out of or relating to any actions, transactions or facts occurring prior to the final adoption of such amendment, termination or repeal.
- 9. Other Rights. The indemnification and advancement of expenses provided by this Article shall not be deemed exclusive of any other rights to which an Indemnitee seeking indemnification or advancement of expenses may be entitled under any law (common or statutory), agreement or vote of stockholders or disinterested directors or otherwise, both as to action in his official capacity and as to action in any other capacity while holding office for the Corporation, and shall continue as to an Indemnitee who has ceased to be a director or officer, and shall inure to the benefit of the estate, heirs, executors and administrators of the Indemnitee. Nothing contained in this Article shall be deemed to prohibit, and the Corporation is specifically authorized to enter into, agreements with officers and directors providing indemnification rights and procedures different from those set forth in this Article. In addition, the Corporation may, to the extent authorized from time to time by its Board of Directors, grant indemnification rights to other employees or agents of the Corporation or other persons serving the Corporation and such rights may be equivalent to, or greater or less than, those set forth in this Article.
- 10. Partial Indemnification. If an Indemnitee is entitled under any provision of this Article to indemnification by the Corporation for some or a portion of the expenses (including attorneys' fees), judgments, fines or amounts paid in settlement actually and reasonably incurred by him or on his behalf in connection with any action, suit, proceeding or investigation and any appeal, therefrom but not, however, for the total amount thereof, the Corporation shall nevertheless indemnify the Indemnitee for the portion of such expenses (including attorneys' fees), judgments, fines or amounts paid in settlement to which the Indemnitee is entitled.
- 11. *Insurance*. The Corporation may purchase and maintain insurance, at its expense, to protect itself and any director, officer, employee or agent of the Corporation or another corporation, partnership, joint venture, trust or other enterprise (including any employee benefit plan) against any expense, liability or loss incurred by him in any such capacity, or arising out of his status as such, whether or not the Corporation would have the power to indemnify such person against such expense, liability or loss under the DGCL.
- 12. *Merger or Consolidation*. If the Corporation is merged into or consolidated with another corporation and the Corporation is not the surviving corporation, the surviving corporation shall assume the obligations of the Corporation under this *Article* with respect to any action, suit, proceeding

Table of Contents

or investigation arising out of or relating to any actions, transactions or facts occurring prior to the date of such merger or consolidation.

- 13. Savings Clause. If this Article or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the Corporation shall nevertheless indemnify each Indemnitee as to any expenses (including attorneys' fees) judgments, fines and amounts paid in settlement in connection with any action, suit, proceeding or investigation, whether civil, criminal or administrative, including an action by or in the right of the Corporation, to the fullest extent permitted by any applicable portion of this Article that shall not have been invalidated and to the fullest extent permitted by applicable law.
- 14. *Definitions*. Terms used herein and defined in Section 145(h) and Section 145(i) of the DGCL shall have the respective meanings assigned to such terms in such Section 145(h) and Section 145(i).
- 15. Subsequent Legislation. If the DGCL is amended after adoption of this Article to expand further the indemnification permitted to Indemnitees, then the Corporation shall indemnify such persons to the fullest extent permitted by the DGCL, as so amended.

ARTICLE IX

The Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, in the manner now or hereafter prescribed by statute, and all rights conferred upon the stockholders herein are granted subject to this reservation.

EXHIBIT B

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF BIOCRYST PHARMACEUTICALS, INC.

ARTICLE I

The name of the corporation is BioCryst Pharmaceuticals, Inc. (the "Corporation").

ARTICLE II

The address of the Corporation's registered office in the State of Delaware is 251 Little Falls Drive, Wilmington, County of New Castle, 19808. The name of its registered agent at such address is Corporation Service Company.

ARTICLE III

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of the State of Delaware (as amended or supplemented from time to time, the "DGCL").

ARTICLE IV

The aggregate number of shares which the Corporation shall have authority to issue is 1,000 shares of capital stock all of which shall be designated "*Common Stock*" and have a par value of \$0.01 per share.

ARTICLE V

In furtherance of and not in limitation of the powers conferred by the DGCL, the Board of Directors of the Corporation (the "Board of Directors") is expressly authorized to make, amend or repeal Bylaws of the Corporation (the "Bylaws").

ARTICLE VI

The business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors. The number of directors which shall constitute the whole Board of Directors shall be fixed by the Board of Directors in the manner provided in the Bylaws. Elections of directors need not be by written ballot unless otherwise provided in the Bylaws.

ARTICLE VII

Except to the extent that the DGCL prohibits the elimination or limitation of liability of directors for breaches of fiduciary duty, no director of the Corporation shall be personally liable to the Corporation or its stockholders for monetary damages for any breach of fiduciary duty as a director, notwithstanding any provision of law imposing such liability. No amendment to or repeal of this provision shall apply to or have any effect on the liability or alleged liability of any director of the Corporation for or with respect to any acts or omissions of such director occurring prior to such amendment.

ARTICLE VIII

The Corporation shall, to the fullest extent permitted by Section 145 of the DGCL, as amended from time to time, indemnify each person who was or is a party or is threatened to be made a party to

A-B-1

Table of Contents

any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that he is or was, or has agreed to become, a director or officer of the Corporation, or is or was serving, or has agreed to serve, at the request of the Corporation, as a director or officer of the Corporation, or is or was serving, or has agreed to serve, at the request of the Corporation, as director, officer or trustee of, or in a similar capacity with, another corporation, partnership, joint venture, trust or other enterprise (including any employee benefit plan), or by reason of any action alleged to have been taken or omitted in such capacity, against all expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by him or on his behalf in connection with such action, suit or proceeding and any appeal therefrom.

Indemnification may include payment by the Corporation of expenses in defending an action or proceeding in advance of the final disposition of such action or proceeding upon receipt of an undertaking by the person indemnified to repay such payment if it is ultimately determined that such person is not entitled to indemnification under this *Article*, which undertaking may be accepted without reference to the financial ability of such person to make such repayment.

The Corporation shall not indemnify any such person seeking indemnification in connection with a proceeding (or part thereof) initiated by such person unless the initiation thereof was approved by the Board of Directors.

The indemnification rights provided in this Article (i) shall not be deemed exclusive of any other rights to which those indemnified may be entitled under any law, agreement or vote of stockholders or disinterested directors or otherwise, and (ii) shall inure to the benefit of the heirs, executors and administrators of such person. The Corporation may, to the extent authorized from time to time by its Board of Directors, grant indemnification rights to other employees or agents of the Corporation or to other persons serving the Corporation and such rights may be equivalent to, or greater or less than, those set forth in this *Article*.

ARTICLE IX

The Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, in the manner now or hereafter prescribed by statute, and all rights conferred upon the stockholders herein are granted subject to this reservation.

A-B-2

Annex B Opinion of Goldman, Sachs & Co. LLC

Table of Contents

PERSONAL AND CONFIDENTIAL

January 21, 2018

Board of Directors Idera Pharmaceuticals, Inc. 167 Sidney Street Cambridge, MA 02139

Ladies and Gentlemen:

You have requested our opinion as to the fairness from a financial point of view to the holders (other than BioCryst Pharmaceuticals Inc. ("BioCryst") and its affiliates) of the outstanding shares of common stock, par value \$0.001 per share (the "Shares"), of Idera Pharmaceuticals Inc. (the "Company") of the Exchange Ratio (defined below) pursuant to the Agreement and Plan of Merger, dated as of January 21, 2018 (the "Agreement"), by and among BioCryst, Nautilus Holdco, Inc., a wholly owned subsidiary of BioCryst ("Holdco"), Island Merger Sub, Inc., a wholly owned subsidiary of Holdco ("Island Merger Sub"), Boat Merger Sub, Inc., a wholly owned subsidiary of Holdco ("Boat Merger Sub"), and the Company. Pursuant to the Agreement, Island Merger Sub will be merged with and into the Company, and each outstanding Share (other than Shares held by BioCryst and its affiliates) will be converted into the right to receive 0.20 of a share of common stock, par value \$0.01 per share (the "Holdco Common Stock"), of Holdco (the "Exchange Ratio"). The Agreement also provides that Boat Merger Sub will be merged with and into BioCryst and each outstanding share of common stock, par value \$0.01 per share (the "BioCryst Common Stock"), of BioCryst, will be converted into the right to receive 0.50 of a share of Holdco Common Stock.

Goldman Sachs & Co. LLC and its affiliates are engaged in advisory, underwriting and financing, principal investing, sales and trading, research, investment management and other financial and non-financial activities and services for various persons and entities. Goldman Sachs & Co. LLC and its affiliates and employees, and funds or other entities they manage or in which they invest or have other economic interests or with which they co-invest, may at any time purchase, sell, hold or vote long or short positions and investments in securities, derivatives, loans, commodities, currencies, credit default swaps and other financial instruments of the Company, BioCryst, any of their respective affiliates and third parties, including Baker Bros Advisors LP, a significant stockholder of the Company and BioCryst ("Baker Bros"), and Pillar Invest Corporation, a significant stockholder of the Company ("Pillar") and any of their respective affiliates and portfolio companies, or any currency or commodity that may be involved in the transaction contemplated by the Agreement (the "Transaction"). We have acted as financial advisor to the Company in connection with, and have participated in certain of the negotiations leading to, the Transaction. We expect to receive fees for our services in connection with the Transaction, the principal portion of which is contingent upon consummation of the Transaction, and the Company has agreed to reimburse certain of our expenses arising, and indemnify us against certain liabilities that may arise, out of our engagement. We have provided certain financial advisory and/or underwriting services to the Company and/or its affiliates from time to time for which our Investment Banking Division has received, and may receive, compensation, including having acted as bookrunning manager with respect to the public offering by the Company of 26,225,243 Shares in October 2016 and the public offering of 38,333,334 Shares in October 2017. We may in the future provide financial advisory and/or underwriting services to the Company, BioCryst, Baker Bros, Pillar and their respective affiliates and, as applicable, portfolio companies for which our Investment Banking Division may receive compensation. Affiliates of Goldman Sachs & Co. LLC also may have co-invested with Baker Bros or Pillar or their respective affiliates from time to time and may have invested in limited partnership units of affiliates of Baker Bros or Pillar from time to time and may do so in the future.

Table of Contents

In connection with this opinion, we have reviewed, among other things, the Agreement; annual reports to stockholders and Annual Reports on Form 10-K of the Company and BioCryst for the five years ended December 31, 2016; certain interim reports to stockholders and Quarterly Reports on Form 10-Q of the Company and BioCryst; certain other communications from the Company and BioCryst to their respective stockholders; certain publicly available research analyst reports for the Company and BioCryst; certain internal financial analyses and forecasts for BioCryst on a stand-alone basis and with respect to its products prepared by its management; certain internal financial analyses and forecasts for the Company and certain financial analyses and forecasts for BioCryst, in each case, on a stand-alone basis and with respect to its products and probabilities of success related thereto, and certain financial analyses and forecasts for Holdco on a pro-forma basis giving effect to the Transaction, in each case, as prepared by the management of the Company and approved for our use by the Company (collectively, the "Forecasts"), including certain operating synergies projected by the management of the Company to result from the Transaction, as approved for our use by the Company (the "Synergies"); and certain analyses prepared by the management of the Company related to the expected utilization by the Company, BioCryst and Holdco of certain net operating loss carryforwards, as approved for our use by the Company (the "NOL Forecasts"). We have also held discussions with members of the senior managements of the Company and BioCryst regarding their assessment of the strategic rationale for, and the potential benefits of, the Transaction and the past and current business operations, financial condition and future prospects of BioCryst and with members of the senior management of the Company regarding their assessment of the past and current business operations, financial condition and future prospects of the Company; reviewed the reported price and trading activity for the Shares and shares of BioCryst Common Stock; compared certain financial and stock market information for the Company and BioCryst with similar information for certain other companies the securities of which are publicly traded; and performed such other studies and analyses, and considered such other factors, as we deemed appropriate.

For purposes of rendering this opinion, we have, with your consent, relied upon and assumed the accuracy and completeness of all of the financial, legal, regulatory, tax, accounting and other information provided to, discussed with or reviewed by, us, without assuming any responsibility for independent verification thereof. In that regard, we have assumed with your consent that the Forecasts, including the Synergies, and the NOL Forecasts have been reasonably prepared on a basis reflecting the best currently available estimates and judgments of the management of the Company. We have not made an independent evaluation or appraisal of the assets and liabilities (including any contingent, derivative or other off-balance-sheet assets and liabilities) of Holdco, the Company or BioCryst or any of their respective subsidiaries and we have not been furnished with any such evaluation or appraisal. We have assumed that all governmental, regulatory or other consents and approvals necessary for the consummation of the Transaction will be obtained without any adverse effect on the Company, BioCryst or Holdco or on the expected benefits of the Transaction in any way meaningful to our analysis. We have assumed that the Transaction will be consummated on the terms set forth in the Agreement, without the waiver or modification of any term or condition the effect of which would be in any way meaningful to our analysis.

Our opinion does not address the underlying business decision of the Company to engage in the Transaction, or the relative merits of the Transaction as compared to any strategic alternatives that may be available to the Company; nor does it address any legal, regulatory, tax or accounting matters. This opinion addresses only the fairness from a financial point of view to the holders (other than BioCryst and its affiliates) of Shares, as of the date hereof, of the Exchange Ratio pursuant to the Agreement. We do not express any view on, and our opinion does not address, any other term or aspect of the Agreement or Transaction or any term or aspect of any other agreement or instrument contemplated by the Agreement or entered into or amended in connection with the Transaction, including, the fairness of the Transaction to, or any consideration received in connection therewith by, the holders of any other class of securities, creditors, or other constituencies of the Company; nor as to the fairness of

Table of Contents

the amount or nature of any compensation to be paid or payable to any of the officers, directors or employees of the Company, or class of such persons, in connection with the Transaction, whether relative to the Exchange Ratio pursuant to the Agreement or otherwise. We are not expressing any opinion as to the prices at which shares of Holdco Common Stock will trade at any time or as to the impact of the Transaction on the solvency or viability of the Company, BioCryst or Holdco, or the ability of the Company, BioCryst or Holdco to pay their respective obligations when they come due. Our opinion is necessarily based on economic, monetary, market and other conditions as in effect on, and the information made available to us as of, the date hereof and we assume no responsibility for updating, revising or reaffirming this opinion based on circumstances, developments or events occurring after the date hereof. Our advisory services and the opinion expressed herein are provided for the information and assistance of the Board of Directors of the Company in connection with its consideration of the Transaction and such opinion does not constitute a recommendation as to how any holder of Shares should vote with respect to such Transaction or any other matter. This opinion has been approved by a fairness committee of Goldman Sachs & Co. LLC.

Based upon and subject to the foregoing, it is our opinion that, as of the date hereof, the Exchange Ratio pursuant to the Agreement is fair from a financial point of view to the holders (other than BioCryst and its affiliates) of Shares.

Very truly yours,

/s/ Goldman Sachs & Co. LLC (GOLDMAN SACHS & CO. LLC)

Annex C Opinion of J.P. Morgan Securities LLC

Table of Contents

January 21, 2018

The Board of Directors BioCryst Pharmaceuticals, Inc. 4505 Emperor Blvd, Suite 200 Durham, NC 27703

Members of the Board of Directors:

You have requested our opinion as to the fairness, from a financial point of view, to the holders of common stock, par value \$0.01 per share (the "Company Common Stock"), of BioCryst Pharmaceuticals, Inc. (the "Company") of the Exchange Ratio (as defined below) in the proposed Transaction (as defined below). Pursuant to the Agreement and Plan of Merger, dated as of January 21, 2018 (the "Agreement"), among the Company, Idera Pharmaceuticals, Inc. (the "Merger Partner"), Nautilus Holdco, Inc. ("Holdco"), Island Merger Sub, Inc. ("Merger Sub A") and Boat Merger Sub, Inc. ("Merger Sub B"), (i) Merger Sub A will merge with and into the Merger Partner and (A) each outstanding share of common stock, par value \$0.001 per share, of the Merger Partner (the "Merger Partner Common Stock") will be converted into the right to receive 0.2 shares of Holdco's common stock, par value \$0.01 per share (the "Holdco Common Stock") and cash in lieu of fractional shares of Holdco Common Stock as contemplated by the Agreement and (B) each outstanding share of Series A convertible preferred stock, par value \$0.01 per share, of the Merger Partner (the "Merger Partner Preferred Stock") will be converted into the right to receive that number of shares of Holdco Common Stock equal to (1) \$1.00 per share plus any accrued but unpaid dividends on the Merger Partner Preferred Stock, divided by the Market Price (as defined in the Agreement) of Merger Partner Common Stock as of the third day prior to the Closing Date (as defined in the Agreement), multiplied by (2) 0.2 and (ii) Merger Sub B will merge with and into the Company, and each outstanding share of common stock, par value \$0.01 per share, of the Company (the "Company Common Stock") will be converted into the right to receive 0.5 shares (the "Exchange Ratio") of Holdco Common Stock and cash in lieu of fractional shares of Holdco Common Stock as contemplated by the Agreement (the transactions described in the immediately preceding clauses (i) and (ii), together and not separately, are ref

In connection with preparing our opinion, we have (i) reviewed the Agreement; (ii) reviewed certain publicly available business and financial information concerning the Merger Partner and the Company and the industries in which they operate; (iii) reviewed certain internal financial analyses and forecasts prepared by the managements of the Merger Partner and the Company relating to their respective businesses, as well as the estimated amount and timing of the cost savings and related expenses and synergies expected to result from the Transaction (the "Synergies"); and (iv) performed such other financial studies and analyses and considered such other information as we deemed appropriate for the purposes of this opinion.

In addition, we have held discussions with certain members of the management of the Merger Partner and the Company with respect to certain aspects of the Transaction, and the past and current business operations of the Merger Partner and the Company, the financial condition and future prospects and operations of the Merger Partner and the Company, the effects of the Transaction on the financial condition and future prospects of the Company, and certain other matters we believed necessary or appropriate to our inquiry.

In giving our opinion, we have relied upon and assumed the accuracy and completeness of all information that was publicly available or was furnished to or discussed with us by the Merger Partner and the Company or otherwise reviewed by or for us. We have not independently verified any such

C-1

Table of Contents

information or its accuracy or completeness and, pursuant to our engagement letter with the Company, we did not assume any obligation to undertake any such independent verification. We have not conducted or been provided with any valuation or appraisal of any assets or liabilities, nor have we evaluated the solvency of the Merger Partner, Holdco or the Company under any state or federal laws relating to bankruptcy, insolvency or similar matters. In relying on financial analyses and forecasts provided to us or derived therefrom, including the Synergies, we have assumed that they have been reasonably prepared based on assumptions reflecting the best currently available estimates and judgments by management as to the expected future results of operations and financial condition of the Merger Partner and the Company to which such analyses or forecasts relate. We express no view as to such analyses or forecasts (including the Synergies) or the assumptions on which they were based. We have also assumed that the Transaction and the other transactions contemplated by the Agreement will qualify as a tax-free reorganization for United States federal income tax purposes, and will be consummated as described in the Agreement. For purposes of conversion of the Merger Partner Preferred Stock, we have assumed at the direction of the Company that the Market Price of shares of Merger Partner Common Stock on the day that is three days before the Closing Date will be \$2.29. We have also assumed that the representations and warranties made by the Company and the Merger Partner in the Agreement and the related agreements are and will be true and correct in all respects material to our analysis. We are not legal, regulatory or tax experts and have relied on the assessments made by advisors to the Company with respect to such issues. We have further assumed that all material governmental, regulatory or other consents and approvals necessary for the consummation of the Transaction will be obtained without any adverse effect on the Merger Partner, Holdco or the Company or on the contemplated benefits of the Transaction.

Our opinion is necessarily based on economic, market and other conditions as in effect on, and the information made available to us as of, the date hereof. It should be understood that subsequent developments may affect this opinion and that we do not have any obligation to update, revise, or reaffirm this opinion. Our opinion is limited to the fairness, from a financial point of view, to the holders of the Company Common Stock of the Exchange Ratio in the proposed Transaction and we express no opinion as to the fairness of any consideration to be paid in connection with the Transaction to the holders of any other class of securities, creditors or other constituencies of the Company or as to the underlying decision by the Company to engage in the Transaction. Furthermore, we express no opinion with respect to the amount or nature of any compensation to any officers, directors, or employees of any party to the Transaction, or any class of such persons relative to the Exchange Ratio applicable to the holders of the Company Common Stock in the Transaction or with respect to the fairness of any such compensation. We are expressing no opinion herein as to the price at which the Company Common Stock, the Holdco Common Stock or the Merger Partner Common Stock will trade at any future time.

We have acted as financial advisor to the Company with respect to the proposed Transaction and will receive a fee from the Company for our services, a substantial portion of which will become payable only if the proposed Transaction is consummated. In addition, the Company has agreed to indemnify us for certain liabilities arising out of our engagement. During the two years preceding the date of this letter, we and our affiliates have had commercial or investment banking relationships with the Company and the Merger Partner for which we and such affiliates have received customary compensation. Such services during such period have included acting as sole bookrunner on the Company's offering of equity securities in March 2017 and joint lead bookrunner on the Company's offering of equity securities in September 2017, and acting as joint lead bookrunner on the Merger Partner's offerings of equity securities in October 2016 and October 2017. In addition, we and our affiliates hold, on a proprietary basis, less than 1% of the outstanding common stock of each of the Company and the Merger Partner. In the ordinary course of our businesses, we and our affiliates may actively trade the debt and equity securities or financial instruments (including derivatives, bank loans or other obligations) of the Company, Holdco or the Merger Partner for our own account or for the

Table of Contents

accounts of customers and, accordingly, we may at any time hold long or short positions in such securities or other financial instruments.

On the basis of and subject to the foregoing, it is our opinion as of the date hereof that the Exchange Ratio in the proposed Transaction is fair, from a financial point of view, to the holders of the Company Common Stock.

The issuance of this opinion has been approved by a fairness opinion committee of J.P. Morgan Securities LLC. This letter is provided to the Board of Directors of the Company (in its capacity as such) in connection with and for the purposes of its evaluation of the Transaction. This opinion does not constitute a recommendation to any shareholder of the Company as to how such shareholder should vote with respect to the Transaction or any other matter. This opinion may not be disclosed, referred to, or communicated (in whole or in part) to any third party for any purpose whatsoever except with our prior written approval. This opinion may be reproduced in full in any proxy or information statement mailed to shareholders of the Company but may not otherwise be disclosed publicly in any manner without our prior written approval, except as otherwise agreed pursuant to the terms of the engagement letter dated January 14, 2018 between J.P. Morgan Securities LLC and the Company.

Very truly yours,

/s/ J.P. Morgan Securities LLC (J.P. MORGAN SECURITIES LLC)

C-3

Annex D Form of Amended and Restated Certificate of Incorporation of Holdco

Table of Contents

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF [•]

Nautilus Holdco, Inc. (the "Corporation"), a corporation organized and existing under the laws of the State of Delaware, does hereby certify as follows:

FIRST: The present name of the Corporation is Nautilus Holdco, Inc. The original Certificate of Incorporation of the Corporation was filed with the Secretary of State of the State of Delaware on January 16, 2018 (the "Original Certificate").

SECOND: This Amended and Restated Certificate of Incorporation (the "*Restated Certificate*") has been duly adopted by the Corporation in accordance with the provisions of the General Corporation Law of the State of Delaware (as amended or supplemented from time to time, the "*DGCL*").

THIRD: This Restated Certificate amends, restates and integrates, in the entirety, the provisions of the corporation's Original Certificate.

FOURTH: The text of the Original Certificate is hereby amended and restated in its entirety to read in its entirety as follows:

ARTICLE I

NAME

The name of the Corporation is [•].

ARTICLE II

REGISTERED OFFICE AND AGENT

The address of the registered office of the Corporation in the State of Delaware is 251 Little Falls Drive, Wilmington, County of New Castle, 19808. The name of its registered agent at that address is Corporation Service Company.

ARTICLE III

PURPOSE AND POWERS

The purpose of the Corporation is to engage in any lawful act or activity for which a corporation may now or hereafter be organized under the DGCL. The Corporation shall have all powers that may now or hereafter be lawful for a corporation to exercise under the DGCL.

ARTICLE IV

CAPITAL STOCK

- (A) Classes of Stock. The total number of shares of stock of all classes of capital stock that the Corporation is authorized to issue is three hundred eighty million (380,000,000) shares. The authorized capital stock is divided into twenty million (20,000,000) shares of preferred stock having a par value of \$0.01 per share (the "Preferred Stock") and four hundred million (400,000,000) shares of common stock having a par value of \$0.01 per share (the "Common Stock").
- (B)

 Preferred Stock. The Board of Directors of the Corporation (the "Board of Directors") is hereby expressly authorized to provide by resolution or resolutions from time to time for the issue of the Preferred Stock in one or more series, the shares of each of which series may have such voting powers, full or limited, or no voting powers, and such designations, preferences and relative, participating,

optional or other special rights, and qualification, limitations or restrictions thereof, as shall be permitted under the DGCL and as shall be stated in the resolution or resolutions providing for the issue of such stock adopted by the Board of Directors pursuant to the authority

Table of Contents

expressly vested in the Board of Directors hereby. Without limiting the generality of the foregoing, the resolution or resolutions providing for the issuance of any series of Preferred Stock may provide that such series shall be superior or rank equally or be junior to the Preferred Stock of any other series to the extent permitted by law. The number of authorized shares of Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the stock of the Corporation entitled to vote, irrespective of the provisions of Section 242(b)(2) of the DGCL.

(C) Common Stock.

- 1. *General*. All shares of Common Stock shall be of one and the same class, shall be identical in all respects and shall have equal rights, powers and privileges, *provided*, *however* that such rights are subject to and qualified by the rights of the holders of the Preferred Stock of any series as may be designated by the Board upon any issuance of the Preferred Stock of any series.
- Vote. Except as otherwise provided for by resolution or resolutions of the Board pursuant to this *Article IV* with respect to the issuance of any series of Preferred Stock or by the DGCL, the holders of outstanding shares of Common Stock shall have the exclusive right to vote on all matters requiring stockholder action. On each matter on which holders of Common Stock are entitled to vote, each outstanding share of such Common Stock will be entitled to one (1) vote. There shall be no cumulative voting. The number of authorized shares of Common Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the stock of the Corporation entitled to vote, irrespective of the provisions of Section 242(b)(2) of the DGCL.
- Dividends. Subject to the rights of holders of any series of outstanding Preferred Stock which may from time to time come into existence and be outstanding, the holders of shares of Common Stock shall have equal rights of participation in the dividends and other distributions in cash, stock or property of the Corporation when, as and if declared thereon by the Board of Directors from time to time out of assets or funds of the Corporation legally available therefor. Any dividends declared by the Board of Directors to the holders of the then outstanding Common Stock shall be paid to the holders thereof *pro rata* in accordance with the number of shares of Common Stock held by each such holder as of the record date of such dividend.
- 4.

 Liquidation. Subject to the rights of holders of any series of outstanding Preferred Stock which may from time to time come into existence and be outstanding, in the event of any liquidation, dissolution or winding up of the affairs of the Corporation, whether voluntary or involuntary, the holders of shares of Common Stock shall have equal rights to receive the assets and funds of the Corporation available for distribution to stockholders and such assets or funds shall be distributed pro rata in accordance with the number of shares of Common Stock held by each such holder.

ARTICLE V

BOARD OF DIRECTORS

- (A) Power of the Board of Directors. The business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors. In furtherance, and not in limitation, of the powers conferred by the DGCL, the Board of Directors shall be expressly authorized to:
 - determine the rights, powers, duties, rules and procedures that affect the power of the Board of Directors to manage and direct the business and affairs of the Corporation;

Table of Contents

- 2. establish one or more committees of the Board of Directors, by the affirmative vote of a majority of the entire Board of Directors, to which may be delegated any or all of the powers and duties of the Board of Directors to the fullest extent permitted by law; and
- 3. exercise all such powers and do all such acts as may be exercised by the Corporation, subject to the provisions of the DGCL, this Restated Certificate, and the Amended and Restated Bylaws of the Corporation (as the same may be amended and/or restated from time to time, the "*Bylaws*").
- (B)

 Number of Directors, Election and Terms. The number of directors constituting the entire Board of Directors shall consist of not less than six (6) nor more than twelve (12) persons. The exact number of directors within the minimum and maximum limitations specified in the preceding sentence shall be fixed from time to time by the Board of Directors pursuant to a resolution adopted by a majority of the entire Board of Directors. Each director shall hold office until the expiration of the term for which elected and until his or her successor is duly elected and qualified, or until his or her earlier death, resignation or removal. Election of the directors need not be by written ballot.
- Vacancies and Newly Created Directorships. Except as otherwise required by law and subject to the rights of the holders of any class or series of Preferred Stock to elect directors, any vacancies on the Board of Directors for any reason, including from the death, resignation, disqualification or removal of any director, and any newly created directorships resulting by reason of any increase in the number of directors shall be filled exclusively by the Board of Directors, acting by the affirmative vote of a majority of the remaining directors then in office, even if less than a quorum, or by the sole remaining director, and shall not be filled by stockholders. Any directors elected to fill a vacancy shall hold office until the next annual meeting of stockholders or until their successors are duly elected and qualified. No decrease in the number of directors constituting the Board of Directors shall shorten the term of any incumbent director.
- (D) *Continuance in Office*. Notwithstanding the foregoing provisions of this *Article V*, any director whose term of office has expired shall continue to hold office until his successor shall be elected and qualified.
- (E)

 Removal of Directors. Except as otherwise required by law and subject to the rights of the holders of any class or series of Preferred Stock then outstanding, any director, or the entire Board of Directors, may be removed from office at any time, with or without cause, only by the affirmative vote of the holders of a majority of the voting power of all of the shares of capital stock of the Corporation then entitled to vote generally in the election of directors, voting as a single class.

ARTICLE VI

LIMITATION OF LIABILITY AND INDEMNIFICATION.

(A)

Limitation of Liability of Directors. A director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director except for liability (i) for any breach of the director's duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or knowing violation of law, (iii) under Section 174 of the DGCL, or (iv) for any transaction from which the director derived any improper personal benefit. If the DGCL is hereafter amended to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director shall be eliminated or limited to the fullest extent permitted by the DGCL, as so amended. No repeal or modification of this Article VI shall adversely affect any right of or protection afforded to a director prior to such repeal or modification.

Table of Contents

(B)

Indemnification. The Corporation shall, through the Bylaws or otherwise, to the fullest extent permitted by Section 145 of the DGCL, indemnify each person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that he is or was, or has agreed to become, a director or officer of the Corporation, or is or was serving, or has agreed to serve, at the request of the Corporation, as a director or officer of the Corporation, or is or was serving, or has agreed to serve, at the request of the Corporation, as director, officer or trustee of, or in a similar capacity with, another corporation, partnership, joint venture, trust or other enterprise (including any employee benefit plan), or by reason of any action alleged to have been taken or omitted in such capacity, against all expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by him or on his behalf in connection with such action, suit or proceeding and any appeal therefrom. Indemnification may include payment by the Corporation of expenses in defending an action or proceeding in advance of the final disposition of such action or proceeding upon receipt of an undertaking by the person indemnified to repay such payment if it is ultimately determined that such person is not entitled to indemnification under this Article VI, which undertaking may be accepted without reference to the financial ability of such person to make such repayment. The Corporation shall not indemnify any such person seeking indemnification in connection with a proceeding (or part thereof) initiated by such person unless the initiation thereof was approved by the Board of Directors of the Corporation. The indemnification rights provided in this Article VI (i) shall not be deemed exclusive of any other rights to which those indemnified may be entitled under any law, agreement or vote of stockholders or disinterested directors or otherwise, and (ii) shall inure to the benefit of the heirs, executors and administrators of such person. The Corporation may, to the extent authorized from time to time by its Board of Directors, grant indemnification rights to other employees or agents of the Corporation or to other persons serving the Corporation and such rights may be equivalent to, or greater or less than, those set forth in this Article VI. Any amendment, repeal or modification of this Article VI shall not adversely affect any rights or protection existing hereunder immediately prior to such amendment, repeal or modification.

ARTICLE VII

STOCKHOLDER ACTION

(A)

Action by Written Consent. Any action required or permitted to be taken by the stockholders of the Corporation must be effected at a duly called annual or special meeting of stockholders of the Corporation and may not be effected by any consent in writing by such stockholders; provided, however, that any action required or permitted to be taken by the holders of any series of Preferred Stock, voting separately as a series or separately as a class with one or more other such series, may be taken without a meeting, without prior notice and without a vote, to the extent expressly so provided by the applicable certificate of designation for such series of Preferred Stock.

(B)

Special Meetings. Except as otherwise required by law and subject to the rights of the holders of any class or series of Preferred Stock, (1) special meetings of stockholders of the Corporation may only be called by the Board of Directors pursuant to a resolution adopted by a majority of the entire Board of Directors, upon not less than ten (10) nor more than sixty (60) days' prior written notice, and (2) the stockholders of the Corporation shall have no right to call a special meeting of stockholders.

Table of Contents

ARTICLE VIII

AMENDMENT OF BYLAWS

- (A)

 Amendment by the Board of Directors. In furtherance, and not in limitation, of the powers conferred upon it by law, the Board of Directors is expressly authorized and empowered to amend, alter, change, adopt or repeal the Bylaws of the Corporation; provided, however, that no Bylaws hereafter adopted shall be inconsistent with the provisions of this Restated certificate or invalidate any prior act of the directors that would have been valid if such Bylaws had not been adopted. The Corporation may in its Bylaws confer powers upon its Board of Directors in addition to the foregoing and in addition to the powers and authorities expressly conferred upon the Board of Directors by applicable law.
- Amendment by Stockholders. In addition to any requirements of the DGCL (and notwithstanding the fact that a lesser percentage may be specified by the DGCL), unless otherwise specified in the Bylaws, the affirmative vote of the holders of a majority of all of the shares of capital stock of the Corporation then outstanding and entitled to vote generally in the election of directors, voting as a single class, shall be required for the stockholders of the Corporation to amend, alter, change, adopt or repeal any Bylaws of the Corporation. The holder of each share of capital stock entitled to vote thereon shall be entitled to cast the same number of votes as the holder of such shares is entitled to cast generally in the election of each director.

ARTICLE IX

FORUM SELECTION

Unless the Corporation consents in writing to the selection of an alternative forum (an "Alternative Forum Consent"), the Court of Chancery of the State of Delaware shall be the sole and exclusive forum for (i) any derivative action or proceeding brought on behalf of the Corporation, (ii) any action asserting a claim of breach of a fiduciary duty owed by any director, officer, stockholder, employee or agent of the Corporation to the Corporation or the Corporation's stockholders, (iii) any action asserting a claim against the Corporation or any director, officer, stockholder, employee or agent of the Corporation arising out of or relating to any provision of the DGCL or this Restated Certificate or the Corporation's Bylaws, or (iv) any action asserting a claim against the Corporation or any director, officer, stockholder, employee or agent of the Corporation governed by the internal affairs doctrine of the State of Delaware; provided, however, that, in the event that the Court of Chancery of the State of Delaware lacks subject matter jurisdiction over any such action or proceeding, the sole and exclusive forum for such action or proceeding shall be another state or federal court located within the State of Delaware, in each such case, unless the Court of Chancery (or such other state or federal court located within the State of Delaware, as applicable) has dismissed a prior action by the same plaintiff asserting the same claims because such court lacked personal jurisdiction over an indispensable party named as a defendant therein. Failure to enforce the foregoing provisions would cause the Corporation irreparable harm and the Corporation shall be entitled to equitable relief, including injunctive relief and specific performance, to enforce the foregoing provisions. Any person or entity purchasing or otherwise acquiring any interest in shares of capital stock of the Corporation shall be deemed to have notice of and consented to the provisions of this Article IX. The existence of any prior Alternative Forum Consent shall not act as a waiver of the Corporation's ongoing consent right as set forth above in this Article IX with respect to any current or future actions or claims.

Table of Contents

ARTICLE X

AMENDMENT OF CERTIFICATE OF INCORPORATION

The Corporation hereby reserves the right at any time and from time to time to amend, alter, change or repeal any provision contained in this Restated Certificate, and any other provisions authorized by the DGCL may be added or inserted, in the manner now or hereafter prescribed by the DGCL, and all rights, preferences and privileges of whatsoever nature conferred on stockholders, directors or any other persons whomsoever therein granted are subject to this reservation; *provided*, *however*, that, notwithstanding any other provision of this Restated Certificate or the Bylaws or any lesser percentage specified therein or in any law (but in addition to any other vote that may be required by applicable law or this Certificate), the affirmative vote of the holders of at least seventy-five percent (75%) of the voting power of the outstanding shares entitled to vote for the election of directors shall be required to amend, alter, change or repeal, or to adopt any provision as part of this Restated Certificate inconsistent with the intent and purpose of *Article IV*, *Article VI*, *Article VII*, *Article VIII*, *Article IX* of this Restated Certificate or this *Article X*.

This Restated Certificate was duly adopted by the Board of Directors in accordance with Section 245 of the DGCL.

Table of Contents

IN WITNESS WHEREOF, the undersigned has duly executed this Amended and Restated Certificate of Incorporation on this [•] day of [•], [•].				
[•]				
By:				
Name: Title: [Signature Page to Amended and Restated Certificate of Incorporation]				
D-7				

Annex E Form of Amended and Restated Bylaws of Holdco

AMENDED AND RESTATED BYLAWS OF [•]

Incorporated Under The Laws of Delaware EFFECTIVE AS OF [•], 2018

Table of Contents

TABLE OF CONTENTS

<u>ARTICL</u>	E I CAPITAL STOCK	<u>E-1</u>
<u>1.1</u>	Certificates	
		<u>E-1</u>
<u>1.2</u>	Shares Without Certificates	<u>E-1</u>
1.3	Record Ownership	<u>E-1</u>
<u>1.4</u>	Transfer of Record Ownership	<u>E-1</u>
1.5	Lost Certificates	<u>E-2</u>
1.6	Transfer Agents; Registrars; Rules Respecting Certificates	
1.7	Stock Transfer Agreement	<u>E-2</u>
1.8	Record Date	<u>E-2</u> E-2
	E II MEETINGS OF STOCKHOLDERS	<u>E-2</u>
AKTICL	E II MEETINGS OF STOCKHOLDERS	Е 2
0.1	DI CM (<u>E-3</u>
<u>2.1</u>	<u>Place of Meetings</u>	F. 2
		<u>E-3</u>
2.2	Annual Meeting	<u>E-3</u>
<u>2.3</u>	Special Meetings	<u>E-3</u>
<u>2.4</u>	<u>Notice</u>	<u>E-3</u>
<u>2.5</u>	<u>List of Stockholders</u>	<u>E-3</u>
<u>2.6</u>	<u>Quorum</u>	<u>E-4</u>
<u>2.7</u>	<u>Organization</u>	<u>E-4</u>
2.8	Voting	<u>E-4</u>
2.9	Inspectors of Election	<u>E-5</u>
2.10	Notification of Stockholder Nominations and Other Busin	
2.11	Proxy Access for Director Nominations	<u>E-10</u>
2.12	Adjourned Meeting; Notice	E-17
	E III BOARD OF DIRECTORS	<u>17 17</u>
AKTICL	E III BOARD OF DIRECTORS	E-18
2 1	N	<u>E-10</u>
3.1	Number and Qualifications	F 10
2.2	TI da im	<u>E-18</u>
<u>3.2</u>	Election and Term	<u>E-18</u>
<u>3.3</u>	Resignation and Removal	<u>E-18</u>
<u>3.4</u>	<u>Vacancies</u>	<u>E-19</u>
<u>3.5</u>	Regular Meetings	<u>E-19</u>
<u>3.6</u>	Special Meetings	<u>E-19</u>
<u>3.7</u>	Notice of Special Meetings	<u>E-19</u>
3.8	Place of Meetings	<u>E-19</u>
3.9	Participation in Meetings by Conference Telephone	E-19
3.10	Ouorum	E-19
3.11	<u>Organization</u>	<u>E-19</u>
3.12	Compensation of Directors	<u>E-20</u>
3.13	Action by Written Consent	E-20
	E IV COMMITTEES OF THE BOARD	<u>L-20</u>
AKTICL	ETV COMMITTEES OF THE BOARD	E 20
4 1	C . 10	<u>E-20</u>
<u>4.1</u>	Creation and Organization	F 20
		<u>E-20</u>
<u>4.2</u>	Audit Committee	<u>E-21</u>
<u>4.3</u>	Compensation Committee	<u>E-21</u>
4.4 4.5 4.6	Corporate Governance Committee	<u>E-21</u>
<u>4.5</u>	Powers Reserved to the Board	<u>E-21</u>
<u>4.6</u>	Meeting and Actions of Committees	<u>E-21</u>
<u>4.7</u>	<u>Minutes</u>	<u>E-22</u>
		E-i

Table of Contents

	V OFFICERS	<u>E-22</u>
<u>5.1</u>	<u>Designation of Officers</u>	F 22
5.2	Plastion and Torm	<u>E-22</u>
<u>5.2</u> <u>5.3</u>	Election and Term Resignation	<u>E-22</u> E-22
	Removal	<u>E-22</u> E-22
<u>5.4</u> <u>5.5</u>	Vacancies	<u>E-22</u> <u>E-22</u>
<u>5.5</u> <u>5.6</u>	<u>Vacancies</u> <u>Chief Executive Officer</u>	<u>E-22</u> <u>E-22</u>
<u>5.0</u> <u>5.7</u>	Executive Vice Presidents	<u>E-22</u> <u>E-23</u>
<u>5.7</u> <u>5.8</u>	Vice Presidents	E-23
<u>5.8</u> <u>5.9</u>	Chief Financial Officer	E-23
5.10	Treasurer	<u>E-23</u> <u>E-23</u>
<u>5.10</u> <u>5.11</u>	Secretary	<u>E-23</u> <u>E-23</u>
<u>5.11</u> <u>5.12</u>	General Counsel	<u>E-23</u>
<u>5.12</u> <u>5.13</u>	Subordinate Officers	E-23
<u>5.13</u> <u>5.14</u>	Representation of Shares of Other Corporations and Entities	<u>E-23</u> E-23
5.1 4 5.15	Authority and Duties of Officers	<u>E-23</u> <u>E-23</u>
<u>5.15</u> <u>5.16</u>	Execution of Corporate Contracts and Instruments	<u>E-23</u> <u>E-23</u>
<u>5.10</u> <u>5.17</u>	Compensation of Officers	<u>E-23</u> <u>E-24</u>
	VI STOCK LEDGER	<u>L-2-</u>
MITTELL	VISIOER EEDOER	E-24
ARTICLE	VII INDEMNIFICATION	<u>L-2-</u> T
MITTELL	VII INDEMINITERATION	<u>E-24</u>
<u>7.1</u>	Indemnification	<u>15-2-1</u>
7.1	<u>Indemnification</u>	E-24
<u>7.2</u>	Expenses Payable in Advance	E-25
<u>7.2</u> <u>7.3</u>	Judicial Determination of Indemnification or Advancement of Expenses	<u>E-25</u>
<u>7.3</u> <u>7.4</u>	Nonexclusivity	<u>E-25</u>
7.5	Insurance	<u>E-25</u>
7.6	Other Indemnification	<u>E-26</u>
7.7 7.7	Continuation of Indemnification	<u>E-26</u> E-26
<u>7.7</u> 7.8	Definitions	<u>E-26</u>
7.9	Survival	<u>E-26</u>
<u>7.10</u>	Repeal, Amendment or Modification	<u>E-26</u>
	VIII NOTICE BY ELECTRONIC SUBMISSION	<u>E 20</u>
MITTELL	THE TOTAL BY ELLEVINOUS SELFINGUICS	<u>E-27</u>
<u>8.1</u>	Notice by Electronic Submission	<u>557</u>
		E-27
8.2	Definition of Electronic Transmission	<u>E-27</u>
	IX MISCELLANEOUS	<u> </u>
	<u> </u>	E-27
<u>9.1</u>	Seal	<u>=</u>
		<u>E-27</u>
9.2	Fiscal Year	<u>E-28</u>
9.3	Waiver of Notice	E-28
9.4	Voting of Stock Owned by the Corporation	E-28
9.5	Forum for Adjudication of Certain Disputes	<u>E-28</u>
9.6	Registered Office	<u>E-29</u>
9.7	Executive Office	<u>E-29</u>
9.8	<u>Dividends</u>	<u>E-29</u>
	X AMENDMENT OF BYLAWS	
		<u>E-29</u>
	E-ii	

Table of Contents

ARTICLE I

CAPITAL STOCK

1.1 Certificates.

- (a) Shares of the capital stock of [], (the "Corporation") may be certificated or uncertificated in accordance with the General Corporation Law of the State of Delaware, as amended (the "DGCL"); provided that, commencing on or prior to the date of these Amended and Restated Bylaws of the Corporation (as the same may be amended from time to time, the "Bylaws"), the shares of common stock, par value \$0.01 per share, of the Corporation shall be uncertificated, as provided by resolutions adopted by the Board of Directors of the Corporation (the "Board of Directors" or "Board"). Certificates for the shares of stock, if any, shall be in such form as is consistent with the Amended and Restated Certificate of Incorporation of the Corporation (as the same may be amended from time to time, the "Certificate of Incorporation") and applicable law. To the extent any certificates are ever issued with respect to any class or series of a class of capital stock of the Corporation, every holder of stock represented by certificates shall be entitled to have a certificate, in such form as may be prescribed by law, the Certificate of Incorporation and the Board of Directors, signed in the name of the Corporation by any two (2) officers authorized to sign stock certificates representing the number of shares registered in certificate form held by such holder. The Chairman of the Board of Directors (the "Chairman"), the Chief Executive Officer, any Vice President, the Treasurer or an Assistant Treasurer and the Secretary or an Assistant Secretary of the Corporation shall be specifically authorized to sign stock certificates. Any or all the signatures on a certificate shall have ceased to be such officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar at the date of issue.
- (b) Each certificate for shares of stock that are subject to any restriction on transfer pursuant to the Certificate of Incorporation, the Bylaws, applicable securities laws or any agreement among any number of stockholders or among such holders and the Corporation shall have conspicuously noted on the face or back of the certificate either the full text of the restriction or a statement of the existence of such restriction.
- 1.2 Shares Without Certificates. Unless the Board of Directors determines that all of the shares of any class or series of stock of the Corporation shall be represented by certificates, the Corporation shall adopt a system of issuance, recordation and transfer of its shares of stock by electronic or other means not involving the issuance of certificates, provided the use of such system by the Corporation is permitted in accordance with applicable law.
- 1.3 Record Ownership. A record of the name and address of the holder of each certificate, the number of shares represented thereby and the date of issue thereof shall be made on the Corporation's books. The Corporation shall be entitled to treat the holder of record of any share of stock as the holder in fact thereof for all purposes, including the payment of dividends and the right to vote with respect to such stock, regardless of any transfer, pledge or other disposition of such stock, until such share has been transferred on the books of the Corporation in accordance with the requirements of these Bylaws. Accordingly, the Corporation shall not be bound to recognize any equitable or other claim to or interest in any share on the part of any other person, whether or not it shall have express or other notice thereof, except as required by the laws of the State of Delaware. If certificated, the certificates of each class or series of a class of stock shall be numbered consecutively.
- 1.4 *Transfer of Record Ownership*. Subject to applicable laws and these Bylaws, transfers of shares of stock of the Corporation shall be made on the books of the Corporation only by direction of the registered holder thereof or such person's attorney, lawfully constituted in writing, and, if such shares are represented by a certificate, only upon the surrender to the Corporation or its transfer agent

Table of Contents

or other designated agent of the certificate representing such shares properly endorsed or accompanied by a properly executed written assignment of, or power of attorney properly executed with respect to, the shares evidenced thereby, with transfer stamps (if necessary) affixed, and all other authentication reasonably required by the Corporation, which certificate shall be canceled before a new certificate or uncertificated shares are issued. No transfer of stock shall be valid as against the Corporation for any purpose until it shall have been entered in the stock records of the Corporation by an entry showing the names of the persons from and to whom it was transferred.

- 1.5 Lost Certificates. Any person claiming a stock certificate in lieu of one lost, stolen or destroyed shall give the Corporation an affidavit as to such person's ownership of the certificate and of the facts which go to prove its loss, theft or destruction, and the Corporation may issue a new certificate of stock or uncertified shares in place of any previously issued certificate alleged to have been lost, stolen or destroyed. Such person shall also, if required by policies adopted by the Board of Directors, give the Corporation a bond, in such form as may be approved by the General Counsel or his or her staff, sufficient to indemnify the Corporation against any claim that may be made against it on account of the alleged loss of the certificate or the issuance of a new certificate or of uncertificated shares.
- 1.6 Transfer Agents; Registrars; Rules Respecting Certificates. The Board of Directors may appoint, or authorize any officer or officers to appoint, one (1) or more transfer agents and one (1) or more registrars. The Board of Directors may make such further rules and regulations as it may deem expedient concerning the issue, transfer and registration of shares of stock of the Corporation.
- 1.7 Stock Transfer Agreement. The Corporation shall have power to enter into and perform any agreement with any number of stockholders of any one (1) or more classes or series of stock of the Corporation to restrict the transfer of shares of stock of the Corporation of any one (1) or more classes owned by such stockholders in any manner not prohibited by the DGCL.

1.8 Record Date.

- (a) The Board of Directors may fix in advance a date, not more than sixty (60) days or less than ten (10) days preceding the date of an annual or special meeting of stockholders and not more than sixty (60) days preceding the date of payment of a dividend or other distribution, allotment of rights or the date when any change, conversion or exchange of capital stock shall go into effect or for the purpose of any other lawful action, as the record date for determination of the stockholders entitled to notice of and to vote at any such meeting and any adjournment thereof, or to receive any such dividend or other distribution or allotment of rights, or to exercise the rights in respect of any such change, conversion or exchange of capital stock, or to participate in any such other lawful action. If the Board so fixes a date, such stockholders and only such stockholders as shall be stockholders of record on the date so fixed shall be entitled to such notice of and to vote at such meeting and any adjournment thereof, or to receive such dividend or other distribution or allotment of rights, or to exercise such rights, or to participate in any such other lawful action, as the case may be, notwithstanding any transfer of any stock on the books of the Corporation after any such record date fixed as aforesaid, unless the Board of Directors determines, at the time it fixes such record date, that a later date on or before the date of the meeting shall be the date for making such determination. If no record date is fixed, (i) the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day before the day on which notice is given, or, if notice is waived, at the close of business on the day before the day on which the meeting is held and (ii) the record date for determining stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights in respect of any change, conversion or exchange of stock, or for any other purpose, shall be at the close of business on the day on which the Board of Directors adopts the resolution relating to such purpose.
- (b) A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board of Directors may fix a new record date for the adjourned meeting and, in such case, shall also fix as the record date for stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote in accordance herewith at the adjourned meeting.

ARTICLE II

MEETINGS OF STOCKHOLDERS

- 2.1 Place of Meetings. All meetings of stockholders shall be held at such place within or outside the State of Delaware as may be designated from time to time by the Board of Directors, or, if not so designated, at the registered office of the Corporation. Notwithstanding the foregoing, the Board of Directors may, in its sole discretion, determine that a meeting of stockholders shall not be held at any place, but may instead be held solely by means of remote communication as authorized by Section 211(a)(2) of the DGCL.
- 2.2 Annual Meeting. The annual meeting of stockholders for the election of members of the Board of Directors (each, a "Director," and collectively, the "Directors") and the transaction of such other business as may properly be brought before the meeting shall be held annually on a date and at a time and place, within or outside Delaware, as determined by the Board of Directors.
 - 2.3 Special Meetings.
- (a) *Purpose.* Special meetings of stockholders for any purpose or purposes may only be called by the Board of Directors or Secretary of the Corporation upon direction of the Board, pursuant to a resolution adopted by a majority of the entire Board of Directors upon motion of a Director, upon not less than ten (10) nor more than sixty (60) days' written notice. The stockholders of the Corporation shall have no right to call a special meeting of stockholders.
- (b) Date, Time and Place. A special meeting shall be held at such date, time and place, within or outside the State of Delaware, as determined by the Board of Directors.
- (c) *Conduct of Meeting*. At any such special meeting, only such business may be transacted as is set forth in the notice of special meeting. The chairman of a special meeting shall determine all matters relating to the conduct of the meeting. Except in accordance with this Section 2.3, stockholders shall not be permitted to propose business to be brought before a special meeting of the stockholders. Stockholders who nominate persons for election to the Board at a special meeting must also comply with the requirements set forth in Section 2.10.
- 2.4 Notice. Unless otherwise provided by law, the Certificate of Incorporation or these Bylaws, notice (either written or as otherwise permitted by the DGCL) of each meeting of stockholders, whether annual or special, stating the date, time, place and the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting shall be distributed in advance of any such meeting. Additionally, the notice of a special meeting shall state the purpose thereof. Each notice shall be distributed (either by the U.S. Postal Service or as otherwise permitted by the DGCL) by the Secretary or Assistant Secretary not less than ten (10) days nor more than sixty (60) days before the date of such meeting to every stockholder entitled to vote thereat. If mailed, notice is given when deposited in the United States mail, postage prepaid, directed to the stockholder at his address as it appears on the records of the Corporation. If electronically transmitted, notice is given as provided in Section 8.1 of these Bylaws. An affidavit of the Secretary or an Assistant Secretary of the Corporation or of the transfer agent or any other agent of the Corporation that the notice has been given by mail or by a form of electronic transmission, as applicable, shall, in the absence of fraud, be prima facie evidence of the facts stated therein.
- 2.5 List of Stockholders. A complete list of the stockholders entitled to vote at any meeting of stockholders (provided, however, that if the record date for determining the stockholders entitled to vote is less than ten (10) days before the date of the meeting, the list shall reflect the stockholders entitled to vote as of the tenth (10th) day before the meeting date), arranged in alphabetical order, and showing the address of each stockholder and the number of shares registered in the name of each stockholder, shall be prepared by the Secretary at least ten (10) days before every meeting of

Table of Contents

stockholders and shall be open to the examination of any stockholder for any purpose germane to the meeting for a period of at least ten (10) days before the meeting during ordinary business hours at the principal place of business of the Corporation. A list of stockholders entitled to vote at the meeting shall be produced and kept at the place of the meeting during the whole time of the meeting and may be examined by any stockholder who is present. Except as otherwise provided by law, the stock ledger shall be the only evidence as to the identity of the stockholders entitled to examine the list of stockholders required by this Section 2.5 or to vote in person or by proxy at any meeting of stockholders.

- 2.6 Quorum. The holders of a majority of the voting power of all of the shares of capital stock of the Corporation then entitled to vote with respect to any one (1) of the purposes for which the meeting is called, present in person or represented by proxy, shall constitute a quorum, except as otherwise required by the DGCL, the Certificate of Incorporation or these Bylaws. In the event of a lack of quorum at a meeting, then (i) the chairman of the meeting or (ii) a majority in interest of the stockholders present in person or represented by proxy may adjourn the meeting from time to time without notice other than announcement at the meeting, until a quorum shall be obtained. At any such adjourned meeting at which there is a quorum, any business may be transacted that might have been transacted at the meeting originally called.
- 2.7 Organization. The Chairman, or, in the absence of the Chairman, the Chief Executive Officer, or, in the absence of both, any Executive Vice President, shall preside at meetings of stockholders as chairman of the meeting and shall determine the order of business for such meeting. The Secretary of the Corporation shall act as secretary at all meetings of stockholders, but in the absence of the Secretary, the chairman of the meeting may appoint a secretary of the meeting. The Board of Directors shall be entitled to make such rules or regulations for the conduct of meetings of stockholders as it shall deem necessary, appropriate or convenient. Subject to such rules and regulations of the Board of Directors, if any, the chairman of the meeting shall have the right and the authority to prescribe such rules, regulations and procedures and to do all such acts as, in the judgment of such chairman, are necessary, appropriate or convenient for the proper conduct of the meeting, including, without limitation, establishing (i) an agenda or order of business for the meeting, (ii) rules and procedures for maintaining order at the meeting and the safety of those present, (iii) limitations on participation in such meetings to stockholders of record of the Company and their duly authorized and constituted proxies, and such other persons as the chairman of the meeting shall permit, (iv) restrictions on entries to the meeting after the time affixed for the commencement thereof, (v) limitations on the time allotted to the questions or comments by participants and (vi) regulation of the opening and closing of the polls for balloting and matters which are to be voted on by ballot. Unless and to the extent determined by the Board of Directors or the chairman of the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure.

2.8 Voting; Proxies.

(a) Subject to all of the rights of the preferred stock provided for by resolution or resolutions of the Board of Directors pursuant to the Certificate of Incorporation or by the DGCL, each stockholder entitled to vote at a meeting shall be entitled to one vote, in person or by proxy (either written or as otherwise permitted by the DGCL), for each voting share held of record by such stockholder. The votes for the election of Directors and, upon the demand of any stockholder, the vote upon any matter before the meeting, shall be by written ballot. Except as otherwise required by the DGCL or as specifically provided for in the Certificate of Incorporation or these Bylaws, in any question or matter brought before any meeting of stockholders (other than the election of Directors), the affirmative vote of the holders of voting shares present in person or by proxy representing a plurality of the votes actually cast on any such question or matter at a meeting in which a quorum is present shall be the act of the stockholders. Directors shall be elected by the vote of a majority of the votes cast at a meeting

Table of Contents

in which a quorum is present; except that, notwithstanding the foregoing, Directors shall be elected by a plurality of the votes cast at a meeting where there is a quorum if, as of the record date for such meeting, the number of nominees exceeds the number of Directors to be elected. For purposes of the foregoing sentence, a majority of the votes cast means that the number of shares voted "for" a Director nominee must exceed the number of shares voted "against" that Director nominee.

- (b) Each stockholder entitled to vote at a meeting of stockholders may authorize another person or persons to act for such stockholder by proxy authorized by an instrument in writing executed by the stockholder or his authorized agent or by a transmission permitted by law filed in accordance with the procedure established for the meeting, but no such proxy shall be voted or acted upon after three (3) years from its date, unless the proxy provides for a longer period. The revocability of a proxy that states on its face that it is irrevocable shall be governed by the provisions of Section 212 of the DGCL. A proxy may be in the form of a telegram, cablegram or other means of electronic transmission which sets forth or is submitted with information from which it can be determined that the telegram, cablegram or other means of electronic transmission was authorized by the stockholder.
- 2.9 Inspectors of Election. In advance of any meeting of stockholders, the Board of Directors or the chairman of the meeting shall appoint one (1) or more inspectors to act at the meeting and make a written report thereof. The chairman of the meeting may designate one (1) or more persons as alternate inspectors to replace any inspector who fails or is unable to act. Each inspector, before entering upon the discharge of his or her duties, shall take and sign an oath faithfully to execute the duties of inspector with strict impartiality and according to the best of his or her ability. At each meeting of stockholders, the inspector(s) shall (i) ascertain the number of shares outstanding and the voting power of each, (ii) determine the shares represented at the meeting and the validity of proxies and ballots, (iii) count and tabulate all votes and ballots, (iv) determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the inspector(s), and (v) certify the inspectors' determination of the number of shares represented at the meeting and the count of all votes and ballots. The inspector(s) may appoint or retain other persons or entities to assist the inspector(s) in the performance of the duties of the inspector(s). Any report or certificate made by the inspector(s) shall be prima facie evidence of the facts stated therein.
 - 2.10 Notification of Stockholder Nominations and Other Business.

(a) Annual Meeting.

- (i) Nominations of persons for election to the Board of Directors and the proposal of business other than nominations to be considered by the stockholders may be made at an annual meeting of stockholders only (A) by or at the direction of the Board of Directors, including by any committee or persons authorized to do so by the Board, (B) by any stockholder of the Corporation who is a stockholder of record at the time the notice provided for in this Section 2.10 is delivered to, or mailed to and received by, the Secretary of the Corporation, who is entitled to vote at such annual meeting and who complies with the notice procedures and disclosure requirements set forth in this Section 2.10, or (C) in the case of stockholder nominations to be included in the Corporation's proxy statement for an annual meeting of stockholders, by an Eligible Stockholder (as defined below) who satisfies the notice, ownership and other requirements of Section 2.11 of these Bylaws. The foregoing clauses (B) and (C) shall be the exclusive means for a stockholder to make nominations of persons for election to the Board of Directors and the proposal of business other than nominations at an annual meeting.
- (ii) For nominations or other business to be properly brought before an annual meeting by a stockholder pursuant to clause (B) of Section 2.10(a)(i), such stockholder must have given timely written notice thereof in proper form to the Secretary of the Corporation and provide updates or supplements to such notice at the times and in the forms required by this Section 2.10, and such proposed business must be a proper subject for stockholder action. To be timely, a stockholder's

Table of Contents

notice must be delivered to, or mailed to and received by, the Secretary at the principal executive offices of the Corporation: not later than the close of business on the ninetieth (90th) day or earlier than the close of business on the one hundred twentieth (120th) day prior to the one-year anniversary of the preceding year's annual meeting; provided, however, that in the event that the annual meeting is called for a date that is not within thirty (30) days before or more than sixty (60) days after the first (1st) anniversary of the prior year's annual meeting, notice by the stockholder in order to be timely must be so delivered, or so mailed and received, not earlier than the close of business on the one hundred twentieth (120th) day prior to such annual meeting and not later than the close of business on the later of (A) the ninetieth (90th) day prior to such annual meeting and (B) the tenth (10th) day following the date on which public disclosure (as defined below) of the date of the annual meeting is first made by the Corporation. In no event shall the public disclosure of an adjournment or postponement of an annual meeting commence a new time period (or extend any notice time period) for the giving of a stockholder's notice as described above. Such stockholder's notice shall set forth:

(A)

as to each person, if any, whom such stockholder proposes to nominate for election or re-election as a Director: (1) all information relating to such person that would be required to be disclosed in a proxy statement soliciting proxies for the election of such nominee as a Director in an election contest (even if an election contest is not involved) or that is otherwise required to be disclosed, under Section 14(a) of the Securities Exchange Act of 1934, as amended (the "Exchange Act") and the rules and regulations promulgated thereunder, (2) the written consent of the nominee to being named in the proxy statement as a nominee (if applicable) and to serving as a Director if elected and a representation by the nominee to the effect that, if elected, the nominee (x) currently intends to serve as a director for the full term for which such person is standing for election and (y) will agree to and abide by all policies of the Board of Directors as may be in place at any time and from time to time, (3) any information that such person would be required to disclose pursuant to paragraph (ii)(C) of this Section 2.10, if such person were a stockholder purporting to make a nomination or propose business pursuant thereto, (4) a description of any direct or indirect material interest in any material contract or agreement between or among any stockholder nominating such person, on the one hand, and each candidate for nomination or his or her respective other participants in such solicitation, on the other hand, including, without limitation, all information that would be required to be disclosed pursuant to Item 404 under Regulation S-K if such stockholder making the nomination were the "registrant" for purposes of such rule and the candidate for nomination were a director or executive officer of such registrant, and (5) a completed and signed questionnaire, representation and agreement as provided in Section 2.10(a)(iii);

(B)

as to any other business that such stockholder proposes to bring before the meeting: (1) a brief description of the proposed business desired to be brought before the meeting, (2) the text of the proposal or proposed business (including the text of any resolutions proposed for consideration and, in the event that such business includes a proposal to amend the Bylaws of the Corporation, the language of the proposed amendment), (3) the reasons for conducting such business at the meeting, (4) any substantial interest (within the meaning of Item 5 of Schedule 14A under the Exchange Act) in such business of such stockholder and the beneficial owner (within the meaning of Section 13(d) of the Exchange Act), if any, on whose behalf the business is being proposed, (5) any other information relating to such stockholder and beneficial owner, if any, on whose behalf the proposal is being made, required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for the proposal and pursuant to and in accordance with Section 14(a) of the Exchange Act and the rules and

Table of Contents

regulations promulgated thereunder, and (6) a reasonably detailed description of all agreements, arrangements, or understandings between or among such stockholder, or any affiliates or associates of such stockholder, and any other person or persons (including their names) in connection with the proposal of such business and any material interest of such stockholder or any affiliates or associates of such stockholder, in such business, including any anticipated benefit therefrom to such stockholder, or any affiliates or associates of such stockholder; provided, however, that the disclosures required by this Section 2.10(a)(ii)(B) shall not include any disclosures with respect to any broker, dealer, commercial bank, trust company or other nominee who is a stockholder giving notice solely as a result of being the stockholder directed to prepare and submit the notice required by these Bylaws on behalf of a beneficial owner; and

(C)

as to the stockholder giving the notice and the beneficial owner, if any, on whose behalf the nomination is made or the other business is proposed: (1) the name and address of such stockholder, as they appear on the Corporation's books, and the name and address of such beneficial owner, if any, on whose behalf the nomination is made, (2) the class and number of shares of capital stock of the Corporation which are, directly or indirectly, owned (beneficially (within the meaning of Rule 13d-3 under the Exchange Act) and of record) by such stockholder and owned by the beneficial owner, if any, on whose behalf the nomination is being made, as of the date of such stockholder' notice, except that such stockholder giving notice and such beneficial owner shall in all events be deemed to beneficially own any shares of any class or series of stock of the Corporation as to which such person has a right to acquire beneficial ownership at any time in the future, (3) a reasonably detailed description of any agreement, arrangement or understanding with respect to the nomination or other business between or among such stockholder or beneficial owner and any other person, including without limitation any agreements that would be required to be disclosed pursuant to Item 5 or Item 6 of Exchange Act Schedule 13D (regardless of whether the requirement to file a Schedule 13D is applicable to the stockholder or beneficial owner) and a representation that the stockholder shall notify the Corporation in writing within five (5) business days after the record date for such meeting of any such agreement, arrangement or understanding in effect as of the record date for the meeting, (4) a reasonably detailed description of any agreement, arrangement or understanding (including any derivative or short positions, profit interests, options, hedging transactions, and borrowed or loaned shares) that has been entered into as of the date of such stockholder's notice by, or on behalf of, such stockholder or beneficial owner or any of their respective affiliates or associates, the effect or intent of which is to mitigate loss to, manage risk or benefit of share price changes for, or increase or decrease the voting power of such stockholder or beneficial owners or any of its affiliates or associates with respect to shares of stock of the Corporation, (5) a representation that such stockholder is a holder of record of shares of the Corporation entitled to vote at the meeting and intends to appear in person or by proxy at the meeting to nominate the person or persons or propose such other business, in each case as specified in the notice, (6) a representation whether such stockholder intends to deliver a proxy statement and/or form of proxy to holders of at least the percentage of the Corporation's outstanding capital stock required to approve the election of the nominee and/or otherwise to solicit proxies from stockholders in support of such election and (7) and, with respect to (2), (4) and (5) above, a representation that such stockholder will promptly notify the Corporation in writing of the same as of the record date for the meeting promptly following the later of the record date or the date notice of the record date is first publicly disclosed.

Table of Contents

- (iii) To be eligible to be a candidate for election as a Director of the Corporation at an annual or special meeting, a candidate must be nominated in the manner prescribed in this Section 2.10 and the candidate for nomination, if not nominated by the Board, must have previously delivered (in accordance with the time period prescribed for delivery in a notice to such candidate given by or on behalf of the Board), to the Secretary of the Corporation at the principal executive offices of the Corporation, (a) a completed written questionnaire (in the form provided by the Corporation) with respect to the background, qualifications, stock ownership and independence of such candidate for nomination and (b) a written representation and agreement (in the form provided by the Corporation) that such candidate for nomination (1) is not and, if elected as a Director during his or her term of office, will not become a party to (A) any agreement, arrangement or understanding with, and has not given and will not give any commitment or assurance to, any person or entity as to how such proposed nominee, if elected as a Director of the Corporation, will act or vote on any issue or question (a "Voting Commitment") or (B) any Voting Commitment that could limit or interfere with such proposed nominee's ability to comply, if elected as a Director of the Corporation, with such proposed nominee's fiduciary duties under applicable law, (2) is not, and will not become a party to, any agreement, arrangement or understanding with any person or entity other than the Corporation with respect to any direct or indirect compensation or reimbursement for service as a Director of the Corporation that has not been disclosed therein or to the Corporation and (3) if elected as a Director of the Corporation, will comply with all applicable corporate governance, conflict of interest, confidentiality, stock ownership and trading and other policies and guidelines of the Corporation applicable to Directors and in effect during such person's term in office as a Director (and, if requested by any candidate for nomination, the Secretary of the Corporation shall provide to such candidate for nomination all such policies and guidelines then in effect).
- (iv) The Corporation may require any proposed nominee to furnish such other information as may reasonably be required by the Corporation to determine the eligibility of such proposed nominee to serve as a Director of the Corporation, including information relevant to a determination whether such proposed nominee can be considered an independent Director or that could be material to a reasonable stockholders' understanding of the independence, or lack thereof.
- (v) A stockholder proposing a Director nominee or other such business as well as the candidate for nomination shall update and supplement its notice to the Corporation of its intent to propose business at an annual meeting, if necessary, so that the information provided or required to be provided in such notice pursuant to this Section 2.10 shall be true and correct as of the record date for stockholders entitled to vote at the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to, or mailed and received by, the Secretary of the Corporation at the principal executive offices of the Corporation not later than five (5) business days after the record date for stockholders entitled to vote at the meeting (in the case of the update and supplement required to be made as of such record date), and not later than eight (8) business days prior to the date for the meeting or, if practicable, any adjournment or postponement thereof (and, if not practicable, on the first practicable date prior to the date to which the meeting has been adjourned or postponed) (in the case of the update and supplement required to be made as of ten (10) business days prior to the meeting or any adjournment or postponement thereof).
- (vi) Notwithstanding anything in these Bylaws to the contrary, no business shall be conducted and no candidate shall be eligible for nomination at an annual meeting unless such business or nomination is properly brought before the meeting in accordance with this Section 2.10. The presiding officer of the meeting shall, if the facts warrant, determine that the business or

Table of Contents

nomination was not properly brought before the meeting in accordance with this Section 2.10, and if he or she should so determine, he or she shall so declare to the meeting and any such business or nomination not properly brought before the meeting shall not be transacted or considered.

- (vii) This Section 2.10(a) shall not apply to a proposal proposed to be made by a stockholder if the stockholder has notified the Corporation of his or her intention to present the proposal at an annual or special meeting only pursuant to and in compliance with Rule 14a-8 under the Exchange Act and such proposal has been included in a proxy statement that has been prepared by the Corporation to solicit proxies for such meeting. In addition to the requirements of this Section 2.10 with respect to any business proposed or Director nomination to be brought before an annual meeting, each proposing stockholder shall comply with all applicable requirements of the Exchange Act with respect to any such business. Nothing in this Section 2.10 shall be deemed to affect the rights of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act.
- (b) Special Meeting. Only such business shall be conducted at a special meeting of stockholders as shall have been brought before the meeting and specified pursuant to the Corporation's notice of meeting. Nominations of persons for election to the Board of Directors may be made at a special meeting of stockholders called by the Board of Directors at which Directors are to be elected pursuant to the Corporation's notice of meeting (i) by or at the direction of the Board of Directors or (ii) provided that the Board of Directors has determined that Directors shall be elected at such meeting, by any stockholder of the Corporation who is a stockholder of record at the time the notice provided for in this Section 2.10(b) is delivered to, or mailed to and received by, the Secretary of the Corporation and at the time of the special meeting, who is entitled to vote at the special meeting and upon such election, and who complies with the notice procedures set forth in this Section 2.10 as to such nomination, provided that the notice required by Section 2.10(a)(ii) shall be delivered to, or mailed to and received by, the Secretary at the principal executive offices of the Corporation not earlier than the close of business on the one hundred twentieth (120th) day prior to such special meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such special meeting or the tenth (10th) day following the day on which public disclosure of the date of the special meeting and of the nominees proposed by the Board of Directors to be elected at such meeting is first made by the Corporation. In no event shall the public announcement of an adjournment or postponement of a special meeting commence a new time period (or extend any time period) for the giving of a stockholder's notice as described above.

(c) General.

(i) Only such persons who are nominated in accordance with the procedures set forth in this Section 2.10 or Section 2.11 shall be eligible to be elected at any meeting of stockholders of the Corporation to serve as Directors and only such other business shall be conducted at a meeting of stockholders as shall have been properly brought before the meeting in accordance with the procedures set forth in this Section 2.10 or Section 2.11, as applicable. The chairman of the special meeting shall have the power and duty to determine whether a nomination or any other business proposed to be brought before the meeting was made or proposed, as the case may be, in accordance with the procedures set forth in this Section 2.10. If any proposed nomination or other business was not made or proposed in compliance with this Section 2.10 or Section 2.11, applicable, then except as otherwise provided by law, the chairman of the meeting shall have the power and duty to declare that such nomination shall be disregarded or that such proposed other business shall not be transacted. Notwithstanding the foregoing provisions of this Section 2.10, unless otherwise required by law, if the stockholder does not provide the information required under clauses (2)-(4) of Section 2.10(a)(ii)(C) to the Corporation within five (5) business days following the record date for an annual or special meeting of stockholders, or if the stockholder (or a qualified representative of the stockholder) does not appear at the annual or special meeting

Table of Contents

of stockholders of the Corporation to present a nomination or proposed other business, such nomination shall be disregarded and such proposed other business shall not be transacted, notwithstanding that proxies in respect of such vote may have been received by the Corporation. For purposes of this Section 2.10, to be considered a "qualified representative" of the stockholder, a person must be a duly authorized officer, manager or partner of such stockholder or authorized by a writing executed by such stockholder (or a reliable reproduction or electronic transmission of the writing) delivered to the Corporation prior to the making of such nomination or proposal at such meeting by such stockholder stating that such person is authorized to act for such stockholder as proxy at the meeting of stockholders.

(ii) For purposes of this Section 2.10, "public disclosure" shall mean disclosure in a press release reported by the Dow Jones News Service, Associated Press, or comparable national news service or any document publicly filed by the Corporation with the Securities and Exchange Commission (the "SEC") pursuant to Section 13, 14 or 15(d) of the Exchange Act. For purposes of clause (1) of Section 2.10(a)(ii)(C), shares shall be treated as "beneficially owned" by a person if the person beneficially owns such shares, directly or indirectly, for purposes of Section 13(d) of the Exchange Act and Regulations 13D and 13G thereunder or has or shares pursuant to any agreement, arrangement or understanding (whether or not in writing): (A) the right to acquire such shares (whether such right is exercisable immediately or only after the passage of time or the fulfillment of a condition or both), (B) the right to vote such shares, alone or in concert with others and/or (C) investment power with respect to such shares, including the power to dispose of, or to direct the disposition of, such shares.

2.11 Proxy Access for Director Nominations.

- (a) *Eligibility*. Subject to the terms and conditions of these Bylaws, in connection with an annual meeting of stockholders at which Directors are to be elected, the Corporation (a) shall include in its proxy statement and on its form of proxy the names of, and (b) shall (or, in the case of clause (C) of the definition of "Additional Information" (as defined below), may) include in its proxy statement the Additional Information relating to, an Authorized Number (defined below) of nominees specified pursuant to Section 2.11(b)(i) for election to the Board of Directors submitted pursuant to this Section 2.11 (each, a "*Stockholder Nominee*"), if:
 - (i) the Stockholder Nominee satisfies the eligibility requirements in this Section 2.11;
 - (ii) the Stockholder Nominee is identified in a timely notice (the "Stockholder Notice") that satisfies this Section 2.11 and is delivered by a stockholder that qualifies as, or is acting on behalf of, an Eligible Stockholder (as defined below);
 - (iii) the Eligible Stockholder satisfies the requirements in this Section 2.11 and expressly elects at the time of the delivery of the Stockholder Notice to have the Stockholder Nominee included in the Corporation's proxy materials; and
 - (iv) the additional requirements of these Bylaws are met.

(b) Definitions.

(i) The maximum number of Stockholder Nominees appearing in the Corporation's proxy materials with respect to an annual meeting of stockholders (the "Authorized Number") shall not exceed the greater of (x) two (2) and (y) twenty percent (20%) of the number of Directors in office as of the last day on which a Stockholder Notice may be delivered pursuant to this Section 2.11 with respect to the annual meeting, or if such amount is not a whole number, the closest whole number (rounding down) below twenty percent (20%); provided that the Authorized Number shall be reduced by any nominees who were previously elected to the Board as Stockholder Nominees at any of the preceding two (2) annual meetings and who are nominated

Table of Contents

for election at the annual meeting by the Board as a Board nominee. In the event that one (1) or more vacancies for any reason occurs after the date of the Stockholder Notice but before the annual meeting and the Board resolves to reduce the size of the Board in connection therewith, the Authorized Number shall be calculated based on the number of Directors in office as so reduced.

(ii) To qualify as an "Eligible Stockholder," a stockholder or a group as described in this Section 2.11 must:

(A)

(1)(x) Own and have Owned (as defined below), continuously for at least three (3) years as of the date of the Stockholder Notice, a number of shares (as adjusted to account for any stock dividend, stock split, subdivision, combination, reclassification or recapitalization of shares of the Corporation that are entitled to vote generally in the election of Directors) or (y) provide to the Secretary of the Corporation, within the time period referred to in Section 2.11(c)(ii), evidence of continuous ownership of such shares for such three (3) year period from one (1) or more securities intermediaries in a form that satisfies the requirements as established by the Commission for a stockholder proposal under Rule 14a-8 under the Exchange Act (or any successor rule) that represents at least three percent (3%) of the outstanding shares of the Corporation that are entitled to vote generally in the election of Directors as of the date of the Stockholder Notice (the "Required Shares"), and

(B) thereafter continue to Own the Required Shares through such annual meeting of stockholders.

For purposes of satisfying the ownership requirements of this Section 2.11(b)(ii), a group of not more than twenty (20) stockholders and/or beneficial owners may aggregate the number of shares of the Corporation that are entitled to vote generally in the election of Directors that each group member has individually Owned continuously for at least three (3) years as of the date of the Stockholder Notice if all other requirements and obligations for an Eligible Stockholder set forth in this Section 2.11 are satisfied by and as to each stockholder or beneficial owner comprising the group whose shares are aggregated. No shares may be attributed to more than one (1) Eligible Stockholder, and no stockholder or beneficial owner, alone or together with any of its affiliates, may individually or as a member of a group qualify as or constitute more than one (1) Eligible Stockholder under this Section 2.11. A group of any two (2) or more funds shall be treated as only one (1) stockholder or beneficial owner for this purpose if they are (A) under common management and investment control OR (B) under common management and funded primarily by a single employer OR (C) part of a family of funds, meaning a group of publicly offered investment companies (whether organized in the U.S. or outside the U.S.) that hold themselves out to investors as related companies for purposes of investment and investor services. For purposes of this Section 2.11, the term "affiliate" or "affiliates" shall have the meanings ascribed thereto under the rules and regulations promulgated under the Exchange Act. For the avoidance of doubt, in the event of a nomination by a group of Eligible Stockholders, any and all requirements and obligations for an individual Eligible Stockholder that are set forth in this Section 2.11, including the minimum holding period, shall apply to each member of such group; provided, however, that the Required Shares shall apply to the ownership of the group in the aggregate. Should any stockholder withdraw from a group of Eligible Stockholders at any time prior to the annual meeting of stockholders, the group of Eligible Stockholders shall only be deemed to own the shares held by the remaining members of the group.

(iii) For purposes of this Section 2.11:

(A)

A stockholder or beneficial owner is deemed to "Own" only those outstanding shares of the Corporation that are entitled to vote generally in the election of Directors as to which the person possesses both (1) the full voting and investment rights pertaining to the

Table of Contents

shares and (2) the full economic interest in (including the opportunity for profit and risk of loss on) such shares, except that the number of shares calculated in accordance with clauses (1) and (2) shall not include any shares (a) sold by such person in any transaction that has not been settled or closed, (b) borrowed by the person for any purposes or purchased by the person pursuant to an agreement to resell, or (c) subject to any option, warrant, forward contract, swap, contract of sale, or other derivative or similar agreement entered into by the person, whether the instrument or agreement is to be settled with shares or with cash based on the notional amount or value of outstanding shares of the Corporation that are entitled to vote generally in the election of Directors, if the instrument or agreement has, or is intended to have, or if exercised would have, the purpose or effect of: (x) reducing in any manner, to any extent or at any time in the future, the person's full right to vote or direct the voting of the shares, and/or (y) hedging, offsetting or altering to any degree any gain or loss arising from the full economic ownership of the shares by the person. The terms "Owned," "Owning" and other variations of the word "Own," when used with respect to a stockholder or beneficial owner, have correlative meanings. For purposes of clauses (a) through (c), the term "person" includes its affiliates.

- (B)

 A stockholder or beneficial owner "Owns" shares held in the name of a nominee or other intermediary so long as the person retains both (1) the full voting and investment rights pertaining to the shares and (2) the full economic interest in the shares. The person's Ownership of shares is deemed to continue during any period in which the person has delegated any voting power by means of a proxy, power of attorney, or other instrument or arrangement that is revocable at any time by the stockholder.
- A stockholder or beneficial owner's Ownership of shares shall be deemed to continue during any period in which the person has loaned the shares, provided that the person has the power to recall the loaned shares on not more than five (5) business days' notice and includes in the Stockholder Notice an agreement that it will (1) promptly recall such loaned shares upon being notified that any of its Stockholder Nominees will be included in the Corporation's proxy materials pursuant to this Section 2.11 and (2) continue to hold such recalled shares (including the right to vote such shares) through the date of the annual meeting of stockholders.
- (D)

 An Eligible Stockholder shall furnish any other information that may reasonably be required by the Board to verify such stockholder's continuous ownership of at least the Required Shares during the three (3) year period referred to above.
- (E)

 No person may be in more than one (1) group of Eligible Stockholders, and if any person appears as a member of more than one (1) group, it shall be deemed to be a member of the group that owns the greatest aggregate number of shares of the Corporation's common stock as reflected in the Stockholder Notice, and no shares may be attributed as owned by more than one (1) person constituting an Eligible Stockholder under this Section 2.11.
- (iv) For purposes of this Section 2.11, the "Additional Information" referred to in Section 2.11(a) that the Corporation will include in its proxy statement is:
 - (A)
 the information set forth in the Schedule 14N provided with the Stockholder Notice concerning each Stockholder
 Nominee and the Eligible Stockholder that is required to be disclosed in the Corporation's proxy statement by the
 applicable requirements of the Exchange Act and the rules and regulations thereunder;

Table of Contents

- (B) if the Eligible Stockholder so elects, a written statement of the Eligible Stockholder (or, in the case of a group, a written statement of the group), not to exceed five hundred (500) words, in support of its Stockholder Nominee(s), which must be provided at the same time as the Stockholder Notice for inclusion in the Corporation's proxy statement for the annual meeting (the "Statement"); and
- (C)
 any other information that the Corporation or the Board of Directors determines, in its discretion, to include in the proxy materials relating to the nomination of the Stockholder Nominee, including, without limitation, any statement in opposition to the nomination and any of the information provided pursuant to this Section 2.11.

Notwithstanding anything to the contrary contained in this Section 2.11, the Corporation may omit from its proxy materials all or any portion of any information or Statement, or may supplement or correct, any information or Statement, that (1) it, in good faith, believes is untrue in any material respect (or omits a material fact necessary in order to make the statements made, in light of the circumstances under which they are made, not misleading), (2)would violate the SEC's proxy rules or any applicable law, rule, regulation or listing standard or (3) would directly or indirectly impugn the character, integrity or personal reputation of, or, without factual foundation, directly or indirectly makes charges concerning improper, illegal or immoral conduct or associations with respect to, any person. Once submitted with a Stockholder Notice, a Statement of a Stockholder Nominee may not be amended, supplemented or modified by the Eligible Stockholder or Stockholder Nominee. Nothing in this Section 2.11 shall limit the Corporation's ability to solicit against and include in its proxy materials its own statements relating to any Eligible Stockholder or Stockholder Nominee.

- (c) Stockholder Notice and Other Informational Requirements.
 - (i) The Stockholder Notice shall set forth all information, representations and agreements required under Section 2.10(a)(ii) above, including the information required with respect to (i) any nominee for election as a Director, (ii) any stockholder giving notice of an intent to nominate a candidate for election, and (iii) any stockholder, beneficial owner or other person on whose behalf the nomination is made under this Section 2.11. In addition, such Stockholder Notice shall include:
 - (A) a copy of the Schedule 14N that has been or concurrently is filed with the SEC under the Exchange Act;
 - (B) the details of any relationship that existed within the past three (3) years and that would have been described pursuant to Item 6(e) of Schedule 14N (or any successor item) if it existed on the date of submission of the Schedule 14N;
 - (C)
 a representation and warranty that the Stockholder Nominee's candidacy or, if elected, Board membership, would
 not violate the Certificate of Incorporation, these Bylaws, or any applicable state or federal law or the rules of any
 stock exchange on which the Corporation's common stock is traded;
 - (D) if desired by the Eligible Stockholder, a Statement;
 - (E)

 a written statement of the Eligible Stockholder (and in the case of a group, the written statement of each stockholder or beneficial owner whose shares are aggregated for purposes of constituting an Eligible Stockholder), which statement(s) shall also be included in the Schedule 14N filed with the SEC, setting forth and certifying to the number of shares of the Corporation entitled to vote generally in the election of Directors that the Eligible Stockholder Owns and has Owned (as defined in Section 2.11(b)(iii) of these Bylaws) continuously for at least three (3) years as of the date of the Stockholder Notice, and agreeing to continue to Own such shares through the annual meeting;

Table of Contents

(F)
the written agreement of the Eligible Stockholder (and in the case of a group, the written agreement of each stockholder or beneficial owner whose shares are aggregated for purposes of constituting an Eligible Stockholder) addressed to the Corporation, setting forth the following additional agreements, representations and warranties:

it shall provide (a) within five (5) business days after the date of the Stockholder Notice, one (1) or more written statements from the record holder(s) of the Required Shares and from each intermediary through which the Required Shares are or have been held, in each case during the requisite three (3) year holding period, specifying the number of shares that the Eligible Stockholder Owns, and has Owned continuously in compliance with this Section 2.11, (b) within five (5) business days after the record date for the annual meeting both the information required under Section 2.10(a)(ii)(C) and notification in writing verifying the Eligible Stockholder's continuous Ownership of the Required Shares, in each case, as of such date, and (c) immediate notice to the Corporation if the Eligible Stockholder ceases to own any of the Required Shares prior to the annual meeting;

it (a) acquired the Required Shares in the ordinary course of business and not with the intent to change or influence control at the Corporation, and does not presently have this intent, (b) has not nominated and shall not nominate for election to the Board at the annual meeting any person other than the Stockholder Nominee(s) being nominated pursuant to this Section 2.11, (c) has not engaged and shall not engage in, and has not been and shall not be a participant (as defined in Item 4 of Exchange Act Schedule 14A) in, a solicitation within the meaning of Exchange Act Rule 14a-1(l), in support of the election of any individual as a Director at the annual meeting other than its Stockholder Nominee(s) or any nominee(s) of the Board, and (d) shall not distribute to any stockholder any form of proxy for the annual meeting other than the form distributed by the Corporation;

(3)
 a statement as to whether or not it intends to continue to Own the Required Shares for at least one
 (1) year following the annual meeting of stockholders (which statement may include an explanation as to why it is unable to make the foregoing statement);

it will (a) assume all liability stemming from any legal or regulatory violation arising out of the Eligible Stockholder's communications with the stockholders of the Corporation or out of the information that the Eligible Stockholder provided to the Corporation, (b) indemnify and hold harmless the Corporation and each of its Directors, officers and employees individually against any liability, loss or damages in connection with any threatened or pending action, suit or proceeding, whether legal, administrative or investigative, against the Corporation or any of its Directors, officers or employees arising out of the Eligible Stockholder's communications with the stockholders of the Corporation or out of the information that the Eligible Stockholder provided to the Corporation, (c) comply with all laws, rules, regulations and listing standards applicable to its nomination or any solicitation in connection with the annual meeting, (d) file with the SEC any solicitation or other communication by or on behalf of the Eligible Stockholder relating to the Corporation's annual meeting of stockholders, one (1) or more of the Corporation's Directors or Director nominees or any Stockholder Nominee, regardless of whether the filing is required under Exchange Act Regulation 14A, or whether any exemption from filing is available for the materials under Exchange Act Regulation 14A, and (e) at the request of the Corporation, promptly, but in any event within five (5) business days after such request (or by the day prior to the day of the annual

E-14

(4)

Table of Contents

meeting, if earlier), provide to the Corporation such additional information as reasonably requested by the Corporation; and

in the case of a nomination by a group, the designation by all group members of one (1) group member that is authorized to act on behalf of all members of the group with respect to the nomination and matters related thereto, including withdrawal of the nomination, and the written agreement, representation, and warranty of the Eligible Stockholder that it shall provide, within five (5) business days after the date of the Stockholder Notice, documentation reasonably satisfactory to the Corporation demonstrating that the number of stockholders and/or beneficial owners within such group does not exceed twenty, including whether a group of funds qualifies as one (1) stockholder or beneficial owner within the meaning of Section 2.11(b)(ii).

All information provided pursuant to this Section 2.11(c)(i) shall be deemed part of the Stockholder Notice for purposes of this Section 2.11.

- (ii) To be timely under this Section 2.11, the Stockholder Notice must be delivered to, or mailed to and received by, the Secretary at the principal executive offices of the Corporation not later than the close of business on the one hundred twentieth (120th) day or earlier than the close of business on the one hundred fiftieth (150th) day prior to the anniversary date on which the Corporation first distributed its definitive proxy materials for the prior year's annual meeting of stockholders; provided, however, that in the event that the annual meeting is called for a date that is not within thirty (30) days before or after the first (1st) anniversary of the prior year's annual meeting, notice by the stockholder in order to be timely, must be so delivered, or so mailed and received, not earlier than the close of business on the one hundred fiftieth (150th) day prior to such annual meeting and not later than the close of business on the later of the one hundred twentieth (120th) day prior to such annual meeting or the tenth (10th) day following the date on which public disclosure (as defined in Section 2.10(c)(ii) above) of the date of the annual meeting is first made by the Corporation. In no event shall the public disclosure of an adjournment or a postponement of an annual meeting commence a new time period (or extend any time period) for the giving of the Stockholder Notice as described above.
- (iii) Within the time period for delivery of the Stockholder Notice, a written representation and agreement of each Stockholder Nominee shall be delivered to the Secretary of the Corporation at the principal executive offices of the Corporation, which shall be signed by each Stockholder Nominee and shall represent and agree (A) as to the matters set forth in Section 2.10(a)(ii)(A), and (B) that such Stockholder Nominee consents to being named in the Corporation's proxy statement and form of proxy as a nominee. At the request of the Corporation, the Stockholder Nominee must promptly, but in any event within five (5) business days after such request, submit all completed and signed questionnaires required of the Corporation's nominees and provide to the Corporation such other information as it may reasonably request. The Corporation may request such additional information as necessary to permit the Board to determine if each Stockholder Nominee satisfies the requirements of this Section 2.11.
- (iv) In the event that any information or communications provided by the Eligible Stockholder or any Stockholder Nominees to the Corporation or its stockholders is not, when provided, or thereafter ceases to be, true, correct and complete in all material respects (including omitting a material fact necessary to make the statements made, in light of the circumstances under which they were made, not misleading), such Eligible Stockholder or Stockholder Nominee, as the case may be, shall promptly notify the Secretary and provide the information that is required to make such information or communication true, correct, complete and not misleading; it being understood that providing any such notification shall not be deemed to cure any defect or

Table of Contents

limit the Corporation's right to omit a Stockholder Nominee from its proxy materials as provided in this Section 2.11.

The information and documents required by this Section 2.11(c) shall be: (a) provided with respect to and executed by each group member, in the case of information applicable to group members; and (b) provided with respect to the persons specified in Instruction 1 to Items 6(c) and 6(d) of Schedule 14N (or any successor item) if and to the extent applicable to a Eligible Stockholder or group member. The Stockholder Notice shall be deemed submitted on the date on which all the information and documents referred to in this Section 2.11(c) (other than such information and documents contemplated to be provided after the date the Stockholder Notice is provided) have been delivered to or, if sent by mail, received by the Secretary of the Corporation. For the avoidance of doubt, in no event shall any adjournment or postponement of an annual meeting of stockholders or the public announcement thereof commence a new time period for the giving of a Stockholder Notice pursuant to this Section 2.11.

(d) Proxy Access Procedures.

- (i) Notwithstanding anything to the contrary contained in this Section 2.11, the Corporation may omit from its proxy materials any Stockholder Nominee, and such nomination shall be disregarded and no vote on such Stockholder Nominee shall occur, notwithstanding that proxies in respect of such vote may have been received by the Corporation, if:
 - (A)
 the Eligible Stockholder or Stockholder Nominee breaches any of its agreements, representations or warranties set forth in the Stockholder Notice or otherwise submitted pursuant to this Section 2.11, any of the information in the Stockholder Notice or otherwise submitted pursuant to this Section 2.11 was not, when provided, true, correct and complete, or the Eligible Stockholder or applicable Stockholder Nominee otherwise fails to comply with its obligations pursuant to these Bylaws, including, but not limited to, its obligations under this Section 2.11;
 - the Stockholder Nominee (1) is not independent under any applicable listing standards, any applicable rules of the SEC and any publicly disclosed standards used by the Board in determining and disclosing the independence of the Corporation's Directors, (2) is or has been, within the past three (3) years, an officer or Director of a competitor, as defined in the Clayton Antitrust Act, (3) is a named subject of a pending criminal proceeding (excluding traffic violations and other minor offenses) or has been convicted in a criminal proceeding (excluding traffic violations and other minor offenses) within the past ten (10) years or (4) is subject to any order of the type specified in Rule 506(d) of Regulation D promulgated under the Securities Act;
 - (C) the Corporation has received a notice (whether or not subsequently withdrawn) that a stockholder intends to nominate any candidate for election to the Board pursuant to the advance notice requirements for stockholder nominees for Director in Section 2.10(a); or
 - (D)
 the election of the Stockholder Nominee to the Board would cause the Corporation to violate the Certificate of Incorporation, these Bylaws, or any applicable law, rule, regulation or listing standard.
- (ii) An Eligible Stockholder submitting more than one (1) Stockholder Nominee for inclusion in the Corporation's proxy materials pursuant to this Section 2.11 shall rank such Stockholder Nominees based on the order that the Eligible Stockholder desires such Stockholder Nominees to be selected for inclusion in the Corporation's proxy materials and include such assigned rank in its Stockholder Notice submitted to the Corporation. In the event that the number of Stockholder Nominees submitted by Eligible Stockholders pursuant to this Section 2.11 exceeds the Authorized Number, the Stockholder Nominees to be included in the Corporation's proxy materials shall be

Table of Contents

determined in accordance with the following provisions: one (1) Stockholder Nominee who satisfies the eligibility requirements in this Section 2.11 shall be selected from each Eligible Stockholder for inclusion in the Corporation's proxy materials until the Authorized Number is reached, going in order of the amount (largest to smallest) of shares of the Corporation each Eligible Stockholder disclosed as Owned in its Stockholder Notice submitted to the Corporation and going in the order of the rank (highest to lowest) assigned to each Stockholder Nominee by such Eligible Stockholder. If the Authorized Number is not reached after one (1) Stockholder Nominee who satisfies the eligibility requirements in this Section 2.11 has been selected from each Eligible Stockholder, this selection process shall continue as many times as necessary, following the same order each time, until the Authorized Number is reached. Following such determination, if any Stockholder Nominee who satisfies the eligibility requirements in this Section 2.11 thereafter is nominated by the Board, thereafter is not included in the Corporation's proxy materials or thereafter is not submitted for Director election for any reason (including the Eligible Stockholder's or Stockholder Nominee's failure to comply with this Section 2.11), no other nominee or nominees shall be included in the Corporation's proxy materials or otherwise submitted for election as a Director at the applicable annual meeting in substitution for such Stockholder Nominee.

- (iii) Any Stockholder Nominee who is included in the Corporation's proxy materials for a particular annual meeting of stockholders but (x) withdraws from or becomes ineligible or unavailable for election at the annual meeting for any reason, including for the failure to comply with any provision of these Bylaws (provided that in no event shall any such withdrawal, ineligibility or unavailability commence a new time period (or extend any time period) for the giving of a Stockholder Notice) or (y) does not receive at least twenty-five percent (25%) of the votes cast in favor of such Stockholder Nominee's election shall be ineligible to be a Stockholder Nominee pursuant to this Section 2.10 for the next two (2) annual meetings.
- (iv) Notwithstanding the foregoing provisions of this Section 2.11, unless otherwise required by law or otherwise determined by the chairman of the meeting or the Board, if the stockholder delivering the Stockholder Notice (or a qualified representative of the stockholder, as defined in Section 2.10(c)(i)) does not appear at the annual meeting of stockholders of the Corporation to present its Stockholder Nominee or Stockholder Nominees, such nomination or nominations shall be disregarded, notwithstanding that proxies in respect of the election of the Stockholder Nominee or Stockholder Nominees may have been received by the Corporation.
- (v) The Board of Directors (and any other person or body authorized by the Board of Directors) shall have the power and authority to interpret this Section 2.11 and to make any and all determinations necessary or advisable to apply this Section 2.11 to any persons, facts or circumstances, including, without limitation, the power to determine (1) whether one (1) or more stockholders or beneficial owners qualifies as an Eligible Stockholder, (2) whether a Stockholder Notice complies with this Section 2.11 and has otherwise met the requirements of this Section 2.11, (3) whether a Stockholder Nominee satisfies the qualifications and requirements in this Section 2.11, and (4) whether any and all requirements of this Section 2.11 (or any applicable requirements of Section 2.10) have been satisfied. Any such interpretation or determination adopted in good faith by the Board of Directors (or any other person or body authorized by the Board of Directors) shall be binding on all persons, including, without limitation, the Corporation and its stockholders (including, without limitation, any beneficial owners).
- (vi) This Section 2.11 shall be the exclusive method for stockholders to include Director nominees for election in the Corporation's proxy materials.
- 2.12 Adjourned Meeting; Notice. When a meeting is adjourned to another time or place, unless these Bylaws otherwise require, notice need not be given of the adjourned meeting if the time, place, if

Table of Contents

any, thereof, and the means of remote communications, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such adjourned meeting are announced at the meeting at which the adjournment is taken. At any adjourned meeting, the Corporation may transact any business that might have been transacted at the original meeting. If the adjournment is for more than thirty (30) days, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. If after the adjournment a new record date for determination of stockholders entitled to vote is fixed for the adjourned meeting, the Board of Directors shall fix as the record date for determining stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote at the adjourned meeting, and shall give notice of the adjourned meeting to each stockholder of record as of the record date so fixed for notice of such adjourned meeting.

ARTICLE III

BOARD OF DIRECTORS

- 3.1 Number and Qualifications. The business and affairs of the Corporation shall be managed by or under the direction of its Board of Directors, who may exercise all of the powers of the Corporation except as otherwise provided by the DGCL or the Certificate of Incorporation. The number of Directors constituting the entire Board of Directors shall be not less than six (6) nor more than twelve (12), as fixed from time to time exclusively by resolution of a majority of the entire Board of Directors. As used in these Bylaws, the term "entire Board of Directors" means the total authorized number of Directors that the Corporation would have if there were no vacancies. In the event of a vacancy in the Board of Directors, the remaining Directors, except as otherwise provided by the DGCL or the Certificate of Incorporation, may exercise the powers of entire full Board of Directors until the vacancy is filled.
- 3.2 Election and Term. The Directors shall be elected at the annual meeting of stockholders by such stockholders as have the right to vote on such election. Directors need not be stockholders of the Corporation. Subject to any rights of holders of preferred stock to elect Directors, each Director shall hold office until the next annual meeting for the election of Directors and until the Director's successor is duly elected and qualified, or until his earlier death, resignation or removal. No reduction of the authorized number of Directors shall have the effect of removing any Director before that Director's term of office expires. The Certificate of Incorporation or these Bylaws may prescribe other qualifications for Directors.
- 3.3 Resignation and Removal. A Director may resign at any time by giving written notice or electronic notice to the Corporation at its principal office or to the Chairman, the Chief Executive Officer or the Secretary. Unless otherwise stated in such notice of resignation, the acceptance thereof shall not be necessary to make it effective, and such resignation shall take effect at the time or upon the happening of an event specified therein or, in the absence of such specification, it shall take effect upon the receipt thereof. When one (1) or more Directors so resigns and the resignation is effective at a future date or upon the happening of an event to occur on a future date, a majority of the Directors then in office, including those who have so resigned, shall have power to fill such vacancy or vacancies, the vote thereon to take effect when such resignation or resignations shall become effective, and each Director so chosen shall hold office as provided in this Section 3.3 in the filling of other vacancies. Except as otherwise required by law and subject to the rights of the holders of any class or series of Preferred Stock then outstanding, any Director, or the entire Board of Directors, may be removed from office at any time, with or without cause, only by the affirmative vote of the holders of a majority of the voting power of all of the shares of capital stock of the Corporation then entitled to vote generally in the election of Directors, voting as a single class.

Table of Contents

- 3.4 Vacancies. Subject to the provisions of the Certificate of Incorporation and the rights of the holders of any class or series of preferred stock to elect Directors, any vacancies on the Board of Directors for any reason, including from the death, resignation, disqualification or removal of any Director, and any newly created directorships resulting by reason of any increase in the number of Directors shall be filled exclusively by the Board of Directors, acting by the affirmative vote of a majority of the remaining Directors then in office, even if less than a quorum, or by a sole remaining Director, and shall not be filled by stockholders. Any Directors elected to fill a vacancy shall hold office until the next annual meeting of stockholders at which term of the Directors for which he has been chosen expires or until their successors are duly elected and qualified, or until his earlier death, resignation or removal. A vacancy in the Board of Directors shall be deemed to exist under these Bylaws in the case of the death, removal or resignation of any Director.
- 3.5 Regular Meetings. Regular meetings of the Board of Directors may be held without further notice on such date and at such time and place as shall from time to time be determined by the Board of Directors. A meeting of the Board of Directors for the election of officers and the transaction of such other business as may come before it may be held without notice immediately following the annual meeting of stockholders.
- 3.6 Special Meetings. Special meetings of the Board of Directors may be called by the Chairman, the Chief Executive Officer or at the request in writing or by the affirmative vote of a majority of the Directors then in office.
- 3.7 Notice of Special Meetings. Notice of the time and place of each special meeting shall be (i) mailed to each Director at least four (4) days before the meeting, or (ii) telegraphed, telecopied or electronically transmitted or delivered personally or by telephone to such Director at least one (1) day before the meeting but such notice may be waived by such Director. Each notice shall be directed to each Director at that Director's address, telephone number, facsimile number or electronic mail address, or other address for electronic transmission, as the case may be, as shown on the Corporation's records. The notice need not state the purposes of the special meeting and, unless indicated in the notice thereof, any and all business may be transacted at a special meeting.
- 3.8 Place of Meetings. The Directors may hold their meetings and have an office or offices within or outside of Delaware as the Board of Directors may from time to time determine.
- 3.9 Participation in Meetings by Conference Telephone. Unless otherwise restricted by the Certificate of Incorporation, members of the Board of Directors, or of any committee thereof, may participate in a meeting of the Board of Directors or committee by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other, and such participation shall constitute presence in person at the meeting.
- 3.10 *Quorum*. At all meetings of the Board of Directors, a majority of the total number of Directors then holding office shall constitute a quorum for the transaction of business. The vote of a majority of the Directors present at any meeting at which a quorum is present shall be the act of the Board, except as may be otherwise specifically provided by statute, the Certificate of Incorporation or these Bylaws. A meeting at which a quorum is initially present may continue to transact business notwithstanding the withdrawal of Directors, if any action taken is approved by at least a majority of the required quorum for that meeting. In the event of lack of a quorum, a majority of the Directors present may adjourn the meeting from time to time without notice, other than announcement at the meeting, until a quorum shall be obtained.
- 3.11 *Organization.* The Chairman, or, in the absence of the Chairman, the Chief Executive Officer, or, in the absence of the Chairman and the Chief Executive Officer, a member of the Board selected by the members present, shall preside at meetings of the Board. The Secretary or an Assistant

Table of Contents

Secretary of the Corporation shall act as secretary, but in the absence of the Secretary or an Assistant Secretary, the presiding officer may appoint a secretary.

- 3.12 *Compensation of Directors.* Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, Directors shall receive such compensation for their services on the Board of Directors and any committee thereof and such reimbursement for their expenses of attending meetings of the Board of Directors and any committee thereof as the Board of Directors may determine from time to time.
- 3.13 Action by Written Consent. Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting, if all members of the Board or committee, as the case may be, consent to the action in writing or by electronic transmission, and the writing or writings or electronic transmission or transmissions are filed with the minutes of proceedings of the Board of Directors or committee thereof. Such filing shall be in paper form if the minutes are maintained in paper form and shall be in electronic form if the minutes are maintained in electronic

ARTICLE IV

COMMITTEES OF THE BOARD

4.1 Creation and Organization. The standing committees of the Board of Directors shall be an Audit Committee, a Compensation Committee and a Corporate Governance Committee, having the respective powers and duties assigned to each in this Article IV and any other powers and duties assigned to such committee by resolution passed by a majority of the entire Board of Directors from time to time. Except as specified herein, each such standing committee shall consist of one (1) or more Directors and such other ex officio members as the Board of Directors shall from time to time determine. The chairman of each standing committee shall be one (1) or more of such committee's members who shall be designated as that committee's chairman by a majority vote of the entire Board of Directors. Members of each standing committee shall be elected by a majority vote of the entire Board of Directors. Vacancies in any standing committee shall be filled by a majority vote of the entire Board of Directors. The Board of Directors may appoint management employees of the Corporation or its subsidiaries to be ex officio members of any standing committee. Ex officio members of standing committees shall be entitled to be present at all meetings of their respective committees and to participate in committee discussions, but shall not be entitled to vote or be counted for quorum purposes. The Board of Directors may designate one (1) or more Directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee. In the absence or disqualification of a member of a committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not such member or members constitute a quorum, may unanimously appoint another member of the Board of Directors to act at the meeting in the place of any such absent or disqualified member. Each standing committee may also establish subcommittees thereof, with such power and authority assigned to such subcommittee as may be provided for in the charter establishing such subcommittee. Each standing committee and subcommittee shall fix its own rules of procedure and shall meet where and as provided by such rules, but the presence of a majority of its members shall be necessary to constitute a quorum. The Board of Directors may from time to time designate one (1) or more additional committees or special committees with such powers and such members as it may designate in a resolution or resolutions adopted by a majority of the entire Board of Directors.

Table of Contents

- 4.2 *Audit Committee.* The Audit Committee shall have the sole authority to appoint or replace the Corporation's independent auditors, subject to shareholder ratification at each annual meeting. The Audit Committee shall assist the Board in monitoring:
 - (a) the integrity of the financial statements of the Corporation;
 - (b) the independent auditor's qualifications, independence and performance;
 - (c) the performance of the Corporation's internal controls and audit function;
 - (d) the application of the Corporation's accounting principles; and
 - (e) the compliance by the Corporation with legal and regulatory requirements.

The Audit Committee shall prepare the report required by the rules of the SEC to be included in the Corporation's annual meeting proxy statement.

- 4.3 Compensation Committee. The Compensation Committee shall discharge the Board's responsibilities relating to the total compensation of the Corporation's Chief Executive Officer as well as all other senior executives in a manner consistent with and in support of the business objectives of the Corporation, competitive practice and all applicable rules and regulations.
- 4.4 Corporate Governance Committee. The Corporate Governance Committee shall consider and report periodically to the Board of Directors on all matters relating to the selection, qualification and compensation of members of the Board and candidates nominated to the Board, as well as any other matters relating to the duties of the members of the Board. The Corporate Governance Committee shall act as a nominating committee with respect to candidates for Directors and will make recommendations to the full Board concerning the size of the Board and structure of committees of the Board. The Corporate Governance Committee shall also assist the Board with oversight of corporate governance matters.
 - 4.5 Powers Reserved to the Board. No committee of the Board of Directors shall have the power or authority to:
- (a) approve or adopt, or recommend to stockholders, any action or matter expressly required by the DGCL to be submitted to stockholders for approval; or
 - (b) adopt, amend or repeal these Bylaws.

No committee of the Board of Directors shall take any action that is required by these Bylaws, the Certificate of Incorporation or the DGCL to be taken by a vote of a specified proportion of the entire Board of Directors.

- 4.6 *Meeting and Actions of Committees.* Meeting and actions of committees shall be governed by, and held and taken in accordance with the provisions of:
 - (a) Section 3.5 (regular meetings);
 - (b) Section 3.6 (special meetings);
 - (c) Section 3.7 (notice of special meetings);
 - (d) Section 3.8 (place of special meetings);
 - (e) Section 3.9 (participation in meetings by conference telephone);

- (f) Section 3.10 (quorum);
- (g) Section 3.13 (action by written consent); and
- (h) Section 9.3 (waiver of notice),

Table of Contents

with such changes in the context of those Bylaws as are necessary to substitute the committee and its members for the Board and its members. *However*,

- (i)
 regular meetings of committees may be held without notice at such time and at such place as shall from time to time be determined by the applicable committee, or otherwise by either resolution of the Board or resolution of the applicable committee;
- (ii) special meetings of committees may also be called by resolution of the Board or the chairman of the applicable committee; and
- (iii) the Board may adopt rules for the governance of any committee to override the provisions that would otherwise apply to the committee pursuant to this Section 4.6, provided that such rules do not violate the provisions of the Certificate of Incorporation or applicable law.
- 4.7 Minutes. Each committee shall keep regular minutes of its meetings and report the same to the Board of Directors when required.

ARTICLE V

OFFICERS

- 5.1 Designation of Officers. The officers of the Corporation appointed by the Board of Directors shall be a Chief Executive Officer, a Chief Financial Officer, a General Counsel, and may also include one (1) or more Executive Vice Presidents, one (1) or more Vice Presidents, one (1) or more Chief Operating Officers, a Secretary, a Treasurer and a Controller. The Board of Directors also may elect or appoint, or provide for the appointment of, and, if delegated to the Chief Executive Officer, the Chief Executive Officer also may elect or appoint, such other officers or assistant officers (including, without limitation, one (1) or more Assistant Treasurers, one (1) or more Assistant Secretaries and one (1) or more Assistant Controllers) as may from time to time appear necessary or advisable in the conduct of the business and affairs of the Corporation. Any number of offices may be held by the same person.
- 5.2 Election and Term. At its first (1st) meeting after each annual meeting of stockholders, the Board of Directors shall elect the officers. The term of each officer shall be until the first (1st) meeting of the Board of Directors following the next annual meeting of stockholders and until such officer's successor is chosen and qualified, unless a different term is specified in the resolution electing or appointing such officer, or until such person's earlier death, disqualification or removal.
- 5.3 Resignation. Any officer may resign at any time by giving written notice or electronic notice to the Corporation at its principal office or to the Chief Executive Officer or the Secretary. Unless otherwise stated in such notice of resignation, the acceptance thereof shall not be necessary to make it effective, and such resignation shall take effect at the time specified therein or, in the absence of such specification, it shall take effect upon the receipt thereof. Any resignation is without prejudice to the rights, if any, of the Corporation under any contract to which the officer is a party.
- 5.4 *Removal.* Except where otherwise expressly provided in a contract authorized by the Board of Directors, any officer elected or appointed by the Board of Directors may be removed at any time with or without cause by the affirmative vote of a majority of the entire Board of Directors.
- 5.5 *Vacancies*. A vacancy in any office for any reason may be filled for the unexpired portion of the term by resolution of the Board of Directors. The Board of Directors may, in its discretion, leave unfilled for such period of time as it may determine, any offices.

5.6 *Chief Executive Officer*. The Chief Executive Officer shall report to the Board of Directors and shall, subject to the Board's supervision, have the general supervision, direction and control over the business of the Corporation. The Chief Executive Officer shall have such other powers and perform such other duties as may be assigned by the Board of Directors from time to time.

Table of Contents

- 5.7 Executive Vice Presidents. The Executive Vice Presidents shall assist the Chief Executive Officer in the management of the business and affairs of the Corporation and shall perform such other duties as may be assigned by the Chief Executive Officer or the Board of Directors.
- 5.8 *Vice Presidents.* Each Vice President shall have such powers and perform such duties as may be assigned by the Chief Executive Officer or the Board of Directors. The Board of Directors may designate one (1) or more Vice Presidents as Senior Vice Presidents.
- 5.9 *Chief Financial Officer*. The Chief Financial Officer shall be the principal financial officer of the Corporation and shall have such powers and perform such duties as may be assigned by the Board of Directors or the Chief Executive Officer.
- 5.10 *Treasurer*. The Treasurer shall have charge of all funds of the Corporation and shall perform all acts incident to the position of Treasurer, subject to the control of the Board of Directors.
- 5.11 Secretary. The Secretary or an Assistant Secretary shall keep the minutes and give notices of all meetings of stockholders and Directors and of such committees as directed by the Board of Directors. The Secretary shall have charge of such books and papers as the Board of Directors may require. The Secretary or any Assistant Secretary is authorized to certify copies of extracts from minutes and of documents in the Secretary's charge, and anyone may rely on such certified copies to the same effect as if such copies were originals and may rely upon any statement of fact concerning the Corporation certified by the Secretary or any Assistant Secretary. The Secretary shall perform all acts incident to the office of Secretary, subject to the control of the Board of Directors.
- 5.12 *General Counsel*. The General Counsel shall be in charge of all matters concerning the Corporation involving litigation or legal counseling. The General Counsel shall have such other powers and perform such other duties as may be assigned by the Board of Directors and shall submit such reports to the Board of Directors as it may request.
- 5.13 Subordinate Officers. The Board of Directors may appoint, or empower the Chief Executive Officer or, in the absence of a Chief Executive Officer, the President, to appoint, such other officers and agents as the business of the Corporation may require. Each of such officers and agents shall hold office for such period, have such authority, and perform such duties as are provided in these Bylaws or as the Board may from time to time determine.
- 5.14 Representation of Shares of Other Corporations and Entities. The Chairman, the Chief Executive Officer, the President, any Vice President, the Treasurer, the Secretary or Assistant Secretary of the Corporation, or any other person authorized by the Board, the Chief Executive Officer, the President or a Vice President, is authorized to vote, represent and exercise on behalf of the Corporation all rights incident to any and all shares of any other corporation or corporations or other entity or entities standing in the name of the Corporation. The authority granted herein may be exercised either by such person directly or by any other person authorized to do so by proxy or power of attorney duly executed by such person having the authority.
- 5.15 Authority and Duties of Officers. All officers of the Corporation shall respectively have such authority and perform such duties in the management of the business of the Corporation as may be provided herein or designated from time to time by the Board of Directors and, to the extent not so provided, as generally pertain to their respective offices, subject to the control of the Board of Directors.
- 5.16 Execution of Corporate Contracts and Instruments. The Board of Directors, except as otherwise provided in these Bylaws, may authorize any officer or officers, or agent or agents, to enter into any contract or execute any instrument in the name of and on behalf of the Corporation; such authority may be general or confined to specific instances. Unless so authorized or ratified by the Board of Directors or within the agency power of an officer, no officer, agent or employee shall have any power or authority to bind the Corporation by any contract or engagement or to pledge its credit or to render it liable for any purpose or for any amount.

Table of Contents

5.17 *Compensation of Officers.* The officers of the Corporation shall receive such compensation for their services as the Compensation Committee may determine.

ARTICLE VI

STOCK LEDGER

A stock ledger consisting of one (1) or more records in which the names of all of the Corporation's stockholders of record, the address and number of shares registered in the name of each such stockholder, and all issuances and transfers of stock of the corporation are recorded in accordance with Section 224 of the DGCL shall be administered by or on behalf of the Corporation. Any records administered by or on behalf of the Corporation in the regular course of its business, including its stock ledger, books of account, and minute books, may be kept on, or by means of, or be in the form of, any information storage device, or method, or one (1) or more electronic networks or databases (including one (1) or more distributed electronic networks or databases), provided that the records so kept can be converted into clearly legible paper form within a reasonable time and, with respect to the stock ledger, that the records so kept (i) can be used to prepare the list of stockholders specified in Sections 219 and 220 of the DGCL, (ii) record the information specified in Sections 156, 159, 217(a) and 218 of the DGCL, and (iii) record transfers of stock as governed by Article 8 of the Uniform Commercial Code.

ARTICLE VII

INDEMNIFICATION

- 7.1 *Indemnification.* The Corporation shall indemnify, to the fullest extent permitted by Section 145 of the DGCL as it presently exists or may hereafter be amended, any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that such person:
 - (a) is or was a Director, officer or employee of the Corporation; or
- (b) is or was serving at the request of the Corporation as a director, trustee, member, member representative or officer of another corporation, partnership, limited liability company, joint venture, trust or other enterprise (including any employee benefit plan)

against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by such person in connection with such action, suit or proceeding if such person acted in good faith and in a manner such person reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe such person's conduct was unlawful. The termination of any action, suit or proceeding by judgment, order, settlement, conviction, or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that the person did not act in good faith and in a manner which such person reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that such person's conduct was unlawful. Notwithstanding the foregoing, except with respect to a proceeding to enforce rights to indemnification or advancement of expenses under this Article VII, the Corporation shall be required to indemnify a person under this Article VII in connection with an action, suit or proceeding (or part thereof) initiated by such person only if such action, suit or proceeding (or part thereof) was authorized by the Board of Directors. The Corporation may, to the extent authorized from time to time by its Board of Directors, grant indemnification rights to other employees or agents of the Corporation or to other persons serving the Corporation and such rights may be equivalent to, or greater or less than, those set forth herein.

Table of Contents

- 7.2 Expenses Payable in Advance. Expenses (including attorneys' fees) incurred by any person who is or was a Director or officer of the Corporation, or any person who is or was serving at the request of the Corporation as a director, trustee, member, member representative or officer of another corporation, partnership, limited liability company, joint venture, trust or other enterprise, in defending or investigating a threatened or pending action, suit or proceeding, whether civil, criminal, administrative or investigative, shall be paid by the Corporation to the fullest extent permitted by Delaware law in advance of the final disposition of such action, suit or proceeding, upon receipt of an undertaking by or on behalf of such person to repay such amount if it ultimately shall be determined that such person is not entitled to be indemnified by the Corporation as authorized in this Article VII. Such expenses (including attorneys' fees) incurred by any person who is or was an employee or agent of the Corporation, or any person who is or was serving at the request of the Corporation as an employee or agent of another corporation, partnership, limited liability company, joint venture, trust or enterprise may be so paid upon such terms and conditions, if any, as the Board of Directors deems appropriate.
- 7.3 Judicial Determination of Indemnification or Advancement of Expenses. If a claim for indemnification (following the final disposition of such proceeding) under this Article VII is not paid in full within sixty (60) days, or a claim for advancement of expenses under this Article VII is not paid in full within thirty (30) days, after a written claim therefor has been received by the Corporation the claimant may apply to any court of competent jurisdiction in the State of Delaware to order indemnification or advancement of expenses to the extent mandated under Sections 7.1 or 7.2 above. The basis of such order of indemnification or advancement of expenses by a court shall be a determination by such court that indemnification of, or advancement of expenses to, such person is proper in the circumstances. Notice of any application for indemnification or advancement of expenses pursuant to this Section 7.3 shall be given to the Corporation promptly upon the filing of such application. The burden of proving that such person is not entitled to such mandatory indemnification or mandatory advancement of expenses, or that the Corporation is entitled to recover the mandatory advancement of expenses pursuant to the terms of an undertaking, shall be on the Corporation. If successful in whole or in part in obtaining an order for mandatory indemnification or mandatory advancement of expenses, or in a suit brought by the Corporation to recover an advancement of expenses pursuant to the terms of an undertaking, such person shall also be entitled to be paid all costs (including attorneys' fees and expenses) in connection therewith.
- 7.4 Nonexclusivity. The indemnification and advancement of expenses mandated or permitted by, or granted pursuant to, this Article VII shall not be deemed exclusive of any other rights to which those seeking indemnification or advancement of expenses may be entitled under the Certificate of Incorporation, any Bylaw, agreement, contract, vote of stockholders or disinterested Directors, or pursuant to the direction (howsoever embodied) of any court of competent jurisdiction or otherwise both as to action by the person in an official capacity and as to action in another capacity while holding such office it being the policy of the Corporation that indemnification of the persons specified in Section 7.1 and Section 7.2 of this Article VII shall be made to the fullest extent permitted by law. The provisions of this Article VII shall not be deemed to preclude the indemnification of any person who is not specified in Section 7.1 or Section 7.2 of this Article VII, but whom the Corporation has the power or obligation to indemnify under Delaware law or otherwise, so long as such person acted in good faith and in a manner such person reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe such person's conduct was unlawful.
- 7.5 Insurance. The Corporation may, but shall not be obligated to, purchase and maintain insurance at its expense on behalf of any person who is or was a Director, officer, employee or agent of the Corporation, or is or was a Director or officer of the Corporation serving at the request of the Corporation as a director, officer, trustee, member, member representative, employee or agent of another corporation, partnership, limited liability company, joint venture, trust or other enterprise against any liability asserted against and incurred by such person in any such capacity, or arising out of the person's status as such, whether or not the Corporation would have the power or the obligation to indemnify such person against such liability under the provisions of this Article VII.

Table of Contents

- 7.6 Other Indemnification. The Corporation's obligation, if any, to indemnify or advance expenses to any person who was or is serving at its request as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, enterprise or non-profit entity shall be reduced by any amount such person may collect as indemnification or advancement of expenses from such other corporation, partnership, joint venture, trust, enterprise or non-profit enterprise.
- 7.7 Continuation of Indemnification. The rights to indemnification and to prepayment of expenses provided by, or granted pursuant to, this Article VII shall continue notwithstanding that the person has ceased to be a director or officer of the Corporation and shall inure to the benefit of the estate, heirs, executors, administrators, legatees and distributees of such person.
- 7.8 Definitions. For the purposes of this Article VII references to "the Corporation" shall include, in addition to the resulting company, any constituent company (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its directors, trustees, members, member representatives, officers, employees or agents, so that any person who is or was a director, trustee, member, member representative, officer, employee or agent of such constituent company, or is or was serving at the request of such constituent company as a director, trustee, member, member representative, officer, employee or agent of another corporation, partnership, limited liability company, joint venture, trust or other enterprise, shall stand in the same position under the provisions of this Article VII with respect to the resulting or surviving company as such person would have with respect to such constituent company if its separate existence had continued. The term "officer" in this Article VII, refers exclusively to a Chairman, the Chief Executive Officer, the Chief Financial officer, the General Counsel, the Executive Vice President, the Treasurer or the Secretary appointed pursuant to Article V of these Bylaws, and to any Vice President, Assistant Secretary, Assistant Treasurer, or other officer of the Corporation appointed by (x) the Board pursuant to Article V of these Bylaws or (y) an officer to whom the Board has delegated the power to appoint officers pursuant to Article V of these Bylaws. The term "other enterprise" as used in this Article VII shall include employee benefit plans. References to "fines" in this Article VII shall include excise taxes assessed on a person with respect to an employee benefit plan. The phrase "serving at the request of the Corporation" shall include any service as a Director, trustee, member, member representative, officer, employee or agent that imposes duties on, or involves services by, such Director, trustee, member, member representative, officer, employee or agent with respect to any employee benefit plan, its participants or beneficiaries.
- 7.9 Survival. The indemnification and advancement of expenses provided by, or granted pursuant to, this Article VII shall continue as to a person who has ceased to be a director, officer, employee or agent of the Corporation, and to a person who has ceased to serve at the request of the Corporation as a Director, trustee, member, member representative, officer, employee or agent of another corporation, partnership, limited liability company, joint venture, trust or other enterprise, and, in each case, shall inure to the benefit of the heirs, executors and administrators of such person.
- 7.10 Repeal, Amendment or Modification. The provisions of this Article VII shall constitute a contract between the Corporation, on the one hand, and, on the other hand, each individual who serves or has served as a Director or officer of the Corporation (whether before or after the adoption of these Bylaws), in consideration of such person's performance of such services, and pursuant to this Article VII the Corporation intends to be legally bound to each such current or former Director or officer of the Corporation. With respect to current and former Directors and officers of the Corporation, the rights conferred under this Article VII are present contractual rights and such rights are fully vested, and shall be deemed to have vested fully, immediately upon adoption of theses Bylaws. With respect to any Directors or officers of the Corporation who commence service following adoption of these Bylaws, the rights conferred under this provision shall be present contractual rights and such rights shall fully vest, and be deemed to have vested fully, immediately upon such Director or officer commencing service as a Director or officer of the Corporation. Any repeal or modification of the

Table of Contents

foregoing provisions of this Article VII shall not adversely affect any right or protection (i) hereunder of any person in respect of any act or omission occurring prior to the time of such repeal or modification or (ii) under any agreement providing for indemnification or advancement of expenses to an officer or Director of the Corporation in effect prior to the time of such repeal or modification.

ARTICLE VIII

NOTICE BY ELECTRONIC SUBMISSION

- 8.1 Notice by Electronic Submission. Without limiting the manner by which notice otherwise may be given effectively to stockholders pursuant to the DGCL, the Certificate of Incorporation or these Bylaws, any notice to stockholders given by the Corporation under any provision of the DGCL, the Certificate of Incorporation or these Bylaws shall be effective if given by a form of electronic transmission consented to by the stockholder to whom the notice is given. Any such consent shall be revocable by the stockholder by written notice to the Corporation. Any such consent shall be deemed revoked if:
- (a) the Corporation is unable to deliver by electronic transmission two (2) consecutive notices given by the Corporation in accordance with such consent; and
- (b) such inability becomes known to the Secretary or an Assistant Secretary of the Corporation or to the transfer agent, or other person responsible for the giving of notice.

However, the inadvertent failure to treat such inability as a revocation shall not invalidate any meeting or other action.

Any notice given pursuant to the preceding paragraph shall be deemed given:

- (i) if by facsimile telecommunication, when directed to a number at which the stockholder has consented to receive notice;
- (ii) if by electronic mail, when directed to an electronic mail address at which the stockholder has consented to receive notice;
- (iii) if by a posting on an electronic network together with separate notice to the stockholder of such specific posting, upon the later of (A) such posting and (B) the giving of such separate notice; and
 - (iv) if by any other form of electronic transmission, when directed to the stockholder.

An affidavit of the Secretary or an Assistant Secretary of the Corporation or of the transfer agent or other agent of the Corporation that the notice has been given by a form of electronic transmission shall, in the absence of fraud, be prima facie evidence of the facts stated therein.

8.2 *Definition of Electronic Transmission.* An "electronic transmission" means any form of communication, not directly involving the physical transmission of paper, including the use of, or participation in, one (1) or more electronic networks or databases (including one (1) or more distributed electronic networks or databases), that creates a record that may be retained, retrieved and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process.

ARTICLE IX

MISCELLANEOUS

9.1 Seal. The corporate seal shall have inscribed upon it the name of the Corporation, the year "2018" and the words "Seal" and "Delaware." The Secretary shall be in charge of the seal and may authorize a duplicate seal to be kept and used by any other officer or person. The Corporation may use

Table of Contents

the corporate seal by causing it or a facsimile thereof to be impressed or affixed or in any other manner reproduced.

- 9.2 *Fiscal Year.* The Board may determine the Corporation's fiscal year. Until changed by the Board, the Corporation's fiscal year shall be the calendar year.
- 9.3 Waiver of Notice. Whenever any notice is required to be given under any provision of the DGCL, the Certificate of Incorporation or these Bylaws to any stockholder or Director of the Corporation, a waiver thereof in writing, signed by the person or persons entitled to the notice, whether before or after the time stated therein, shall be deemed equivalent thereto. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting, except when the person attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any regular or special meeting of the stockholders need be specified in any written waiver of notice or any waiver by electronic transmission unless so required by the Certificate of Incorporation or these Bylaws.
- 9.4 Voting of Stock Owned by the Corporation. Powers of attorney, proxies, waivers of notice of meeting, consents and other instruments relating to securities owned by the Corporation may be executed in the name of and on behalf of the Corporation by the Chief Executive Officer, any Executive Vice President, any Vice President or the General Counsel. Any such officer may, in the name of and on behalf of the Corporation, take all such action as any such officer may deem advisable to vote in person or by proxy at any meeting of security holders of any corporation in which the Corporation may own securities and at any such meeting shall possess and may exercise any and all rights and powers incident to the ownership of such securities and which, as the owner thereof, the Corporation might have exercised and possessed if present. The Board of Directors may from time to time confer like powers upon any other person or persons.
- 9.5 Forum for Adjudication of Certain Disputes. Unless the Corporation consents in writing to the selection of an alternative forum (an "Alternative Forum Consent"), the Court of Chancery of the State of Delaware (the "Chancery Court") shall be the sole and exclusive forum for (i) any derivative action or proceeding brought on behalf of the Corporation, (ii) any action asserting a claim of breach of a fiduciary duty owed by any Director, officer, stockholder, employee or agent of the Corporation to the Corporation or the Corporation's stockholders, (iii) any action asserting a claim against the Corporation or any Director, officer, stockholder, employee or agent of the Corporation arising out of or relating to any provision of the DGCL or the Corporation's Certificate of Incorporation or these Bylaws, or (iv) any action asserting a claim against the Corporation or any Director, officer, stockholder, employee or agent of the Corporation governed by the internal affairs doctrine of the State of Delaware; provided, however, that, in the event that the Chancery Court lacks subject matter jurisdiction over any such action or proceeding, the sole and exclusive forum for such action or proceeding shall be another state or federal court located within the State of Delaware, in each such case, unless the Chancery Court (or such other state or federal court located within the State of Delaware, as applicable) has dismissed a prior action by the same plaintiff asserting the same claims because such court lacked personal jurisdiction over an indispensable party named as a defendant therein. Failure to enforce the foregoing provisions would cause the Corporation irreparable harm and the Corporation shall be entitled to equitable relief, including injunctive relief and specific performance, to enforce the foregoing provisions. Any person or entity purchasing or otherwise acquiring any interest in shares of capital stock of the Corporation shall be deemed to have notice of and consented to the provisions of this Section 9.5. The existence of any prior Alternative Forum Consent shall not act as a waiver of the Corporation's ongoing consent right as set forth above in this Section 9.5 with respect to any current or future actions or claims.

Table of Contents

- 9.6 Registered Office. The address of the registered office of the Corporation in the State of Delaware, and the name of its registered agent at such address, shall be as set forth in the Corporation's Certificate of Incorporation.
- 9.7 Executive Office. The corporate headquarters for the Corporation and its subsidiaries will be located in Exton, Pennsylvania. The primary location for research and development for the Corporation and its subsidiaries will be located in Birmingham, Alabama. The Corporation may have additional offices at any place or places, within or outside the State of Delaware, as the Board of Directors may from time to time establish or as the business of the Corporation may require.
- 9.8 *Dividends*. The Board of Directors, subject to any restrictions contained in either (i) the DGCL or (ii) the Certificate of Incorporation, may declare and pay dividends upon the shares of its capital stock. Dividends may be paid in cash, in property or in shares of the Corporation's capital stock. The Board of Directors may set apart out of any of the funds of the Corporation available for dividends a reserve or reserves for any proper purpose and may abolish any such reserve. Such purposes shall include but not be limited to equalizing dividends, repairing or maintaining any property of the Corporation and meeting contingencies.

ARTICLE X

AMENDMENT OF BYLAWS

The Board of Directors is expressly authorized and shall have the power to amend, alter, change, adopt and repeal the Bylaws of the Corporation or waive the observance of any bylaw (either generally or in a particular instance, and either retroactively or prospectively) at any regular or special meeting of the Board of Directors at which there is a quorum by the affirmative vote of a majority of the total number of Directors present at such meeting, or by unanimous written consent. The stockholders also shall have power to amend, alter, change, adopt and repeal the Bylaws of the Corporation or waive the observance of any bylaw (either generally or in a particular instance, and either retroactively or prospectively) at any annual or special meeting subject to the requirements of these Bylaws and the Certificate of Incorporation by the affirmative vote of the holders of a majority of the voting power of all of then-outstanding shares of capital stock of the Corporation then entitled to vote at an election of Directors, voting as a single class.

Table of Contents

$[\quad \bullet \quad]$ Certificate of Amendment and Restatement of Bylaws

The undersigned hereby certifies that he is the duly elected, qualified and acting [Title] of [•], a Delaware corporation (the "*Corporation*"), and that the foregoing Amended and Restated Bylaws were approved on [•], 2018, effective as of [•], 2018 by the Corporation's board of directors.

IN WITNESS WHEREOF, the undersigned has hereunto set his hand this [•] day of [•], 2018.

Name: Title:

. q IF YOU HAVE NOT VOTED VIA THE INTERNET OR TELEPHONE, FOLD ALONG THE PERFORATION, DETACH AND RETURN THE BOTTOM PORTION IN THE ENCLOSED ENVELOPE. q Proxy | IDERA PHARMACEUTICALS, INC. PROXY SOLICITED BY THE BOARD OF DIRECTORS Special Meeting of Stockholders | [], 2018 Those signing on the reverse side, revoking all prior proxies, hereby appoint(s) Mr. Vincent J. Milano and Mr. Louis. Arcudi, III, or each or any of them with full power of substitution, as proxies for those signing on the reverse side to act and vote all shares of stock of Idera which the undersigned would be entitled to vote if personally present at the Special Meeting of Stockholders of Idera (the Meeting) and at any adjournment or postponement thereof as indicated upon all matters referred to on the reverse side and described in the joint proxy statement/prospectus for the Meeting, and, in their discretion, upon any other matters which may properly come before the Meeting. Attendance of the undersigned at the Meeting or at any adjournment or postponement thereof will not be deemed to revoke this proxy unless those signing on the reverse side shall revoke this proxy in writing. THIS PROXY WHEN PROPERLY EXECUTED WILL BE VOTED IN THE MANNER DIRECTED BY THE STOCKHOLDER(S) SIGNING THE REVERSE SIDE AND IN THE DISCRETION OF THE PROXIES UPON ANY OTHER MATTERS WHICH PROPERLY COME BEFORE THE MEETING. IF NO INDICATION IS MADE, THE PROXIES SHALL VOTE FOR PROPOSALS NUMBERED I AND 2. PLEASE VOTE, DATE AND SIGN ON OTHER SIDE AND RETURN PROMPTLY IN ENCLOSED ENVELOPE. YOUR VOTE IS IMPORTANT. PLEASE VOTE TODAY.